Egyptian Agriculture in Transition: Farmer Perceptions of Risks and Adaptation Opportunities in a Changing Climate

Ayah Roushdy Abdulaty Omar

 BA, Agricultural Sciences – Community Development, Faculty of Agriculture, Cairo University, Egypt.
 MSc, Rural Sociology, Faculty of Agriculture, Cairo University, Egypt.

A thesis submitted in fulfilment of the degree of Doctor of Philosophy Department of Geography, Environment, and Population (GEP) School of Social Sciences Faculty of Arts

The University of Adelaide

July 2021

TABLE OF CONTENTS

TABLE OF CONTENTS	I
LIST OF TABLES	V
LIST OF FIGURES	X
ABSTRACT	XIII
DECLARATION	XV
ACKNOWLEDGMENTS	XVI
ABBREVIATIONS AND ACRONYMS	XVIII
Chapter One: Introduction	1
1.1 Introduction	1
1.2 Purpose and objectives	
1.3 Research Background	
1.3.1 Climate change and the MENA region: An area of concern	
1.3.1.1 Effects on water resources	5
1.3.1.2 Impacts on agriculture and food security	5
1.3.1.3 Impacts on socio-economic and political stability within the region	٥6
1.4 Conceptual Basis for the thesis	7
1.4.1 Agrarian Transition	7
1.4.2 Risk Perceptions	
1.4.3 Adaptation to climate change	9
1.5 Research Gap	
1.6 Research questions	11
1.7 Organisation of the thesis	11
Chapter Two: Review of Literature	13
2.1 Introduction	
2.2 Climate change in the MENA region: current challenges and t	future risks.13
2.2.1 Water	16
2.2.1.1 Dimensions of water crisis in Egypt	
2.2.2 Agriculture and food security	
2.2.2.1 Agriculture and food security in Egypt	
2.2.3 Demographic pressure, unemployment and security	
2.3 Agrarian transition	

2	.3.1 hange ir	Historical development of some land and agricultural policies that drove agrari	an 39
C	2311	The 1952 Revolution and land reform policies	39
	2312	The open-door economic policy	42
	2313	F Economic reform privatisation and liberalisation	43
2	3.2	The paths of agrarian transition in Egypt and the resultant class structure	47
24	Ric	k nercentions	50
2.7		antation to climate change	57
2. 3	5 1 In s	vitu adaptation	57
2	5.2 Fr	situ adaptation	58
2	252Lx	Immobility and trapped populations	50 67
	2.5.2.1	Culture of migration	65
26	2.3.2.2	mate alongo percention and adoptation studies in Equat	69
2.0	Sum	mate change perception and adaptation studies in Egypt	UO 71
2.7	Sun		/1
Ch	apter 1	hree: Research Methodology	13
3.1	Introd	uction	73
3.2	Adopt	ing an epistemological position	73
3.3	Resear	ch Approach: Mixed Methods	74
3.4	Data c	ollection approaches	76
3	.4.1 Uni	t of Analysis and levels and dimensions of data	76
3	.4.2 Sur	vey	76
3.4.2.1 Justification of the case study areas		77	
	3.4.2.2	Sampling	83
	3.4.2.3	Questionnaire Development	84
	3.4.2.4 Fieldwork		86
3	.4.3 In-c	lepth Interviews	88
3.5	Second	lary data	89
3.6	Data A	Analysis techniques	90
	3.6.1 🤇	Quantitative data analysis	90
	3.6.2 🤇	Qualitative data analysis	91
3.7	Ethica	l considerations	91
3.8	Post-st	tudy reflections and limitations of the research	92
3.9	Summ	ary	94
Cha Dai	apter 1 mietta	Four: An overview of the livelihood conditions of rural households Governorate	in 95

4.1 Introduction	95
4.2 General characteristics of the studied households	95
4.2.1 Household size, gender and age	95
4.2.2 Education and employment conditions	98
4.2.3 Livelihood and economic conditions of households	103
4.2.3.1 Social Capital	103
4.2.3.2 Income of households	108
4.2.3.3 Health and housing conditions	110
4.3 Agriculture in Damietta	. 113
4.3.1 Land size and patterns of production in the studied area	117
4.3.2 Households' satisfaction with the agricultural situation in the study area	120
4.3.3 Households' exposure to agricultural-related shocks and risks	124
4.4 Ecological environment in Damietta	. 127
4.5 Summary	. 128
Chapter Five: Perceptions of climate change and other risks associated	with
working in the agricultural sector	. 130
5.1 Introduction	. 130
5.2 Households' perceptions of climate variability and extremes	. 130
5.3 Households' perceptions of working in the agricultural sector	. 138
5.4 Perception of future risks related to the agricultural sector	. 141
5.5 Introducing household groups based upon their income from agricult activities	tural . 144
5.5.1 Agricultural dependency and perceptions of climate variability and extremes	146
5.5.2 Household group perceptions of employment in the agricultural sector	148
5.5.3 Agricultural dependency and perceptions of future risks in the agricultural sector	149
5.5.4 The socio-economic conditions of household groups	151
5.7 Summary	. 159
Chapter Six: Households' adaptation to climate variability and extreme Demiette	s in 160
6.1 Introduction	160
6.2 Household adoptation to long term alimate variability and short t	• 100
environmental shocks in Damietta	erm, .161
6.2.1 Climate change adaptation according to the dependence of households on agriculture	ulture
	167
6.3 Current migration pattern in the study area	. 170
6.4 Culture of migration in rural Damietta	. 175

6.4.1 Household culture of migration in relation to their dependence on agriculture
6.5 Migration plans and intentions of the studied households in Damietta
6.5.1 Household migration plans in relation to their dependence on agriculture
6.6 Summary
Chapter Seven: Livelihood challenges and dimensions of the vulnerability of
agrarian communities in Egypt185
7.1 Livelihood challenges for rural households in Damietta
7.1.1 Socio-economic challenges
7.1.2 Environmental challenges
7.1.3 Agricultural-related challenges
7.2 Perceptions of climate change risks
7.3 Adaptation to climate change
7.3.1 In situ adaptation methods
7.3.2 <i>Ex situ</i> adaptation methods
7.4 Climate change vulnerability across the agrarian transition process
7.4.1 Group Three: Agricultural dependence and climate change
7.4.1.1 A potentially trapped population? Place attachment and migration culture as barriers to <i>ex situ</i> adaptation
7.4.1.2 Structural issues that are intertwined with being "trapped" in place 203
7.4.2 Group Two: Transitioning rural households
7.4.3 Group One: Households moving to independence from agriculture
7.5 Summary
Chapter Eight: Conclusion
8.1 Introduction
8.2 Major findings
8.3 Implications for theory
8.4 Implications for policy and practice
8.5 Future research
8.6 Summary
Appendix 1: Survey on climate change perceptions and adaptation in Damietta 2018 223

LIST OF TABLES

Tab	le 3.1: Dimensions and level of data collection
140	
Tab adm	le 3.2: Area of cultivated agricultural lands and number of holdings by central inistrations as in 2017
Tab	le 3.3: The distribution of sampled households by survey locations
Tab	le 3.4: Summary of content of the questionnaire
Tab	le 4.1: The relationship of the respondents to the household's head
Tab	le 4.2: Distribution of all household members by gender96
Tab	le 4.3: Distribution of governorate's rural population by gender
Tab	le 4.4: Age groups of the respondents97
Tab	le 4.5: Distribution of all household members on the different age groups
Tab	le 4.6: Distribution of respondents by educational level in Damietta, 2018
Tab 201	le 4.7: Distribution of other adults within the households by educational level in Damietta, 8
Tab	le 4.8: Location of primary occupations of employed members101
Tab	le 4.9: Distribution of households by forms of social relationships
Tab ever	le 4.10: Distribution of households by frequency of participation in community
Tab prin	le 4.11: Cross-tabulation between respondents' level of social participation and their nary occupations
Tab adm	le 4.12: Cross-tabulation between the level of social participation and the studied local inistrations
Tab	le 4.13: Forms of support accessed from the local community (multiple responses)106
Tab	le 4.14: Categories of households' monthly income109
Tah	le 4.15: Households satisfaction with local health services 110

Table 4.16: Basic analysis for the distances (in kilometres) between houses and several local destinations
Table 4.17: Households satisfaction with their houses 113
Table 4.18: Distribution of cultivated lands by farm size and holders in Damietta in 2017115
Table 4.19: Total agricultural land area owned, shared and/or rented by households117
Table 4.20: The distance between agricultural farmland and the households
Table 4.21: Damietta respondents' agricultural land use from early May–late September2018.118
Table 4.22: Main marketing channel adopted by households for selling agricultural produce
Table 4.23: Cross-tabulation between marketing channels and rate of product selling119
Table 4.24: Household dependency on family members and paid labourers in agricultural operations (mean percentages) 120
Table 4.25: Respondent households' exposure to sudden shocks during 2016 and 2017 (in percentages)
Table 4.26: Predominant environmental problems in Damietta
Table 5.1: The perceived timing of the most observed climate variability and extreme weather events in Damietta by 2018 (multiple-response question) 131
Table 5.2: Household overall perceptions of the adverse impacts of climate variability and extremes
Table 5.3: Distribution of households by their overall perceptions of the adverse impacts of climate variability and central administration
Table 5.4: Respondents comments on the adverse impacts of climate variability
Table 5.5: Distribution of households by the likelihood of getting exposed to problems in theagricultural sector (n=350)
Table 5.6: Distribution of households having members working as waged labourers by the likelihood of getting exposed to employment problems in the agricultural sector (n=88)

Table 5.7: Distribution of households by agreement with aspects of working in agriculture(n=350)
Table 5.8: Distribution of households by perceptions of future risks in the agricultural sector
Table 5.9: Distribution of overall household perceptions of future risks by centraladministration
Table 5.10: Respondents' most important open-ended comments on their perceived futurerisks within the agricultural sector
Table 5.11: Distribution of households by income from agricultural activities 145
Table 5.12: The distribution of agricultural dependence groups by the respondents' primary occupations in Damietta
Table 5.13: The distribution of households living in the two studied areas by the respondents'primary occupations in Damietta
Table 5.14: Agricultural dependence groups' mean rankings of household perceptions of climate change
Table 5.15: Agricultural dependence groups' mean rankings of household perceptions of climate variability and extremes adverse impacts
Table 5.16: Agricultural dependence groups' mean rankings of household perceptions of working in agriculture
Table 5.17: Agricultural dependence groups' mean rankings of household perceptions offuture potential risks within the agricultural sector
Table 5.18: Agricultural dependence groups' mean rankings of household demographic and economic conditions 152
Table 5.19: Agricultural dependence groups' mean rankings of household expenditure items
Table 5.20: Agricultural dependence groups' mean rankings of household social participation and relationships
Table 5.21: Distribution of agricultural dependence groups by respondent's membership in syndicates
Table 5.22: Distribution of agricultural dependence groups by respondent's membership infarmers' organisations or unions

Table 5.23: Agricultural dependence groups' mean rankings of housing and other living conditions in the study area. 156
Table 5.24: Agricultural dependence groups' mean rankings of household reliance onunpaid/paid labourers for agricultural operations
Table 5.25: Agricultural dependence groups' mean rankings of aspects related to the satisfaction with the agricultural situation in the study area
Table 6.1: On-farm adaptation strategies adopted by the households to reduce losses or adaptto climate variability and extremes from 2012 to 2017
Table 6.2: Livelihood adaptation strategies adopted by the households to reduce losses or adapt to increasing climate variability and extremes from 2012 to 2017163
Table 6.3: Agricultural dependence groups' mean rankings of climate change adaptation levels
Table 6.4: The distribution of agricultural dependence groups by their choice of plantingdifferent crop varieties to reduce losses from 2012 to 2017
Table 6.5: The distribution of agricultural dependence groups by their choice of changingplanting dates to reduce losses from 2012 to 2017
Table 6.6: The distribution of agricultural dependence groups by their choice of mulching toreduce losses from 2012 to 2017
Table 6.7: The distribution of agricultural dependence groups by their choice of increasingthe use of chemicals/fertilizers/insecticides to reduce losses from 2012 to 2017
Table 6.8: The distribution of agricultural dependence groups by their choice of improvingtheir observation of meteorological information from 2012 to 2017
Table 6.9: Nature of migration in Damietta in 2018
Table 6.10: Place of destination for household members migrating from Damietta in2018
Table 6.11: Reasons for migration from Damietta 2018 (multiple-response question)172
Table 6.12: Uses of remittance money as stated by the studied households in Damietta 2018(multiple-response question)
Table 6.13: Reasons for not migrating from Damietta 2018

Table 6.14: Respondents' opinion about occupational mobility and migration in Damietta(n=350)
Table 6.15: The distribution of the respondents by their migration culture Damietta(n=350)
Table 6.16: The distribution of respondents residing in the two local administrations by their agreement with: "If I am financially capable, another family member or I would definitely migrate"
Table 6.17: Agricultural dependence groups' mean rankings of household cultures ofmigration in the study area181
Table 6.18: Household respondents' future migration intentions in Damietta
Table 6.19: Agricultural dependence groups' mean rankings of household migrationintentions in the study area183

LIST OF FIGURES

Figure 1.1 Map of the Middle East and North Africa (MENA) region4
Figure 2.1: Projected percentage of Middle East and North Africa national populations affected by sea-level rise impact
Figure 2.2: Projected impact of sea-level rise impact on the Middle East and North Africa national GDP
Figure 2.3: Water stress ration in the Middle East and North Africa region17
Figure 2.4: Renewable water resources per capita in Egypt, 1958–1962 to 2013–2017
Figure 2.5: The role of the agricultural sector in the Egyptian economy, 1995–201924
Figure 2.6: Egypt imports from wheat, 1994–201825
Figure 2.7: Stages of agricultural transformation based upon workforce and contribution of the agricultural sector in GDP
Figure 2.8: Relationship between vulnerability to environmental change and mobility representing trapped population
Figure 2.9: Schematic representation of (im)mobility categories67
Figure 3.1: Local Administration Levels of Egypt
Figure 3.2: The Egyptian Nile Delta region
Figure 3.3: Damietta location within the Nile Delta region
Figure 3.4: Map of Damietta80
Figure 3.5: Agricultural land area relative to the total inhabited land area of the five central administrations of Damietta (km ²)
Figure 3.6: Distribution of the total rural population over the five central administrations
Figure 3.7: The location of the studied villages in Damietta governorate map
Figure 4.1: Distribution of households by household size in Damietta, 201896
Figure 4.2: Distribution of households' members by occupation in Damietta, 2018100

Figure 4.3: Distribution of non-income earning adult household members by reasons for unemployment
Figure 4.4: Distribution of respondents by membership in organisations107
Figure 4.5: Distribution of households according to the access to information sources108
Figure 4.6: Distribution of households based upon dependency on multiple income sources
Figure 4.7: Distribution of households by the number of rooms within the houses111
Figure 4.8: Distribution of houses by basic facilities in Damietta, 2018112
Figure 4.9: Quality of public services provided in the local area as rated by the households
Figure 4.10: The area of cultivated agricultural lands within each administration area in Damietta (<i>feddans</i>)115
Figure 4.11: Distribution of total cultivated land area by winter crops and vegetables 2017–2018
Figure 4.12: Distribution of total cultivated land area by summer crops and vegetables 2018
Figure 4.13: Household respondents' satisfaction with the different factors of agricultural production in Egypt
Figure 4.14: Coping strategies adopted by households to reduce losses from sudden shocks
Figure 4. 15: Water canal pollution in Hagaga village- Faraskur central administration, Field survey 2018
Figure 4.16: The pollution of irrigation water in Kafr El-Arab village- Faraskur central administration, Field survey 2018
Figure 5.1: Distribution of households by the observed forms of climate variability and extremes in Damietta by 2018 (n=315)
Figure 5.2: Distribution of households by respondents' perceptions of the degree of severity of the adverse impacts of climate variability and extremes
Figure 6.1: Distribution of households by their adaptation level to climate variability and extremes (in percentage)

Figure 6.2: Household strategies for coping with income losses as a result of climate variability and extremes that took place in 2016 - 2017 (multiple-response question).....165

ABSTRACT

The world is witnessing major changes in climate, progressively causing environmental, socio-economic, and political challenges. Climate variability and change are affecting agricultural sectors across the globe and are impacting the economic and social conditions of many agricultural households in rural areas, particularly in developing countries. The agricultural sector in Egypt is already under stress due to population growth and critically low water availability. Egypt is likely to suffer further negative impacts of climate change with key concerns in food security, inundation of coastal areas, environmental degradation, land and freshwater resource depletion, and declining local production and economic conditions. Adaptation to climate change is critical to reduce or avoid the more extreme adverse effects of environmental change, meet the future needs of a growing population, and maintain secure livelihoods for those who depend on agriculture for their existence.

This study aims to understand the links between the phases of agrarian transition concerning perceptions of and adaptation to climate change to generate knowledge and enhance current and future local adaptation to climate change within rural Egypt. This research focuses on three components: (1) Farming households' perceptions of risks from climate change and other changes in the agricultural sector in Egypt; (2) the *in situ* and *ex situ* strategies that farmers and their households consider suitable while adapting to climate change within their socio-economic and cultural contexts and; (3) the cultures of migration that could influence current and potential migration decisions of rural households.

The study was conducted in four villages in Damietta, one of the governorates of the Nile Delta Region. This region was chosen because it is one of the most vulnerable to the impacts of climate change in Egypt, particularly concerning the potential threat of sealevel rise and water resource depletion. A mixed-method research approach is adopted, with both qualitative and quantitative data accessed. Primary quantitative data were collected from 350 small landholder farmers and waged agricultural labourers using a questionnaire implemented via face-to-face interviews from April 2018 to September 2018. A further 11 in-depth interviews were conducted with key respondents to gather qualitative data about the pressing challenges facing farming households in the study area. Small landholder farmers in Damietta are experiencing climate change. They have high levels of perception of its adverse impacts on crop productivity and quality, and household incomes, yet their adaptation levels remain low. Stemming from the different paths or phases of the agrarian transition, three groups of households emerged from the initial analysis to form the basis for grouping three types of climate change adapters based on the relative contributions of agriculture to household income. The social and economic impacts of climate change appear largely unimportant for households who depend primarily on non-agricultural activities for a living. While some households are adapting *in situ* in Damietta, but those are the relatively wealthy households who are not wholly dependent on agriculture for their livelihoods. In contrast, those households who depend primarily on agriculture for a living have a strong perception of the negative impacts of climate change, yet, they are implementing the fewest adaptation responses either in situ or ex situ, and are more reluctant to leave the agricultural sector or their communities. Consequently, relatively poor farming households appear to be less capable of adapting to climate change even though they form a more vulnerable cohort and potentially constitute a trapped population with little capacity to respond effectively to risk.

Considering the growing demographic, economic, social, and environmental challenges in Egypt and other parts of the Middle East and North Africa (MENA) region, many small landholder farming communities might find it difficult to sustain a living from agriculture or exploit livelihood opportunities elsewhere. This situation will be particularly critical for trapped populations who are unable or unwilling to relocate, as the repeated climate change stressors can continue to undermine their already fragile economic livelihoods and erode their asset base, making them less able to adapt either *in situ* or *ex situ*. As has already been experienced in several MENA countries, struggling rural households may form a potential latent group that could continue to drive radical social changes in Egypt if conditions for agriculture continue to deteriorate. Developing marginal communities' capacities to adapt *in situ* through investing in rural education and micro-economy is of key importance both to maintain local livelihoods and the broader social fabric.

DECLARATION

I certify that this work contains no material which has been accepted for the award of any other degree or diploma in my name, in any university or other tertiary institution and, to the best of my knowledge and belief, contains no material previously published or written by another person, except where due reference has been made in the text. In addition, I certify that no part of this work will, in the future, be used in a submission in my name, for any other degree or diploma in any university or other tertiary institution without the prior approval of the University of Adelaide and where applicable, any partner institution responsible for the joint-award of this degree.

I give permission for the digital version of my thesis to be made available on the web, via the University's digital research repository, the Library Search and also through web search engines, unless permission has been granted by the University to restrict access for a period of time.

I acknowledge the support I have received for my research through the provision of an Australian Government Research Training Program Scholarship.

Sign: _

Date: ____1 July 2021_____

ACKNOWLEDGMENTS

"All the praise is for God, the One Who has guided us to this"

First and foremost, I would like to express my sincere gratitude to my principal supervisor A/Prof Douglas Bardsley for his extraordinary supervision, support, and guidance. His immense knowledge and enthusiasm have significantly improved this thesis and pushed me farther than I thought I could. Without his valuable support, it would not have been possible for me to come this far in my journey. I would also like to extend my deepest appreciation to my co-supervisor, A/Prof Yan Tan, for supporting my scholarship application and for her insightful suggestions and invaluable contribution to this thesis.

I gratefully acknowledge the financial contribution of the Department of Geography, Environment and Population and the Charles and Frank Fenner Award, which funded my fieldwork and data collection in Egypt. I thank all staff and colleagues in the Department of Geography, Environment and Population. My most profound indebtedness goes to Dr. Jennifer Bonham, the department's former postgraduate coordinator, for her great support during my candidature. I am also grateful to the Adelaide Campus Child Care Centre for taking good care of my daughter during my PhD journey. I wish to thank my editor, Emma Short, for her efforts and generous advice on formatting.

I must also thank all those who facilitated my fieldwork in Damietta – Eng. Mohamed Abou Attia, Eng. Hassan El-Sharkawy, Mr. Hazim, Eng. Saber and Eng. Sherif Negm. Special thanks to the enumerator team, especially Ehab Hassan, and my colleagues and professors at Cairo University. My sincerest gratitude goes to the hundreds of farmers who participated in this study. I am thankful for their cooperation, time commitment, and wonderful hospitability and generosity.

To my father, Roushdy Omar, and my mother, Samar Ali, whom I deeply miss. Thank you for your endless love, support, and prayers and for your sacrifices for educating my sisters and me. You have always encouraged me to work hard and not think that anything was beyond me. I would like to thank my sisters and best friends, Soha and Shaza, and their lovely sons, Anas and Badr, for their incredible love and words of encouragement. Many thanks to my family in law for their support.

To my little girl, Nour "Light", you are my soul, sunshine, and the best thing ever happened to me. Thank you for getting me through the hard days and filling my life with happiness despite all difficulties. To my husband, Mohamed, thank you for your patience, understanding and encouragement throughout this project. Thank you for putting up with my mood swings, stress and many times of depression.

Last but not least, I thank my colleagues and close friends Hannah, Julia, Bridie, Bay, Jeremiah and Govind, who provided much-needed laughter, advice and many great memories that I will cherish forever. To my Egyptian friends, whom I consider my second family here in Australia, I am very thankful for your friendship and love.

ABBREVIATIONS AND ACRONYMS

CAPMAS	Central Agency for Public Mobilisation and Statistics (Egypt)
СТА	Technical Centre for Agricultural and Rural Cooperation
EEAA	Egyptian Environmental Affairs Agency (Egypt)
FAO	Food and Agriculture Organisation
FIAN	Food-First Information and Action Network
GRD	Grand Renaissance Dam
HH	Household
IFAD	International Fund for Agricultural Development
IOM	International Organisation for Migration
IPCC	Intergovernmental Panel on Climate Change
IRIN	Integrated Regional Information Networks
MALR	Ministry of Agriculture and Land Reclamation (Egypt)
MENA	The Middle East and North Africa
NGO	Non-governmental Organisation
OECD	Organisation for Economic Co-operation and Development
SIDA	Swedish International Development Cooperation Agency
UNDP	United Nations Development Programme
UNFCCC	United Nations Framework Convention on Climate Change
UNICEF	United Nations Children's Fund
UNOCHA	The United Nations Office for the Coordination of Humanitarian Affairs
WWAP	United Nations World Water Assessment Programme

Chapter One: Introduction

1.1 Introduction

There are a range of political, economic and institutional dynamics that shape the agricultural sector (Pradeep & Shane 2003). Yet, environmental factors, especially climate, are considered the primary drivers that directly influence agricultural production in most places (Houghton 2009). Today, the world is witnessing significant changes in climate that are expected to affect most of the world adversely, but especially the poor who depend on agriculture as a primary source of living (IFAD 2010). The Middle East and North Africa (MENA) region appears as the global region most vulnerable to climate change, due to the expansion of aridity in many areas (Al Taweel, Ugursal & Boodlal 2015; Alboghdady & El-Hendawy 2016; Waha et al. 2017; Ozturk et al. 2018) and the lack of water resources (Droogers et al. 2012; Terink, Immerzeel & Droogers 2013; Alboghdady & El-Hendawy 2016). Consequently, the region is considered "one of the hot spots for worsening extreme heat, drought and aridity conditions under climate change" (Waha et al. 2017, p.1). The agricultural sector, in particular, is highly vulnerable to climatic variability (Wasimi 2010), and such vulnerability will have critical consequences as the sector employs most of the labour force and plays a significant role in meeting national food demand (Verner 2012).

The agricultural sector in the MENA region generally, and Egypt specifically, is already under stress due to population growth and critically low water availability (MALR 2009; Koocheki 2010; Sowers, Vengosh & Weinthal 2011; Verner 2012; Waha et al. 2017). This situation would be further challenged with the projected future population increases and resultant increasing demand for food and water (Nigatu & Motamed 2015; Waha et al. 2017). Climate change will exacerbate pressures on the increasingly scarce resources, and its effects upon the agricultural sector will be significant (Cline 2007; Sowers, Vengosh & Weinthal 2011). Such environmental pressures will affect many rural households who depend on agriculture for their livelihood and thus reinforce conditions of unemployment and poverty that are already prevalent in the region, leading potentially to social and political unrest and conflict over resources (Sowers & Weinthal 2010; IFAD 2010; Sowers, Vengosh & Weinthal 2011; Nigatu & Motamed 2015; Waha et al. 2017).

Therefore, adaptation to climate change is critical to attempt to reduce or avoid the adverse effects of climate change; to meet the future needs of a growing population; and to maintain secure livelihoods for those who depend on agriculture for their existence. For adaptation to be achieved, farmers first need to perceive or be made aware of the current and future risks associated with climate change (Smithers & Smit 1997; Weber 1997; Vedwan & Rhoades 2001; Maddison 2007; Patt & Schröter 2008; Weber & Johnson 2009; Gbetibouo 2009; Bardsley & Hugo 2010; Spence et al. 2011; Koerth et al. 2013; Bagagnan, Ouedraogo & Fonta 2019). This thesis aims to analyse Egyptian farmers' perceptions of climate change impacts, including risks for the agricultural sector and from climate change directly, and the measures to adapt *in situ* or *ex situ* with the household. The thesis also considers the socio-economic factors that influence households' adaptive capacities and perceptions of risk. The research takes into consideration how the climate risk perceptions and actions are influenced by households at different phases of the agrarian transition, by grouping households based upon their dependency on agriculture for the generation of their livelihoods. This categorisation of households aims to support an in-depth analysis and understanding of the vulnerability and the adaptive capacities of households within different socio-economic groups in an evolving Egyptian rural society.

The study adopts a mixed-methods approach in which quantitative and qualitative data were collected from four villages in Damietta, one of the governorates of the Nile Delta Region in Egypt. This region was chosen as being one of the most vulnerable areas to the effects of climate change in Egypt, particularly concerning the potential threat of rising sea-levels as it is bordered by the Mediterranean Sea (El-Raey 2010; EEAA 2010, 2016; Frihy & El-Sayed 2013; Wöppelmann et al. 2013; Elshinnawy & Almaliki 2021). Egypt is expected to be badly affected by climate change, as such change could have serious environmental, economic and social effects on the different sectors, but particularly the agricultural sector (EEAA 2010; 2016). Due to growing land fragmentation in Egypt, agricultural lands are dominated by small landholder farmers (Aboulnaga et al. 2017) many of whom are considered to be extremely vulnerable to climate change (IFAD 2010). The general scope of this research, therefore, includes a discussion on the climate change implications for the Egyptian agricultural sector with a particular emphasis on smallholding farmers and landless agricultural labourers. It continues on to examine how different groups across the rural society of Egypt may experience and respond to climate change.

This introductory chapter begins by outlining the aims and objectives of the research. It outlines the persistent social, economic, demographic and environmental challenges in the MENA region in general, and in Egypt specifically, highlighting the key issues surrounding climate change with a particular focus on its impact upon the agricultural sector. The chapter then introduces some key theoretical concepts of risk perception, adaptation to climate change with its two approaches, *in situ* and *ex situ*, and agrarian transition theory. The author then emphasises some of the research gaps and outlines the current research questions. Finally, she frames how the remainder of the thesis will be organised to answer the research questions.

1.2 Purpose and objectives

This study aims to analyse the links between the phases of the agrarian transition and farmer perceptions and adaptation to climate change, to generate knowledge and to enhance current and future local adaptation to climate change within rural Egypt, MENA countries and other parts of the world experiencing climate change as they pass through the agrarian transition process.

Specific objectives of this study are as follows:

- Identify the degree to which Egyptian rural households, particularly within small landholder communities, perceive current and future risks associated with climate change and employment in the agricultural sector;
- Explore the current strategies that farmers and their households consider suitable while adapting to climate change within their socio-economic and cultural contexts; and
- Identify the extent to which households' dependency on agricultural activities for their livelihood and other socio-economic factors influence their perception of, and level of, adaptation to climate change.

1.3 Research Background

1.3.1 Climate change and the MENA region: An area of concern

The MENA region encompasses twenty countries, as defined by UNICEF, located in the Middle East and northern part of Africa (Figure 1.1). In 2018, the region's population reached 484 million, with an average growth of 2.0 percent per year, exceeding the world's average by 0.7 percent (UNICEF 2019). Egypt and Iran are the two most

populated countries in the region, containing nearly 40 percent of the region's population. By 2050, the MENA region's population is expected to reach 724 million, and Egypt will experience a large population increase, with an additional 26 million people expected between 2015 and 2030, and a further 60 million by 2050 (UNICEF 2019). Mostly poor and fragile countries or countries in conflict tend to have faster population growth (Walker 2016). In Egypt, for instance, roughly one-third of the population lived below the national poverty line (around \$1.45 per day) in 2018, with 6.2 percent of Egyptians living in extreme poverty (less than a dollar per day) (CAPMAS 2019a). Together, the rapid population growth and widespread poverty already ensure that many people are extremely vulnerable across the country, even before the implications of climate change are examined.



Figure 1.1 Map of the Middle East and North Africa (MENA) region

Source: UNICEF (2019)

The MENA region is one of the most economically diverse areas of the world, ranging from the high-income, oil-rich countries in the Persian Gulf, to poorer countries with scarce resources and high populations, such as Egypt, Yemen, and Morocco (Terink, Immerzeel & Droogers 2013; OECD-FAO 2018). This economic diversity entails variability in the vulnerability to climate change risks and the adaptive capacities across the region, especially between the Arab Gulf States, including Saudi Arabia, United Arab Emirates, Kuwait, Qatar, Bahrain and Oman, and the other MENA countries (Waha et al. 2017). Many developing countries already face severe environmental threats, governance challenges, and problems in their social and economic structures (Adger et al. 2003; Lemos & Agrawal 2006); consequently, developing countries within the region are likely to bear the worst effects of climate change (Stern 2006).

Arid and low-lying coastal areas, prevailing through most of the MENA countries, are considered to be particularly exposed to the effects of climate change (Al Taweel, Ugursal & Boodlal 2015), as their flood-prone coastal zones hold the majority of economic activities and population, and the region is highly dependent on climate-sensitive agriculture (Alboghdady & El-Hendawy 2016). Therefore, the MENA is one of the most vulnerable regions to the risks of climate change in the world with potential adverse effects on water resources, the agricultural sector, and the socio-economic and political conditions of the region as is explained below (Alboghdady & El-Hendawy 2016; Waha et al. 2017; Ozturk et al. 2018).

1.3.1.1 Effects on water resources

The MENA region is one of the most arid, inhabited areas on the globe (Droogers et al. 2012; Terink, Immerzeel & Droogers 2013; Alboghdady & El-Hendawy 2016). This situation alone, coupled with the projected exponential population growth in the region, is increasing water stress (Koocheki 2010; Sowers, Vengosh & Weinthal 2011). However, the most significant effect could become evident in those countries that depend upon surface water supplies from rivers, such as Egypt relies upon the Nile, and the flow regimes are altered over time (Sowers, Vengosh & Weinthal 2011). Egypt's control over Nile water has been threatened by the Grand Renaissance Dam's construction in Ethiopia - a situation that could further put water availability in Egypt at enormous risk if sufficient water supply remains unavailable for downstream countries (Nunzio 2013; Power 2014). Together, climate change coupled with demographic challenges, water pollution, inefficient irrigation systems, and the proliferation of upstream developments along the Nile River is likely to mean that the water sector in Egypt will require significant reform in the future (Power 2014). In particular, farmers and communities directly reliant upon the Nile waters are likely to need to adjust their production and livelihood systems to adapt to these changing circumstances.

1.3.1.2 Impacts on agriculture and food security

The MENA region has many environmental systems within which there are limits of productive agriculture due to the scarcity of resources (land and water) and are experiencing ongoing degradation of agricultural lands caused by unsustainable farming practices and soil erosion (Koohfkan 2001; Koocheki 2010; OECD-FAO 2018). Population growth, together with water stress, has increased the dependence on imported

food within this area (Waha et al. 2017); for instance, Egypt imports about 40 percent of its grain supply (Koocheki 2010; Nigatu & Motamed 2015).

Climate-related effects on agriculture are expected to be particularly significant in the MENA region given the current environmental, governance, and demographic challenges, which will be exacerbated by a changing climate (Cline 2007). Natural constraints limit the ability of MENA countries to increase their agricultural production meaning that their reliance upon food imports is likely to increase with population growth (Woertz 2017). The high dependence on food imports already raises critical questions about the reliability of global food markets to supply countries' needs, and that risk will increase with a changing climate (Nigatu & Motamed 2015; Woertz 2017; Waha et al. 2017; OECD-FAO 2018). Subsequent changes in agricultural production will have huge consequences on rural livelihoods, poverty, national economies and food security (Breisinger et al. 2012) and exemplify a significant challenge for sustainable development in the region, which will have increasingly global consequences if not managed effectively (Sowers, Vengosh & Weinthal 2011).

1.3.1.3 Impacts on socio-economic and political stability within the region

Political insecurity, economic inequalities, poverty, unemployment, and conflicts are key features in many parts of the MENA region (Mohtar, Assi & Daher 2017; Waha et al. 2017; SIDA 2019). Such conditions are likely to increase the vulnerability of many countries in the region to the negative effects of climate change (Elasha 2010) - and arguably those processes of destabilisation are already underway. Climate change is already driving violence and conflicts in many parts of the world (Selby & Hoffmann 2014; Schleussner et al. 2016; Von Uexkull et al. 2016; Detges 2016; Feitelson & Tubi 2017; Ide et al. 2021). With a growing population and limited economic diversification in countries such as Egypt, national labour markets face challenges of absorbing the expanding workforce, intensifying the problem of unemployment in the region, and particularly in the growing urban centres (Akhtar, Bolle & Nelson 2013; Nigatu & Motamed 2015; Bilgili & Marchand 2016). Verner (2012) argues that the adaptation challenge in MENA's rural areas is problematic in large part due to the prevailing poverty, with more than 34 percent of its rural population being classified as poor, which raises a critical question for this thesis – what role will migration play in supporting communities to adapt to climate change?

Climate change will not necessarily result in more migration within the MENA region (Bilgili & Marchand 2016), especially for those individuals who have extremely limited resources (Gemenne 2015). The opportunity or choice for people to migrate or not to adapt to climate change would unarguably interact with many other economic, social and political factors and could exacerbate current social and political instability within the region (Sowers, Vengosh & Weinthal 2011; Nigatu & Motamed 2015; Schilling et al. 2020). This situation could potentially create social conflict, disturbing human security within the region (Sowers & Weinthal 2010; Gleick 2014; Waha et al. 2017). Egypt has already recently experienced a period of protests and revolution, and with the growing threats of climate change and accelerated water scarcity due to the Renaissance Dam, the future of political stability in the country remains uncertain.

1.4 Conceptual Basis for the thesis

This section briefly outlines the main theoretical and conceptual basis for the thesis that will be further comprehensively discussed in the literature review in Chapter 2.

1.4.1 Agrarian Transition

The agrarian transition is usually paired with "structural transformation" that is widely considered key to the economic development process through which low-income societies become high-income societies (Barrett, Christian & Shiferaw 2017). Three central themes are involved in such transition of economies, including:

- 1. Agricultural intensification;
- 2. The existence of two-way linkages between farm and nonfarm sectors; and
- The emergence of a strong relationship between agricultural production systems, demographic change and dietary shifts (Johnston & Mellor 1961; Timmer 1988, 2009; Dercon & Gollin 2014; Barrett, Christian & Shiferaw 2017).

The classical views of Byres (1977), articulate the transition from an agrarian society to be the result of several factors, including the penetration of capitalism as the primary mode of production, the subsequent development of a diversified class structure, the improvement of new agricultural techniques, and the emergence of urbanised societies. The process of agrarian transition has transformed many wealthy farmers into capitalist farmers, and many poor farmers into waged labourers or an urban proletariat, leading to a distinctive class structure within modern societies (Byres 1977) – a process that is still underway across the MENA region.

The World Bank Group (2015) suggests five stages of agricultural transformation based upon the contribution of the agricultural sector to the society's GDP and the percentage of labour force working in the sector:

- Agricultural-based societies
- Pre-transition societies
- Transition societies
- Urbanised societies
- Developed countries.

According to these phases, Egypt is considered a borderline transition and urbanising country, as Egyptian agriculture contributes 11.22 percent of the GDP and employs 21.7 percent of the labour force (CAPMAS 2020a).

Hopkins (1987) identifies two pathways of agricultural or rural transformation resulting in the dominant class structure in the Egyptian lands of the Delta. The first path is called the "capitalist path", which depends upon the ability of large landholder farmers to accumulate capital and to acquire machinery and other means of production. The second path, on the other hand, is called the "petty commodity producers path" and involves small landholders and small-scale producers. In addition, there is an "agrarian bourgeoisie" another key class of landowners, who largely make their living from nonagricultural sectors present within the social structure of rural villages (Weinbaum 1982; Springborg 1990).

This thesis focuses on adopting the phases of agrarian transition suggested by the World Bank Group (2015) and the associated class structure presented by Hopkins (1987) and Springborg (1990) to assist in reporting and interpreting the results of the study. Taking the different paths inherent to the agrarian transition process into account allows for an in-depth understanding of the perceptions of risk associated with climate change, vulnerabilities, and decisions of rural households about risk and adaption within the different sectors of the Egyptian agrarian community.

1.4.2 Risk Perceptions

People have already developed complex perceptions about climate change (Mertz et al. 2009b; Nyanga, Johnsen & Aune 2011; Spence et al. 2011; Deressa, Hassan & Ringler 2011; Chaudhary & Bawa 2011; Akter & Bennett 2011; Omar et al. 2015).

Understanding perception is fundamental to the social sciences because the perception of issues influences people's willingness to act in specific ways or support decisions being made by others. Hence, adaptation responses in large depend on the ability of individuals to perceive climate change and to take action (Maddison 2007; Antle 2009; Weber & Johnson 2009; Bardsley & Hugo 2010; Bagagnan, Ouedraogo & Fonta 2019). This thesis takes into account farmers' perceptions of several current and future risks associated with climate change and risks related to working in the agricultural sector under present and upcoming environmental challenges.

1.4.3 Adaptation to climate change

Adaptation to climate change refers to the process of adjusting to the actual or expected effects of climate change (IPCC 2007a), which might eventually reduce people's vulnerability through an enhanced resilience to these changes (UNFCCC 2007). Reuveny (2007) classifies adaptation into three broad categories: "*in situ*", "migration or *ex situ*", and "no response". *In situ* adaptation includes localised measures that occur within an affected place, while *ex situ* adaptation implicates the movement of people, assets and/or whole systems from a vulnerable location (Bardsley & Hugo 2010). Not responding to actual or perceived climate change is generally related to a lack of adaptive capacity of individuals or systems to overcome the effects of climate change (Smit & Pilifosova 2003; Adger, Arnell & Tompkins 2005; Gallopín 2006).

Within the agricultural sector, *in situ* adaptations include practices that aim at adjusting or improving agricultural production, natural resources including water and land, and farm management (Smit & Skinner 2002; Grothmann & Patt 2005; Deressa et al. 2009; Marshall 2010; Osberghaus, Finkel & Pohl 2010). Some studies consider climate-induced migration as a last choice in which individuals fail to adapt to new conditions locally (Warner et al. 2010, Laczko & Piguet 2014), while some other researchers claim that migration might be considered an effective and successful response to climate change (Tacoli 2009; Bardsley & Hugo 2010; Black et al. 2011b; Lücke 2011; Gemenne 2013; Adger & Adams 2013; Baldwin & Gemenne 2013). It is also important to mention that options regarding migration might be limited under climate change conditions, especially for poor people with limited resources (Gemenne 2015), forcing many people to become "trapped" within their place or community (Black et al. 2011b; Black & Collyer 2014).

This thesis explores both *in situ* adaptation measures, with a particular focus on on-farm agricultural adaptation practices, and *ex situ* adaptation responses involving both occupational and geographical mobility-related decisions of the rural households in Damietta. It is important to mention that the culture of migration is believed to be an essential factor that influences individuals' migration-related decisions (Cohen & Sirkeci 2011). Hence, this thesis takes account of households' cultures of migration as a critical component that could influence their current and potential migration decisions (discussed in Chapter 2, Section 2.5.2.2).

1.5 Research Gap

Globally, researchers have been able to identify the impacts of climate change upon the agricultural sector and predict future consequences across the different economic sectors and regions around the world. As part of that process, there has been a lot of work suggesting approaches for minimising or avoiding the various direct negative effects and associated broader consequences of climate change on agriculture. However, the degree to which these suggested approaches are consistent with what farmers understand and have to potential to implement has not been studied thoroughly for Egyptian society.

Studies regarding climate change in Egypt have focused more on modelling the ecological physical, and economic effects of climate change on the agricultural sector (e.g. see Eid, El-Marsafawy & Ouda 2007; El-Raey 2010; El-Ramady, El-Marsafawy & Lewis 2013; Hassanein, Elsayed &Khalil 2012; Morsy, El-Sayed & Ouda 2016; Zohry & Ouda 2016; Ahmed et al. 2021). However, few studies have integrated the human aspects of the effects of climate change and addressed the current vulnerabilities to rural society and adaptive practices on the micro-level (Kassem et al. 2019).

Some studies have addressed local awareness and perceptions of climate change with less focus on adaptive practices and their implications for future environmental and socioeconomic pressures (see, for example, Omar et al. 2015; Froehlich & Al-Saidi 2018; Hafez 2020). Some studies have addressed adaptive practices followed by rural farmers, although some were more concerned with *in situ* agricultural adaptive practices followed by farmers, with little to no emphasis on *ex situ* measures (Omar 2015; Kassem et al. 2019), while yet others focused on *ex situ* adaptation and decisions by rural households regarding migration and attempted to find links between environmental change and migration in Egypt (Warner et al. 2008; Afifi 2010; Adoho & Wodon 2014a). Although studies concerned with *ex situ* adaptive measures have found little evidence that environmental change causes migration in Egypt, only minor attention has been paid to possible reasons for immobility, particularly in relation to the local cultures of migration. Although the culture of migration is an important aspect that influences people's decisions about moving and the proportion of the population who might be trapped due to environmental change, it is under-researched in relation to Egyptian rural communities in relation to the effects of climate change, and development of that knowledge is one of the significant contributions of this thesis.

1.6 Research questions

The following questions will be addressed to achieve the objectives of the research:

- 1. What patterns of climate change have been experienced by rural households?
- 2. How do rural households perceive climate change risks and other risks associated with working in the agricultural sector?
- 3. Do rural households perceive additional future risks associated with climate change?
- 4. What kinds of adaptation strategies have been followed by farmers and their households to reduce the risks of climate change impacts?
- 5. What are the socio-economic, demographic and cultural characteristics that influence the capacity of farmers to adapt to the risk associated with climate change? and
- 6. Does the degree of dependency of a household on agriculture for a living influence its level of perception of and adaptation to climate change?

1.7 Organisation of the thesis

This thesis is organised into eight chapters, as shown in Table 1.1. The first chapter introduces the key arguments and the context for the research. Chapter 2 presents the literature review on agrarian transition, climate change, risk perceptions, and adaptation to climate variability. Chapter 3 outlines the research methods undertaken to generate empirical results, including the selection of data collection and analysis techniques. The justification for selecting the study area of the Egyptian delta is also discussed in this chapter. Chapters 4 to 6 discuss the results from the case study. Chapter 4 provides a general background to the livelihood conditions of the studied rural households, including their demographic characteristics and the ecological environment where they live.

Chapter 5 describes local perceptions of climate change patterns, impacts, and potential risks for agriculture. Chapter 6 discusses the adaptation strategies followed by rural households to reduce the risks of climate variability, including sudden shocks related to extreme weather events. Chapter 7 discusses the significant findings of the study and their implications, and then the conclusion is presented in Chapter 8.

In conclusion, this thesis focuses on perceptions of current and potential risks associated with climate change and other challenges associated with working in the agricultural sector in Egypt. It also explores the current adaptive practices of rural households, whether *in situ* or *ex situ*, and takes into account the local cultures of migration, particularly concerning the dominant socio-economic situations of farmer groups. Importantly, this thesis integrates the agrarian transition and acknowledges the differences in the vulnerabilities and adaptive capacities of rural households within the same agrarian system according to their reliance on agriculture as the source of their livelihood. The research discusses the implications of current adaptation practices within the broader context of current and future environmental, demographic, political and socio-economic pressures within Egypt and the MENA region.

Chapter Two: Review of Literature

2.1 Introduction

This chapter elaborates upon the current pressing risks and dimensions of vulnerability in the Middle East and North Africa (MENA) region under climate change conditions, looking at the current socio-economic, environmental, demographic and political issues prevailing in the region, with a particular focus on Egypt. It also presents conceptual descriptions of the key terms used in this study, as previously introduced in Chapter 1 (Section 1.4). It discusses the theoretical framework of the agrarian transition and addresses the nature and drivers of agrarian change in Egypt. It highlights some key policies that shaped the agricultural sector in Egypt, and the different pathways of agricultural transformation and the resultant class structure in the Egyptian rural areas. A discussion about the influence of risk perceptions on shaping the different responses to climate change follows. The chapter then presents a broad review of the key concepts related to vulnerability, adaptive capacities, and adaptation strategies to climate change, including in situ and ex situ measures within rural societies. The relevant literature on populations trapped within the context of climate change is also reviewed in relation to the factors that contribute to human immobility. The role of migration cultures is introduced, emphasising knowledge about their potential importance in guiding decisions about environmentally induced mobility.

2.2 Climate change in the MENA region: current challenges and future risks

As introduced in Chapter 1, the MENA region, including Egypt, is one of the areas especially vulnerable to climate change, particularly concerning the deteriorating circumstances within which the agricultural system currently functions. This following section will discuss in more detail the dimensions of the vulnerability of the MENA region, with a particular focus on Egypt, to the effects of climate change. It will also address the implications of climate change on national and regional food security, political instability and social unrest.

A 0.2°C warming has been observed in the MENA region over the period from 1961 to 1990, with a faster rate of warming since then (Waha et al. 2017). The Mediterranean coast, Algeria, Libya and large parts of Egypt are projected to face substantial warming

with regional warming of 3°C by the end of the century (Waha et al. 2017). A severe increase in the number of high-temperature extremes has also been recorded since the 1960s (Seneviratne et al. 2012), with an increase in the heatwave intensity index particularly in North Africa and the Eastern Mediterranean (Kuglitsch et al. 2010). This trend is expected to continue in the MENA region over time (Lelieveld et al. 2016; Waha et al. 2017; Varela, Rodríguez-Díaz & deCastro 2020; Ozturk, Saygili-Araci & Kurnaz 2021). Similarly, several local studies in Egypt have recorded a general trend in warming since the 1990s (Saber 2009; Met Office et al. 2011; Seyam 2011; Hereher 2016), with more extreme weather events (Met Office et al. 2011; Nashwan, Shahid &Abd Rahim 2018).

Drought has been a "normal" characteristic of much of the Middle East semi-arid and arid climates (UNDP 2011), but studies have reported an enhanced drying trend over the region since the 1960s due to climate change (Hoerling et al. 2011; Sousa et al. 2011; Lelieveld et al. 2012; Donat et al. 2014; Lange 2019), which is projected to cause an average decline of 20 percent in precipitation throughout the 21st century in the MENA region (Conway & Hulme 1996; Arnell 1999; Sanchez et al. 2004; Milly, Dunne & Vecchia 2005; Suppan et al. 2008; Evans 2009; Ozturk et al. 2018). Such warmer and drier climatic conditions would increase the incidence of severe prolonged drought, even though there may be simultaneously more extreme and variable rainfall events, raising the likelihood of both flooding and desertification (Sowers, Vengosh & Weinthal 2011; Lelieveld et al. 2012; Ozturk et al. 2015; Ozturk et al. 2018). Egypt, in particular, has been suffering from more severe hydrologic droughts, which entail a decline in surface and subsurface water as a result of climate change (Hameed, Ahmadalipour & Moradkhani 2020).

Coastal areas are particularly vulnerable to the negative effects of climate change (IPCC 2007b; El-Raey 2010; Waha et al. 2017). In 2010, the population of MENA's coastal cities was roughly 60 million and is expected to reach 100 million by 2030 (World Bank 2011). Direct inundation resulting from slow-onset, rising sea-levels, increased erosion, saltwater intrusion into coastal aquifers, floods, and damage caused by storms and storm surges are considered the key effects of climate change projected for the coastal zones (El-Raey 2010; Hunt & Watkiss 2011; Brecht et al. 2012). Egypt, Libya, Tunisia, and the United Arab Emirates are among the countries at extreme risk of rising sea-levels worldwide based upon the affected percentage of population and land (Waha et al. 2017).

Egypt, in particular, is projected to be the most severely affected country in the MENA area (Kreimer, Arnold & Carlin 2003; Dasgupta et al. 2007; El-Raey 2010; EEAA 2016; Elshinnawy & Almaliki 2021). It is estimated that 13 percent of the Egyptian agricultural area will be lost (compared to 1.15 percent of the region's agricultural lands) with a rise in sea-level of 1m, affecting 10 percent of Egypt's population and causing a potential loss of about 6 percent of the national GDP (Figures 2.1 and 2.2) (Dasgupta et al. 2007; Dasgupta et al. 2009).

Figure 2.1: Projected percentage of Middle East and North Africa national populations affected by sea-level rise impact



Source: Dasgupta et al. (2007)

Figure 2.2: Projected impact of sea-level rise impact on the Middle East and North Africa national GDP



Source: Dasgupta et al. (2007)

Land subsidence will further aggravate the effects of rising sea levels in Egypt's Nile Delta, especially in the eastern part of the delta where Damietta, the area under current investigation, is located (Frihy & El-Sayed 2013; Wöppelmann et al. 2013). The rising sea-level is expected to have other serious effects besides direct inundation in terms of the salinisation of groundwater through saltwater intrusion, rising water tables, and reduction in soil drainage and soil quality (Werner & Simmons 2009; Hunt & Watkiss 2011; Elshinnwy & Almaliki 2021; Omar, Moussa & Hinkelmann 2021). Saltwater intrusion into coastal aquifers has already been a problem in some MENA countries, including Egypt (Weinthal et al. 2005; Kouzana, Mammou & Felfoul 2009; WWAP 2012; Agoubi 2021), as it is also associated with the over-extraction of water for irrigation, a lack of replenishment by freshwater, and increased anthropogenic contamination (Bouchaou et al. 2008; Waha et al. 2017). Sea-level rise, hence, is only aggravating the existing critical condition of groundwater resources (Vengosh 2003; Weinthal et al. 2005; Kouzana, Mammou & Felfoul 2009).

Another predictable effect of climate change is a higher intensity of storms (Knutson et al. 2010). More intense storms combined with rising sea levels will likely result in more intense storm surges, which would have a devastating effect on coastal areas in the eastern Mediterranean (Dasgupta et al. 2009). For example, Dasgupta et al. (2011) suggested that about 2.7 million people in the Egyptian city of Alexandria alone could be affected by storm surges in the future. The following subsections will discuss the critical environmental, socio-economic, demographic and political pressures already prevalent in the region, with a particular focus on Egypt.

2.2.1 Water

The MENA region has the greatest water scarcity on the globe (Droogers et al. 2012; Terink, Immerzeel & Doogers 2013; Alboghdady & El-Hendawy 2016), with the ten countries suffering most water stress on Earth concentrated in the region (as shown in Figure 2.3). Water stress in the MENA region is already evident, with severe challenges in terms of over-extraction of surface and groundwaters, demographic growth, rising sea levels, water pollution, and soil salinisation (Koocheki 2010; Sowers, Vengosh & Weinthal 2011). Climate change will very probably further profoundly affect the quality and quantity of water resources in the region with regards to all of these issues (Conway & Hulme 1996; Sanchez et al. 2004; Milly, Dunne & Vecchia 2005; Suppan et al. 2008; Evans 2009; Sowers, Vengosh & Weinthal 2011).



Figure 2.3: Water stress ration in the Middle East and North Africa region

As mentioned earlier, North Africa is already suffering from the salinisation of coastal aquifers due to seawater intrusion and over-extraction of water from aquifers (Lofgren & Richards 2003; Koocheki 2010). If less water flows down the Nile, its ability to prevent the intrusion of saline seawater will be compromised due to less fresh water being available to recharge these aquifers, exacerbating the problem caused by rising sea levels, and placing the water resources in the region under even more pressure (Vengosh 2003; Weinthal et al. 2005). Water pollution from agricultural, urban and industrial wastes is another escalating challenge in the MENA region affecting the quality of water resources (Lofgren & Richards 2003).

Countries in the MENA region are generally classified into three groups based upon water resources and availability, as suggested by Terink, Immerzeel & Doogers (2013). The first group includes those countries that have sufficient quantities of renewable water; however, the range of annual and within-country variability still generates considerable challenges. Such countries with these characteristics include Iran, Morocco, Lebanon and Tunisia. The second group includes countries that depend primarily on non-renewable groundwater and desalinisation of seawater, such as Yemen, Libya, Jordan, and other Gulf countries that have limited natural sources of renewable water. The third group of

Source: Oxford Analytica (2015)
MENA countries are dependent on the inflow of transboundary exotic rivers, like the Nile in Egypt and the Tigris and the Euphrates in Syria and Iraq. This latter group of countries is of crucial concern as the inflow of rivers could be altered or controlled by other countries, in a manner similar to that which led to the recent conflict between Egypt, Sudan and Ethiopia that is discussed below.

2.2.1.1 Dimensions of water crisis in Egypt

Egypt has one of the highest population growth rates in the Middle East, with a population exceeding 100 million inhabitants (CAPMAS 2020a), which has nearly doubled over the past 30 years. Statistics show that the country's per capita freshwater resources per year have fallen from 2,000 m³ in 1958 to less than 700 m³ in 2013, which is well under the 1,000 m³ threshold believed essential by the United Nations to provide an adequate amount of water for drinking and agriculture (Figure 2.4) (FAO 2017). In other words, Egypt is already below the water "poverty line", with an average of 570 m³ per person in 2019 due to massive population growth, and is heading towards absolute water scarcity with less than 500 m³ per capita in 2025 (Ebrahim 2019).

Population growth increases water stress by increasing the water requirements for both domestic and agricultural consumption to meet higher food demands (Dakkak 2020). Water demand is also expected to rise owing to the government's plans to expand land reclamation projects, industrial activities and urban centres into the desert (Swain 2011; El Bedawy 2014). Several other factors are likely to further reduce the availability of water in Egypt, including inefficient irrigation systems, pollution and the lack of a well-maintained water-delivery infrastructure (Soussa 2010; Karajeh et al. 2011; El Bedawy 2014; Power 2014; Dakkak 2020). It is estimated that Egypt's irrigation system operates at only 50 percent efficiency (Karajeh et al. 2011), placing the country in the bottom 10 percent of MENA region (Soussa 2010; Power 2014).



Figure 2.4: Renewable water resources per capita in Egypt, 1958–1962 to 2013–2017

Source: Adapted from FAO AQUASTAT (2017)

Agricultural pesticides, poor sewerage, and municipal and industrial discharge all contribute to water pollution in Egypt (Abdel-Shafy & Aly 2002; Barakat 2004; Gaballah et al. 2005; Negm, Saaverda & El-Adwy 2017). Persistent water pollution further aggravates the Egyptian water crisis and damages agricultural production by decreasing the quantity of suitable clean water available (Abdelaal & Thilmany 2019). As an illustration, it was estimated that the annual amount of waste water discharged into the Nile, both directly or indirectly through the agricultural drainage systems reached about 18.9 billion m³ in 2016 (EEAA 2016). About 25 percent of the total volume of sanitary waste in Egypt is discharged into irrigation and drainage canals without any previous treatment, and the remaining 75 percent is released into freshwater canals with only primary treatment (Mohamed 2017). In addition, the huge investments in land reclamation projects not only cause the depletion and pollution of groundwater but also raise the question of whether Egypt has enough water to sustain new lands and any further reclamation (Kassim et al. 2018).

The history of agricultural policies has contributed to the critical water situation in Egypt. In 1984, the government declared water as a public good delivered to farms at no cost, so consequently farmers were discouraged from investing in water-conserving irrigation systems and still use traditional flood irrigation techniques (Kassim et al. 2018). With the implementation of economic reforms and especially a range of liberalisation policies, farmers were able to cultivate freely whatever they perceived as being the more economically profitable commodities. As a result, liberalisation increased the cultivation of rice, a water-demanding crop, and thereby affected the capability of the Ministry of Water Resources and Irrigation to uphold and fulfil the water delivery system (Kassim et al. 2018). Consequently, in the 2000s, the government reduced the area permitted for rice cultivation in the Delta by about half to save irrigation water (Ibrahim & Ibrahim 2003). Then in 2016, the government limited rice cultivation to an area of 2 million *feddans* (1 ha = 2.381 *feddans*), and after a further year the area was decreased to 1.7 million *feddans*, distributed over eight governorates in the Delta (Kassem et al. 2019). Farmers who grew rice outside the pre-defined boundaries would be fined and potentially imprisoned (Kassem et al. 2019). Subsequently in 2018, the Egyptian government announced its intention to import more rice to support such an arrangement (Reuters 2018).

Rice cultivation is essential to wash away the saline soils found in the north of the Delta through extensive leaching (Ahmed 1998; Mohamed 2017; El Nour 2018). As a result, reducing rice cultivation in the Nile Delta could lead to the loss of one-third of agricultural lands in the Delta due to salinity (El Nour 2018). Rice is considered a subsistence crop for rural dwellers, and is a main cash crop with high economic returns (Ahmed 1998; El Nour 2018), so limiting rice cultivation in the Delta will have a dramatic impact on the livelihood of millions of farmers and other labourers who work in the hundreds of rice factories across the region (El Nour 2018).

In addition to the broad range of internal pressures introduced above, Egypt is fundamentally dependent on external Nile water flowing from other African countries (Terink, Immerzeel & Doogers 2013) – a situation that increases Egypt's risk of losing access to its full-allocation of Nile waters. Throughout most of its history, Egypt has reserved its power over the control of the Nile, as the dominant user of Nile water (Whittington, Waterbury & Jeuland 014; Pemunta et al. 2021). The 1929 Anglo-Egyptian Treaty between Egypt and Great Britain, as the colonial power in eastern African countries, including Uganda, Sudan, Kenya, Tanzania and Ethiopia (Abdulrahman 2018). The treaty gave Egypt the right to monitor the use of the Nile's waters by upstream countries and veto any projects that would threaten Egyptian interests (Lumumba 2007; Ferede & Abebe 2014). Nile basin countries have increasingly seen this arrangement as unfair, ignoring their water rights in favour of Egypt's interests (Tesfaye 2013). The 1959 bilateral agreement between Egypt and Sudan consolidated their control of the river waters and allocated the entire flow of the river between the two countries, but again did not include any of the other riparian states, even Ethiopia, from which 80 percent of water originated (Lumumba 2007; Tesfaye 2013; Whittington, Waterbury & Jeuland 2014; Abdulrahman 2018). Until 2011 there had been no serious threat to Egypt's historical rights and apart from negotiating with Sudan, the *status quo* prevailed in the Nile basin (Whittington, Waterbury & Jeuland 2014). However, Ethiopia benefited from the turmoil of 2011 uprisings in Egypt and announced the construction of the Grand Renaissance Dam (GRD) (Whittington, Waterbury & Jeuland 2014). In July 2020, Ethiopia announced the completion of the initial filling of the GRD with 4.9 billion cubic meters of water in the dam's reservoir so far (Salam 2020).

The GRD could affect Egypt adversely in three ways. First, Egypt might not be able to withdraw enough water to supply all its agricultural needs if the filling of the GDR reservoir was to occur during a sequence of years in which the Blue Nile flowed and the Aswan High Dam reservoir itself was low. Secondly, after the completion of the GRD, Egypt might run deficits of water during a sequence of drought years if the operation of the GRD was not coordinated with the Aswan High Dam. Finally, Egypt could be negatively affected by withdrawal of water for irrigation taking place in upstream countries (Whittington, Waterbury & Jeuland 2014; El-Nashar & Elyamany 2018). Power (2014) claims that the construction of the GRD would lead to a loss of 3 billion cubic meters of water each year due to evaporation disturbing the flow of water downstream to Egypt. Such a situation would also affect the quality of downstream Nile waters through increasing salinity (Ramadan et al. 2013).

Future changes in precipitation in the Ethiopian Highlands due to climate change could also affect Nile river water flow for Egypt and Sudan (Conway 2005). Higher temperatures coupled with climate change already entail higher rates of evaporation of Lake Nasser, the large reservoir behind the Egyptian Aswan High Dam, and of the White Nile river in southern Sudan reducing the annual river discharge (Sowers, Vengosh & Weinthal 2011). Climate change can potentially change water supply and demand patterns in the basin, and hence, the allocation of the already scarce Nile water resources may become a serious security challenge in the future (Swain 2011; Pemunta et al. 2021). As a result, it is argued that the current governance of countries' access to Nile water could trigger an interstate conflict, which could destabilise the whole North African region if the situation is not managed well (Nunzio 2013; Pemunta et al. 2021).

Climate change is also projected to affect water availability further in Egypt through the increasing demand for irrigation water for many crops (El-Marsafawy 2008; Elshamy, Sayed & Badawy 2009; Kim & Kaluarachchi 2009; Ahmed at al. 2021). As already mentioned, there is the potential for rising sea-levels to threaten low-lying Nile Delta lands in particular with increased saltwater intrusion causing detrimental effects to freshwater supplies and agricultural productivity (McCarl et al. 2015; Elshinnawy & Almaliki 2021).

In brief, water security in Egypt is threatened by water scarcity regionally; however, with Egypt specifically, upstream Nile projects, population growth, and risks associated with resource extraction, governance and climate change are all expected to intensify challenges with water scarcity (Tellioglu & Konandreas 2017). Given these escalating water-related issues, Egypt could face mounting food insecurity and unemployment, which, in turn, could spark grievances against the state or even lead to political instability and conflict in the Nile basin region (Climate Diplomacy n.d-a; Pemunta et al. 2021).

2.2.2 Agriculture and food security

The agriculture and food security situation in the MENA region is very challenging. Current environmental systems within the region limit agricultural productivity due to the scarcity of resources (land and water) and the ongoing degradation of agricultural lands caused by unsustainable farming practices and erosion (OECD-FAO 2018). Although the agricultural sector does not contribute significantly to the national economies of many countries in the MENA region, it still employs a significant percentage of the labour force in most places (Sowers, Vengosh & Weinthal 2011; Woertz 2017). As discussed, the lack of water availability is considered the critical constraint in the agricultural sector in the MENA region (Koocheki 2010), as the agricultural sector is already the largest consumer of water in the area, consuming an average of 80 percent of the water budget, leaving little for urban or other commercial uses (Sowers, Vengosh & Weinthal 2011; Woertz 2017). With the water scarcity in the region, it is expected that less water would be allocated to the agricultural sector in the future, as has already happened in Yemen, Jordan and Libya, because the economic returns from the use of water and the political power

within the region's states are increasingly concentrated in urban areas (Sowers, Vengosh & Weinthal 2011).

Population growth, together with water stresses on agricultural production, have increased dependence on imported food within the MENA region (Waha et al. 2017). For instance, Egypt now imports about 40 percent of its grain supply (Koocheki 2010; Nigatu & Motamed 2015). Economic and population growth have changed dietary preferences and raised food consumption and demand across the region, making the MENA region a key influence over world trade in agricultural commodities (Nigatu & Motamed 2015). The reliance on food imports is likely to continue and is predicted to increase over time as current trends continue (Woertz 2017). This high reliance on food imports makes the region vulnerable to fluctuations in global markets and food production in and from other world regions (Woertz 2017; Waha et al. 2017; OECD-FAO 2018). For instance, in the years 2007-2008 and 2010-2011, Egypt experienced a massive spike in global food prices as a result of trading conditions and adverse weather events that affected several major food-exporting countries (Sternberg 2013; Climate Diplomacy n.d-a). Furthermore, ongoing geopolitical conflicts and the potential for many new forms of economic or environmental instability generate considerable uncertainties regarding both food demand in and supply to the region (OECD-FAO 2018).

The expected degradation in the quality of irrigation water due to climate or non-climate stressors would decrease the quality of agricultural lands within the region (GLOWA 2009). Another challenging problem that faces the sustainable use of agricultural lands is the unbalanced use of fertilizers and chemical biocides in the region (Koohfkan 2001; Koocheki 2010). The excessive use of inorganic fertilizers and chemicals, which might further increase with climate change as farmers respond to more challenging growing conditions, has already caused environmental pollution, pest resistance to chemicals and weakening responses to fertilizers (Koohfkan 2001; Koocheki 2010). Chemical pesticides, for example, have been extensively used in the Egyptian agriculture sector since the 1950s with a total of about 1 million metric tons of pesticides being used over the intervening 50 years (Mansour 2008). This rate is expected to increase, given the growing population that requires the production of sufficient food.

The lack of government control measures for monitoring the use of chemical pesticides together with a lack of awareness about the acceptable agricultural practices needed for

handling and using pesticides contribute primarily to the health and environmental hazards caused by them in Egypt, as well as in many developing countries around the world (Mansour 2004; Mansour 2008). Sheahan, Barrett & Goldvale (2017) found that the application of pesticides was associated with increased health costs and time lost from work due to illness, although it correlated with high agricultural output value. Their findings highlight that pesticides were becoming more extensively used in several African countries.

2.2.2.1 Agriculture and food security in Egypt

Agriculture represents one of the most important sectors driving the Egyptian economy. It contributes to the overall food needs of the country, provides raw material for the domestic market, and generates income for agricultural labour, as well as labour for related businesses such as wholesalers, exporters, and transporters of agricultural produce (Tellioglu & Konandreas 2017). In 2019, the sector contributed roughly 11 percent of the national GDP, a decline of about 7 percent since the mid-1990s (World Bank 2020). Employment in agriculture has declined less dramatically with the sector still considered as being labour-intensive, employing 23.8 percent of the total national labour force as in 2019 (Figure 2.5) (World Bank 2020). Moreover, agriculture provides the livelihood for more than 57 percent of the population living in rural areas of Egypt in 2018 (IFAD 2019).



Figure 2.5: The role of the agricultural sector in the Egyptian economy, 1995–2019

Recently, Egypt has become a growing importer of basic foodstuffs, and the world's largest importer of wheat reaching more than 10 million tons in 2018 (Figure 2.6)

Source: Adapted from World Bank (2020)

(Koocheki 2010; Nigatu & Motamed 2015; Tellioglu & Konandreas 2017; FAO 2020). This can be attributed to not only the Camp David agreement (further discussed in Section 2.3.1.2) but also, the weakening agricultural reforms, population growth, and severe water stress; hence, the low capacity to produce domestically (Bush 2007).



Figure 2.6: Egypt imports from wheat, 1994–2018

Source: Adapted from FAOSTAT (2018)

Climate change scenarios in Egypt show that warmer and drier weather is expected to increase the water requirements of many crops and lower their productivity particularly with wheat, rice and other grain crops by an average of 18 percent by 2050 (Abou-Hadid 2006; Eid, El-Marsafawy & Ouda 2007; El-Marsafawy 2008; El-Ramady, El-Marsafawy & Lewis 2013; Ahmed et al. 2021). Production of livestock will also be affected by climate change, through reducing the quality and quantity of feed and water available, changes in livestock diseases and the escalating heat stress (Thornton et al. 2009; Ahmed et al. 2021). Moreover, Ahmed et al. (2021) predict that many Egyptian agricultural labourers could lose their jobs in the near future due to climate change. For illustration, it is estimated that 3.5 million landholders will be affected by temperature increase in the Nile Delta by 2030 (Gouda 2020). In additional, roughly 2090 landholders and their households in four governorates in Damietta by 2030 (Gouda 2020). A total of 153 agricultural labourers are predicted to lose their jobs by 2030 due to sea-level rise in the Northeast Delta, in addition to the loss of 3162 direct and indirect jobs (Gouda 2020).

2.2.2.1.1 Land scarcity and fragmentation

Arable land is scarce in Egypt, and is highly fragmented into small plots, limiting the capacity to develop efficiency in production and sustainable agricultural development in general (Tellioglu & Konandreas 2017). In the old lands of the deltaic and riverine regions of the Nile, the overall average land area per holding decreased from 6.3 *feddans* (1 *feddan* = 0.42ha) in 1950 to 3.2 *feddans* in 1960, and further to 2.1 *feddans* according to the 1999–2000 agricultural census (MALR 2009). In 2013, the average land area per holding per rural household was 1.6 *feddans*, or only 0.67ha (Nawar & Abdel-Hakim 2013). Land reform laws have exacerbated the fragmentation of agricultural landholding in the country (discussed in Section 2.2.1.1), made worse by rapid population growth and the practising of Islamic inheritance laws (Kassim et al. 2018).

Inheritance laws applied in Muslim countries facilitate or demand the subdivision of holdings among all heirs (Golmohammadi 2020). Consequently, when inheriting land, this land becomes a joint property belonging to several heirs or it is divided into predefined parts, which leads to fragmentation of the land (Golmohammadi 2020). What exacerbates the problem of land fragmentation in Egypt is the current tax law of agricultural lands. As illustration, farmers who own less than 2 *feddans* are exempt from paying tax on their lands, which discourages the sale and consolidation of farmland (Tellioglu & Konandreas 2017).

Land fragmentation obstructs the conservation of natural resources due to preventing collective crop rotations and the organisation of agricultural production services (Aboulnaga et al. 2017). Furthermore, it is often economically unfeasible to introduce extension and technology to each small farming unit (Tellioglu & Konandreas 2017). Consequently, land fragmentation has adverse effects on agricultural production, food security and efficiency in the use of natural resources, as well as on the living conditions of small landholders and their households (Abouelnaga et al. 2017). For example, a recent study in Iran found that land fragmentation and small holdings were considered main barriers for utilising modern technologies, increasing agricultural productivity and achieving food security and self-sufficiency (Golmohammadi 2020).

The small and fragmented holdings are usually held onto tightly by individual farmers who have limited resources and hence, are incapable of managing their farms in an efficient manner, which in turn puts them at high risk from environmental change and economic fluctuations (EEAA 2010). To illustrate this point, evidence from Egypt shows that small landholder farmers are often characterised by low literacy levels, low economic status and modest standard of living, high dependence on the farm as the main source of income and weak decision making and political influence (Aboulnaga et al. 2017).

2.2.2.1.2 Land degradation

Many factors cause land degradation in Egypt, particularly in the Nile Delta, including salinity, alkalinity, nutrient depletion, waterlogging, pollution, seawater intrusion and urban encroachment (Mohamed 2017). Desertification is also one of the challenges that affects large swathes of land in the Eastern and Western delta, where approximately 800,000ha have been affected by the active encroachment of desert sands and dunes (Warner 2010). Soil salinity is also considered a key challenge in the Nile Delta, especially in the northern part of the Nile Delta where Damietta is located, due to poor quality irrigation water, shallow water tables and seawater intrusions (Mohamed 2017). Due to the lack of sufficient freshwater for irrigation, farmers have no choice but to reuse agricultural wastewater, especially those whose farms are located at the tail of the water canal (Mohamed 2017), a situation that causes further deterioration of soil properties. Urban encroachment is another critical problem that causes land degradation and threatens agricultural land available in the Nile Delta of Egypt (Negm, Saaverda & El-Adwy 2017). It is estimated that 2,536.3km² of agricultural lands were lost to urbanisation over the period from 1984 to 2006, which represents roughly 7.5 percent of the total agricultural area in the Nile Delta as in 2002 (Shalaby 2012).

2.2.3 Demographic pressure, unemployment and security

The MENA region hosts one of the fastest growing populations in the globe. Population numbers have more than quadrupled in four decades, raising from 138 million in 1970 and reaching 456 million (World Bank 2019b). According to the medium-fertility scenario of the United Nations, the population of the region is expected to reach 600 million by 2050, and 845 million in 2100. Around 50 percent of the population is considered to be less than 25 years of age (Forouheshfar, El Mekkaoui & d'Albis 2020), making the region one of the most youthful areas in the globe (Tür 2018). The large proportion of youth in the population, known as the "youth bulge", is a central fact conditioning the region's social, economic, political and cultural development (UNDP 2016).

Demographic pressures are a leading cause of high youth unemployment rates in the region (UNDP 2016; Tür 2018; Al-Shammari & Willoughby 2019; Forouheshfar, El Mekkaoui & d'Albis 2020). The labour market of the region is characterised by a continuing job crisis particularly among youth and it has been unable to provide sufficient job opportunities to absorb the new entrants (Dimova, Elder & Stephan 2016). Youth unemployment rates in most MENA countries are double the global average (World Bank 2019), with higher rates of unemployment found amongst educated youth (Barsoum, Ramadan & Mostafa 2014; Assaad & Krafft 2015). For example, unemployment in Egypt reached 11.2 percent in 2018, with 75 percent of the unemployed aged 15-29 years (IFAD 2019). However, youth unemployment rates in Egypt are rather misleading because they often underestimate the number of poor youths who are forced to work at whatever job they find as they cannot afford to be unemployed (Bremer 2018). They may, thus, be working at lower pay (or in some cases at no pay at all in a family business for example), fewer hours, and with less security (Bremer 2018). Underemployment is, hence, another problem that young people are increasingly facing today in the region (Sika 2019; Assaad Krafft & Yassin 2020). In order to stabilise youth unemployment, the region needs to generate more than 60 million jobs in the upcoming decade to absorb the large number of job entrants (UNDP 2016), a challenging situation when one considers the massive number of young workers expected to enter the labour market (Forouheshfar, El Mekkaoui & d'Albis 2020).

The critical situation of unemployment in the region goes beyond the mismatch between labour supply and demand. The educational attainment is argued to be inconsistent with the labour market's demand for skills, owing to a deficient education system prevalent in the region (Barsoum, Ramadan & Mostafa 2014; Dimova & Stephan 2020). Inequality of opportunity is also a contributing cause for the malfunctioning of labour markets in the MENA region (Dimova & Stephan 2020). Educational attainment depends substantially on parents' background and community characteristics (Salehi-Isfahani, Hassine & Assaad 2014). Consequently, young people from advantaged backgrounds are found to have better access to higher-skilled job opportunities when they become available (Dimova & Stephan 2020). The presence of social "connections" in securing job opportunities is also another dimension of inequality (World Bank 2014; Sika 2019; Assaad & Krafft 2020; Bremer 2020). According to Sika (2019), unemployment in Egypt

is mainly associated with highly educated young men from low-income families who lack the social connections to boost their chances of employment.

Moreover, recent evidence from Egypt shows that social class plays a role in determining the success or the failure of the transition from school to work (Assaad & Krafft 2020). Assaad & Krafft (2020, p.1) suggest "whether youth successfully make transitions to formal jobs, embark on such transitions and fail, or pursue more traditional careers in informal employment or family businesses or farms depends on a complex and changing interaction between their own educational attainment and the resources of their families".

The high level of youth unemployment in the region is consistent with the global trend of premature deindustrialisation that hinders the capacity of developing countries to create jobs for their labour market entrants (Rodrik 2016, 2017). Almost all the labour markets in MENA countries are characterised by dominant service sectors, which have historically been less successful in developing job opportunities than a typical growing manufacturing sector (Sika 2016; Dimova & Stephan 2020). Moreover, Egypt saw limited gains in productivity from sectoral labour reallocation over the past three decades (Morsy & Levy 2020). Not only did labour fail to shift from agriculture towards high-productivity manufacturing and private sectors, but it stayed limited to low-value-added activities such as the public sector and construction, locking Egypt into a "low value trap" (Morsy & Levy 2020). On the contrary, many emerging economies have improved their per capita income and created decent job opportunities by rapidly reallocating labour from less to more productive sectors as has happened in China for example (Roncolato & Kucera 2014).

The high percentage of young people reaching over 40 percent of the total population marks the problem of school-to-work transition and its patterns of vital priority for both the economic policy discussion in Egypt and the political stability of the country (Selwaness & Roushdy 2019). The country has not completed its demographic transition with no definitive reduction in birth rates (Bremer 2018). Accordingly, the number of population under 14 will continue to rise quickly, resulting in a second youth bulge double the size of the youth cohort at the time of 2011 mass demonstrations (Bremer 2018).

Over the last decade, several countries in the MENA region have already experienced the political instability that is still being witnessed in recent times, as in Syria, Yemen, and

Libya, and high youth unemployment rates are argued to be one of the chief causes of political instability and important trigger for the Arab Spring in the region (World Bank 2014; Heyne & Gebel 2016; Al-Shammari & Willoughby 2019; Selwaness & Roushdy 2019; Forouheshfar, El Mekkaoui & d'Albis 2020).

Unfortunately, the Arab Spring has not led to any real, significant improvements in employment opportunities that the youth might have hoped for in Egypt (Assaad & Krafft 2014; Heyne & Gebel 2016; Selwaness & Roushdy 2019). The political instability that came after the revolutions was associated with economic stagnation and a deterioration in the quality of jobs with mounting informality and irregularity of employment (Amer 2015; Heyne & Gebel 2016; Paciello & Pioppi 2020). The Egyptian government in particular, has showed a lack of commitment to the large-scale growth and real structural change that would facilitate a transformation in employment opportunities in the country (Bremer 2018; Morsy & Levy 2020).

Climate change has also been a driver of political unrest in the region, as evidence shows that it has contributed to political instability in 16 countries and conflicts in 15 countries of the MENA region from data analysed over the period 1985 to 2016 (Sofuoğlu & Ay 2020). Arguably, the experiences in Syria over the last decade are an actual example of the effects of climate change on human mobility. In this case, the severe drought that took place in 2006–2007 caused agricultural failures that affected millions of farmers and herders, and their communities, leading to a massive increase in the rates of migration to urban areas in search of better income prospects (Gleick 2014; Kelley et al. 2015; Suter 2017). Evidence suggest that the uncontrolled rural-urban migration already aggravates underemployment and unemployment conditions in urban areas (Chaudhuri 2000; Herrmann & Svarin 2009; Zenou 2011). Consequently, the internal mismanaged displacement and the few employment opportunities in the major cities added more pressure on the socio-economic tensions already existing in the area, and helped to trigger the mass protests of 2011 and subsequent conflict (De Châtel 2014; Gleick 2014; Kelley et al. 2015; Suter 2015; Suter 2017).

This story is no different from what is currently happening in Yemen where water is scarce and its unsustainable use together with climate change has augmented its depletion at a faster pace (Glass 2010; Sowers & Weinthal 2010; Suter 2017). With population growth and higher rates of urbanisation, water became even more scarce and unaffordable

for most of the poor population, aggravating tribal conflicts, the rural–urban gap, and violence over the control of water in several areas (Suter 2017). Recently, the control over limited water is steadily becoming a war tactic employed by both sides of the conflict, the Houthi and Saudi-led coalition (Suter 2017). Both factions have been blocking humanitarian aid deliveries of food and water to the other side (Human Rights Watch 2020) – a situation that is preventing millions of Yemenis from accessing adequate and safe drinking water (UNOCHA 2019).

Evidence shows that income inequality, unemployment, and poverty were the main triggers of the Egyptian uprising in 2011 (Verme et al. 2014; Ianchovichina, Mottaghi & Devarajan 2015). However, there is also a growing argument that the implications of climate change and environmental stress on food security were a hidden, indirect factors for the revolution in Egypt (Null & Prebble 2013; Sternberg 2013; Ayeb & Bush 2019). In 2010 for instance, drought and heatwaves affected the world's largest producers of wheat, such as Russia, Ukraine, Australia and China, playing an essential role in the rapid increases in global food prices (Sternberg 2013). Despite high subsidies, especially on bread, food prices in Egypt rose sharply in 2011, significantly affecting food security and living standards of many people across the country (Ghoneim 2012). Several commentators have argued, the increasing food process played a role in sparking the Egyptian uprisings in 2011 (Grossman-Cohen 2011; Null & Prebble 2013; Sternberg 2013; Maystadt, Trinh Tan & Breisinger 2014).

The Egyptian government is still at risk of a severe crisis of legitimacy due to declining agriculture caused by water scarcity (Climate Diplomacy n.d-b; Power 2014). Dwindling water resources due to both climatic and non-climatic factors may further intensify present grievances related to deficient water management infrastructure, poor management of the Egyptian water sector, as well as unequal distribution of water (Cunningham 2012). Recent years have already witnessed several protests in response to water shortages, pollution and water-intensive land reclamation projects in Egypt (IRIN 2010; Swain 2011; Pacific Institute 2020). For illustration, in 2007 and 2008 thousands of people staged demonstrations in the governorates of Fayyoum, Daqahliya and Ismailia to protest against severe shortages in irrigation and drinking water (OOSKAnews Correspondent 2008). Several protests were recorded in the summer of 2012 in which rural protestors blocked highways in Menoufiya and Fayyoum over a shortage in irrigation water (OOSKAnews Corrspondent 2012a). Another incident in Beni Sueif

governorate, resulted in one farmer being killed while others were injured during a conflict over irrigation water (OOSKAnews Correspondent 2012a). Water contamination was the basis for a protest by Egyptian villagers in Menoufiya in 2012 (OOSKAnews Correspondent 2012b), while in 2015, residents in an urban area in Giza protested against ongoing water cuts and police using tear gas had to be deployed (Hamama & Charbel 2015; Local Press Report 2015). Such grievances could intensify with further depletion of the available water resources. Ultimately, water shortage and political instability in Egypt may affect the entire Nile basin, either unintentionally due to Egyptian unrest or as a deliberate course of action to blame upstream countries (Climate Diplomacy n.d-b).

Egypt has witnessed a series of riots and protests in rural areas over the past century. In 1919, there was one of the biggest small farmers' revolts, not only against the British occupation, but also against the big landowners who were seen to have been supported by, and in turn supported, the British regime (Goldberg 2009; Shaarawy 2012; El Nour 2015). Similarly, the 1952 revolution was preceded by unrest between 1944 and 1951 due to the increase in rental prices for agricultural land by big landowners, low wages and heavy taxes (Berque 1967 cited by El Nour 2015). As a result, one of the main actions that took place after the 1952 revolution was the restructuring of the Egyptian agrarian system in an attempt to put an end to the power of big landowners (Abdel-Fadil 1975; Bush 2007).

However, in the early 1970s, a counter-revolution to these reforms has been ongoing in rural Egyptian villages (de Lilles 2020). Policies promoted the gradual elimination of subsidies on agricultural inputs, the liberalisation of markets for seeds and pesticides, privatisation of irrigation, a demolished role of agricultural cooperatives, and a massive shift towards state-led reclamation actions aimed at developing large-scale agricultural projects simultaneously accompanied with neglect in investment in smallholder agriculture, as further discussed in Section 2.2.1.2 (Bush 2007; El Nour 2015). In 1977, Egypt witnessed widespread protests driven by the threat of reducing food subsidies and increasing bread prices (Bush 2007). The 1992 and subsequent land tenancy reform laws were considered the turning point in agricultural policies, sparking violent movements and protests in the Egyptian countryside that increased after the implementation of the law in 1997 (Bush 2007; Bush 2011; El Naggar 2012; Keshk 2012; El Nour 2015). The tenancy law caused the formation of "networks of resistance" among agrarian communities, solidarity, activist and political organisations (Bush 2011). Farmers

continued to protest against the change in tenancy law after 2000, and by 2010 there were estimated 2,000 arrests, more than 1,500 injuries and 200 deaths resulting from land disputes and clashes with farmers (Bush & Martiniello 2017). These farmers' protests drove successive waves of protests by workers and employees in the cities, eventually leading to the 2011 revolution (Sharaawy 2012; El Nour 2015).

Protests by members of the working-class and farmers in January 2011, were therefore provoked by general frustration at the high prices of food (Bush & Martiniello 2017). Therefore, it is claimed that the 2011 revolution in Egypt was an outcome of a decade of widespread protests at all levels of society (de Lellis 2019); however, the participation of rural areas in the revolution was, significantly, often outside the focus of the media (Abu-Lughod 2012). The trouble in rural communities continued after the revolution with a total of 158 farmers protesting against shortages in resources, in particular, irrigation water and fertilizers in 2012 (Land Centre for Human Rights 2012 cited by El Nour 2015). The regime was resilient against the forces of change and was largely able to restore itself after the 2011 revolution (Gervasio & Manduchi 2020). The agrarian system also remained the same, including its dependency on international markets to provide its food needs rather than striving for self-sufficiency (Breisinger et al. 2012).

In conclusion, evidence supports the view that climate change generates, accelerates and deepens current instabilities in the countries of the MENA region (Sofuoğlu & Ay 2020). Climate change could have huge negative effects on the Egyptian economy, putting the country at risk from decreased food production and associated high food costs potentially giving rise to increased malnutrition and unemployment (Smith et al. 2014). Simultaneously, Egypt has a long history of food protests and riots in its modern history, with small landholder farmers often being at the heart of economic and political struggles (Bush & Martiniello 2017). Riots in Egypt, especially in rural areas, have been driven primarily by food crises and shortages in resources, particularly water and land. With the growing water crisis, land depletion and losses in rural livelihoods due to climate change, as well as the ever-increasing unemployment, rural poverty and marginalisation of rural areas (further discussed in Section 2.3.1), Egypt could potentially face a chronic food crisis and a growing threat of political instability.

2.3 Agrarian transition

The concept of the agrarian transition refers to the process through which agrarian societies are exposed to rapid and profound societal and environmental transformations as a result of the growing capitalist drivers of change facilitating their shift into industrial societies (Castella 2012). The change is experienced across most of society, including economic policies, market integration, environmental regulations and demographic change. The classical views of Byres (1977, 1986) describes the agrarian transition as the penetration of capitalist relations of production, the modernisation of agricultural production, the emergence of a consequential class structure, and the expansion of urban-industrial societies. He argues,

"capitalism became the dominant mode of production in agriculture: growing out of simple commodity production, here via a landlord class and there via a peasantry which gradually became differentiated (so providing, at the extremes, a stratum of rich peasants who ultimately became capitalist farmers and a stratum of poor peasants who were transformed into agricultural labourers or who joined urban proletariat)" (Byres 1977, p.258).

The crucial component in the development of capitalism in agriculture is the production of surpluses - in labour, food, raw material and financial (Byres 2016). Land use, changing technologies, tenure, class relations and differentiation, and the role of the government are the major themes that are in flux during the agrarian transition of societies, which have been addressed in a range of studies (Scott 1976; Scott 1985; Kerkvliet 1990; De Koninck 1992; Pincus 2000; Rigg 2001). However, Kelly (2011) suggests that there is a more complex and broader set of processes driving the transition of agrarian societies during the global era, rather than merely one process of capitalist commodification, such as the form the agrarian transition takes within any particular developing country will vary significantly. These processes include agricultural expansion into new production zones; agricultural production intensification through mechanisation and genetic modification; urbanisation and industrialisation of land use and livelihoods; changing regulatory frameworks; environmental change; and, the impact of migration on agrarian production and the social and cultural processes in which it is embedded (Rigg 2006; Kelly 2011). Hence, a significant component of any agrarian transition is the increasing urbanisation of populations, and the associated deagrarianisation and diversification of livelihoods within rural locales (Kelly 2011; Castella 2012; Barrett, Christian & Shiferaw 2017; Barrett et al. 2017). Labour movement out of agriculture into non-farming activities within rural communities and secondary towns is argued to be more effective in reducing poverty than migrating to big cities, especially if rates of industrial development are unable to provide individuals with employment opportunities (Barrett et al. 2017). Hence, the effectiveness of any transition out of agriculture for reducing poverty is highly dependent on where off-farm jobs are available and whether immigrants have the capacity to move into new occupations (Christiaensen, De Weerdt & Todo 2013; Dorosh & Thurlow 2014; Christiaensen & Kanbur 2016).

The agrarian transition extends beyond an agricultural transformation to affect broader structural transformations (Barrett, Christian & Shiferaw 2017). That process can be described as the mechanisms through which low-income agrarian societies become highincome developed societies having relatively small, yet productive, agricultural sectors (Barrett et al. 2017). Economic development in most developing societies is typified by a decline in agriculture's share of both GDP and employment over time as a result of income growth, urbanisation, demographic change and poverty reduction (Barrett et al. 2017). The macro-level descriptors of the transition of economies involves three central dimensions (Barrett et al. 2017). First, the early stages of transition involve agricultural intensification through the extended use of purchased modern inputs and technology and increased market-oriented production, which consequently improves the productivity of farms and labour (Johnston & Mellor 1961; Timmer 1988). Second, the transition leads to the existence of new forms of two-way linkages between farm and non-farm sectors in which agricultural productivity growth encourages progress in the non-agricultural sector, and households start to diversify their income sources through the non-farm sector, often then reinvesting further capital into agricultural intensification (Johnston & Mellor 1961; Barrett, Reardon & Webb 2001). And third, the process involves the emergence of a strengthened relationship between agricultural production systems, broad societal demographic changes and dietary shifts that are commonly paired with increases in income growth and urbanisation (Barrett, Christian & Shiferaw 2017; Barrett et al. 2017).

There are different ways to interpret the stages of transition. Timmer (1988) identifies four stages of agricultural transformation, starting with "Getting Agriculture Moving", which involves huge investments in the agricultural sector particularly in research, technology and rural infrastructure. The second phase "Agriculture as a Contributor to Growth" consists of the transformation of the agricultural sector into a key contributor to economic growth, and consequently, a state of de-equilibrium occurs between agriculture and industry, which widens the gap between both sectors (Kuznets 1966; Chenery & Taylor 1968; Chenery & Syrquin 1975). Thirdly, comes "Integrating Agriculture into the Macro Economy", which involves a narrowing of the gaps between secondary industry (urban) and primary agricultural (rural) sectors, through the improvement of more efficient labour and credit markets that link the two sectors together. In other words, within Timmer's framework, the transformation also involves a transfer of labour and capital from the agricultural sector to other industries or services over time. Although the evolution of a more complex economy provides flexibility for individuals who might otherwise be trapped in low-income occupations, rural labour productivity may stagnate during the transformation if industrial development is unable to absorb the proletariat effectively, resulting in both rural-urban and urban-urban income distribution problems across societies (Timmer 1988). The integration of the agricultural sector into a capitalist macro-economy can make agricultural communities more susceptible to fluctuations in prices and trade systems (Schuh 1976). The fourth and final phase of the agricultural transformation is "Agriculture in Industrial Economies" (Timmer 1988). By this stage, the share of the agricultural labour force has generally declined below 20 percent and the amount of money spent on food by urban household's drops to around 30 percent. Some political issues might also arise as agriculture becomes less important to the overall economy (Anderson 1983).

This last situation has often led to a rural–urban tension where low farm incomes due to declining returns for their products, encourage farmers to redirect resources from agriculture, and the emergent industrial sectors in urban areas are often incapable of sustainably absorbing rural migrants (Timmer 1988). The outcome as seen throughout many countries towards the final stages of their agrarian transition, and across Africa in particular due to industrial production not keeping pace with the liberation of labour from farming areas, is a rise in unemployment, forcing agricultural labour to remain in the sector, often in association with rising rural landlessness and poverty (Timmer 1988; van Neuss 2019; Wood 2019).

Similar to Timmer's analysis, the World Bank (2015) suggests five stages of agricultural transformation based upon the contribution of the agricultural sector to the society's GDP and the percentage of labour force working in the sector (Figure 2.1), as shown below.





Source: World Bank (2015)

The World Bank Group (2015) classification reflects five different stages of agricultural transformation within countries as follows:

- 1. **Agriculture-based countries**: agriculture contributes to over 25 percent of GDP and employs more than 50 percent of the economically active population;
- 2. **Pre-transition countries**: agriculture still employs more than 50 percent of the workforce, however, its contribution to GDP decreases to less than 25 percent;
- 3. **Transition countries**: agriculture supports a smaller labour force, employing 25 percent to 50 percent of the economically active population, and less than 25 percent of GDP;
- 4. **Urbanised countries**: agriculture still contributes to less than 25 percent of GDP, whereas it employs between 10 to 25 percent of the workforce; and
- 5. **Developed countries**: workforce engagement in agriculture drops to less than 10 percent and its contribution to the GDP also drops to below 10 percent.

According to the stages in Figure 2.7, Egypt as a whole is considered to be reaching the border between a transition and urbanising country, as Egyptian agriculture contributes 11.05 percent of the GDP and employs 23.3 percent of the labour force (World Bank 2020). In such a context, within every single region, there will be significant social

complexity as different groups take on new roles or are stuck in entrenched positions, with limited opportunities either in rural or urban settings.

The agricultural transformation process is an integral part of the dominant development approach or the process of modernisation through which societies pass. The dynamics of modernisation create new global risks and threats, like environmental risks, which are the unintended consequences of scientific, technological and economic developments (Beck 1989; 2008). Risks constitute the driving force of modern industrial society, and consequently, late modernity can be described as "risk society" (Beck 1992; 2002). Accordingly, the main feature of risk society theory is that modernisation is increasing the likelihood of new future threats, instead of overcoming societal risks (Bardsley 2015).

Modernisation causes risks not only associated with the irreversible endangering of human, animal and plant life, as occurs with the use of chemical substances in agricultural production (Beck 1989), but also with the limitations of modern systems to foresee, control and overcome these risks (Beck 1995, 2002, 2008; Cohen 1997). Risk society is affected by an uncontrollable set of threats in the areas of ecology, economy and politics that typically generate conflicts (Beck 2002, 2008). Even if modern systems proved a success in managing environmental risks, climate change is now threatening these accomplishments (Bardsley 2015). Environmental risks, as well as other risks caused by the modernisation of society, become social problems (Beck 1992) and create new inequalities either between different social classes and groups of the same society or among other international countries (Beck 1989, 2010). The emergent disparities and the socially recognised risks have the power to become political and to develop new social movements (Beck 1989). Beck, Giddens & Lash (1994) believe that reflexive modernity is a critical pathway that might allow societies to move beyond the redevelopment of risks as they develop.

Consequently, for agricultural communities to successfully respond to increasing risks associated with the modern world, they will need to build "complex, resilient and sustainable modern system[s] in light of future climate change and resources constraints" (Bardsley & Knierim 2020 p.4). In order to establish resilient agrarian systems, farmers will need to work closely with researchers, policy-makers, national government and non-government organisations in order to adapt in a sustainable way (Pretty et al. 2018; Bardsley & Knierim 2020). The following section will discuss the historical evolution of

land use, tenure and agricultural policies in Egypt that shaped the dominant paths of agrarian change. It also highlights some of the risks generated by the national agricultural development interventions.

2.3.1 Historical development of some land and agricultural policies that drove agrarian change in Egypt

There are several social factors that have driven agricultural change in rural Egypt over the last two centuries. These factors include commoditisation, migration, capital investment, irrigation, electricity, road networks, transportation, mechanisation, education, urbanisation, and government policy toward agricultural and rural development (Hopkins 1987). There are several essential policy milestones in the history of Egyptian agriculture that had enormous influence over the transformation of the sector and rural areas in Egypt as described below.

2.3.1.1 The 1952 Revolution and land reform policies

In the first half of the twentieth century, Egypt had one of the most unequal distributions of land ownership in the world (Woertz 2017). Land in Egypt was not valued for its income-generating capacity alone, but possession of land always stood as a symbol of political power and social prestige (Oweis 1971 as in Abdel-Fadil 1975). In the 1940s, 0.5 percent of landholders owned only 37.1 percent of the cultivated land, while 70 percent owned 12.4 percent (Cliff 1946). The number of landless families was also increasing, and big landlords monopolised land and water resources raising the prices of land rents leaving little profit for the tenant farmers or sharecroppers (Abdel-Fadil 1975). This substantial disproportionality in land distribution intensified poverty and social inequality across rural areas, which led, in part, to the 1952 revolution that promised a radical social reform program for Egypt (Abdel-Fadil 1975; Bush 2007).

The successful revolutionary process of transformation in Egypt involved significant changes to the agrarian structure through land reforms that started in the 1950s. The land reforms of 1952 and later had three main objectives: the redistribution of rural resources to give landless and small-farmers' rights to land; attempts to shift the balance of political power in the countryside away from an extreme minority and; the desire to allocate capital surpluses from agriculture to subsidise urban growth and industrialisation (Abdel-Fadil 1975; Bush 1994, 2007, 2011). To achieve these goals, the government redistributed agricultural lands, mandated crop rotation schedules and crop area allocations, forced

obligatory delivery quotas for crops at fixed prices that were considerably below international prices, and subsidised consumer prices for basic food commodities (Cassing et al. 2009).

The first land reform in 1952 limited land ownership to 200 *feddans* (84 hectares), the maximum that could be reserved by the owner for himself, with an additional 100 *feddans* allowed for any dependent children (Marii 1954; Bush 1994; Bush 2011). Following that initiative, a second Land Reform Law was enacted in 1961 limiting ownership to 100 *feddans* per person (Abdel-Fadil 1975). The law was then modified again in 1969 to reduce levels of single household ownership further to 50 *feddans* (Abdel-Fadil 1975). Another significant aspect of these state policies since 1952, relates to the expansion of agricultural land through land reclamation programs (Bromley & Bush 1994). Land reclamation started with about 78,000 *feddans* based on small-farmer ownership (Lewis 2008). Since 1952, the land chosen for reclamation has been in two areas; on the fringes of already cultivated areas in the Nile Delta and valley where reclamation was largely only a matter of installing pumping systems so that Nile water could irrigate raised areas of land, and in the desert and coastal areas which otherwise depended on groundwater and rainfall (Voll 1980).

During this period, agricultural cooperatives were also established to provide farming inputs and machinery, grant agricultural loans to rural members, provide advice for farmers on the selection of crops, efficient land cultivation, and destruction of pests, support sales of the principal crops, and provide for other agricultural services, including the provision of canals and drainage (Margold 1957). Agricultural cooperatives were mandated mainly to determine cropping patterns, provide inputs to farmers and organise the purchase of outputs from them at fixed prices (Bush 2007; Cassing et al. 2009).

Cooperatives coordinated the provision of credit and output allocations with agricultural credit banks, which together became an essential instrument of agricultural trade and allocation of finance (Cassing et al. 2009; Kassim et al. 2018). The system also attempted to provide a legal and institutional rural framework for the first time, setting fixed minimum wages and land rents, and improving tenancy agreements in an attempt to break the power of big landlords, in order to benefit the peasantry (Bush 1994). The new legislation even gave tenants rights of inheritance, which had not been identified previously (Bush 2007).

The most notable feature of the agrarian reforms in Egypt was the fact that after two major attempts in redistributing land, the landholding size distribution remained very similar over time, leaving the relative numbers of big landowners, medium-scale and small landowners almost unchanged (Abdel-Fadil 1975). The poor peasants and landless saw minimal real gains during the reform periods (Woertz 2017). It was mostly the middle and upper-middle segment of peasants who benefited from such reforms by increasing their holdings, controlling the benefits to be derived from the agricultural cooperative system, and boosting their market access and influence (Bush 1994; Woertz 2017). Arguably, the major outcome of the post-colonial agrarian reform policies was the shift of near-subsistence family farms into petty-commodity producers, who became highly dependent on state subsidies for production and subsistence (Araghi 2009).

As part of the socialist regime adopted in the 1950s, the food subsidy system was introduced to provide basic commodities at a subsidised price to all Egyptians (Kassim et al. 2018). Bread was the cornerstone of the subsidy system in which the government was involved extensively in all stages of production from procurement of wheat to milling into flour, to production in bakeries, to produce bread for eligible customers at subsidised prices (Tellioglu & Konandreas 2017). The number of subsidised commodities increased gradually, reaching 18 food items by the early 1970s (Ahmed et al. 2001; Cassing et al. 2009).

In response to the ongoing low profitability of agriculture in the late 1960s and early 1970s, yields decreased, self-sufficiency gaps grew wider, exports fell, the rural–urban gap became wider, and labour began to move out of the agricultural sector to seek better opportunities in the non-agricultural sectors (Cassing et al. 2009). Farmers became frustrated as taxes consumed a large part of their earnings and yet simultaneously, the food subsidy system represented a considerable burden on a budget of the government (Cassing et al. 2009). With the expansion of non-farm employment as well as more government public positions becoming available, and mechanisation, young men became more engaged with education in the search for employment opportunities away from agriculture, resulting in a further labour shortage in the sector (Commander & Hadhoud 1986).

2.3.1.2 The open-door economic policy

In response to slower growth in the agricultural sector, the government started an opendoor international policy to attract foreign investments (Lewis 2008; Cassing et al. 2009). With the beginning of the open-door policy after the 6th of October War in 1973, a shortterm boom in the Egyptian economy was driven by high oil prices, labour remittances and high rents from the Suez Canal leading to further neglect of agriculture because it was seen as a less profitable enterprise (Bush 2007; Cassing et al. 2009; Abdou 2013). The agricultural sector experienced further labour shortages due to the migration of rural inhabitants to the oil-producing Gulf Countries (Commander & Hadhoud 1986). Simultaneously, cities indicated that they had limited capacity to absorb rural migration flows sustainably (Woertz 2017).

The open-door policy encouraged foreign investment in reclaimed lands and made it legal for joint-venture (Egyptian and foreign) companies to own thousands of *feddans* on reclaimed lands (Dixon 2020). The private sector was, hence, allowed to reclaim desert land, often in a disorganised manner, for agricultural use and at the same time, urban encroachment on agricultural land expanded prominently with the growth of remittances (Lewis 2008).

Consequently, agriculture's share of the GDP and employment declined significantly over this period (Cassing et al. 2009), and the sector lost its dominant economic position (Lewis 2008). The neglect of agricultural development during the early phase of the opendoor policy also caused a severe decline in exports as the sector converted into a freemarket with severe levels of intra-state competition (Abdou 2013). One of the favourable outcomes of the open-door policies was the transfer of technology and scientific cooperation from western and international organisations leading to remarkable improvements in yields (Abdou 2013). However, trade liberalisation also shifted Egypt and many other developing countries from food self-sufficiency to import dependency, especially as the US and EU were seen to subsidise agriculture to maintain global prices of strategic grains at less than the costs of production in the developing world (Bush & Martiniello 2017). By 1976, Egypt was the world's third-largest importer of grain, increasing its external debt (Bush 2007). Accordingly, in 1977, the government attempted to reduce the total expenditure on subsidies for some food items, which caused substantial, violent rioting because people had grown dependent on state-subsidised food (Alderman 1986; Ahmed et al. 2001). The country then experienced a food crisis in the 1980s resulting from low local grain production and a growing population necessitating further increases in imports (Bush 1994).

Another significant feature of this era was the Camp David agreement signed between Egypt and Israel in 1978 and mediated by the United States. The agreement came after years of ongoing tension following the first Arab–Israeli war in 1948 after Israel declaration of independence. According to the agreement, Egypt took back its land in Sinai, and benefited from the foreign aid that was allocated to the Egyptian government by the United States as a condition of the agreement (Bani Salameh, Bani Salameh & Al-Shra'h 2012).

Accordingly, Egypt became significantly more dependent upon the United States' economic and military assistance after 1979 (Sharp 2020). The situation was seen as essential to keeping peace and funding a political regime that supported American interests in a critical region of the world (Bani Salameh, Bani Salameh & Al-Shra'h 2012). The agreement also involved cooperation between Egypt and Israel in the development of agricultural research and technology, tourism, transport, and trade (Pohoryles 2009). Accordingly, the agreement ended the economic boycott, and it ensured the free movement of people and goods between Egypt and Israel and vice versa, and cheap, highquality Israeli goods began to out-compete the Egyptian products on the free market (Hamdan 1989). Simultaneously, also as a result of this agreement, which had been seen to accede considerable sovereign power to the US, the Arab countries imposed economic and political sanctions on Egypt and cut off trade and democratic relations (Bani Salameh, Bani Salameh & Al-Shra'h 2012). With the decline in domestic food supply of main strategic crops, including wheat, maize, rice and sugar in relation to total consumption (Bruton 1983), Egypt has become increasingly dependent on imports from the US to supply the needs of a growing population.

2.3.1.3 Economic reform, privatisation and liberalisation

The state-interventionist policies began to be gradually dismantled in 1986, as the government's role was curtailed in favour of the private sector (Cassing et al. 2009; Woertz 2017; Kassim et al. 2018). The prices for the main strategic crops were partially or entirely liberalised, input subsidies were cut, and forced delivery of the strategic crops to the state were eliminated or contracted out, thereby decreasing the government monopoly on agricultural inputs and outputs (Cassing et al. 2009). The strategy removed

pricing constraints to encourage farmers, purchasers and processors of agricultural commodities to invest in agricultural productivity (Bush 2007).

Liberalisation policies advanced further in the 1990s, and by 1994, only sugar and cotton stayed under government regulation (Woertz 2017). In 1991, the government agreed to adopt economic reform and a structural adjustment program, with the help of the International Monetary Fund (IMF) and the World Bank. The program sought to promote an export-led agricultural development and market-oriented economy through the production of less nutritious foodstuffs with high-value for export (Bush 2007). Simultaneously, Gulf investments in land reclamation increased in Egypt, with the aim of producing agri-food commodities needed to ensure food security for people living in Gulf countries (Henderson 2017). Cotton marketing was then liberalised in 1994, all remaining state input subsidies were removed, and the private sector was encouraged to play a more significant role in agricultural trading (Cassing et al. 2009). However, the attempts to shift Egypt to an export-led agricultural economy were mostly unsuccessful, and the cultivation of crops of high value for exports have not contributed much to the economy or the local incomes of farmers (Bush 2007).

Another significant aspect of the liberalisation of Egypt's rural areas has been the implementation of Law 96 of 1992 entitled, "Reform of the tenancy relationship between owner and tenant" (Cassing et al. 2009; El Nour 2015). The law aimed to promote land consolidation, increase agricultural productivity and promote rural stability (Bush 2000). However, the law made the extension and continuity of rental contracts of farming lands invalid for the first time since 1952, and it regulated for the total suspension and replacement of existing land-holding agreements after a five year transitional period (FIAN 2000). It also gave landlords the right to levy rents at market rates and allowed contracts to be reduced to 12 months and revoked without notice (Bush 2000). In 1997, after the full implementation of the law, landlords had the right to evict current tenants and lease the land to new tenants at higher rates (Ibrahim & Ibrahim 2003; Adriansen 2009). Consequently, the law led to increasing insecurity for tenants and raised rents for agricultural lands more than threefold, meaning that many farmers were unable to pay their rents (Kassim et al. 2018).

As a result of the reconsolidation, landlords were able to regain their lands that they had lost during previous reforms and access new lands, triggering a process of domestic land grabbing (Fautras & Iocco 2019). Yet land appropriation was generally not for agricultural purposes, with many agricultural lands taken for public infrastructure projects and private wealth gains, such as the establishment of tourists resorts (Dixon 2020). About one million tenants lost their land across Egypt and became sharecroppers, renting small plots of land from a landlord in return for a part of their crop to be given to the landlord (Kassim et al. 2018).

The law resulted in more rural poverty, landlessness, and increased the desire for migration by younger family members of ex-tenants, often at a time of fewer opportunities in urban areas (Bush 2000; El Nour 2015). The law also sparked violent protests by farmers in more than 100 villages across Egypt's countryside (El Nour 2015) that were repressed by the government (Saad 2002). Market liberalisation, hence, created greater poverty and unemployment in many Egyptian rural areas (Bush 2004). However, farmers managed this deteriorating economic situation through greater dependence on family resources and assets (Bush 2004).

Importantly, the implementation of the neo-liberal reforms since the 1980s and 1990s radically altered labour conditions resulting in a labour class that was insecure, fragmented, increasingly precarious, and dispossessed of substantial socio-economic rights (Paciello & Pioppi 2020). The economic reform and structural adjustment program reduced the role for the public sector in the economy and labour market, which became unable to absorb the growing numbers of job entrants (Bremer 2018). Although the goal of these reforms was to create a strong private sector, Egypt was unable to create decent, formal jobs in the private sector (World Bank 2014). With the expansion of urban economic opportunities and the rise in rural populations, peasants migrated to urban areas in the last decade of the twentieth century to become absorbed into the informal economy (Bremer 2018).

Adding to these economic challenges, Egyptian farmers face two main contemporary issues that threat their livelihoods. The first, the Grand Renaissance Dam in Ethiopia under development since 2014 when Egypt, under the current president, signalled its acceptance that there is little it could do to prevent the dam from being completed (The Economist 2015; Barnes 2020). The second, farmers being dispossessed from natural resources through the process of domestic land grabbing driven by members of the local elite and capitalist investors who have political and economic power at the national level

(El Nour 2019). Cochrane (2016 p.1) defines land grabbing as "the application of force to coerce individuals to illegally give up their land or the other wise illegal dispossession of land".

It can be concluded that rural areas in Egypt have been suffering from decades of marginalisation and growing poverty (Bush 2000; Bush 2004; El Nour 2015; Bush 2016). For instance, recent statistics show that 32.5 percent of the population in Egypt (32 million citizens) live below the national poverty line (around US\$ 1.45 per day) in 2018 (CAPMAS 2019a), an increase of 15.8 percent since 1999 (World Bank 2017). Poverty rates are generally high in rural areas reaching 51.9 percent in some regions in Egypt in 2018 (Armanious 2018).

De Lellis (2019) claims that government policies have been working against the agriculture labour force for at least 30 years, dispossessing them of economic and natural resources. He argues,

"Since the late 1980s, dispossession of natural resources and marginalisation of the poor peasants have been a constant feature of agrarian transformations in North Africa. This has affected the whole landscape of societies, engendering phenomena of migration, urbanisation, environmental degradation, impoverishment and unemployment. Dispossession in Egypt has been gradual and the result of different forms of commoditisation of land and of other crucial agricultural inputs and services" (de Lellis 2019, p.583).

Dispossession is not only about physical elements but also about the intertwined social order, the customs and representations connected with means of accessing and controlling resources (Fautras & Iocco 2019). Indeed, small food producers in many developing countries have faced increased dispossession from land throughout the process of transitioning to commercial agriculture and commoditisation (Desmarais 2007). Many small-scale farmers in the developing world have been unable to combat the effects of trade liberalisation and compete with subsidised large-scale agricultural production of the developed countries (Bush & Martiniello 2017), a process that is referred to as "depeasantisation" (Dixon 2020). Likewise, the process of "de-agrarianisation" refers to the pushing of small farm households into non-farm sectors to supplement low

agricultural income that is often reduced under capitalism (Bryceson 2004; Bush & Martiniello 2017).

These two processes contribute to a structural crisis and accumulated resistance, often sparking protests, among dispossessed farmers and other rural people who find limited employment opportunities in towns and countryside (as explained earlier in Section 2.2.3) (Bush 2016; Bush & Martiniello 2017). Recent decades have witnessed growing social movements in many agrarian societies that engaged in struggles against liberalisation and the mass dispossession caused by capitalism (Borras, Edelman & Kay 2009; Murari 2015; Vergara-Camus 2014; Tilzey 2020; Engles 2021).

Moore (2010) also advocates that the production of cheap food became an integral aspect of modernity over the expense of ecological sustainability. He links patterns of capital accumulation and the crisis of capitalism in agriculture with climate change and other ecological crises, as industrialised agriculture has generated "negative value" and caused depletion of natural resources (Moore 2015). For instance, the introduction of chemicals as a feature of modern agriculture has undermined the sustainable management of soil nutrition in many places (Bush & Martiniello 2017) and eradicated natural enemies of pests (Foster & Magdoff 1998). This mismatch between capitalism and resource depletion, which is, again, a characteristic of a risk society (Beck 1992), has made food riots and protests a persistent feature of modern times (Moore 2010).

The modern global food system has commoditised food to the extent that the landless poor only have access to nutrients just adequate for survival if they can purchase it (Bush & Martiniello 2017). Consequently, poor and ecologically marginal countries that fail to ensure adequate local food production, or are unable to buy food and distribute it at affordable prices for poor people, will likely face persistent food crises and associated political opposition (Bush & Martiniello 2017).

2.3.2 The paths of agrarian transition in Egypt and the resultant class structure

To better understand the agricultural sector in Egypt, we need to recognise the history and trajectories of agrarian change that have produced the current situation of agriculture at local and regional scales. Hart, Turton & White (1989) advocate the importance of conducting local-specific studies and even further recognizing the significance of nonfarm incomes and production processes within the household as a whole when

addressing agrarian change. On a local scale, the nature of rural villages in Egypt are different from one another based upon their location (near or far from river or water canal, city, desert), history, economic diversity, cultural sophistication, or the internal economic structures (Hopkins & Mehanna 1981 cited in Hopkins 1987). More importantly, rural villages are also distinguished from one another based upon the "paths of rural transformation" (Keyder 1983). Hopkins (1987) identifies three primary paths of agrarian transformation in rural Egypt as follows:

- 1. The "capitalist" path, which results from the ability of a stratum of large landholder farmers to acquire machinery and other means of production and to accumulate funds. These capitalist farmers rely on machinery and hired labour, thus reproducing capitalist relations of production in the villages they dominate.
- 2. The "petty commodity producers" path. These include small-scale producers having control over several *feddans*; however, they have less access to other means of production than the large landholders. They typically rely on family labour, but also sometimes hire labour to supplement the household labour supply, and hire tractors and pumps to complete their farm work. Such small farmers are losing some of their decision-making over the agricultural process, and they are "deskilled'. In sum, the petty commodity path exists, but the producers are, to some extent, falling under the influence of capitalist farmers.
- 3. The third path of agrarian transformation revolves around the role of the state in newly reclaimed lands or agrarian reform areas. In these areas, the state controls both land and water, whether through cooperatives or state companies. Villages within the newly reclaimed areas have large landholdings and their dependence on government policy support is high. People resettle there from elsewhere, and most are smallholders, recent graduates or officials. Finally, they are villages that depend predominantly on labour from outside the villages.

Individual villages are often typified by one or another of these dominant paths of agrarian transformation; however, within each village, particularly in the old lands of the Delta, both capitalist farmers and petty commodity producers can be found within single locations (Hopkins 1987). More importantly, the class structure of virtually every village also contains a substantial number of people who might possess a small farm area or none at all, and who earn their living mainly through working outside the village or the agricultural sector (Commander & Hadhoud 1986; Hopkins 1987). Springborg (1990)

refers to this group of people who earn from non-farm sectors as a new class of landowners called "agrarian bourgeoisie" who emerged following the implementation of agrarian reforms.

The fundamental sources of status and income for the agrarian bourgeois are commonly generated from outside agriculture and beyond the local area (Weinbaum 1982). This category of rural dwellers relies on superior technical and managerial skills to exploit capitalist opportunities and often have more educational qualifications than the traditional capitalist farmers (Springborg 1990). Therefore, this category of individuals is found in the agrarian class structure of the village, and they continue to reside in the village and have political and social influence (Hopkins 1987).

It can be concluded then, that villages in Egypt are composed of distinct groups of farmers and their households each of which could represent or reflect the different stages of the agrarian transition process (Table 2.1). This heterogeneity requires more attention when studying farmers' vulnerabilities and capacity to adapt in response to climate change, because different social groups will have different levels of vulnerability to socioecological change and have variable opportunities to adapt. It is therefore important to integrate these social categories into the data analysis and the explanation of the study results.

Table	2.1	Dominant	class	structure	within	the	Egyptian	rural	areas	and	the
representing stage of the agrarian transition process											

Stage of agrarian transition	Dominant class	Characteristics				
Agricultural-based societies	Petty Commodity producers	- Small landholders or landless, agricultural waged labourers				
		- High dependency on family labour				
		- Low political and social power				
Pre-transition	Capitalist farmers	- Relatively larger landholders				
societies		- High dependency on machinery and hired labour				
		- High political and social power				
Transition societies	Agrarian bourgeois	- Earn largely from non-agricultural sources				
		- Living in rural areas and reserving some social and political powers				
		- High educational qualifications				

Source: Adapted from Hopkins 1987; Springborg 1990; World Bank Group 2015

This integration allows for an explanation of the links between the broader context of perceptions of risk caused by climate change, and the social and economic characteristics of these groups, and it will help to frame the particular vulnerabilities and decisions to adapt within the Egyptian agrarian community under investigation. The following section will, therefore, present a broad review of key concepts linked to risk perceptions and adaptation to climate change.

2.4 Risk perceptions

The study of risk perception refers to the analysis of the instinctive judgment of individuals and groups about the features and severity of risk (Slovic 1987). Risk generally involves "the degree of uncertainty in a given situation" (Roumasset, Boussard & Singh 1979 p.4). Environmental perception is the way by which individuals seek to comprehend their surrounding environment, and it can either occur directly through personal experience with environmental events (storms, droughts and floods), or indirectly through other individuals and community members, government agencies, media, and scientists (Koubi, Stoll & Spilker 2016).

O'Connor, Bard & Fisher (1999) argue that risk perception is a significant factor for predicting behavioural intentions. One of the key factors that significantly influences individuals' decisions regarding adaptation is their perception of the risks linked to the adverse effects of climate change (Weber 1997; Vedwan & Rhoades 2001; Patt & Schröter 2008; Gbetibouo 2009; Smithers & Smit 2009; Weber & Johnson 2009; Bardsley & Hugo 2010; Spence et al. 2011; Koerth et al. 2013; Bagagnan, Ouedraogo & Fonta 2019). Likewise, Maddison (2006) suggests that adaptation requires two steps: the first is perceiving that climate change has occurred and the second is deciding whether or not to adopt a certain adaptation measure. Similarly, Antle (2009) suggests that farmers' perceptions of current and future climate change-related risks together with farming conditions and their socio-economic characteristics, were key factors affecting their decisions to adopt particular adaptation measures.

Consequently, to develop an appropriate and effective response to climate change, it is essential first to understand whether farmers have observed changes in climatic conditions and how they perceive the risks associated with these changes (Wheeler, Zuo & Bjornlund 2013; Li et al. 2017). Assessing perceptions of risk is also crucial in

understanding individuals' preparedness to adopt adaptation strategies and hence, assist appropriate policy design in a given setting (Li et al. 2017).

Perceptions of environmental change not only depend on individual's exposure to climatic events but they are also mediated by his or her ability to mediate or cope with environmental stressors, given their individual and household circumstances (Koubi, Stoll & Spilker 2016). Actions that follow perceptions of climate change risks are informed by different factors such the resource base, cultural values, political and institutional environment and there is no assurance that perceptions of climate change would result in effective adaptation responses (Weber 2010). Therefore, most research studying individual or household decisions about migration-related adaptation emphasises the importance of understanding perceptions of environmental change, as it allows a better explanation of why some people decide to migrate, for example, when exposed to a given environmental stress, while others do not (Mortreux & Barnett 2009; Bardsley & Hugo 2010; Black, Kniveton & Schmidt 2011; Black et al. 2013; Hunter, Luna & Norton 2015).

Numerous studies have suggested that people, whether in developed or developing countries, have already developed their own perceptions about climate change (Mertz et al. 2009b; Akter & Bennett 2011; Chaudhary & Bawa 2011; Deressa, Hassan & Ringler 2011; Nyanga, Johnsen & Aune 2011; Spence et al. 2011; Nnko et al. 2021). There are several socio-economic, demographic and biophysical factors that shape individuals' perceptions of climate change including geographic location and soil types (Maddison 2007; Gbetibouo 2009; Omar et al. 2015), income (Semenza et al. 2008), ethnic background (Leiserowitz 2006), gender (Leiserowitz 2006; Rothermich et al. 2021), age (Diggs 1991; Maddison 2007; Ishaya & Abaje 2008; Rothermich et al. 2021), education and access to extension services (Maddison 2007; Gbetibouo 2009; Omar et al. 2015; Nnko et al. 2021), irrigation infrastructure (Niles & Mueller 2016) and the source of the information about climate change (Sampei & Aoyagi-Usui 2009; Weber 2010). Adger et al. (2013) also suggest that culture is one of the factors that mediate people's risk perceptions and adaptive behaviour as will be further addressed in the following discussions about adaptation to climate change (Section 2.5).

2.5 Adaptation to climate change

The rewards of a whole year's effort in farming are largely dependent on two fundamental variables that are outside the farmers' direct control: the weather and prices (Timmer 1988). The sensitivity of the agricultural sector to climate makes the sector one of the most vulnerable sectors to the impacts of climate change (Parry & Carter 1989; Smit & Skinner 2002; Mendelsohn 2008). Therefore, adaptation to climate change is crucial for sustaining the agricultural sector and the livelihood of people working within the sector. Human adaptation to climate change is defined as "the process of adjustment to actual or expected climate and its effects by seeking to moderate or avoid harm or exploit beneficial opportunities" (IPCC 2014 p.5).

Adaptation is influenced by socio-economic, cultural, political, geographical, ecological and institutional factors that shape the human-environment interrelationships (Eriksen et al. 2011). Smithers & Smit (1997) argue that individuals and systems (social, economic, institutional and ecological systems) can and do adapt to changing environment. However, sustainable and effective adaptation depends on the adaptive capacity of those involved (IPCC 2007a). Many factors influence the adaptive capacity of farming communities such as knowledge about, and perceptions of, climate change (Adger et al. 2003; Adger et al. 2007; IPCC 2007a), education (Abid et al. 2019; Ekembonye et al. 2020), social capital and networks (IPCC 2007a; Smit & Wandel 2006; Tinch et al. 2015; Jones, Ludi & Levine 2010; Abid et al. 2017), household size (Ekemhonye et al. 2020), off-farm income (Ekemhonye et al. 2020; Khan et al. 2021), access to information and appropriate technology (Abid et al. 2019; Awazi et al. 2020; Khan et al. 2021), access to markets (Ado et al. 2020; Marie et al. 2020), household income (Awazi et al. 2020), land size and tenure (Defiesta & Rapera 2014; Abid et al. 2019; Khan et al. 2021), and access to credit (Abid et al. 2016; Awazi et al. 2020; Khan et al. 2021). Lack of availability and access to assets, including both tangible capitals (natural, physical and financial) and intangible ones (human and social), may significantly limit the capacity of a system to adapt to climate change (Jones, Ludi & Levine 2010). Adger et al. (2003 & 2007) stress the influence of values and traditions in shaping the adaptive capacity of communities.

The resilience of livelihoods and their alternatives in terms of diversification and financial remittances and mobility is another critical factor that influenced the adaptive capacity of communities (Barnett 2001; Sutherland et al. 2005; IFAD 2008; Sobczak-Szelc & Fekih 2020; Benveniste, Oppenheimer & Fleurbaey 2020; Maharjan et al. 2020). Income

diversification both within the same sector and across the different economic sectors has often been considered a risk mitigation strategy (Barrett, Reardon & Webb 2001; Barrett, Bezuneh & Aboud 2001) and is essential to avoid poverty traps (Barrett, Reardon & Webb 2001; Barrett, Bezuneh & Aboud 2001; Haggblade, Hazell & Reardon 2007; Marenya & Barrett 2007; Bezu, Barrett & Holden 2012; Stephens et al. 2012). However, many smallholder households are incapable of seizing high-return, non-farm opportunities due to their limited human capital and other forms of capital including non-land capital assets, and poor access to markets and financial services (Barrett, Reardon & Webb 2001; Barrett, Bezuneh & Aboud 2001; Amare & Shiferaw 2017).

Adaptive capacity is a critical component for reducing the climate change vulnerability of any system (Smit & Wandel 2006). Vulnerability often refers to the "degree to which a system is susceptible to, and unable to cope with, adverse effects of climate change" (IPCC 2007a p.52). The IPCC's definition of vulnerability highlights exposure, sensitivity and adaptive capacity as the three essential features. The term vulnerability generally covers physical or natural, social, and economic vulnerabilities (UNDP 2014). Physical vulnerability includes aspects of location, place and geography (Wilbanks 2003; Anjum & Fraser 2021), whereas social vulnerability often relates to the characteristics of people affected by climate change (Piguet, Pecoud & de Guchteneire 2011). Several scholars have identified common aspects of social vulnerability including race (Fothergill, Maestas & Darlington 1999; Elliott & Pais 2006; Bolin 2007; Cutter & Finch 2008; Elliott & Pais 2010; Drzewiecki et al. 2020; Woo et al. 2020), education (Drzewiecki et al. 2020; Poudel et al. 2020), age (Wisner 2006; Bartlett 2008; Peek 2008; Peek & Stough 2010), socio-economic status (Bolin 2007; Ray-Bennett 2009; Ribot 2013; Poudel et al. 2020), gender (Sen 1983; Ray-Bennett 2009; Poudel et al. 2020), migration and housing tenure (Cutter & Finch 2008), and health conditions (McGuire, Ford & Okoro 2007; Peek & Stough 2010). Vulnerability is the outcome of exposure to climate change as well as inequalities in socio-economic status and income (IPCC 2014). Waly Ayad & Saadallah (2020) suggest that high levels of poverty, inequality, and problems related to housing, unemployment, access to basic public facilities such as sanitation and drinking water define the social vulnerability of communities.

The vulnerability of a society is subjective to its level of development, distribution of resources, physical exposures, previous pressures and social and government institutions (Kelly & Adger 2000; O'Brien et al. 2004; Smit & Wandel 2006). Political stability and
good governance both influence the adaptive capacity of societies as adaptation is viewed as a transformative socio-political process (Ribot 2011; Eriksen, Nightingale & Eakin 2015; Nightingale 2017). People living in areas where conflicts are prevalent are particularly vulnerable to climate change (Adger et al. 2014). All societies can adapt to some variation in climate, yet adaptive capacities are unequally distributed, both across countries and within communities (Adger et al. 2007).

For instance, developing countries, like Egypt, are more likely to suffer from the effects of climate change (Ayers & Huq 2009; Mertz et al. 2009a; Loayza et al. 2012; IPCC 2014), in part, due to the exposure to climate change highlighted previously. This greater vulnerability is also due to several social factors prevailing in developing countries including:

- 1. Economic structures and livelihoods of a large population depend on climatesensitive agriculture (Noy 2009; Loayza et al. 2012);
- 2. High rates of poverty in which poor people have low adaptive capacity, and
- 3. Inadequate economic and technical capacity needed for successful adaptation to climate change (Mertz et al. 2009a).

The effects of climate change would significantly influence most impoverished and marginalised people as they are least able to adapt, resulting in them becoming poorer (Bohl, Downing & Watts 1994; Adger et al. 2003; Thomalla et al. 2006; Parry et al. 2007; Mirza 2011; IPCC 2014; Warner, Hoffmaister & Milan 2015; Islam & Winkel 2017; Otto et al. 2017). Small landholders and the landless are among the most vulnerable to climate change (Morton 2007; Oppenheimer et al. 2014; Warner, Hoffmaister & Milan 2015). Clay & King (2019) also suggest that small landholders have uneven vulnerabilities and capacities to adapt to the impacts of climate change.

Even within both developing and developed countries, there are some regions, communities, social groups or individuals that are lacking the capacity to adapt to climate change (O'Brien et al. 2007). For instance, communities located in geographically remote areas are more vulnerable to climate change (Kohler et al. 2010). Rural households and communities, especially in developing countries, are the most vulnerable because their livelihoods are very dependent on primary resources (Paavola & Adger 2002; Agrawal & Perrin 2009; Dasgupta et al. 2014). Rural households are also exposed to other non-climate stressors, including land degradation, problems with policy regarding land and

natural resources, and under-investment in agriculture (Field et al. 2014). Füssel & Klein (2006) argue that non-climatic stressors are considered an essential determinant in shaping people's vulnerability. Below, Schmid & Sieber (2015) categorise non-climatic factors into three groups: socio-economic factors at the household level (e.g. gender, size, age structure, health status, income, ownership of livestock, and access to land), biophysical (e.g. soil quality, water availability) and institutional factors (e.g. the presence of extension services, access to markets, prices of farm inputs and produce). It is hence important to note that the adaptive capacity of societies is mostly dependent on the relative distribution of health, education and wealth (Feenstra et al. 1998). As a result, if a society or community includes a larger number of poor, unhealthy and poorly-educated people, the more vulnerable it is likely to be to climate change and the less likely it will be to adapt (Feenstra et al. 1998). Not responding to perceived or actual climate change risk is not always related to lack of adaptive capacity; it can be considered as proactive means when people have strong resilience and do not consider climate change as a significant risk (Mortreux & Barnett 2009).

Within this context, Warner, Hoffmaister & Milan (2015) classified rural households into:

- 1. Resilient households, which are those who are poor but possess a variety of assets (social, financial, political, educational) and who have better access to adaptation or risk-management measures (including mobility measures) and options to diversify their livelihoods.
- Vulnerable households, which are those who are land scarce or landless, and poor with very few opportunities for livelihood diversification and *in situ* adaptation. They usually manage climate-related risk through selling assets, consuming less food of a poorer quality, and seeking help from other community members.

Climate change can have a detrimental effect on the agricultural sector in the absence of adaptation; however, with the implementation of successful adaptation measures, vulnerability can be significantly reduced (Easterling et al. 1993; Smit & Skinner 2002). Consequently, adaptation can be seen to have the overarching aim of reducing vulnerability or improving resilience to climate change and variability (Smit & Pilifosova 2003). Decisions about adaptation do not only depend on perceptions of risk held by the society, as discussed earlier (see Section 2.4), but also on the places and cultures of a given society that may limit the array of adaptation measures (Adger et al. 2009).

Adaptation takes several forms at various social and spatial scales (Bryant et al. 2000), and research on it has employed different typologies to describe its multi-dimensional and multi-scale nature. Burton, Kates & White (1993) classified adaptation by its specific objectives as:

- 1. share the loss
- 2. bear the loss
- 3. modify the events
- 4. prevent the effects
- 5. change the use, or
- 6. change location.

Adaptations can also be classified according to their timing (reactive vs proactive); motivation of adaptation (economic improvement or safety); duration of adaptation (longterm vs short-term); impact of adaptation (worsened or enhanced resilience); ways of implementation (autonomous vs planned), etc. (Klein 1998; Smit et al. 2000; Klein 2003; Adger, Arnell & Tompkins 2005; Bryan et al. 2009; Smithers & Smit 2009). Other researchers typify adaptation as adjustments that occur in place (i.e. in situ) vs those that have mobility elements (i.e. ex situ) (Bardsley & Hugo 2010). Another approach developed by Park et al. (2012), suggested two central categories of adaptation: incremental and transformational. Incremental adaptation aims to retain the "integrity of a system or process at a given scale". In contrast, transformational adaptation suggests necessary changes "in the biophysical, social or economic attributes of a system from one form, function or location to another" Park et al. (2012 p.5). Wheeler, Zuo & Bjornlund (2014) proposed three main means for incremental adaptation; either by expanding land or water use, or decreasing irrigated land or water use; or accommodating the new conditions through improving irrigation efficiency and changing the mix of crops. Transformational adaptation comprises activities that are:

- 1. New to a system or region;
- 2. At a larger scale of intensity; and
- 3. In new places and locations (Kates, Travis & Wilbanks 2012).

The agricultural sector globally has been mostly adopting incremental adaptation strategies such as agricultural intensification, improvements to crop varieties and livestock breeds, or improving water and soil management practices, and transformational adaptation means are less applied in practice so far (Dowd et al. 2014). Incremental

adaptation might be inefficient or ineffective in promoting sustainable development of the agricultural sector on the long term (Howden, Crimp & Nelson 2010; Rickards & Howden 2012), and it could even lead to "maladaptation" (Barnett & O'Neill 2010). Maladaptation refers to adaptation measures that do not reduce vulnerability, but increase it instead (UNFCCC 2007). It can be concluded then, that farmers' responses to weather variations have aggregate consequences (Roumasset, Boussard & Singh 1979). Therefore, adaptation to climate change plays a key role in enabling or restraining development in the agricultural sector.

This study explains adaptation measures within the broader categorisation of *in situ* and *ex situ* as will be discussed in further detail in the following sections (2.5.1 and 2.5.2).

2.5.1 In situ adaptation

In situ adaptation strategies refer to localised adjustments to climate change and other environmental pressures (Yang et al. 2019). *In situ* adaptation practices in the agricultural sector include strategies aiming to alter or improve agricultural production, farm management, living habits and personal lifestyles, and the use of natural resources particularly land and water (Smit & Skinner 2002; Grothmann & Patt 2005; Deressa et al. 2009; Marshall 2010; Osberghaus et al. 2010).

This thesis explains *in situ* adaptation as localised adjustments in farming activities that improve agricultural production and farm management. The definition could include cases of income diversification within the agricultural sector while remaining in the original farming occupation. Income diversification activities within agriculture could consist of livestock production, trade, or paid labour (Below et al. 2010). Hassan & Nhemachena (2008) found that better access to markets, extension and credit services, farm assets (labour, land and capital), technology and information about adaptations, including technological and institutional methods, affect adaptation to climate change. Consequently, to ensure effective *in situ* adaptation practices that reduce vulnerability in the long run, it is argued that governments need to provide infrastructure, improve access to financial services, cash returns and social safety nets, and provide information and incentives to improve economic transformation (Castells-Quintana, Lopez-Uribe & McDermott 2018).

Several studies address specific agricultural adaptation measures that could reduce the vulnerability of the sector in Egypt. These measures include changing sowing dates, altering the crop mix to minimise the area cultivated with high-water use crops like sugarcane and rice, improving the efficiency of irrigation systems, increasing irrigation requirements and enhancing technical progress for fast-growing crop varieties (Eid et al. 2007; Attaher, Medany & Abou-Hadid 2009; Attaher, Medany & El-Gindy 2010; McCarl et al. 2015).

2.5.2 Ex situ adaptation

As populations and economies grow, people and their systems can move from one place to another to seize opportunities, improve the quality of their lives and to avoid risks. Migration is an essential social and economic phenomenon in all societies (Adger et al. 2015). There is diverse evidence that economic growth and demographic changes, particularly in developing countries drive migration (de Haas 2010b; Hugo 2011); and theories of migration have formed a basis for understanding the reasons behind human migration.

For instance, the neoclassical economic theory claims that wage disparities between sectors or countries, particularly labour-rich countries and capital-rich countries drive migration (Massey et al. 1993). Likewise, the dual labour market theory argues that migration is primarily caused by "pull" factors (i.e. the demand for cheap, unskilled labour) in developed, receiving countries rather than "push" factors in the sending countries (Piore 1979; Massey et al. 1993). The social networks theory of migration, however, focuses on the crucial role of social networks and ties (i.e. social capital) between former migrants and non-migrants at their origin, in identifying people's decisions and their ability to migrate (Massey 1990; de Haas 2010a). Another theory, the new economics of labour migration, explains migration as a risk-sharing decision made collectively by families or households rather than individually (Massey et al. 1993; de Haas 2010b). According to this theory, households make decisions to migrate to reduce their risk of loss of income due to crop failure, market-related failures (labour, credit, insurance), price fluctuations, and unemployment and to improve the household's overall economic well-being (Massey et al. 1993). Stark & Levhari (1982) consider migration of family members is a "clear strategy" rural households follow whenever there are either no alternatives for diversification of income, or if there are alternatives, then they are insufficient for their needs.

Migration theories have focused fundamentally on the economic, social, institutional, demographic and cultural factors of migration (Massey, Axinn & Ghimire 2010); nevertheless, it is widely agreed that environment is one of the key factors that influence human mobility (Hugo 1996; Warner et al. 2008; Jäger et al. 2009; Warner et al. 2009; Warner et al. 2010; Massey, Axinn & Ghimire 2010; Black et al. 2011a & 2011b; Foresight 2011; Afifi et al. 2016; Raha & Gayen 2021). In recent years, the effects of environmental change have already manifested themselves in several regions of the world, such as droughts in Yemen, flooding and landslides in Nepal and India, cyclones and storm surges in Pakistan, and several other incidences (Martin, Weerasinghe & Taylor 2014). As a result of the economic losses caused by these extreme weather events and natural disasters (Neumayer, Plümper & Barthel 2014), and the systematic environmental changes, migration and environment has become an evolving research discipline (Adger et al. 2015).

Climate change is expected to generate human movements not only due to increased frequency and intensity of natural disasters, but also due to potential new conflicts over scarce resources (Martin, Weerasinghe & Taylor 2014). The usual environmentally-related risks such as natural disasters, drought, crop failures, and the social and economic changes that take place within countries during development generate an economic environment where uncertainty and unpredictability prevail in both urban and rural areas (Massey 1990). Migration is, therefore, considered a well-known strategy to spread risks under challenging environmental conditions (Massey 1990; Adger & Adams 2013).

The influence of environmental factors on migration patterns is closely related to other factors – demographic, environmental, economic and socio-political – and hence, such a relationship entails complexity and multiple causalities (Foresight 2011). Climate change impacts has a multiplier effect on other drivers that influence human migration (Afifi & Warner 2008). In other words, the effects of any environmental events are multiplied by other structural elements like poor governance, high levels of poverty and inequality, insufficient access to basic services, conflict, poor health, unemployment, economic constraints, and weaknesses in both the local and national capacity to act against risks (Kniveton et al. 2008; Foresight 2011; Martin, Weerasinghe & Taylor 2014).

Consequently, migration as an adaptation to climate change is considered a complex phenomenon that is shaped by various drivers and is hardly triggered by a sole cause (Kniveton et al. 2008; Foresight 2011).

Although migration has also been viewed as a failure to adapt in several studies (Conisbee & Simms 2003; Renaud et al. 2007; Stern 2007; Warner et al. 2009), evidence has shown that migration as a consequence of climate change is not necessarily a negative outcome (Bardsley & Hugo 2010; Black et al. 2011a & 2011b; Piguet, Pecoud & de Guchteneire 2011; Adger & Adams 2013; Baldwin & Gemenne 2013; Gemenne & Blocher 2017). Migration is often an important strategy for households to diversify their incomes that is much needed in the context of environmental change (Foresight 2011; Sobczak & Fekih 2020). Within the agricultural sector, migration might help support agricultural production (Rigg 2007). Studies show that remittances sent to rural areas represent a significant source of capital to purchase land and machinery, adopt new crops, and even to expand non-farm employment in rural areas (Banzon-Bautista 1989; Gibson, Law & McKay 2001; McKay 2003; Asfaw et al. 2019; Sobczak & Fekih 2020).

When studying migration environment nexus, it is important to consider the extent to which migration is considered as a rational option of an array of adaptation alternatives or a failure of *in situ* adaptation measures i.e. whether people are forced to move or migrate as a precautionary adaptation to environmental changes (Bardsley & Hugo 2010). Climate change induced migration can take many forms, involve various degrees of voluntariness, and generate different outcomes (Hugo et al. 2009; Piguet, Pecoud & de Guchteneire 2011). Generally, there are three forms of movements as a response to any humanitarian crisis (Martin, Weerasinghe & Taylor 2014):

- 1. **Displacement**: includes people who are obliged to move as a result of being directly threatened by certain events.
- Anticipatory movement: encompasses people who choose to move as a result of some potential future risks to their living conditions. Such future risks would include slow-onset events such as rising sea-levels or increasing desertification, or drought.
- 3. **Relocation for trapped populations**: includes those individuals who are directly affected by certain crisis but who do not, or cannot, move and hence are in need of relocation.

Environmental change influences patterns of migration by altering the mix of economic activities at the location and altering the location itself (Adger & Adams 2013). Consequently, this study refers to *ex situ* adaptation through the broader context of mobility, both occupational and geographical mobility. Mobility entails the movement of people across jobs and physical space (Long & Ferrie 2006). Internal and international movements are thus understood as geographical mobility, while movement from the agricultural sector to other non-farm economic sectors is referred to as occupational mobility. Geographical mobility can also be further classified based upon the distance (short-distance or long-distance), and type of movement (forced or voluntary) (Long & Ferrie 2006). In this sense, migration is considered one of the forms of mobility and hence the terms "mobility" and "migration" are used interchangeably in the thesis.

The study of Rigg (2006) states that five main forces push rural households away from agriculture are:

- 1. The reduction in the profitability of small-lands;
- 2. The rise of non-farm employment;
- 3. Changes in the cultural and social context of rural areas such as migration and education;
- 4. Environmental losses; and
- 5. Shortages in resources.

Another critical dimension that should be considered when studying environmental induced migration, is the nature of the climate change events, whether slow-onset or sudden events (Bardsley & Hugo 2010). Slow onset events paired with environmental degradation and climate change may cause a gradual loss of people's resilience and governments' capacity to protect them (Bronen 2015). The study of Koubi, Stoll & Spilker (2016) suggests that slow-onset environmental events such as drought decreased the likelihood of migration. Consequently, understanding individuals and their household's adaptation strategies to slow-onset events, particularly migration decisions, if any, is crucial in understanding the "tipping point" or threshold of any environmental crisis. The likelihood that some people could be unable to migrate or move away from a specific environmental risk emphasises the importance of explaining the nature and causes of trapped populations that are discussed in the next section.

2.5.2.1 Immobility and trapped populations

One of the earliest attempts to explicitly theorise immobility was Standing's (1981) research. In his study, he noted that the global changes in the economic and political systems had reduced mobility and hence he challenged the assumptions of migration theories entailing the inevitable movement of poor and unemployed due to wage disparities. In one example, Standing (1981) believed that very poor households were unable to finance migration of their members or unable to spare the labour and suggesting that it is the most marginalised households that are more likely to become trapped.

Researchers have agreed that environmental factors play a key role in human mobility (Jäger et al. 2009; Warner et al. 2009; Van der Geest, Vrieling & Dietz 2010; Black & Collyer 2014; Afifi et al. 2016; Thornton et al. 2019; Raha & Gayen 2021). However, attention to immobility has gained momentum in recent years, particularly within the broader concerns about the impacts of climate change (Black et al. 2011a, 2011b; Black et al. 2013; Adger et al. 2015; Afifi et al. 2016; Ayeb-Karlsson, Smith & Kniveton 2018; Farbotko et al. 2020). Although many people move as a result of acute events or to avoid potential future risks, some others lack the resources or capacities to migrate (Martin, Weerasinghe & Taylor 2014; Warner & Afifi 2014; Black et al. 2013; Adams 2016).

Access to resources or capital generally influences the ability of individuals or households to migrate (Black & Collyer 2014), and lack of one or more forms of capital (social, cultural, human, economic, geographical and political) can be a burden to human mobility (Kothari 2003). Therefore, migration may not be a consequence of environmental change if people do not have sufficient economic resources, networks and capital (Adger & Adams 2013; Black & Collyer 2014). Environmental change can be barrier to movement rather than an encouraging factor (Black et al. 2011a; Foresight 2011; Gary & Mueller 2012). Climate change is expected to undermine already fragile economic livelihood, especially in those populations which are actually or potentially trapped in place due to poverty (Black et al. 2011a; Foresight 2011; Gray & Mueller 2012; Nawrotzki & DeWaard 2018). Climate change might bring about less migration not only through removing the resources needed for migration but also through increasing the need for labour at the place of origin (Foresight 2011; Gray & Mueller 2012). Such loss of individuals' assets and capital are among the main reasons that cause people's immobility and trapped populations (Black et al. 2013).

Black & Collyer (2014) identify "trapped" populations as those people who lack control over their decisions to move away or escape potential risk or threat. In their study, they highlighted the theoretical problem arising from distinguishing between individuals who aspire to move but are unable to, and those who do not wish to move. However, it is of great importance to advance our understanding of immobile populations as they are the most vulnerable to being "trapped" in more severe environmental crises. Being immobile magnifies the vulnerability of those individuals who are unable to move or incapable of coping; therefore, vulnerable populations will be more likely to be trapped and to bear the ultimate burdens and risks (Black & Collyer 2014).

Immobility leaves vulnerable populations who do not have the resources to migrate at increased risks (Adger & Adams 2013). Figure 2.8 shows the dynamics of vulnerability to environmental change, well-being, and mobility patterns. People who are most exposed and vulnerable to the impacts of climate change have the least capacity for migrating (Foresight 2011; Adger & Adams 2013). Similarly, low levels of capital increase the vulnerability of individuals to crisis and decrease their ability to move (Black & Collyer 2014). Therefore, it has been claimed that people are trapped by their lack of mobility (Black et al. 2013), and those trapped populations are the most to suffer a significant injustice (Adger & Adams 2013). Research emphasises that some people who depend on natural resources for living such as farmers, herders, fishermen and pastoralists are more exposed to climate stressors and maybe the least able to move (Betts 2010; Black et al. 2011a). These potentially "limited mobility" populations could face declining livelihood conditions within their local communities, with fewer possibilities for moving to more desirable places in a safe and effective manner (Adger & Adams 2013).

Figure 2.8: Relationship between vulnerability to environmental change and mobility representing trapped population



Source: Adger & Adams (2013)

Information on the presence and features of trapped populations is crucial to advise policy interventions (Martin, Weerasinghe & Taylor 2014; Zickgraf et al. 2016). Although trapped populations or those who choose to stay in areas vulnerable to climate change represent a significant policy concern, empirical work addressing such populations is very limited (Zickgraf 2019). Current information about the mechanism of trapped populations is inadequate to suggest any clear measures to be specified in policy aiming at reducing their vulnerability or enabling them to move when they feel they need to (Black & Collyer 2014). Consequently, advancing understandings of the reasons contributing to individuals' immobility could help guide policy responses to consider their real situation and needs (Black & Collyer 2014).

Researchers have criticised existing migration theories as they are more concerned with drivers of migration and they fail to address why populations stay (Hammer & Tamas 1997), and to distinguish between not being able to migrate and not wanting to and (Carling 2002). The study of Lubkemann (2008), for instance, found that females who were left behind in rural Mozambique were forced to be immobile during drought-prone seasons as a result of civil war that prevented them from their normal mobility during these times. Hence, recognising the fact that some people do not, and prefer not to, migrate could guide efforts in developing policy to assist trapped populations to adapt *in situ* (Findlay 2011).

Zickgraf (2018) argues that immobility is an outcome of the interaction between several factors including political, economic, social, environmental, demographic and cultural

factors Consequently, advancing research about the characteristics of immobile populations requires examining not only what constrains mobility (in terms of low assets and capital that reduce the capacity of individuals to migrate), but also what convinces people to migrate or stay. The following section will further discuss some determinants of immobility culturally embedded within some communities.

2.5.2.2 Culture of migration

Black at al. (2011a p.447) argue, "the greatest risks will be borne by those who are unable or unwilling to relocate". This unwillingness to migrate can be derived by culture. Adger et al. (2013) studied the cultural dimension of the effects of climate change and adaptation and argued that society's response to every dimension of climate change was influenced by culture. Migration is embedded in the identity and culture of societies (Hugo 2006; Adger & Adams 2013). Some communities and households choose to migrate while others experiencing similar conditions do not (McLeman & Smit 2006). The structure and history of migration and its continuity are factors that characterise past and present movements for most "sending" communities and create a "culture of migration" that supports movers and non-movers and enables network creation between origin and destination populations (Cohen 2004; Cohen & Sirkici 2011). Massey & España (1987) believe that people in a community from which a large number of its members have migrated should be more likely to migrate, whereas, people are unlikely to migrate from communities where migration is relatively unusual. Accordingly, some societies are more mobile than others, and in some countries, migration constitutes a fundamental basis to many aspects of cultural, economic and political influence in rural areas such as in the Philippines and across South East Asia (Lee 1985; Quisumbing & McNiven 2005; Kelly 2011; Knerr 2017). In Nepal for instance, the emigration rate for 2011 was estimated at 10.77 migrants per 1,000 population (IOM 2019). On the contrary, the rate in Egypt was only 1.9 per 1,000 population in 2017 (Forouheshfar, El Mekkaoui & d'Albis 2020). In 2020, it was estimated that 3.6 million Egyptian were residing abroad representing only 3.5 percent of the population (IOM 2020a). This percentage is considered to be lower than other countries in the MENA region such as Jordan and Lebanon where 8.1 percent and 10.7 percent of their populations were residing abroad in 2020, respectively (IOM 2020b, 2020c).

In Egypt, permanent emigration is mostly restricted to highly educated individuals working in highly skilled occupations such as health and engineering (EACH-FOR 2009). A recent study in the MENA region found that workers with higher levels of education are more likely to migrate than less qualified workers as their potential gains are usually higher (Ramos 2019). Temporary and circular migration, however, in Egypt has been traditionally important and continues to occur, but with much less significance (Zohry 2005). The number of temporary migrants to Gulf countries increased in 1974 after the oil boom, until the mid-1980s, with the Iran-Iraq war reducing oil prices and pushing down the number of Egyptian migrants (Zohry 2002, 2007 & 2014). The number of Egyptians emigrating continued to decline, and simultaneously, the flow of returning migrants from Gulf countries increased after the 1990 Gulf War between Iraq and Kuwait (Zohry 2007, 2014). Since the 2000s, the increased competition of immigrant workers from Asia as well as the desire of Gulf countries to replace their non-national labour force with citizen workers has further stagnated the share of Egyptian migrants in exploiting work opportunities in the Gulf (Zohry 2007). The rate of internal migration is also considered low due to the stagnant rate of urbanisation in Egypt since the 1970's which limits rural-urban migration flows (World Bank 2014).

Another key aspect that contributes to immobility, besides the capacity of individuals to migrate is their aspirations (Carling 2002). Carling (2002) suggests that immobility results from the lack of either the ability or the aspirations to migrate. Hence, people might have the capacity to migrate but lack the aspiration to migrate and *vice versa*. As a result of the two critical aspects of mobility/immobility (capacity and aspirations), there are four general categories of mobility (Figure 2.9):

- 1) Mobility (i.e. having both the ability and aspiration to migrate);
- 2) Voluntary immobility (i.e. having the ability but not the aspiration to migrate);
- 3) Involuntary immobility (i.e. having the ambition but not the ability to migrate); and
- Acquiescent immobility (i.e. neither having the aspirations nor the ability to migrate) (Carling 2002; Schewel 2019).

Low levels of aspiration to move may limit the prospects of what people visualize for their futures. Development processes and modernisation within societies tend to increase individuals' aspirations which thereby coincide with his or her increased desire to migrate (De Haas 2007, 2014). Migration itself is both a result and a cause of high aspirations;

even when taking into consideration socio-economic and demographic factors such as younger age, wealth, and better education, migrants tend to have higher aspirations than non-migrants (Czaika & Vothknecht 2014). Within the MENA region, Ramos (2019) found that the aspirations of qualified youth to migrate abroad differed from one country to another. Interestingly, more qualified youth in Palestine and Jordan, for instance, have more aspirations to migrate than those in Egypt (Ramos 2019).





The degree of attachment to a specific place is another cultural dimension that could influence decisions of individuals and communities to migrate or stay, and could lead to a situation of immobility. High levels of attachment to place may drive individuals to stay in a risky location (Adger & Adams 2013; Adams 2016; Singh et al. 2020). For example, Farbotko (2018) and Farbotko et al. (2020) found that many indigenous people in the Pacific are "voluntary immobile", expressing a high preference to stay in their lands for cultural reasons, even with the continuing deterioration in health and livelihoods as a result of climate change. Besides, the study of De Sherbinin et al. (2011) found that individuals resisted planned resettlement schemes even when they perceived that the risks of staying were high.

Fie (1939) was one of the lead researchers who tried to theorise immobility through founding the "Earthbound compulsion" theory to describe the attachment of Chinese peasants to their lands that contributed to low rates of migration in China at that time. The "rootedness of people" as the standard and favourable state of affairs is assumed to be a reason for immobility (Malkki 1992; Bakewell 2008). People develop knowledge and

Source: Schewel (2019)

skills relevant to a particular place or occupation and accordingly, staying in place can make economic sense, even if higher incomes may be found in another area (Schewel 2019). The individual sees a "location-specific capital" in the place of origin (DaVanzo 1980), and develops "bounded rationality" in which people lean towards choosing alternatives that are "good enough" instead of the best option among all those available (Simon 1982).

In conclusion, there can be social and cultural limits to adaptation that may well be related to the differences found among individuals in their experiences, interpretations and responses to climate change (Adger et al. 2007). Individuals may have different vulnerabilities and preferences about adaptation measures to risk, depending on their set of values, beliefs and their understanding about the world (Adger et al. 2007). Pre-existing migration patterns will have significant influence on future migration decisions of the different societies (Laczko & Piguet 2014). Recognising these cultural dimensions is critical for planning and governing successful adaptation responses to current and future risks associated with climate change.

2.6 Climate change perception and adaptation studies in Egypt

There is quite a large number of studies addressing perceptions of global, environmental change conducted in many regions of the world (Pyhälä et al. 2016). However, risk perception of climate change is a relatively new area of research in the MENA region and in Egypt specifically, and in the past few years, studies on public perceptions of climate change have been gaining momentum. The following section reviews the findings from previous studies addressing public perceptions of climate change, particularly in rural areas in the MENA region, including Egypt.

The study by Adoho & Wodon (2014b) addressed the perceptions of climate change held by householders in five MENA countries, including Egypt, Yemen, Morocco, Algeria, and Syria. The study showed that most households across all of the countries had perceived increases in temperature and changes in rainfall patterns, in particular, more frequent droughts and more erratic rainfall with significant adverse effects on crop and livestock production. Households also believe that diseases were increasing for livestock and that there were more insects and pests in crops. The study suggests that poor or lesswealthy households were more likely to suffer from crop and income losses due to climate change. On the local level, households from Egypt were less likely to perceive climate change and its adverse effects in comparison to other countries, specifically Syria, with higher levels of perceptions.

Omar et al. (2015) studied farmers' perceptions of climate change in Egypt through a survey conducted on 197 interviewees. Results of the study revealed that farmers had noticed an increase in temperatures, humidity and incidence of heatwaves, and a decrease in rainfall. Most farmers also perceived significant adverse effects on agricultural and livestock production in addition to increased severity of pest infestation and disease infection such as foot-and-mouth disease in cattle. Education, age and experience in agriculture were the key factors influencing farmers' perception of climate change. Similar results were revealed in the study by Froehlich and Al-Saidi (2018). They found that changes in temperature and patterns of rainfall, as well as increased incidence of extreme weather events, were the three most perceived changes in climate in the studied areas. Their study also suggested that farmers in rural areas experienced more animal diseases and crop pests, water shortages, and health problems/exhaustion due to climate change.

Hafez (2020) addressed perceptions of the effects of climate change on agriculture, livelihoods and women on 200 households in Egypt. Results showed that nearly half of the interviewees observed an increase in temperature, as well as in incidence of sudden wind and rainfall. On the contrary, the majority were not fully aware of climate change as a phenomenon and of means for reducing its associated risks. Similarly, the results of the study of Kassem et al. (2019) on farmers' awareness of climate change revealed that most of the farmers studied were not aware of climate change and consequently, its adverse effects.

Regarding adaptation, some attempts have been made to analyse how farmers adapt to climate change in Egypt and the MENA region. Most of these studies have addressed and confirmed that rural households were adopting several agricultural practices *in situ* to reduce the associated risks. In one example, Adaho & Wodon (2014c) found that rural inhabitants in the five MENA countries had mostly adopted four agricultural adaptation strategies including increasing the use of fertilisers and pesticides, changing farm production technology, storing grains and water, and seeking off-farm income. The study also found large differences between countries in the adoption of adaptation practices,

with households in Egypt and Syria making fewer changes to their modes of livelihood in comparison with Algeria, Yemen and Morocco.

Similarly, the study by Omar (2015) focused only on agricultural, adaptation measures *in situ* to reduce the risks of climate change. The study found that only a small percentage of farmers in rural Egypt have adapted to climate change, and the most-adopted measures included changing sowing dates, changing irrigation timing and increasing the frequency of irrigation. Her study found that adaptations to climate change positively correlated with education and perceptions of climate change risks.

Kassem et al. (2019) also found that most of the studied farmers in Egypt fell in the low adopter category, regarding *in situ* adaptation measures. The most popular on-farm practices included maximising the use of manure, changing cropping patterns, crop rotation, and cultivating drought-resistant varieties. The study also found that education, farm size, farm income diversification within the agricultural sector, and membership in water user associations were vital factors that influenced the likelihood of adoption of adaptation measures.

On *ex situ* adaptations, unlike other regions of the world, there is very little empirical evidence that environmental change and shocks influence migration in the MENA region, including Egypt (Laczko & Piguet 2014). Most of the existing research confirms the importance of the socio-economic and political factors in driving decisions about migration rather than environmental problems. For example, the study by Adaho & Wodon (2014b) on five MENA countries suggested that climate change and extreme weather events resulted in a higher possibility of migration; however, the role of climate as a driver of migration remained smaller as compared to socio-economic drivers and job prospects in the cities.

Likewise, the study of Afifi (2010) found that the effects of climate change alone are less likely to drive decisions to migration in rural Egypt. Despite the small sample size, his study suggested that people in Egypt would be migrating from one place to another within the country in the case of water shortage and/or land degradation only under certain conditions. These conditions included not owning land, being socially and financially capable of leaving their place of origin and/or being forcibly displaced by government or landowners. He concluded that rural people would not be willing to leave their homes as long as they were not facing a sudden natural disaster, such as earthquakes or floods. Hence, the study suggests that slow-onset climate events do not drive migration and also gave a notion about the importance of integrating the culture of migration of rural dwellers in the analysis of their adaptation preferences or decisions. Similarly, Warner et al. (2008) study of rural-urban migrants in Egypt found that problems with unemployment and poverty were the main factors influencing their decisions to migrate. Most importantly, they highlighted that land degradation, and water shortages in the place of their origin had caused these problems, although the migrants did not mention it explicitly.

In conclusion, perceptions of, and adaptation to, climate change are relatively new disciplines in the MENA region, and specifically in Egypt. Most of the local studies on these perceptions have focused on the level of perception and/or awareness of people (either in rural or urban areas) to climate change and its current related risks. This thesis, however, considers farmers' perceptions of several current and future risks associated with climate change in addition to those risks related to working in the agricultural sector under present and upcoming environmental challenges. Moreover, studies on adaptation in Egypt and MENA region have mostly focused on agricultural adaptation practices *in situ* that farmers follow, while only a few have addressed *ex situ* adaptation, providing evidence that climate change has less influence on their decision to migrate than other socio-economic factors. Consequently, this thesis aspires to address these gaps in current knowledge, focusing expressly on both *in situ* and *ex situ* adaptation practices followed by rural households in the study area.

This thesis also aims to integrate the culture of migration as an essential factor that could influence rural households' current and future decisions to migrate. More importantly, the study aims at integrating the paths of agrarian transition and the resultant class structures with risk perceptions and adaptations to climate change. It also pinpoints the likelihood of the presence of potentially trapped rural populations in Egypt and other countries with similar conditions in the developing world.

2.7 Summary

The chapter discussed the main theoretical approaches and conceptual basis for this thesis, looking at the current socio-economic, environmental, demographic and political issues in the MENA region. It then deliberated on the theoretical framework of the agrarian transition and then discussed the policy evolution of land and tenure in Egypt,

the different pathways of agricultural transformation and the resultant class structure in Egypt.

Previous local agrarian studies show that the agricultural and economic policies had already imposed several economic, social, political and often environmental risks on the agricultural sector. With the internal demographic, economic, social, political and environmental pressures prevalent in the MENA region and in Egypt specifically, climate change is further jeopardising the agricultural sector and the livelihoods of many rural dwellers in the region.

The chapter addressed the concept of perceptions of risk and their importance in guiding adaptation actions to climate change. It also explored, based on previous research, the contextual factors that influence individuals' and communities' perception of risk. Afterwards, the chapter outlined the concept of climate change adaptation and the related concepts of vulnerability and adaptive capacity. It broke down adaptation into *in situ* and *ex situ* measures and then emphasising the reasons behind immobility and trapped populations.

Chapter Three: Research Methodology

3.1 Introduction

This chapter describes the methodology adopted to investigate the research questions and meet the research objectives. This study adopts a mixed-method approach of data collection and has a deductive nature as it begins with formulating research questions that guided the data collection process (Neuman 2014). The chapter starts with a brief discussion of the epistemological and philosophical foundations of the study, including the reasoning behind the selection of the mixed-methods approach. Following that, there is a detailed discussion of the specific quantitative and qualitative methods and techniques used in the data collection process. Justification for choosing the study area is also included. The techniques and tools employed for data entry, processing, and analysis are introduced and will link across to the significant findings presented in the results chapters (Chapters 4, 5 and 6). Finally, the chapter concludes with the researcher's perspective and experiences in conducting the research data, including a brief outline of the research limitations.

3.2 Adopting an epistemological position

Epistemology concerns the theory of knowledge and is essential for framing the research approach because each researcher will hold a particular opinion about what constitutes acceptable knowledge (Bryman 2016). Epistemological and philosophical assumptions are defined as researchers' "assumptions about how they will learn and what they will learn during the inquiry" (Creswell 2003 p.3). Teddlie and Tashakori (2009) identify three major epistemological schools of thought in behavioural and social science research, including positivism, constructivism, and pragmatism. Among the three schools, pragmatism is the one that focuses primarily on the problem being researched and the consequences of the research itself (Feilzer 2010). Pragmatists hence, do not insist on a particular method or mix of methods and do not exclude any based on a fixed epistemological position (Feilzer 2010). Instead, they support the view that the choice of methodology should be the best to meet the needs of the research (Creswell 2003; Johanson & Onwuegbuzie 2004; Teddlie & Tashakori 2009).

Feilzer (2010) asserts that pragmatism is the best path through the dichotomy of positivism and constructivism. Positivism relies on empirical evidence, whereas

constructivism depends on learning from social interactions and psychological processes (Armstrong 2013). Consequently, pragmatism could be considered as the most suitable epistemological position for a mixed-methods approach, which in this case combines both a numerical, positivist approach and a narrative-based, constructivist approach to generating and analysing the data needed to answer the research questions (Bryman 2006; Teddlie & Tashakkori 2009; Feilzer 2010).

3.3 Research Approach: Mixed Methods

Following the developments of quantitative and then qualitative research, a mixedmethods approach is referred to as the "third methodological movement" (Tashakkori & Teddlie 2003, p.5), the "third research movement" or "third wave" (Johnson & Onwegbuzie 2004, p.17). Mixed methods research has become an accepted approach that has been increasingly adopted in social research (Bryman 2016). Johnson and Onwegbuzie (2004, p.17) define mixed-methods research as "the class of research where the researcher mixes or combines quantitative and qualitative research techniques, methods, approaches, concepts or language into a single study". Likewise, Tashakkori and Teddlie, (2003, p.711) define mixed methods as "a type of research design in which qualitative and quantitative approaches are used in types of questions, research methods, data collection and analysis procedures, and/or inferences".

Mixed-methods research entails mixing of at least one qualitative method and one quantitative method in the same research project (Hesse-Biber 2015). Quantitative research is more concerned with the collection and analysis of numerical data, whereas qualitative research emphasises words in the collection and analysis of research data (Bryman 2016). Quantitative research methods focus on the collection of numerical data that can be analysed by statistical techniques. In contrast, qualitative research methods are more concerned with exploring how individuals understand and put meanings to their social world (Walter 2013).

Mixed-methods research gives more evidence for investigating a research problem than either qualitative or quantitative research can provide solely (Creswell & Clar 2017). Mixed methodologies are useful in obtaining data that are more consistent and comprehensive and help broaden the scope and dimension of the study (Morse 2003; McKendrick 2009). The mixed-methods approach allows the researcher to utilise the strengths of both quantitative and qualitative data jointly, to obtain a better understanding of the research problems than using either approach alone (Creswell 2015). The use of both qualitative and quantitative methods either sequentially or independently improves the findings of a study (Teddlie & Tashakkori 2009). Moreover, Neuman (2014) suggests the use of more than one method of data collection whenever the research has a descriptive and explanatory nature to ensure that a holistic analysis is developed. Similarly, Bryman (2016) suggests that using mixed methods enables the researcher to address the research questions from multiple dimensions and to cross-check data gathered from the field through a process of triangulation, hence strengthening the validity and credibility of the study results.

The primary aim of this research is to understand how agricultural communities perceive and respond to the risks associated with climate change. Applying a mixed-methods approach to the collection and analysis of the data helps to provide a more complete or in-depth analysis of the agrarian communities in rural Egypt. Given that a mixed-methods approach is adopted, there is a considerable effort made to try and consolidate the knowledge generated by comparing and contrasting the quantitative and qualitative data. Although this study prioritises quantitative collection and analysis of data, qualitative data and findings will illustrate, check and correct the quantitative conclusions. Primary quantitative data were collected from the survey, whereas primary qualitative data were gathered using in-depth interviews with farmers within the areas under investigation. The core findings of the study were obtained from the analysis of survey data, while data collected from in-depth interviews supported the development of an understanding of farmers' perception of risk and adaptation choices within the Egyptian context. The following section (3.4) will discuss in further detail the approaches taken for primary data collection during the study.

3.4 Data collection approaches

3.4.1 Unit of Analysis and levels and dimensions of data

This study employed a multilevel model (individual, household, community levels) in designing the tools to collect data for different levels of information through household surveys and in-depth interviews with farmers. The study is mainly concerned with the perceptions of small landholder and landless farmers, who work as waged agricultural labourers, and their households in northern Egypt. Small landholders in this study are defined as those farmers who own, rent or share agricultural lands with less than 5 *feddans* (11.9 ha). Individual and household information was gathered through household surveys using a structured questionnaire in face-to-face interviews. Further qualitative in-depth data were collected from interviews with farmers regarding their local experiences with climate change, and their opinions of the biggest local environmental problems and of the government interventions aiming at developing their community and reducing risks of climate change. Table (3.1) summarises the levels and dimensions of data sought.

		r		
Data collection tool	Information sought	Level of data		
		Individual	Household	Community
Questionnaire	Perceptions of climate change (observed	Х	Х	
survey	ey patterns, frequency, and impacts)			
	Adaptation responses (in situ and ex situ)		Х	
	History of migration	Х	X	
	Migration culture	X	Х	
	Demographic characteristics	X	Х	
In-depth interviews	Experiences with climate change	X	X	Х
	Local environmental challenges	Х	Х	X
	Government interventions to reduce risks associated with climate change	X	X	X

Table 3.1: Dimension	ns and level o	of data collection
----------------------	----------------	--------------------

3.4.2 Survey

Surveys are the main method for collecting quantitative data (Bryman 1984). "Surveys are arguably the most-used tool for social scientists, market research and a variety for others looking for information regarding people's attitudes, behaviours and experiences"

(Abbott & McKinney 2013 p.36). Walter (2006) refers to survey research as, "the collection and analysis of respondents' answers to the same set of structured questions". One of the significant advantages of the survey is that it allowed the researcher to generate data from a large number of respondents within a short time (Neuman 2014). Surveys typically provide information on facts observed by respondents, their behaviours, perceptions and attitudes (Dane 1990 p.121). Hence, the survey would be the most suitable quantitative data collection method for this study as it aims to gain insights on small landholder farmers' perceptions of risk, adaptation behaviours and experiences with climate change, in addition to their opinions about working in the agricultural sector under current and future environmental changes.

The survey was started on 22 April 2018 and ended on 30 September 2018. Before the survey was conducted, several preparatory steps took place, including applying for ethics approval, selecting the study area and designing the questionnaire, sampling methods and population. Pre-testing the questionnaire and choosing and providing training to enumerators also took place, as well as planning and handling fieldwork. These steps are introduced in order below.

3.4.2.1 Justification of the case study areas

There are three levels of administration in Egypt: governorates, districts and villages. Egypt is divided into 26 governorates with four that are administratively considered as Urban Governorates (Cairo, Alexandria, Suez and Port Said) with no rural population. Whereas, each of the other 22 governorates is subdivided into urban and rural areas (UNICEF 2018). The Nile Delta region of Lower Egypt holds nine of these governorates, while eight governorates are located in the Nile Valley of Upper Egypt, and five are located on the eastern and western boundaries of Egypt. The Egyptian administrative levels are illustrated in Figure 3.1.

Figure 3.1: Local Administration Levels of Egypt



Source: El Megharbel (2011)

The geographical area of study was chosen from rural areas of the Nile Delta region. The Nile delta is located in the northern part of Egypt overlooking the Mediterranean Sea having a total area of about 20,000 km² (Figure 3.2). The total length of the delta coast is 250 km^2 extending from Alexandria (west) to Port Said (east) governorates.

Figure 3.2: The Egyptian Nile Delta region



Source: Google maps

This study chose the Nile Delta region to conduct fieldwork for three reasons. Firstly, this region is strategically and economically important as it contains the core fertile agricultural lands of Egypt and produces almost half of all national crop production (MALR 2009). Besides, the region hosts major economic hubs, including the ports of Alexandria, Rosetta, Damietta and Port Said.

Secondly, as introduced in chapter 2, Egypt's coastal zones are reported as being particularly vulnerable to the various effects of climate change including rising sea levels and the other potential effects that would affect their water and coastal resources, agricultural productivity, tourism, and human settlements (EEAA 2010). As sea level rises, the area between Damietta and Rosetta is expected to disintegrate into distinct

islands surrounded by water due to the uneven topographical nature of the coastal area (EEAA 2016).

Thirdly, besides the high rate of population growth (hosting 50% of the total Egyptian population of 93 million), the Nile Delta region is suffering from many problems including high rates of urbanisation, excessive erosion rates, and land subsidence, relatively low elevation, excessive pollution and degradation of the natural ecosystem and weak institutional management systems (MALR 2009).

Out of the ten governorates located in the Nile Delta Region, Damietta governorate was the focus for the fieldwork. Damietta governorate is situated along the branch of the Damietta Nile. Damietta governorate's coastal location makes it one of the most vulnerable to sea-level rise (SLR), with projections that parts of Damietta will be inundated between 2040 and 2050 (EEAA 2016; Elshinnawy & Almaliki 2021). It is a peninsula embraced from the north and west by the Mediterranean Sea, and to the east by El Manzala Lake, as shown in Figures 3.3 and 3.4.





Source: Google maps

The governorate's economy relies on a diverse number of economic activities besides agricultural production. Damietta is well-known for its furniture industry, fishing, production of cheese and other dairy products, and patisserie (SIS 2009).

Figure 3.4: Map of Damietta



Source: El-Gammal, Ali & Eissa (2014)

In 2018, the total population of Damietta was estimated as being 1,496,765, out of which 39.4 percent of the population are living in urban areas, whereas the majority (60.6 percent) live in rural areas (Damietta's Census Data for 2018). The governorate consists of five central administrations, ten cities, 47 local units, 85 villages and 486 hamlets.

The five central administrations (*marakez*) are Damietta, Faraskur, Elzarqa, Kafr Saad, and Kafr Elbateekh. Figure 3.5 shows the total land area and agricultural land area within each central administration in square kilometres and including fallow/uncultivated land areas.

Figure 3.5: Agricultural land area relative to the total inhabited land area of the five central administrations of Damietta (km²)



Source: Adapted from Information and Decision-making Support Unit of Damietta (2017)

Faraskur and Kafr Saad were chosen as the two central administrations for the survey for three reasons. Firstly, the two central administrations together hold the largest agricultural land areas: representing 55.8 percent of the total agricultural area of the governorate (18.4 percent in Faraskur and 37.4 percent in Kafr Saad) as shown in Table 3.2. Secondly, roughly 46 percent of the total rural population of the governorate lives in Faraskur and Kafr Saad (25.3 percent for Kafr Saad and 20.6 percent for Faraskur) as shown in Figure 3.6. Thirdly, the two local administrations together hold 53.2 percent of the governorates' small-landholder farmers. Kafr Saad has the highest number of small-land holders with 33,340 holders (representing 37.0 percent), and Faraskur comes in third place with 14,611 small-land holder farmers (representing 16.2 percent).





Source: Adapted from Damietta's census data (2018)

Central Administration	Farm size (<i>feddan</i> *)	Agricultural Area		Holdings	
		Feddan	Percentage	Holders	Percentage
Damietta	< 5	9,502	67.9	14,821	97.2
	≥5	4,490	32.1	431	2.8
Total		13,992	100	15,252	100
Faraskur	< 5	11,451	61.3	14,611	95.6
	≥5	7,223	38.7	667	4.4
Total		18,674	100	15,278	100
Kafr Elbateekh	< 5	12,438	72.3	13,161	95.9
	≥5	4,754	27.7	559	4.1
Total		17,192	100	13,720	100
Kafr Saad	< 5	26,537	69.8	33,340	96.6
	≥5	11,498	30.2	1,183	3.4
Total		38,035	100	34,523	100
Elzarqa	< 5	8,730	63.0	14,320	97.2
	≥5	5,125	37.0	411	2.8
Total		13,855	100	14,731	100
Total Governorate		101,748		93,504	

 Table 3.2: Area of cultivated agricultural lands and number of holdings by central administrations as in 2017

Source: Adapted from Damietta's Department of Agriculture data (2017)

During the preliminary and pre-testing stages of the fieldwork, it was revealed that several villages of Faraskur were suffering from severe water pollution, as the water used for both irrigation and drinking is mixed with sewage water and agricultural wastewater. Therefore, the area was found to be interesting to target an investigating of how current environmental problems could further interact with the impacts of climate change and affect the livelihood of farmers living there. A detailed description of the conditions of the study areas and the studied households will be further discussed in Chapter 5.

Four villages have been included in this study; two villages affiliated to each central administration (pinpointed in Figure 3.7). The villages were chosen based upon the variation in crops cultivated within each village, the availability of a local facilitator from each village who was willing to help, and the ease of accessibility of the village to decrease costs and efforts related to data collection. The study villages are Hagaga, Kafr El-Arab, Kafr Suliman El-Bahary, and Kafr Saad El-Balad. Hagaga was specifically chosen due to the persistence of severe water pollution in the village, which was observed during the preliminary visits.



Figure 3.7: The location of the studied villages in Damietta governorate map

Source: Damietta's Department of Agriculture (2018)

3.4.2.2 Sampling

Based upon the statistics sample size calculator tool¹, a random sample size of 384 respondents was calculated with a 5% margin error and a level of significance equivalent to 95% in the two selected central administrations with a population of more than 500,000. However, after data cleaning, 350 valid responses were used in the analysis.

¹ The tool used to calculate the sample size can be found in <u>http://www.raosoft.com/samplesize.html</u>

The sample units involved:

- 1. Smallholder farmers (farmers who are either renting or owning less than 5 *feddans* (=2.1 ha.); and
- 2. Landless agricultural labourers both women and men aged between 18 years of age and over.

A de-identifiable list of farmers was obtained from the local agricultural organisations, and respondents were chosen by simple random sampling. However, the main methodological issue is involved with the lack of a sample frame of agricultural labourers especially temporary workers. Therefore, snowball sampling was adopted to provide a sample of workers. Snowball sampling consists of gathering respondents by the referral methods by which each respondent who volunteers to join the survey is asked to identify one or more people with certain characteristics who would be willing to participate in the study (Babbie 2011; Hibberts, Johnson & Hudson 2012). The distribution of sampled households by the four villages is shown in Table 3.3.

Local Administration	Village	Sample size
Faraskur (n=180)	Hagaga	110
	Kafr El-Arab	70
Kafr Saad (n=170)	Kafr Saad Elbalad	85
	Kafr Suliman Elbahary	85
Total		350

Table 3.3: The distribution of sampled households by survey locations

Source: Field survey 2018

3.4.2.3 Questionnaire Development

Questionnaires are the most suitable data collection tools for providing more descriptive and comprehensive information about human actions and behaviour patterns (Preston 2009; Sahu 2013). In this study, a structured questionnaire was designed for rural households, with a focus on hired agricultural labourers and small landholder farmer respondents. The questionnaire was formulated to gather individual and household data adopting the insights of the New Economics of Migration theory (NELM), which states that migration is largely a household decision, rather than just an individual one. The questions were organised into several sections (as shown Table 3.4) in line with the main objectives of this study to seek information on:

- a) the social, economic and demographic conditions and other related aspects of rural agricultural households;
- b) previous migration experience and current/future migration intentions;
- c) employment conditions and job satisfaction; and
- d) environmental factors particularly climate variability and extremes and current decisions regarding adaptation.

-	-	
Section	Sub-section	Information sought
А	Household information	Individual information: gender, educational level, marital status, age
	Employment situation	Primary and secondary occupations, distance to work, the reason for
		unemployment, type of employment
	Perception of working in	The degree of exposure to issues related to working in the sector
	agriculture	
В	Previous movements (2012-	Individual movements, time, duration, destination, reasons for
	2017)	migration / non-migration
	Migration culture	Opinions on migration and encouraging family members to migrate,
		the likelihood of migrating / leaving the agricultural sector
	Remittances	The situation, regularity and uses.
С	Perception of climate variability	Experience with climate variability; patterns and frequency. Climate
	and extremes and their impacts	change impacts on crop quality and productivity, health, income,
		housing, employment, livestock productivity, labour productivity,
		and communication.
	Adaptation to climate variability	The measures adopted by the household to adapt to climate
	and extremes	variability and extremes, including on-farm / in situ adaptation
		practices and other household practices, including <i>ex situ</i> actions.
D	Social capital	Social relationships, the institution, and/or people who provide help
		to the households in times of need, participation in community
		events, membership in any formal or informal groups,
		communication means, etc.
	Economic conditions	Household's monthly expenditure, the contribution of income
	TT 1.1 11.1	sources to the total annual income, assets ownership.
	Health conditions	Health ranking of household members, health problems, satisfaction with local health services
	Housing and transportation	Property: type size facilities present and satisfaction
	flousing and transportation	Transportation: Distance to some places as primary school market
		town, main road, public transport, etc.
Е	Land Use and patterns of	Land tenure, size, current use, marketing channels, members
	production	responsible for the different agricultural operations, satisfaction with
		inputs, short-term agricultural problems encountered and adaptation
		response, etc.
	Perception of future risks and	The most significant risk in agriculture in the next 10 years, current
	related governmental actions	government support and its impact on farm business and future
		actions needed. Opinion on the required government actions in the
		future.
F	Movement intentions and plans	Individual and household occupational and geographical movement
		plans (temporary and permanent) for the next 12 months.

Table 3.4: Summary of content of the questionnaire

The questionnaire was first designed in English and then translated into Arabic, which is the official language of Egypt. The final Arabic version of the questionnaire incorporated feedback from the pilot study, which was later translated into the English version that can be found in Appendix 1. Closed-ended questions were the main types of questions included in the questionnaire; however, there are several open-ended questions requesting open-ended responses dispersed amongst them. Open-ended questions allow the respondent who is being interviewed to elaborate on any issues and answer on their terms without being forced to choose from pre-determined choices (Bryman 2016). Open-ended questions were used to enable the generation of qualitative data from respondents shedding light on responses to specific closed-ended questions. For example, respondents were asked if they would like to comment further on their perceptions of climate change and the adaptation strategies followed by the households to reduce risks of climate change. The partially open-ended responses were in the form of "others" so that respondents could provide answers that might not be mentioned. For illustration, in the section measuring social capital, respondents were asked to mention "other" individuals or institutions that could have provided help in times of difficulties other than the mentioned categories. Several filter questions were also included in the questionnaire so that the respondents could only answer relevant questions.

3.4.2.4 Fieldwork

Organising and training interviewers: The researcher had to work with a team of enumerators given the size of work and the limited time frame for data collection. The team of enumerators consisted of six graduates from the Department of Rural Sociology, Faculty of Agriculture, Cairo University, and the researcher. The members of the team were chosen based on their previous experience in data collection and their sound knowledge of the nature of rural agricultural areas in Egypt. As suggested by Bryman (2016), the enumerators need training in contacting potential respondents and providing an introduction to the study, using suitable probing styles and reading out questions as written and recording the exact answer. Since the questionnaire was somehow complex and comprehensive, it took three days to train the enumerators. The training introduced the research aims and questions to the team. It gave an item-by-item review of the questionnaire to explain how questions should be delivered and what kind of explanations should be provided to respondents. It also provided clarification for ambiguous questions and procedures related to ethics and safety considerations in the field.

This review was followed by a mock interview in which pairs practised the questionnaire together. In the beginning, the interviewer took from 45 to 60 minutes to complete the interview, but after several attempts, each enumerator was expected to finish the questionnaire within 30 minutes only. This training was crucial for decreasing variations among interviewers during the administration of the survey.

Contact and cooperation with local agricultural organisations: Being a staff member of Cairo University, the researcher requested three formal letters from its Dean to the general directors of Damietta's Department of Agriculture, the Information and Decisionmaking Unit, and Agricultural Extension Unit. These letters were essential for conducting fieldwork in Egypt; otherwise, no one would have cooperated. The letters formally introduced the researcher and her work, and requested the specified institutions to cooperate in facilitating the fieldwork and in providing the required secondary data. The researcher personally travelled for this initial contact and to introduce the aims and contents of the survey. A copy of the questionnaire was also handed in so that officials could review it and have a better picture of the goals of the research. It was also essential to obtain consent from the higher levels of government before the survey, otherwise, the local authorities could have terminated the survey at any time. The director of Damietta's Department of Agriculture, with whom the researcher has a good relationship as the result of previous work in Damietta, referred her to the responsible persons in the agricultural association in the selected local administrations so that they provided her with the required support.

Face-to-face interviews: The questionnaires were administered by face-to-face interviews. Interviews are the most common modes of data collection and allow direct interaction between the researcher and the participants. A face-to-face interview also has several advantages over other modes of data collection, including the higher response rate and the lower possibility of questions being misinterpreted (Neuman 2012; Walter 2013; Davies & Hughes 2014). Face-to-face interviews are beneficial whenever the respondents are illiterate or have low educational levels, which is common in Egyptian rural areas. All interviews were conducted in public places to ensure the safety of the interviewers. Having a facilitator from local agricultural associations accompanying the interviewers helped in gaining the trust of rural dwellers and ensured their safety as well. Most of the respondents were very friendly and cooperative and showed respect to the enumerators' team. Before each interview, the interviewer provided a verbal briefing on the scope of

the study, and respondents were assured that they would remain anonymous with the right to withdraw from the interview at any point. People who were 18 years of age or older were invited to join the survey. Only two respondents terminated the interview halfway through, which reflects the high response rate of face-to-face interviews.

3.4.3 In-depth Interviews

In-depth interviewing is considered a fundamental method of exploring social meanings in social science research (Walter 2013). In-depth interviewing is more like an openended conversation between an interviewer and a participant, and it is usually guided by a set of general themes (Travers 2006). Such interviews tend to be flexible and allow the interviewer to ask additional questions, explore issues as the interviewee raises them, and express his or her opinion where appropriate (Travers 2006). Qualitative interviews allow the interviewer to even adjust the emphasis in the research as a result of any emerging significant issues to generate qualitative data that is focused on particular concerns or opportunities (Bryman 2016).

The interviews were conducted to collect and justify information about individual and household opinions and experiences with the effects of climate change on their livelihoods and their current and potential migratory behaviour. The interviews also gathered community-level information on topics related to current problems facing the communities, significant future environmental threats, government adaptation measures, and expected community and governmental adaptation actions. It was essential in this study to adopt in-depth interviews to allow the interviewees, the local farmers, to express their opinions about the challenges within the agricultural sector and issues related to adaptation to climate change, particularly those *in situ* measures.

In total, eleven in-depth interviews were undertaken in Damietta governorate. During the survey, respondents were asked if they were willing to participate in a follow-up, in-depth interview. The selection of interviewees was random. The interviewees were un-identifiable and anonymous, and are referred to in the text by their "interview number". All interviews were completed in a one-on-one format by the researcher alone and were conducted in public areas. The average length of an interview was about 20 minutes. Most of the interviewees consented to having the interview audio-recorded, with only two interviewees preferring not to be audio-recorded, so the interviewer had to take manual

notes. All interviews were conducted in Arabic, and later they were transcribed and translated into English.

The interviews were used to capture detailed experiences on pre-determined themed topics like farming, climate change, and adaptation, including *ex situ* experiences. An interview guide involving the general themes of the research was set before the fieldwork started. The first theme involved the main challenges within the Egyptian agricultural sector. The next theme was more concerned with local environmental problems and climate change experiences, while the third theme discussed adaptation measures followed with an emphasis on local migration experiences. The fourth theme was related to projections about the future of agriculture in Egypt and whether people would leave agriculture or migrate if the situation became more intolerable for farmers and labourers. And the final theme was concerned with knowledge and opinions about government actions that were present, if any, that were aimed at improving local livelihoods of rural households in the light of climate change.

3.5 Secondary data

Pawar (2004) argues that the collection of secondary data is as crucial as collecting primary data to achieve meaningful research. The primary qualitative and quantitative data gathered through the survey and interviews were complemented by secondary data collected from mainly four sources as follows:

- Reviewing academic literature around climate change in the region, agricultural practices, adaptation and coping mechanisms, and migration within the Egyptian context. The main aim of the secondary data review was to help the researcher establish a basis for the research, based upon methods and tools in the field, and theories that have been applied for similar studies (Bryman 2016);
- Government documents such as annual reports, policies, and planning reports. These data were useful in describing the broad institutional context within which the Egyptian agricultural sector is operating;
- 3. Public data, such as census data and demographic data. These data were instrumental in understanding the general characteristics of the population; and
- 4. Local records and reports from government officials in Damietta. During the fieldwork, the researcher visited local government offices, including Damietta's Department of Agriculture and the local Information and Decision-making
Support Unit to access a range of data, reports, and maps of the study area. These included land tenure data, agricultural profiles, and types of crops cultivated in each central administration in Damietta over the five years that preceded the survey.

Secondary data sources were fundamental to helping the understanding of the current situation in the study region, and in building the complex arguments in the research.

3.6 Data Analysis techniques

3.6.1 Quantitative data analysis

For social data to be analysed statistically, they need first to be converted into a numerical form (Babbie 2013). The questionnaire was pre-coded, meaning that each question was assigned a variable code, and every possible response was assigned a particular value. This process of assigning codes or values to data is referred to as coding (Bryman 2016; Babbie 2016). SPSS 21 software was used to create a database for the questionnaire survey data by first defining the variables within the questionnaire and assigning their specific codes to facilitate the following stage of the data entry. Open-ended questions were translated into English and were also post-coded before data entry. Data entry was conducted by the researcher solely, and the complete database was checked and cleaned carefully to ensure the accuracy of the data.

Quantitative data analysis is usually either descriptive or explanatory (Babbie 2013). The study used two approaches in analysing the quantitative data, which were descriptive analysis and non-parametric statistical analysis. Descriptive analysis was used to summarise perceptions of climate change effects, demographic characteristics, the culture of migration, opinions on the agricultural sector, future risks, and so forth. Univariate analysis was used to analyse and represent data in the form of tables, bar charts, and pie charts. Continuous variables (ratio or interval) such as age, land area, and other variables measured on a 10-point scale were re-coded into ordinal variables for easier representation and interpretation.

Non-parametric inferential statistics were employed to identify relationships and establish an association between variables. Non-parametric statistical methods are robust and easier to use (McHugh 2013). The Kruskal–Wallis test or one-way analysis of variance (ANOVA) on-ranks test was the primary technique used in the study to compare between two or more independent groups of samples of equal or different sample sizes. The identification of three categories or groups of rural households was introduced and justified in Chapter 5. The test assumes that the population was not normally distributed and tests against the null hypothesis that the median values of the three groups were equal. The test was employed to compare the three groups of households and capture any significant differences found among them in some demographic characteristics, land area acquisition, perceptions of the effects of climate change and levels of adaptation, and opinions about migration. The results of the Kruskal–Wallis test are explained in Chapters 5 and 6. The dependent variables used for this test were measured at a continuous level which was conditional on validating the use of the Kruskal–Wallis test.

A Chi-squared test was also conveyed to test the significance of ordinal or nominal data. This was done, for example, to identify significant differences between households living in each local administration in their level of perception of the effects of climate change. The techniques mentioned above for quantitative analysis were used extensively throughout this study, and their results were supplemented with qualitative evidence to explain further and validate the key findings.

3.6.2 Qualitative data analysis

Qualitative data for this study were primarily gathered through in-depth interviews. The interviews were transcribed and translated into English, and the key concepts and similar experiences related by the interviewees were identified and coded as themes. Direct quotes from the interviews have been used in the results chapters to illustrate, supplement, and validate quantitative results when needed. Quotations from the interviews were used mainly to reflect farmers' perceptions of the effects of climate change and other risks associated with working in agriculture, and are given in Chapter 5. They were also used to supplement quantitative data to explain local adaptation measures, particularly those decisions related to *ex situ* adaptations in Chapter 6.

3.7 Ethical considerations

This study was conducted under the *Australian Code for the Responsible conduct of Research*. The research was approved by the Human Research Ethics committee (HREC) of the University of Adelaide on 17th April 2018, approval no. H-2018-012. The researcher sought the consent of all participants in the survey and interviews before the

commencement. The principle of consent is concerned with ensuring that the respondents are fully informed about the nature of the research and their participation, and that they would not be forced to participate (Bryman 2016). Each participant was given the details of the survey, the participant information sheet, and the signed consent form to retain in the Arabic language. Before each interview, the enumerator team and the researcher explained to the respondents that they were under no obligation to participate in the study, and they could withdraw at any point during the interview. The researcher also sought permission from participants to audio-record their interviews.

To ensure anonymity, the researcher did not collect any identifying data such as names, addresses, or phone numbers from the respondents. Consequently, individual respondents were protected from identification as participants in the study. The researcher also ensured the confidentiality of respondents so the information they provide could not be linked to them. This was very important as some questions might have touched on specific political issues, such as expressing opinions about government actions. The researcher clearly explained that the research was being conducted through an independent university and that her work was not related to any governmental bodies. All interviews were conducted in public areas, such as the local agricultural association or a local coffee shop, instead of private spaces, to ensure the safety of respondents and the enumerators' team and to avoid any feeling of discomfort among the participants.

3.8 Post-study reflections and limitations of the research

Being an Egyptian and having prior experience in fieldwork in Damietta helped the researcher to communicate easily with local authorities and gain access to the study sites. Moreover, informing the local authorities formally through Cairo University was extremely helpful in facilitating their participation. It is also important to mention that although the researcher was able to communicate and mingle with the locals actively, the researcher was still considered a "female outsider" with an urban cultural background, as she was born and raised in Cairo. Typically, most of the time, the researcher had to share her name and home location, and in many incidences, she was asked about her marital status and family situation. Stemming from the different rural cultural background, the respondents adopted certain attitudes and perceptions of roles that females should typically perform. Hearing comments like, "Does your husband agree to you travelling on your own for doing research?"; "You should be beside your daughter"; "Why are you

exerting so much effort travelling from Cairo to do research?" and "Why are you doing that to yourself?" all the time was the most challenging component of the fieldwork. Although this situation was very discomforting for the researcher, conversely, it helped encourage farmers to participate in the study due perhaps to a feeling that they needed to help the researcher, a mother, to return to her toddler.

The fieldwork was conducted during spring and summer, during which several intense heatwaves occurred. These extreme events could have influenced/supported some responses regarding perceptions of climate change that are explained in detail in Chapter 5. Travelling from Cairo to Damietta and commuting between the study sites at a temperature reaching 45°C was very difficult for the researcher and the accompanying team.

Fieldwork started at the beginning of summer crop cultivation, particularly after the declaration of limiting rice cultivation by the government to conserve Nile water resources (Reuters 2018; Kassem et al. 2019). This policy was of significant concern to local farmers, and it was their primary focus of complaint during the survey and interviews. The researcher had to continuously refocus the attention of the respondents on the main objectives of the study and interview, while assuring them that some questions would cover this issue, and that more notes would be taken in the open-ended questions. Moreover, the researcher and the accompanying team were mistakenly understood to be government officials by non-participating farmers, who approached us on many occasions during interviews to complain about the rice policy, thinking that we could voice their opinions to higher authorities.

Secondary data about migration in Damietta was minimal, and the only data that could have been obtained was the number of overseas work permits provided by Damietta's Police Department. Overseas work permits were very difficult to obtain for two reasons; firstly, the safety of the researcher given that she is affiliated to a foreign independent university was not assured, and secondly, the need for national security approval that could take up to one year to be completed. Moreover, any data would only capture international migration data, and data on internal migration would still be limited.

The questionnaire was conduct in a face-to-face interview mode due to the low literacy rate of the study participants, who found it challenging to understand the questions thoroughly. The interviewers provided explanation and clarification of questions as needed during the interviews. Lastly, the in-depth interviews were all conducted in an Arabic dialect and then translated into English. Some information may be misinterpreted in the process of translation, given the difficulty of the process from the local dialect. However, the researcher read and compared the transcribed and translated information several times and discussed some issues with local people to avoid this problem.

3.9 Summary

This chapter has discussed the design, approach, and methods used to conduct this research and meet its aims, as well as justifying the choice of a mixed-methods approach to answer the research questions best. The chapter also provided an overview of the study area, its administrative divisions, and reasons for selecting the study sites. Primary quantitative and qualitative data were collected using a questionnaire survey and in-depth interviews. In total, 350 respondents were surveyed in a face-to-face interview, and further 11 in-depth interviews were conducted over the period from April to October 2018. The SPSS 21 software was used for quantitative data entry and analysis.

The main statistical techniques used to analyse quantitative data involved descriptive statistical analysis and the use of other non-parametric statistical tests, particularly the Kruskal–Wallis and Chi-squared test to measure significant relationships between dependent and independent variables, while the qualitative data were transcribed, translated and analysed using a thematic approach. Triangulation was achieved through the use of various methods in data collection and analysis, which increased the validity of the research findings. The results of the analysis will be discussed in detail in Chapters 4, 5 and 6.

Chapter Four: An overview of the livelihood conditions of rural households in Damietta Governorate

4.1 Introduction

This chapter provides an overview of the livelihood conditions of households in the study area. It starts with a comprehensive analysis of the key demographic data for the sampled households. The analysis also outlines the social, economic and environmental conditions of the studied villages in Damietta governorate to provide a general understanding of the local environment and the agricultural situation in the villages. Hence, this chapter is essential for providing the context for discussing farmers' perceptions of risks caused by climate change mentioned in Chapter 5 and the options for adaptation to climate variability and extremes discussed in Chapter 6. In addition, it provides a justification for the importance of distinguishing between different groups of rural households based upon their income from agricultural activities, which is the basis of the detailed statistical analysis adopted in this study.

4.2 General characteristics of the studied households

4.2.1 Household size, gender and age

A total of 322 respondents (representing 92.0 percent) out of the 350 respondents are the heads of their households (Table 4.1). Other respondents include spouses (5.4 percent), sons or daughters (1.1 percent) and parents and parents-in-law (0.9 and 0.6 percent, respectively). This result suggests that the respondents are mostly the household heads, and are therefore the main decision-makers within their immediate context.

Table 4.1: The relationsh	p of the respondents to	the household's head
---------------------------	-------------------------	----------------------

		Frequency	Percentage
Respondents	Head of the household	322	92.0
	Spouse	19	5.4
	Son/daughter	4	1.1
	Parent	3	0.9
	Parent-in-law	2	0.6
	Total	350	100.0

The studied households have an average of approximately 5 members each, with a maximum of 13 household members and a minimum of 1 member. Most households (62.0 percent) are medium-sized (4–6 members) for rural Egypt, as displayed in Figure 4.1. Large households (7 members and more accounted for 13 percent of the sample, and 25.0 percent are small households (3 members or fewer). Similarly, the census of Damietta (2018) also shows that most rural households (62.2 percent) are medium-sized.



Figure 4.1: Distribution of households by household size in Damietta, 2018

Within the 350 interviewed households, 54.1 percent of the 1630 members are males, and 45.9 percent are females (Table 4.2). These percentages relate to the general statistics presented in the governorate's census data of 2018, in which the percentage of males is higher than females corresponding to 51.2 and 48.8 percent of the whole population living in governorate's rural areas, respectively (Table 4.3).

Table 4.2: Distribution of all household members by gender

		n	Percentage
Total Gender of household members	Male	882	54.1
	Female	748	45.9
Total		1630	100.0

Source: Field Survey 2018; Damietta's Census data 2018

		Counts	Percentage
Rural population by gender	Males	464,964	51.2
	Females	442,578	48.8
Total rural population		907,542	100.0

Table 4.3: Distribution of governorate's rural population by gender

Source: Damietta's census data 2018

Results from the descriptive analysis show that the maximum age found for respondents is 85 years old, and the minimum is 20 years of age, with a mean of approximately 51 years across the cohort. The majority of respondents, or 45.4 percent, fall in the category of 45 to 60 years of age, whereas 29.1 and 25.4 percent of the cohort fall in the first and third categories with ages 18 to less than 45, and 60 or above, respectively (Table 4.4).

Table 4.4: Age groups of the respondents

		Frequency	Percentage
Age groups	18-45	102	29.1
	45-60	159	45.4
	≥ 60	89	25.4
	Total	350	100.0

Source: Field survey 2018

For the age categories of all members within the households, including children, (n=1,630) (Table 4.5), a majority of 63.3 percent could be considered to be economically active, falling in the age categories between 16 and 60 years of age, followed by 27.7 percent being younger than 16 years of age, then 9.0 percent who are 60 years old or more. Therefore, more than one-third of household members could be considered economically dependent.

Table 4.5: Distribution of all household members on the different age groups

		Frequency	Percentage
Age groups	< 16	452	27.7
	16–60	1,032	63.3
	≥ 60	146	9.0
	Total	1630	100.0

4.2.2 Education and employment conditions

Results reveal that the average length of formal education experienced by the respondents is approximately 7 years, with a minimum of zero and a maximum of 23 years. The percentage of respondents who are illiterate or who haven't received any formal education is equivalent to 30.0 percent, while 10.0 percent had some formal schooling but left without completion (Table 4.6). This result suggests that 40.0 percent of all respondents from Damietta have no or very low educational attainment. About 11.4 percent and 3.4 percent of respondents, respectively, have completed primary or preparatory schooling, suggesting that they have completed a basic level of education. Almost one-third of respondents (representing 32.3 percent) have completed secondary education, whereas 12.9 percent have university degrees or higher.

At the household level, over half of the other adult members within the household (aged 16 or above) had completed secondary education or higher. In contrast, approximately 25.0 percent of the cohort of all household members have received no or fewer than 6 years of formal education (Table 4.7). This result could suggest that although most household heads have low educational attainment, they tend to encourage younger household members to complete at least secondary education, especially as opportunities for formal education have improved in the region. Hence, education may be taken as means of improving individuals' employability, income and social mobility (Barnett & Adger 2007), and consequently, improve the living conditions and, by association, the adaptive capacities of rural households in the studied area.

	Frequency	Percentage
Illiterate / no formal education	105	30.0
Some formal education without a degree	35	10.0
Primary degree	40	11.4
Preparatory degree	12	3.4
Secondary and post-secondary degree	113	32.3
University degree or higher	45	12.9
Total	350	100.0

 Table 4.6: Distribution of respondents by educational level in Damietta, 2018

Table 4.7: Distribution of other adults within the households by educational levelin Damietta, 2018

	Frequency	Percentage
Illiterate / no formal education	175	21.1
Some formal education without a degree	43	5.2
Primary degree	73	8.8
Preparatory degree	80	9.7
Secondary and post-secondary degree	367	44.3
University degree or higher	90	10.9
Total	828	100.0

Source: Field survey 2018

Household members who were over the age of 16 were asked whether they are working in any paid occupation. The employment data is presented in-depth because it provides a key component of the later analysis of perception and adaptation data and gives a clear reflection of the modes of income diversification within the rural community studied.

Primary occupations (that provide the biggest portion of income) were categorised into four broad groups, including:

- 1. Agricultural-related occupations, mainly involving farmers and waged agricultural labourers; and
- 2. Unskilled non-agricultural occupations for those who possess no particular skills nor related educational background;
- 3. Semi-skilled non-agricultural occupations that require less academic or technically skilled backgrounds; and finally:
- 4. Highly skilled non-agricultural occupations which include jobs that require higher educational level and/or higher technical background (such as doctors, teachers, lawyers, government officials, etc.).

As shown in Figure 4.2, most of the working respondents (n=332), representing 64.8 percent, work primarily as farmers and/or waged agricultural labourers in the study area. The subsequent most prevalent occupation were high-skilled non-agricultural occupations (21.4 percent), whereas 9.6 percent and 4.2 percent of respondents work in semi-skilled and unskilled non-agricultural occupations, respectively.

These results can be conceptualised slightly differently at the household level. There are 555 household members in total (47.1 percent) who have paid work, out of 1,178

members who are 16 years of age or above. Most members, representing 50.3 percent, work as farmers or waged agricultural labourers. Only 21.6 percent of members work in highly-skilled, non-agricultural occupations and 18.4 percent work in semi-skilled non-agricultural occupations. Finally, only 9.7 percent of members work in unskilled non-agricultural occupations. Consequently, the ratio of individuals who work in the agricultural *versus* non-agricultural sectors is roughly one to one. Notably, the majority of households' members other than the respondents work primarily in non-agricultural sectors.



Figure 4.2: Distribution of households' members by occupation in Damietta, 2018

Therefore, it can be implied that rural households have been diversifying their income sources through other household members, mostly younger individuals working in non-agricultural sectors. This result might also suggest that in later years, if younger family members inherit farmland, they would then probably be operating agricultural activities as a secondary job or occupation. For illustration, a farmer (Interview 10, Kafr Saad), speaking about his sons' occupations, stated:

"My children are not working in agriculture, one is a driver, and the other has a private dairy-products shop in Damietta town and commutes there every day. They told me, "we don't want to work in agriculture; you stay working in it until you die, and we shall see what we would do after" ... they are not even helping me with our land".

Source: Field survey 2018

Another farmer (Interview 1, Faraskur) was asked whether he would encourage his young sons to work in agriculture in the future responded in the following way:

"No, no, no, no, I already told them [i.e. his children] no one works in agriculture. I didn't make anything from working in agriculture; what comes from this business is barely equal to what has been paid. I want them to complete their education and work in any other sector".

Due to such variation in the primary occupations of household members, more attention should be given to the contribution of each of the agricultural and non-agricultural sectors to the total income of households, as will be presented in section 4.2.3.2. Identifying the degree to which each household depends on agricultural activities to generate their livelihood could have a key influence on their decisions regarding various aspects of their production patterns and climate change adaptation responses which will be discussed later in the following chapters.

Most primary occupations of household members are undertaken in the same village as their residence, representing 75.9 percent of responses (Table 4.8). Following that, 12.0 percent and 8.2 percent of employed members work in towns or other villages within the governorate, respectively. Finally, 2.7 percent are employed overseas and 1.3 percent outside Damietta.

		Frequency	Percentage
Location of primary occupation	In the same village	418	75.9
	Another village within the governorate	45	8.2
	Town	66	12.0
	Another governorate	7	1.3
	Overseas	15	2.7
Total		551	100.0
Missing		4	

 Table 4.8: Location of primary occupations of employed members

Source: Field survey 2018

Several reasons were reported for those household members above the age of 16 who are not working in any paid job (n=623) (Figure 4.3). The majority are housewives representing 64.7 percent of the unpaid cohort, followed by 20.1 percent who are full-time students. Approximately 7 percent of household members reported that they do not wish to work (mostly females) and 5.2 percent were unemployed and looking for a job.

Only a few (3.4 percent) reported that household members are too old or have a disability, which would not allow him/her to work. So even if those household members are economically dependent, they are still performing essential household functions.



Figure 4.3: Distribution of non-income earning adult household members by reasons for unemployment

Another remarkable result is that most household members aged 16 and above do not regularly participate in unpaid agricultural activities or help on the farm, with only 22.3 percent regularly participating in farming. More specifically, 66.8 percent of household members between the age of 16 and 40 (n=644) do not regularly participate in unpaid agricultural activities. Taking gender into consideration, most of the household members aged 16 and above who participate in unpaid farm activities are males (85.2 percent), while females only represent 14.8 percent of the agricultural labour cohort. This result is of particular interest as agriculture in Egypt is traditionally considered to be a family business, with all family members contributing (Abou-Hadid 2008), but the differentiation of household labour may be indicative of a broader transformation in agriculture which will be discussed later. Importantly, the results suggest that younger members of rural households in Damietta, especially males, are becoming less engaged with work in the agricultural sector or even assist in farming activities. Younger males are increasingly engaged with off-farm occupations and/or education, meaning that they have limited free time to participate in household agricultural activities actively, or perhaps they are merely unwilling to participate.

Moreover, only 14.7 percent of household members under the age of 16 (n=452) are engaged in some form of child labour. The majority (87.7 percent) of the cohort voluntarily help on the farm without getting paid. Very few children within respondent

Source: Field survey 2018

households work as paid labourers in agricultural or non-agricultural sectors -3.1 and 9.2 percent, respectively. It needs to be acknowledged that there could be a limitation to the analysis, as this age group might not provide a true representation of child labour either because it includes very young children (for example babies and toddlers) who will not be able to participate in any form of labour anyway, or people may be less willing to reveal how much their children are working on their farm.

4.2.3 Livelihood and economic conditions of households

This section describes the living conditions of households, with a particular focus on their social relationships and participation and other economic and physical aspects of their lives. Questions on social capital, economic conditions, particularly income, health conditions, and housing and transportation conditions are outlined. It is essential to understand the social context of households as it could have a considerable influence on their adaptive capacity and perceptions of risks related to climate change (Isham 2002; Adger 2003; Armitage 2005; Deressa et al. 2009; Thomas, William & Tobias 2018; Mekonnen & Kassa 2019; Shinbrot et al. 2019).

4.2.3.1 Social Capital

Respondents were asked to rate their relationships with other family members, relatives and friends, neighbours and village fellows, colleagues, and government cadres to generate an indicator of social capital. Almost all respondents representing 96.9 percent, 94.3 percent, and 82.0 percent have good relationships with other family members, relatives and friends, and neighbours and village fellows, respectively (Table 4.9). Also, almost three-quarters of respondents (74.0 percent) stated that they have good relationships with their colleagues. In contrast, their relationship with government employees was valued as good by only 51.4 percent of respondents. The perceived social distance to government officials could have important implications for the acceptance or the implementation of policy in the studied area; specifically, a good relationship with government officials is likely to facilitate and promote adaptation to climate variability (Pelling et al. 2008; Fatti & Patel 2013).

		Percentage		
	Very poor and Poor	Neutral	Good and Very good	Mean
Family members	0.9	2.3	96.9	9.26
Relatives and friends	0.3	5.4	94.3	8.93
Neighbours and village fellows	0.6	17.4	82.0	8.59
Colleagues	1.1	24.9	74.0	8.16
Government cadres	25.1	23.4	51.4	6.15

Table 4.9: Distribution of households by forms of social relationships

Source: Field survey 2018

Another aspect of understanding the social capital of households is the degree of participation in local community events (Table 4.10). Respondents were asked how often their households discuss public events, organise public activities, give advice to others, solve conflicts with others, and attend community events. Results revealed that 43.4 percent of households regularly participate in the discussion of public events; 61.1 percent of them always provide advice to others, and 64.6 percent frequently help to solve conflicts among others.

Table 4.10: Distribution of households by frequency of participation in community events

	Percentage			
	Never and rarely	Neutral	Often and Always	Mean
Discussion of public affairs	34.0	22.6	43.4	5.05
Organising public activities	46.9	26.0	27.1	3.90
Giving advice to others	11.1	27.7	61.1	6.97
Solving conflicts	12.9	22.6	64.6	7.10
Attendance of community	58.3	17.1	24.6	3.61
events				

Source: Field survey 2018

However, respondents' social participation is less common in organising and attending public activities and community events, with the majority being 46.9 percent and 58.3 percent, respectively. Hence, results suggest that the social participation of most households in community activities is not particularly high, except for aspects of involvement related to giving advice and solving conflicts among others.

A social participation index was developed to identify the overall level of social participation of respondents through a compilation of their reported scores for each of the 5 aspects of the previously mentioned social participation criteria. Then an average of

social participation was calculated on 10 point scale (0=no participation to 10= high participation). Results from cross-tabulation reveal that social participation is particularly low amongst respondents whose primary occupation is farming and waged agricultural labouring, or who are working in unskilled, non-agricultural occupations, a result that is significantly less than the other two occupation categories (Pearson Chi-squared p= >0.001) (Table 4.11). Therefore, the degree of social participation of respondents is likely to be interconnected with their primary occupations. Once again, this result might have implications for the application of adaptation policy because people who depend on agricultural livelihoods may also have the least social capital on which to rely.

Table 4.11: Cross-tabulation between respondents' level of social participation and their primary occupations

Social participation

		50	boolai participation			
		Low	Medium	High	Total	
Farmers and waged labourer	Count	24	148	43	215	
	% within primary occupation	11.2	68.8	20.0	100.0	
Unskilled non-agricultural	Count	5	6	3	14	
occupations	% within primary occupation	35.7	42.9	21.4	100.0	
Semi-skilled non-agricultural	Count	5	19	8	32	
occupations	% within primary occupation	15.6	59.4	25.0	100.0	
High skilled non-agricultural	Count	8	31	32	71	
occupations	% within primary occupation	11.3	43.7	45.1	100.0	
Total	Count	42	204	86	332	
	% within primary occupation	12.7	61.4	25.9	100.0	

Source: Field survey 2018

Results also suggest that there was no significant difference in the distribution of households in the two studied local administrations in terms of their degree of social participation (Pearson Chi-squared p=0.25) (Table 4.12).

		Social participation				
			Low	Medium	High	Total
Local	Faraskur	Count	21	102	57	180
administration		% within	11.7	56.7	31.7	100.0
		Local Unit				
	Kafr Saad	Count	28	109	33	170
		% within	16.5	64.1	19.4	100.0
		Local Unit				
Total		Count	49	211	90	350
		% within	14.0	60.3	25.7	100.0
		Local Unit				

Table 4.12: Cross-tabulation between the level of social participation and the studied local administrations

Source: Field survey 2018

Another important element that reflects a household's social capital is the amount of support (financial, psychological, help in daily activities) that household members receive from individuals other than their family members whenever they are in need. Results show that almost 70 percent of households receive help from others, and most respondents (46.3 percent) suggested that support is mainly in the form of financial assistance, i.e. lending money or goods, help in finding a job, help in agricultural operations, etc. (Table 4.13). About 37 percent and 16.6 percent of responses indicated that help is provided in the form of psychological support, i.e. emotional support or giving advice, and assistance in daily activities, i.e. taking care of children or patients, respectively.

Table 4.13: Forms of support accessed from the local community (multiple responses)

		Responses	
		Frequency	Percentage
Kinds of support offered	Financial support	206	46.3
	Psychological support	165	37.1
	Help in daily activities	74	16.6
Total		445	100.0

Source: Field survey 2018

Respondents were also asked whether or not they were members of any of the following organisations: syndicates, farmers or labour organisations or unions, non-governmental organisations, government agencies, and political parties. Syndicates are considered as formal groups or organisations for individuals having a similar profession. The key

mandate of syndicates in Egypt is to protect the professional interest of their members, such as conditions of work, salaries and pensions. There are also some political functions for syndicates, such as recruiting political leaders and advocacy. However, historically, syndicates in Egypt were considered primarily as a means by which the government could control the behaviours of their members and, in many instances, mobilise professionals to support the régime (Springborg 1978). Moreover, several recent professional organisations and unions for farmers were established or came into force after the Egyptian revolution in 2011 (Nawar & Abdel-Hakim 2013). The main objectives of these unions were set around particular goals such as providing their members with agricultural inputs and services; resolving economic, marketing and land tenure problems; providing farmers with a pension, health insurance, and social services, especially to marginalised categories; and improving agricultural extension activities in rural Egypt (Nawar & Abdel-Hakim 2013).

As illustrated in Figure 4.4, results show a very low participation rate amongst respondents, even though they are primarily household heads, with only a few being members of one or more associations. Only one-third of respondents are members of syndicates, and 8.3 percent of respondents reported being members of farmers' unions or organisations. This low level of participation in formal bodies could have a considerable influence on the vulnerability of many rural households and the recognition of their political, economic and social circumstances by decision-makers in Egypt.



Figure 4.4: Distribution of respondents by membership in organisations

Source: Field survey 2018

Information about access by household members to the various information sources, including internet, TV, mobile phones, radio broadcast, newspapers, and land-line phones, was also gathered. As illustrated in Figure 4.5, televisions and mobile phones are the most common and accessible sources of information for most households. More specifically, the majority (97.7 percent) reported that televisions were the primary source from which they obtain their information.



Figure 4.5: Distribution of households according to the access to information sources

Followed by mobile phones (80.6 percent), internet (45.7 percent), radio (40.6 percent) and land-line phones (21.4 percent). Newspapers, as a source of information, are considered the least accessible to households. Low literacy rates could be a reason for depending more upon audio-visual information sources other than those that could require higher reading abilities or even learning a new technological skill, such as using a smartphone or a computer to obtain information.

4.2.3.2 Income of households

From personal experience working in Egyptian rural communities, it is more suitable to ask about the income of rural dwellers by identifying their expenditure, because rural people often feel uncomfortable responding to direct questions related to income or economic earnings. In fact, consumption expenditure can provide a good reflection of the permanent income of households (Freidman 1975; DeJuan & Seater 1999; Sweeney et al. 2018). Therefore, the incomes of households were estimated by adding together the monthly expenditure on various household items including food, clothes, health, education, house maintenance, cigarettes and mobile phone recharging, electricity, gas,

Source: Field survey 2018

transportation, rent, gifts and donations, water, entertaining activities, non-food groceries, and any other uncategorised item. Due to the way respondents' financial capital was established, expenditure and income concepts will be used interchangeably throughout the thesis. Results suggest that the minimum monthly income of households is equivalent to EGP 1,150 (AUD 1 = EGP 11.87, i.e. AUD 97 per month), while the maximum income is EGP 21,005 (i.e. AUD 1,770 approximately) with an average of about EGP 5,530 (i.e. AUD 466) of monthly income. For more straightforward interpretation of the results, income was then categorised into four groups (1= below EGP 2,404; 2= from EGP 2,404 to less than 5,530; 3= from EGP 5,530 to below EGP 8,656 and finally; 4= equivalent to EGP 8,656 and higher), calculated through the mean and standard deviation. Most households (60 percent) have income below the average monthly income of EGP 5,530 per month, including about 7 percent of respondent households with very low income, corresponding to approximately less than AUD 210 per month (Table 4.14). At the same time, 28.6 percent and 11.4 percent of households fall in the third and fourth income categories, respectively. This result shows that most households have low-income levels.

	Frequency	Percentage	Cumulative Percentage
EGP 0-2404	24	6.9	6.9
EGP 2404.0 - 5530	186	53.1	60.0
EGP 5530.0 - 8656	100	28.6	88.6
EGP > 8656.0	40	11.4	100.0
Total	350	100.0	

Table 4.14: Categories of households' monthly income

Source: Field survey 2018

Most households, or 65 percent, have more than one income source (Figure 4.6). Within those households, the contribution of agricultural activities varied significantly from 10 percent to 99 percent of the total household income. Approximately 35.0 percent depend solely on agriculture, having no other secondary occupations or sources of income. This result also supports a broader argument regarding the importance of categorising households based upon their dependency on agricultural activities for their livelihoods. This categorisation could be a key influencer on their perceptions about the agricultural situation in general and, consequently, may affect their decisions. That point will be taken up and discussed in detail in Chapter 5.

Figure 4.6: Distribution of households based upon dependency on multiple income sources



Source: Field survey 2018

4.2.3.3 Health and housing conditions

Respondents were asked to rate their health condition on a 10-point scale (1=very poor to 10=very good). Descriptive analysis shows that the minimum rate they reported is 1, and the maximum is 10, with an average of about 7, which means that the health of most respondents is relatively good. Around three respondents in ten have chronic diseases, mostly liver, kidney and heart-related diseases. About 30 percent of respondents reported having health insurance, with only 30.8 percent of those who have chronic diseases having health insurance. At the household level, 10.2 percent of the respondents' other household members suffer from chronic diseases. Health services in the studied areas are inadequate, as suggested by the fact that 60 percent of households were dissatisfied with the health services provided in the village (Table 4.15).

Table 4.15: Households satisfaction with local health services

	Frequency	Percentage
Very dissatisfied and dissatisfied	212	60.6
Neutral	64	18.3
Satisfied and very satisfied	74	21.1
Total	350	100.0

Source: Field survey 2018

Several questions were framed to assess the physical structure and location of respondents' houses in Damietta. Firstly, the housing arrangement, as almost all households own their residences, with only 12 households representing 3.4 percent are leasing their homes. Secondly, the number of rooms within the house reflects the size of the house. Results show that the average number of rooms per house is 3, with a minimum

of 1 room and a maximum of 18 rooms. The majority of households have three rooms within their houses (53.4 percent), followed by 20.6 percent of houses with four rooms (Figure 4.7). Only 17.4 percent of houses are comprised of two rooms or less, whereas 8.1 percent of houses have five rooms and above.



Figure 4.7: Distribution of households by the number of rooms within the houses

Thirdly, the different types of facilities found on properties included tap water, electricity, toilet, separated room for poultry and livestock and, finally, sanitation. Except for just one house, all houses of the studied households have tap water, electricity and toilets (Figure 4.8). About 70 percent of houses have separate rooms for rearing poultry and livestock, and almost 60 percent have a sewerage sanitation system. More specifically, 172 houses out of 350 are not connected to either a public or a private sewerage system, which could be one of the reasons for the prevailing local problems with water pollution in the study area, and which will be discussed in more detail later in this chapter (Section 4.4). This result was also consistent with the governorate's census data obtained during fieldwork, indicating that 55.1 percent of households in the two studied local administrations are connected to a public sewerage system. On the contrary, 44.9 percent of the households living in the two local areas are connected to either a private network, a cesspit or an open field.

Source: Field survey 2018



Figure 4.8: Distribution of houses by basic facilities in Damietta, 2018

Fourthly, respondents were asked to indicate the average distance between their residential houses and the nearest essential amenities such as primary schools, public transportation, medical units, markets, main roads or towns. For most households, the mean distance for almost all services is approximately 2 kilometres, except for the distance to town, which is approximately 6.0 kilometres (Table 4.16).

 Table 4.16: Basic analysis for the distances (in kilometres) between houses and several local destinations

	n	Minimum	Maximum	Mean
Primary school	346	0.02	6.00	1.1163
Public transportation	344	0.00	8.00	1.0854
Medical unit	345	0.02	15.00	1.7238
Market	337	0.02	17.00	2.2630
Main road	344	0.00	20.00	1.0247
Town	336	0.00	35.00	5.7076

Source: Field survey 2018

Finally, respondents were requested to indicate the level of satisfaction with their houses and to rate the quality of the services provided in their area, including water, electricity, sewerage system, and removal of refuse. The majority or 81.1 percent of households demonstrated a general satisfaction with their houses (Table 4.17).

Source: Field survey 2018

	Frequency	Percentage
Very dissatisfied and dissatisfied	14	4.0
Neutral	52	14.9
Satisfied and very satisfied	284	81.1
Total	350	100.0

Source: Field survey 2018

Regarding the quality of services, most households are generally satisfied with the quality of the water and electricity services provided in the local area (62.0 and 76.6 percent, respectively). However, almost 50.0 percent of households rate the quality of sewerage systems and refuse removal as bad (Figure 4.9). This result again could be indicative of the severe pollution that is predominant in the study area.

Figure 4.9: Quality of public services provided in the local area as rated by the households



Source: Field survey 2018

4.3 Agriculture in Damietta

Two dominant types of rural communities can be found in Egypt, and it is essential to differentiate between them:

- 1. The "old" rural communities located around the Nile valley and Delta that were established several centuries or millennia ago; and
- The "new" rural communities which are built settlements established across reclaimed areas (desert or dried lakes areas) within the last few decades (Nawar & Abdel-Hakim 2013).

The total cultivated land area in Egypt is 8.9 million *feddans* (approximately 3.7 million ha) (CAPMAS 2018). The newly reclaimed lands represent only one-third of these agricultural land areas and are mostly unregistered, while most are old, registered agricultural areas. The average size of farms in the old rural communities is generally much smaller than in the newly reclaimed areas (Nawar & Abdel-Hakim 2013). Egyptian agriculture is characterised by the prevalence of individual smallholdings and the high rate of land fragmentation (MALR 2009; Aboulnaga et al. 2017). According to the latest agricultural census (2010), 92.1 percent of the total cultivated land in Egypt are considered individual landholdings farms, with the majority of less than 5 *feddans* (2.1 ha) (Aboulnaga et al. 2017).

In Damietta, the total cultivated land area is 117,151 *feddans* (approximately 49,203 ha) including 7,440 *feddans* (3,125 ha) of gardens and excluding 5,756 *feddans* (2,417 ha) of urban encroachment on to agricultural lands. Old farmland accounts for 94.0 percent of the total land area in Damietta, and only 6.0 percent is newly reclaimed land. The two local administrations selected for the study, namely Faraskur and Kafr Saad, together hold 57.0 percent of the governorate's total cultivated land area (Figure 4.10). There are 6,672 *feddans* of agricultural lands in Faraskur that are newly reclaimed land, and mainly includes the dried areas of El-Manzala Lake, which is a brackish water lake (Department of Agriculture in Damietta 2017). This situation is a specific problem for Faraskur, where higher levels of soil salinity prevail in some areas.

Regarding land tenure, 93,504 individuals hold a total agricultural land area of nearly 100,754 *feddans* (42,317 ha) in Damietta, with a mean area of 1.08 *feddan* per holder (0.76 ha) (Table 4.18). Most occupiers of farmland, representing 70.8 percent of all landholders in Damietta, have very small farms of less than 1 *feddan* and what they hold only represents 20.2 percent of Damietta's total cultivated agricultural area. There is also 21.1 percent of landholders who hold farms of between 1 and less than 3 *feddans*, and what they hold represent 31.6 percent of total cultivated land areas in Damietta. Almost 5 percent of landholders occupy about 16.3 percent of the total cultivated land area in Damietta of size between 3 and less than 5 *feddans*. These results indicate that the vast majority of holders (nearly 97 percent) are considered small landholders occupying less than 5 *feddans* of agricultural land.

Figure 4.10: The area of cultivated agricultural lands within each administration area in Damietta (in *feddans*)



Source: Department of Agriculture in Damietta, December 2017

Table 4.18: Distribution of cultivated lands by farm size and holders in Damietta in2017

Farm size	Are	ea	Holdi	ngs	Average
(feddan*)	Feddan	%	Holders	%	(feddan)
Less than 1	20,396	20.2	66,192	70.8	0.31
1-3	31,875	31.6	19,699	21.1	1.61
3-5	16,392	16.3	4,362	4.7	3.75
5-10	15,389	15.3	2,305	2.5	6.67
10-20	9,446	9.4	696	0.7	13.57
20+	7,256	7.2	250	0.2	29.02
Total	100,754	100	93,504	100	1.08

Source: Department of Agriculture in Damietta, December 2017

Note: *1 feddan=0.42 ha

During the winter season of 2017–2018, the most important cultivated field crops were wheat, linen, beans and clover, whereas vegetables, including potatoes, tomatoes and onions were widely cultivated (Figure 4.11). For summer field crops (2018 season), the largest land areas were cultivated with cotton, rice, maize, green forage (maize, elephant grass and torpedo grass) and watermelon, whereas sweet potatoes, tomatoes, eggplants and bell peppers were the most widely cultivated vegetable crops (Figure 4.12). Some

areas were cultivated with fruit trees, including guava, palm trees, oranges, mangoes, and grapes within Damietta as in 2017 (Department of Agriculture in Damietta). In 2011, the total number of large ruminants (i.e. cattle and buffaloes) was 121,120 head, and the total number of poultry was equivalent to 17,306,097 in Damietta (FAO 2018).

Figure 4.11: Distribution of total cultivated land area by winter crops and vegetables 2017–2018



Source: Department of Agriculture, December 2017





Source: Department of Agriculture, December 2017

4.3.1 Land size and patterns of production in the studied area

The sample data analysis shows that the mean area of owned agricultural land in the studied areas is approximately 1.8 feddans (0.756 ha). In comparison, the mean area of rented or shared agricultural lands is roughly equivalent to 0.7 (0.294 ha) and 0.2 feddans (0.084 ha), respectively. There are 23 households who are landless and are only engaged in agriculture as waged agricultural labourers. The total land area (owned, rented and shared) was categorised into three categories as follows 1=less than 2.6 feddans (i.e. 1.092 ha), 2=equal to 2.6 and less than 5.5 feddans (i.e. 1.092 ha to less than 2.31 ha), and 3=equal to or greater than 5.5 *feddans* (i.e. 2.31 ha). Most households (58.0 percent) fall in the first category, which also includes landless households, whereas 37.1 percent of households fall into the second category (Table 4.19). Only 4.9 percent of households own, share or rent an area greater than 5.5 feddans of agricultural land. Hence, most households have access to small areas of land and are considered small landholders. According to Aboulnaga et al. (2017), farms with less than 3 feddans generate income that does not provide livelihoods that exceed the international poverty line (USD 1.25 per day per capita). Therefore, many experts suggest that agriculture-dependent households holding less than 3 *feddans* in Egypt are the highly vulnerable category of farmers. In addition, results revealed no significant spatial differences between the two studied local administrations in the distribution of land areas (Pearson Chi-Square p=0.007).

Table 4.19: Total agricultural land area owned, shared and/or rented by households

	Frequency	Percentage
<2.6	203	58.0
2.6-5.5	130	37.1
≥ 5.5	17	4.9
Total	350	100.0

Source: Field survey 2018

About 71.0 percent of households report that their farmland is considered to be near their residential houses. In comparison, 17.4 percent live some distance away, and 11.6 percent have mentioned that their agricultural lands are 'far away' (Table 4.20).

Table 4.20: The distance between agricultural farmland and the households

	Frequency	Percentage
Near	232	70.9
Not very far	57	17.4
Far away	38	11.6
Total	327	100.0

Source: Field survey 2018

Results from multiple response analyses show that most farmers, representing about 62.0 percent of the total cohort were cultivating field crops at the time of the fieldwork, with around half of them using the land for rearing livestock as well (i.e. mixed-use) (Table 4.21). Around 5 percent of farmers were cultivating vegetables either solely or in combination with other field crops and/or fruits. Only 1.5 percent and 1 percent of responses included cultivation of fruits and/or medicinal and aromatic plants, respectively.

Table 4.21: Damietta respondents' agricultural land use from early May–late September 2018

	Responses	
-	Frequency	Percentage
Crops	321	62.1
Livestock	159	30.8
Fruits	8	1.5
Vegetables	27	5.2
Fallow	1	0.2
Medicinal and aromatic plants	1	0.2
Total	517	100.0

Source: Field survey 2018

Most households, or 71.7 percent, sell their agricultural products mainly through tradesmen or middlemen (Table 4.22). About a quarter of households (25.5 percent) sell their products directly, while very few households mentioned collective marketing or government-arranged purchases as their primary marketing channels, with percentages of 1.2 and 0.9, respectively. Finally, 0.6 percent of households use their production solely for domestic consumption.

Table 4.22: Main marketing channel adopted by households for selling agricultural produce

	Frequency	Percentage
Self-selling	82	25.5
Tradesman	231	71.7
Collective marketing	4	1.2
Government-arranged purchase	3	0.9
No selling/ household use	2	0.6
Total	322	100.0

Source: Field survey 2018

Table 4.23: Cross-tabulation between marketing channels and rate of product selling

			Rate your products selling			
			Very bad		Good and	-
			and bad	Neutral	very good	Total
Marketing	Direct	Count	4	22	56	82
channel	selling	% within self-	4.9	26.8	68.3	100.0
adopted by		selling				
the	Tradesman	Count	51	47	130	228
households		% within	22.4	20.6	57.0	100.0
		tradesman				
	Collective	Count	0	4	0	4
	marketing	% within collective	0.0	100.0	0.0	100.0
		marketing				
	Governme	Count	2	0	1	3
	nt-arranged	% within	66.7	0.0	33.3	100.0
	purchase	government				
		purchase				
Total		Count	57	73	187	317
		% within total	18.0	23.0	59.0	100.0

Source: Field survey 2018

Almost six out of ten households rated their agricultural produce selling over the past five years as good, whereas 18.0 percent of households were not satisfied and rated it as poor, while 23.0 percent were neutral. To shed more light on this point results from cross-tabulation reveal that households who self-sell their products are more likely to be satisfied with their methods of sales than those who sell through a third party such as a tradesman or through government (Pearson chi-square p=>000.1) (Table 4.23). Hence, although most (71.7 percent) choose tradesmen to sell their produce, 43.0 percent of that

cohort tend to be generally dissatisfied with the arrangement, rating the quality of the process as six or lower on a 10-point scale.

Households were requested to indicate the percentage of their dependency on family members' paid or unpaid labour in different agricultural operations. Some households indicated their full dependence on family members, while some others always hire paid labourers to take care of their land. Several households have an equal reliance on family members and paid labour. Table 4.23 shows the mean percentage of dependency on paid agricultural labour *versus* family members in households' different farming operations. The degree of dependence on paid agricultural labour was higher in operations related to land preparation (61.5 percent), planting (60.4 percent) and harvesting (66.6 percent). In contrast, the percentage of dependency on family members was higher in irrigation (69.5 percent) and crop operations (52.8 percent).

 Table 4.24: Household dependency on family members and paid labourers in agricultural operations (mean percentages)

Agricultural operations	Family members	Paid agricultural labour	Unpaid labour	
a. Land Preparation	35.3	61.5	3.2	
b. Planting	36.6	60.4	3.0	
c. Irrigation	69.5	27.8	2.7	
d. Crop operations	52.8	44.1	3.1	
e. Harvesting	29.3	66.6	4.1	

Source: Field survey 2018

4.3.2 Households' satisfaction with the agricultural situation in the study area

Households were requested to rate their degree of satisfaction with several key factors related to agricultural production, including the availability and quality of the different resources and inputs such as soil, water, fertilisers or pesticides. In addition, the satisfaction with agricultural policies imposed by the government and the agricultural situation, in general, were explained.

About 66.0 percent of households are satisfied with the soil quality, and 63.0 percent of the cohort are satisfied with the quality of their agricultural drainage systems. Approximately 58.0 percent of households are happy with the availability of water and agricultural inputs, whereas only about half are satisfied with their quality. On the

contrary, and essential for generating context during later discussion, three in four households are dissatisfied with the prices of the inputs and the agricultural policies imposed by the Egyptian government. Furthermore, as a final point, just about 65.0 percent of households are generally dissatisfied with the overall agricultural situation in Egypt (Figure 4.13).



Figure 4.13: Household respondents' satisfaction with the different factors of agricultural production in Egypt

To highlight the issue related to input availability and prices, a farmer (Interview 1, Faraskur) speaking about the major problems in the agricultural system in his area stated that:

"Fertilizers are not available, and if found, they would be very expensive ... very high in prices – 270 or 280 pounds per bag (50kg) is definitely expensive for farmers. For farmers' crop production, the country is not subsidising their crop, so when the farmer spends money on their land, plant, and fuel ... and you know that fuel prices have risen ... so the crop prices don't bring even what has been paid. The farmers are losing, and this is the problem; nothing is easy for them; everything is difficult".

Source: Field survey 2018

Another farmer (Interview 2, Kafr Saad) elaborated on the problems in relation to the agricultural inputs stated that:

"Yes, we have several problems. First of all, they cheat on pesticides; the pesticides are not effective at all. Another thing is the high prices of fuel, the prices of fuel increased, and this increased the cost of irrigation. Generally, the prices of all inputs have risen; we don't know what we shall do. And, the water, we face water shortages in the cultivation season between 1st May and 15th June, the water is not enough. I know some farmers who left their lands fallow; they couldn't cultivate it".

A critical issue for agricultural policy raised by several farmers is the government's plan to reduce rice cultivation in the Nile Delta region to conserve irrigation water (Kassem et al. 2019). Limiting rice cultivation is a particular concern for Damietta as high rates of soil salinisation already prevail. Farmers reported that they need to wash out the soil through the use of irrigation water during rice cultivation. Another important aspect is the economic returns households obtain from rice production. Rice is a short-growing season crop with good financial returns, which can also be used for household consumption (Tarek 1998). As an illustration, a farmer from Faraskur mentioned (Interview 8) the following:

"The soil here is saline, the lands are considered to be newly reclaimed and have a percentage of salinity. So, cultivating rice is good for the soil. This year the government said we should cultivate vegetables. I am a farmer, and I best know what is suitable for my land. I have two feddans; at least they allow me to cultivate only a feddan of rice. As farmers, we cannot buy rice from outside, and our food depends on the rice we cultivate. The government said farmers will pay a fine of 8,000 to 10,000 pounds for each violated feddan and will be imprisoned for a year. I cultivated maize this year; it cost me 4,700 pounds for one feddan from seeds to ploughing to cultivation, and I wanted to sell it, and no one wanted to buy it as fodder for livestock. The cobs started to dehydrate and dry out, so I made silage for my livestock. I cultivated the other feddan a bit later, and a trader came, and he will pay 300 pounds per ton. The feddan produced 11 tonnes; hence I will sell it with 3,300, so I lost, and I will keep losing as long as the minister of agriculture wants us to cultivate dry crops".

In one further example, a farmer (Interview 3, Kafr Saad) said:

"The same thing happened with rice, before the limitation of its cultivation, Egypt produced 6.5 million tons, we use only 4 million, and the rest is being exported. Now, the government wants to import rice, and we are self-sufficient; this is just crazy. The government needs to understand that it is irrational to stop cultivating rice, especially in the areas where the soil has a percentage of salinity".

Another farmer (Interview 11, Faraskur) explained the socio-economic and ecological threats of not cultivating rice and stated:

"We are a population of 100 million, and the government needs to feed us. We eat bread and rice- the basic food we eat. I hope that government officials can provide them to citizens. The person who says don't cultivate rice, and he is sitting on his desk and office away from the real world. Does he understand the nature of my soil? Does he know the quality of our water? Does he know the quality of my drainage system?

I am currently cultivating a dry crop (i.e. vegetables), I swear to god, I swear to god, it is not even giving the costs I paid for cultivating it. I will not even tell you the price of the land rent; I am only saying the costs of its cultivation only. I am forced to pay the rent to the landlord in all cases; the land got revenue or not, it's none of his business. We sell assets, borrow money or do anything to give the landlord the rent".

For more illustration on the perceptions of policy, a farmer (Interview 2, Kafr Saad) when he was asked about current governmental programs aiming at improving the agricultural situation in the area, responded:

"The government!? The government doesn't do anything, oh ... it only does one thing that is to put more burdens on farmers. The policies are sick. We cultivate four crops per year, and the government provides fertilisers only twice a year. And more, we normally cultivate two rotations of potatoes, and the government says, "No, you must cultivate only one rotation". And now the government is annoying us with the rice policy. It is disgusting".

Another farmer (Interview 11, Faraskur) added to this point when he mentioned:

"As far as I know, and I am a man with intermediate education and I don't understand politics, but I do understand in everyday life I am living in. Does Egypt have money? Yes, it does, and the government knows very well that the country is rich, but who has the money and how it is spent? For example, the new Suez Canal has been completed more than a year now, what are its benefits to the citizens? Only God knows. The huge expenditure on roads and bridges and other things, what do we get as citizens from them? I am a farmer, how it is useful for me? Do they facilitate the marketing of my produce? The tradesman who come and buy from me is the one who profits not me.

The government should first monitor the prices of everything and the money spent on mega projects that are being done. The citizens need the money the most, and citizens' quality of life is the most important aspect the government should consider. Even education, we don't want education, because eventually what shall the graduate do with his degree, many educated people can't find jobs anyway".

It is also important to mention that no significant difference was found between Faraskur and Kafr Saad concerning farmers' satisfaction with the agricultural situation in general (Pearson Chi-squared p=0.201). Rather, there was a general agreement that the prices of inputs are high and that the governmental context within which farmers are functioning is not very encouraging. Moreover, in every other factor of agriculture production, there is a large group of households, reaching 50 percent in relation to several aspects, who rated their satisfaction 6 or less on a 10-point scale, and hence there is a wide variation in the opinions of households about the agricultural situation

4.3.3 Households' exposure to agricultural-related shocks and risks

This section highlights the significant economic, social and environmental shocks that surveyed households were exposed to during the two years that preceded the survey (2016

and 2017). After identifying the particular shocks, households who experienced a shock were asked to mention how they dealt with it.

Most of the households have been exposed to crop or income losses as a result of extreme weather events (52.8 percent), increase in the prices of agricultural inputs (57.9 percent) and increases in the cost of renting farmland (61.0 percent) (Table 4.25). Approximately 40.0 percent of households have been exposed to a lack of cash liquidity when purchasing inputs and 37.0 percent experienced reductions in yields as a result of pest infestation and disease infections. More than 35.0 percent of households have experienced a drop in crop prices upon selling their products, and roughly one-third of households have experienced low crop returns due to water and labour shortages. Finally, 23.0 percent of households reported having a reduction in crop yields as a result of irregular agricultural operations. Only 15.5 percent have experienced food shortages within their household, but that does suggest that low food availability and access do have local impacts. To conclude, most of the sudden shocks experienced by the households were economic (particularly increased cost of inputs) and shocks linked with the negative effects of climate change on income and crop production.

Table 4.25: Respondent households' exposure to sudden shocks during 2016 and2017 (in percentages)

	Percentage	
	Yes	No
Poor crop returns due to water shortage (n= 304)	30.3	69.7
Crop/income losses due to extreme weather events (n= 324)	52.8	47.2
Reductions in yields as a result of pests and diseases (n= 305)	37.4	62.6
Reductions in crop yields due to irregularities in agricultural operations (n= 304)	23.0	77.0
Crop prices dropped upon selling agricultural products (n= 304)	35.5	64.5
Food shortages in the household (n= 329)	15.5	84.5
Labour shortages (n= 304)	30.3	69.7
Increase in the land rent prices (n= 100)	61.0	39.0
Increase in the prices of the agricultural inputs (n= 304)	57.9	42.1
Lack of cash liquidity when purchasing production inputs (n= 304)	40.5	59.5

Source: Field survey 2018

Borrowing money from relatives or friends and selling livestock are the two most frequent strategies for reducing the risks associated with these shocks, as reported by 27 percent
of households (Figure 4.14). About 23.0 percent of households did not act in response to these events, and hence they just bore the losses. Approximately 10.0 percent of households' responses were taking advance payments from a tradesman, whereas taking a loan was a coping strategy for 5.3 percent of households. Around 2.0 percent of responses regarding coping strategies involved working extra hours on the farm to overcome the labour shortage and working in a secondary occupation. A few households decreased the number of meals consumed and changed the types of food consumed, representing 1.7 percent and 1.0 percent of the responses, respectively. Several other strategies were mentioned by a small number of those sampled, involving postponing activities until labourers were founding (1.0 percent), intensifying agricultural production (0.8 percent), a child or woman starting to work outside the home (0.8 percent), selling assets from the house (0.7 percent), changing occupation (0.7 percent), increasing hiring rate for labourers (0.6 percent), taking donations from people or religious institutions (0.4 percent), decreasing the amounts of inputs used until the required money had been saved (0.4 percent). The final strategy mentioned was the migration of a family member (0.3)percent). This result suggests that mobility (whether geographical or occupational) as a coping strategy to short term sudden shocks is very rarely adopted by rural households in the study area.

Figure 4.14: Coping strategies adopted by households to reduce losses from sudden shocks



Source: Field survey 2018

4.4 Ecological environment in Damietta

The ecological and natural resource management within Damietta governorate, primarily involves aspects of the soil and water in the area. There are generally six categories for soil in Egypt, based upon its type, properties, and productivity. The soils in Damietta governorate are commonly sandy and calcareous, and almost 70 percent of them are classified as average to low productivity, especially where salinity and alkalinity problems are common (EEAA 2003). Nile water is considered the sole source for irrigation in Damietta (Department of Agriculture in Damietta 2017). Consequently, the area is particularly vulnerable to the effects of climate change as there are no alternative irrigation sources. The Nile Delta is enclosed within two Nile River branches; one, namely Damietta Branch, is in the governorate and ends downstream near Damietta city.

Water pollution is the most predominant environmental problem that was reported by households in the study area. As illustrated in the photos in Figures 4.15 and 4.16, the main open canal that holds water for agricultural irrigation and household use is polluted with farm wastewater and sewage waste from the surrounding households. The problem was more apparent in the local unit of Faraskur in comparison to Kafr Saad. A remarkable comment from a farmer during the fieldwork that was mentioned in one of the questionnaire's open-ended questions stated: "*We are consuming our own wastes*".



Figure 4.15: Water canal pollution in Hagaga village- Faraskur central administration, Field survey 2018



Figure 4.16: The pollution of irrigation water in Kafr El-Arab village - Faraskur central administration, Field survey 2018

During in-depth interviews, another farmer (Interview 1, Faraskur) speaking about water problems in Faraskur stated the following:

"People living here drink sewage water, which is the same as what they irrigate from. Two days ago, we had no water for a whole day ... whole day ... By the end of the day, I opened the faucet I found black water coming out of it. Here [i.e. in the village], we have the highest percentage of people who have kidney failure".

The water quality problem has been persistent since 2003, as the environmental report of Damietta governorate (EEAA 2003), emphasises five crucial environmental issues related to water pollution (Table 4.34).

Problem	Dimensions of the problem
1. Solid-waste management	Getting rid of the waste by throwing it in open plains and water canals or through burning – blockage of drainages and canals – water pollution – widespread related health problems.
2. Sewerage system	Absence of sewerage system that covers all areas within the governorate – exposed banks – pollution of drinking water – lack of system maintenance – pollution of water canals – widespread related health problems.
 Drinking water-related services and water quality 	Drinking water shortage – low water pressure – inefficient water pumps – power outages – lack in the number of water treatment units – poor maintenance of water pipes.
4. Water pollution	Deficiency in sewerage systems – exposed (uncovered) water canals – irrigation and drainage water pollution – widespread diseases such as schistosomiasis.
5. Lack of public environmental awareness	Weak coordination between the different governmental administrations responsible for providing the services – lack of public awareness about acceptable environmental practices.

Table 4.26: Predominant environmental problems in Damietta

Source: EEAA (2003)

4.5 Summary

Initial key results relating to demographic, economic, social, production and policy perceptions provide key information that will be elaborated on in future chapters. The data showed that most heads of households have either no or limited schooling, with more than half either illiterate or having completed only primary education. Regarding the primary occupations of respondents, most work predominantly as farmers or waged

agricultural labourers; however, on the household level, there is a notable percentage of households' members who work in non-agricultural occupations, which will help to frame the discussion in the following chapters. Since most households depend on more than one source of income, it is necessary to understand the importance of farming activities in contributing to the income of households, especially given that agriculture is one of the most vulnerable sectors to climate change. Therefore, households of this study are grouped into three groups based upon the contribution of agricultural activities to their overall income and will be discussed in detail in the next Chapter 5 (Section 5.5). Most of the respondents have good relationships with their family members, relatives and other rural dwellers. In contrast, their social participation in community events is low, particularly amongst full-time farmers and waged labourers and unskilled labourers in other non-agricultural sectors. There is also general agreement that local public services providing health services, sewerage systems and refuse removal in the studied villages are poor. These issues add to the problem of water pollution predominant across the study area. Many householders also complained about the prices of farming inputs and the agricultural system in Egypt, suggesting a broad dissatisfaction with governance arrangements relevant to their places and production activities.

Chapter Five: Perceptions of climate change and other risks associated with working in the agricultural sector

5.1 Introduction

As mentioned earlier (Section 2.4), individuals' experiences and perceptions of environmental change are considered as intermediary factors that influence their decisions to adapt (Maddison 2007; Weber & Johnson 2009; Bardsley & Hugo 2010; Bagagnan, Ouedraogo & Fonta 2019). Perception, on the other hand, is shaped by several factors, including characteristics of individuals and households such as age, education, income, food security and other elements (Diggs 1991; Maddison 2007; Ishaya & Abaje 2008). Hence, understanding how rural households experience and perceive climate variability and extremes could allow us to understand their adaptation behaviour, whether *in situ* or *ex situ*, in relation to environmental change.

Since many rural households depend on the agricultural sector for their livelihoods, their perception of working in the sector is crucial for understanding the general circumstances within which they generate resources to enable the household to function. Knowledge of perceptions of agricultural risk is also vital in determining whether a farming family might move to a different area, especially because farming activities are directly dependent on environmental conditions and sensitive to minor changes in climatic stresses (Parry & Carter 1989; Smit & Skinner 2002).

This chapter describes how rural households perceive of their work in the agricultural sector, in addition to their perceptions of climate variability and extreme weather events. Moreover, it addresses the significant future risks and uncertainties of working in, and a reliance on, agriculture from the rural households' perspectives. Finally, it discusses the different attitudes towards climate change, and other future uncertainties found among the studied households' groups.

5.2 Households' perceptions of climate variability and extremes

This section focuses on households' experiences and perceptions of climate variability and its adverse impacts on agriculture. Results revealed that the majority of the interviewed households (90 percent) had experienced remarkable climate variability and extreme weather events by 2018. Results from the descriptive analysis suggest that the minimum number of times the households have experienced unusual climate variability and extreme weather events in a year is one, and the maximum is 30 times, with an average of about four events per year. Most of the climatic events have been experienced during summer and winter according to 48.2 percent and 42.2 percent of household responses, respectively (Table 5.1).

Table 5.1: The perceived timing of the most observed climate variability and extreme weather events in Damietta by 2018 (multiple-response question)

	Responses		
	Frequency Percenta		
Summer	216	48.2	
Winter	189	42.2	
Spring	30	6.7	
Autumn	6	1.3	
Between seasons	7	1.6	
Total	448	100.0	

Source: Field survey 2018

The households who stated that they had experienced climate variability and extremes (n=315), were further requested to indicate the perceived patterns of change (Figure 5.2). The majority of households, or 91.1 percent, recognised that they were experiencing higher temperatures, with roughly three-quarters of households perceiving more frequent heatwaves. More than half of the cohort observed a decrease in the number of rainy days each year. About 46.0 percent and 44.8 percent of households stated they were experiencing more frequent bouts of frost or cold spells and more erratic rain, respectively. Finally, 37.8 percent of households mentioned changes in the timing of the rainy season and 27.9 percent perceived more frequent sand storms.

Figure 5.1: Distribution of households by the observed forms of climate variability and extremes in Damietta by 2018 (n=315)



Source: Field survey 2018

To further illustrate households' experiences with climate variability and extreme in the studied area, a farmer (interview 4, Faraskur) stated:

"Yes, the weather had changed a lot from the past. It has become much hotter, and such hot weather burns crops and decreases their productivity".

Another farmer (interview 10, Kafr Saad) mentioned:

"There are certainly more frequent heatwaves and cold spells, but we can withstand the hot weather, the cold weather, the sun, and everything. This is what God wants, what we shall do, we get used to that".

Moreover, a farmer (interview 11, Faraskur) stated the following when he was asked about the environmental problems that prevail in his local community:

"There are problems in everything, especially farming; it is full of problems. For instance, sometimes you find the weather is really hot, so the next day you find worms all over the crops due to the heatwave. Such weather conditions are something uncontrollable, and not my fault or anyone's fault. Consequently, you run to buy pesticides and apply them to the affected crop. Some other times you find it raining in an unexpected time of the year, this also negatively impacts crops". Regarding household perceptions of the adverse effects of climate variability and extremes (Figure 5.2), the majority of respondents, or 73.3 percent, believed that there were important adverse effects on crop productivity. Roughly 70.1 percent of households perceived that their income had been badly affected, and 68.9 percent of the cohort also perceived much reduced productivity of individuals or labourers. Similarly, crop productivity was perceived to be greatly reduced while the amount of time spent on agricultural operations increased as indicated by approximately 69.0 percent of households. Furthermore, livestock productivity, health conditions, and soil fertility or quality were all greatly affected, as stated by 63.2 percent, 59.6 percent and 46.2 percent of households, respectively. About 60.0 percent of households perceived severe impacts upon health conditions, and nearly half of the households (47.9 percent) mentioned that transportation and communication within their local communities were also badly affected by this variation in climate and its extremes.





Source: Field survey 2018

On the contrary, roughly 60 and 63 percent of households perceived less severe impacts on housing conditions and the quality and quantity of irrigation water, respectively.

An index of climate change perception was developed by compiling the indicated score for each of the 12 impact items then calculate an average for the overall perception level (0=less severe adverse impacts to 10=more severe adverse impacts). Results revealed that most households, or 48.9 percent, perceived that climate change was having important impacts on the agricultural and living conditions in the studied rural villages (Table 5.2), whereas 34.6 percent of households were neutral, and only 16.5 percent of the cohort perceived few or low levels of impact. Importantly, this means that almost half of the studied households have perceived that climate change is strongly negatively affecting their livelihood conditions.

Table 5.2: Household overall perceptions of the adverse impacts of climate variability and extremes

		Frequency	Percentage
Perception of adverse	Less severe adverse impacts	52	16.5
impacts			
	Neutral	109	34.6
	High severe adverse impacts	154	48.9
Total		315	100.0

Source: Field survey 2018

Results also revealed no significant spatial difference (Pearson Chi-squared p=0.407) between the two studied central administrations of Faraskur and Kafr Saad, in relation to perceptions or the adverse impacts of climate variability and extremes (Table 5.3). Hence, households within the studied areas seem to have a similar level of experience with climate variability and extremes.

Table 5.3: Distribution of households by their overall perceptions of the adverse impacts of climate variability and central administration

				impacts		
			Less severe	Neutral	High severe	Total
Central	Faraskur	Count	30	57	73	160
Administration		% within Local Unit	18.8	35.6	45.6	100.0
	Kafr	Count	22	52	81	155
	Saad	% within Local Unit	14.2	33.5	52.3	100.0
Total		Count	52	109	154	315
		% within Local Unit	16.5	34.6	48.9	100.0

Categories of the total perception of adverse

Source: Field survey 2018

In addition, households had the opportunity to discuss the adverse climate change impacts in answers to open-ended questions. After re-coding and categorising responses, results revealed that most of these unstructured responses related to the negative effects of particular heatwaves and frost events on the productivity of several crops, including wheat, beans, potatoes, onions, tomatoes, and clover (Table 5.4). For instance, some respondents have mentioned that due to heatwaves, the productivity of wheat decreased. More specifically, the wheat grains were "burnt" (or, in other words, were smaller and darker), and the stem becomes more fragile and easily broken in hot conditions. Some additional comments noted a decline in the milk production of livestock and an increase in infections with foot and mouth disease among cattle due to the warmer climate. Three cases have also reported losing huge numbers of chickens due to heatwaves.

Item affected	Form of climate variability	Comments
Wheat production	Warmer climate / hotter	A sharp decline in production
	temperature / heatwaves	"Burnt" and the grains are darker and smaller than
		normal
		Fragile stems / easily broken and burnt
	Irregular rainfall	Adversely affected wheat at harvesting stage
Rice production	Warmer climate / hotter	Grains are darker and severely affected by heatwaves
	temperature / heatwaves	
Vegetable production	Warmer climate / hotter	A sharp decline in production
(particularly tomatoes,	temperature / heatwaves	
onions, and potatoes)	Frost bouts / Cold spells	A sharp decline in production
	Not specified	Vegetables are specifically vulnerable to any minor
		changes in climate
Corn production	Warmer climate / hotter	An increase in worm infestation
	temperature / heatwaves	
Clover production	Warmer climate / hotter	Badly affected by heatwaves
	temperature / heatwaves	
Livestock and poultry	Warmer climate / hotter	A decline in milk production
production	temperature / heatwaves	Increase in foot and mouth disease among cattle
		Huge losses / deaths among chickens due to heatwaves
Cotton Production	Warmer climate / higher	A decline in production
	temperature / heatwaves	
Road quality	Irregular rainfall	Severely affected with rainfall
Soil quality	Less rainfall	Reduced quality of agricultural lands particularly where
		salinity prevails
Productivity of	Warmer climate / hotter	It was hard to work in hot weather
individuals	temperature / heatwaves	Didn't complete the farming activities I was assigned to
		do and had to leave due to very hot weather

Table 5.4: Respondents comments on the adverse impacts of climate variability

Source: Field survey 2018

To shed more light on the adverse impacts of climate variability, a farmer (interview 2, Kafr Saad) mentioned the following when being asked about any observed climate variability:

"Yes, definitely, the climate has changed, specifically the temperatures in the summer season. Cold spells also harm potatoes; for instance, the productivity of potatoes decreased by 50 percent in the last season. I am afraid that we would lose more in the upcoming years, even cattle are affected by the warmer weather.

For me, I own and rent around 25 feddans. My losses are severe, to clarify, during the previous cold spell, I lost 15,000 Egyptian pounds per feddan on potatoes".

Another farmer adding more to the cold bouts (interview 1, Faraskur)

"Of course, we have experienced many climate variability incidences in our area. For instance, last winter season, we witnessed a severe cold spell that negatively affected clover and destroyed all of the potatoes crops I was cultivating".

Another farmer (interview 3, Kafr Saad) stated:

"Yes, humidity and high heat. This variability in weather affects agricultural production, especially vegetables. Nowadays, it is not worth cultivating vegetables, and we don't get any profit from it; it barely gives back what we pay for. I think I would probably stop cultivating vegetables. To add more to that, last season, the weather destroyed the wheat crop; hot weather and lack of rainfall decreased the productivity by almost half. The productivity was 10-12 Ardab per feddan (1,500-1,800 Kg) instead of 22-24 Ardab (3,300-3,600 Kg), which is what we would normally get. What made the problem even worse, the selling price was very low, so I didn't sell it; I decided to feed the cattle with it".

Another farmer (interview 5, Faraskur) speaking about his experience with climate variability and extremes mentioned:

"......for instance, a severe heatwave happened in April of this year, negatively affecting the productivity of tomatoes, potatoes, and onions".

Another farmer (interview 6, Faraskur) stated:

"I wish it would rain, instead of this bad water quality. It happens that we experience severe heat waves that destroy wheat, the grain dries and consequently affects productivity".

Another farmer (interview 11, Faraskur) speaking about heatwaves and irregular rainfall stated:

"Very recently, it happened and affected wheat. Last year the productivity was 16 Ardab per feddan (2,400 Kg). This year, the very high temperature decreased the productivity to 7 Ardab only (1,050 Kg). Also, irregular rainfall adversely impacted on wheat spikes during the filling stage of the grains".

In conclusion, the studied households have high levels of perception of climate change and associated risks linked to its adverse impact on agriculture and their livelihoods.

5.3 Households' perceptions of working in the agricultural sector

The agriculture sector in Egypt is already facing challenges that hinder its progression (MALR 2009); however, the extent to which rural households perceive such challenges is vital in explaining the real experienced situation of Egyptian agriculture from the local perspective. This section, therefore, outlines the opinions of the studied households about the agricultural sector and how they perceive of their work and reliance on the sector. As in Table 5.5, results revealed that most respondents or 58.3 percent are very likely or likely to experience income losses due to climate variability and extremes. This result was also reflected with the high level of perceptions of climate change and its risks, as mentioned earlier (Section 5.2). Approximately 46.0 percent of the cohort mentioned that it is very likely or likely to get poisoned due to the application of pesticides. On the contrary, 54.6 percent of all respondents mentioned that it is unlikely to experience accidents during commuting to and from their farms.

Table 5.5: Distribution of households by the likelihood of getting exposed to problems in the agricultural sector (n=350)

	Very unlikely	Neutral	Likely and very
	and unlikely		likely
Income losses as a result of climate variability	25.1	16.6	58.3
	20.1	14.6	16.0
Poisoning as a result of applying pesticides	39.1	14.6	46.3
Accidents during moving to or from farm	54.6	18.3	27.1
	Income losses as a result of climate variability Poisoning as a result of applying pesticides Accidents during moving to or from farm	Very unlikely and unlikelyIncome losses as a result of climate variability25.1Poisoning as a result of applying pesticides39.1Accidents during moving to or from farm54.6	Very unlikely and unlikelyNeutral and unlikelyIncome losses as a result of climate variability25.116.6Poisoning as a result of applying pesticides39.114.6Accidents during moving to or from farm54.618.3

Percentage

Source: Field survey 2018

For households who have one member or more working as a waged agricultural labourer (n=88), more than half of households (56.8 percent) mentioned that they are unlikely not to be paid for the work they had done (Table 5.6). In additional, 47.7 percent of households reported that it is unlikely to be paid less than what was agreed upon with the landlord. Finally, equal percentages of households mentioned that they were 'likely' and 'unlikely' not to find paid agricultural jobs easily, with 43.2 percent of respondents in each category.

Table 5.6: Distribution of households having members working as waged labourers by the likelihood of getting exposed to employment problems in the agricultural sector (n=88)

		Percentage		
		Very unlikely		Likely and
		and unlikely	Neutral	very likely
Likelihood	Not being paid for work	56.8	12.5	30.7
of				
	Being paid less than was agreed upon	47.7	13.6	38.6
	Not finding paid agricultural roles easily	43.2	13.6	43.2

Source: Field survey 2018

As shown in Table 5.7, the majority of respondents disagreed with two statements outlining that wages of the agricultural sector are good relative to other sectors (44.9 percent), and that current agricultural policies are reinforcing and improving agriculture in Egypt (78.9 percent). There were nearly equal percentages of respondents who agreed (41.1 percent) and who disagreed (40.9 percent) that working in the agricultural sector is more secure and stable in comparison with other sectors. Finally, nearly half of the cohort (46.9 percent) agreed that the future of the Egyptian agricultural is uncertain. This result would indicate that many households are generally not content with working in the agricultural sector and are worried about their future livelihoods.

Table 5.7: Distribution of households by agreement with aspects of working in agriculture (n=350)

		Percentage	
	Strongly disagree and disagree	Neutral	Agree and strongly agree
Wages of agricultural labourers are relatively good in comparison with other sectors	44.9	21.1	34.0
Working in the agricultural sector is more stable and secure than other sectors	40.9	18.0	41.1
Current agricultural policies are reinforcing and improving the Egyptian agricultural situation	78.9	12.6	8.6
The future of agriculture in Egypt is uncertain	36.0	17.1	46.9

Source: Field survey 2018

To further illustrate the problem of wages in the agricultural sector, a farmer (interview 5, Faraskur) stated that:

"When we talk about the problem of labourers, as an example, the farmer sells one kilogram of strawberries for an average of 5 to 7 pounds, the labourer who is harvesting makes 2 pounds for every kilogram collected. From the farmers' viewpoint, the wages of labourers are high. If you calculate production costs, the farmer has to pay for labourers, pesticides application, water usage, and fertiliser. However, from the viewpoint of labourers, the wages are very low; paid labour from women makes 50 pounds per day and men 100 pounds per day. What will she do with 50 pounds and she works from 6 am to 11 am? So, it is a waste of money for both sides, for the farmer as he loses money because he has to pay for many other things, and for the labourer himself, as it wouldn't be sufficient for living".

Another farmer (interview 8, Faraskur), also speaking about the wages, said:

"Waged labourers are available but expensive; for instance, if I want to bring a labourer to hoe the corn, he asks for a hundred pounds. The labourer works from 6 am to 8 am, i.e. two or two and a half hours at most and he wants 100 pounds. Yet, what would he do with 100 pounds? Living conditions are expensive for him as well".

One farmer (interview 10, Kafr Saad) also added to the same issue:

"No one wants to work in agriculture; people think it is not worth working in the sector. The waged labourer works for 80 pounds per day; at the end of the day, he will get 2kg of rice with 20 pounds and think about the other expenses, like children, schools, and clothes. If someone works as a worker in the construction field, he makes about 150 pounds per day, do you think he will come work for me with 70?"

As an initial conclusion, therefore, the results suggest that many households are likely to be exposed to loss of income due to climate variability and are generally not content either with working in the agricultural sector or with the policy structure in Egypt.

5.4 Perception of future risks related to the agricultural sector

This section introduces results focusing on household perceptions of future risks paired with their employment in the agricultural sector. Respondents were asked about what they considered the most significant threats to their businesses or work in agriculture in the next five years by identifying a level for each risk on a 10 point scale (0=no risk to 10=high risk). As Table 5.8 shows, about 56.4 percent have high levels of concern about future environmental degradation (soil erosion, desertification, salinisation) and about 60.0 percent of the cohort perceived higher future risks related to climate variability and extreme weather events.

The greatest threat that was perceived by 79.8 percent of respondents is the risk of financial stress and related consequences. About 60.0 percent of households had high levels of concerns about the future lack of governmental support, wage reduction and lower purchasing power, and risks of unemployment. Risks related to a decrease in demand for a household's crop production were also perceived to a high level by 45.9 percent of households. Finally, only 37.4 percent of households perceived that they could be at greater risk from the effects of large agricultural businesses or corporations. This low concern about the effects of large corporations on their small business could be related to the fact that companies, cooperatives, government, and other entities hold only 7.87 percent of total agricultural land areas in Egypt, which is equivalent to 766 thousand feddans. In comparison, individual household landholdings still represent 92.73 percent of the total cultivated land area, constituting an area of 8965 million feddans in the country (Aboulnaga et al. 2017). Therefore, rural household farms continue to represent the most important production structure in the Egyptian agricultural system, and hence respondents are not reflecting a high level of concern about the influence of large corporations on their farming businesses.

Regarding future threats linked to the deterioration of resources, 71.8 percent of households perceived high future risks from pressures of urban development, and 65.7 percent of the cohort had high levels of concern about the risk of water shortages from the River Nile in the future. Approximately 60.0 percent of households perceived future threats related to a decline in their capacity to access resources (electricity, fuel and land), while the risk of losing agricultural lands was perceived as important by 56.6 percent of the cohort.

Table 5.8: Distribution of households by perceptions of future risks in theagricultural sector

			Percentage	e	
		Low	Medium	High	Mean
Environmental related risks	Environmental degradation (soil erosion, desertification, salinisation) (n=321)	29.6	14.0	56.4	5.94
	Climate variability and extremes (n=320)	29.4	14.7	55.9	6.13
Economic and	Reduction in demand for your production (n=320)	36.6	17.5	45.9	5.08
risks	Impacts of large agricultural businesses or corporations (n=302)	37.4	25.2	37.4	4.85
	Wage reduction or lower purchasing power (n=329)		14.9	58.4	6.35
	Lack of governmental support (n=326)	21.8	13.8	64.4	6.82
	Financial stress and related impacts (n=326)	8.0	12.3	79.8	7.88
	Unemployment (n=325)	27.4	15.1	57.5	5.94
Resources deterioration related risks	Shortage in Nile river water (n=329)	20.7	13.7	65.7	6.85
	Decrease of resources (electricity, fuel and land) (n=302)	24.7	16.3	59.1	6.38
	Loss of agricultural land (n=318)	29.6	13.8	56.6	5.93
	Urban development pressure (n=319)	16.0	12.2	71.8	7.22

Source: Field survey 2018

Results also suggest that again, there were no significant differences in perceptions of future agricultural risks between households living in Faraskur and those who live in Kafr Saad (Pearson Chi-square p=0.572) (Table 5.9). Therefore, households within the studied areas seem to be experiencing very similar levels of future risk related to the agricultural sector, including threats from climate variability and extremes.

Table 5.9: Distribution of overall household perceptions of future risks by central administration

			Low	Neutral	High	Total
	Faraskur	Count	18	78	74	170
		% within Local Unit	10.6	45.9	43.5	100.0
	Kafr Saad	Count	13	74	80	167
		% within Local Unit	7.8	44.3	47.9	100.0
Total		Count	31	152	154	337
		% within Local Unit	9.2	45.1	45.7	100.0

Source: Field survey 2018

Risks related to the deterioration of local natural resources have been of significant concern to several respondents, who commented on the above-mentioned future threats of agriculture in Egypt (Table 5.10). In this regard, some respondents emphasised the risks of water deficits. For instance, a farmer (interview 4, Faraskur) raised concerns about water shortages in the local area and mentioned:

"Currently, there is one general issue on Damietta's level, which is the water in relation to rice cultivation. We want to cultivate rice, and the water is deficient, that's why the government has decreased rice cultivation to reach 36 percent of the total cultivated land area this year ... this is because of the "Renaissance Dam", and wait till it works with its full capacity, the area will be only 12 percent, and as a proof to what I say is that the government is planning to import all types of Indian rice".

The deterioration of soil quality due to the excessive dependency on polluted water for irrigation was a major future concern to some respondents as well. Moreover, the imposed national rice policy has forced farmers to limit their rice cultivation and, therefore, raised further concerns about the deterioration of the soil quality where high levels of salinity prevail. Some other risks were related to lack of government support, high input prices, and the quality and availability of inputs.

Table 5.10: Respondents' most important open-ended comments on their perceived
future risks within the agricultural sector

Future risks	
Economic and business-related risks	 High input prices Low quality of pesticides Lack of agriculture extension specialists Lack of governmental support Government doesn't provide enough agricultural inputs Rice policy entailing cutting down areas cultivated with rice
Resources deterioration related risks	 Water deficiency Weak policies will enforce farmers to produce less and build on lands Residents should stop polluting the water canal as it affects crops
Environmental related risks	 Foot and mouth disease in cattle due to warmer climate Desertification for lands as a result of irrigation from sewage water The policy requiring no rice cultivation on coastal regions will increase soil salinity degrading lands The land can't be cultivated with crops other than rice due to using sewage water and high salinity

Source: Field survey 2018

Again, an initial conclusion is that most of the studied rural households perceive of high levels of risk from future climate variability and extreme weather events. They also have high levels of concern about future economic losses and the deterioration of the natural resources base in the studied areas.

5.5 Introducing household groups based upon their income from agricultural activities

As discussed in Chapter 4, the majority of respondents work primarily as farmers or waged agricultural labourers. However, at the household level, there is a notable percentage of household members who work in other non-agricultural occupations. Although most households depend on more than one income source, dependency on agricultural activities to generate livelihoods varies from as low as 10 percent to 100 percent. This wide variation suggests variable levels of potential climate change vulnerability based on the recognition of the differences between households in terms of their reliance on farming activities and the percentage of income coming from this sector. By grouping households into three groups based upon the contribution of agricultural activities to the total income of the household, the levels of perceived risk and opportunities for adaptation were analysed further. The three groups are as follows:

- 1. **Group 1** includes those households that have one or more sources of income, and income from agricultural activities represent less than 50 percent of their total income. Meaning that for Group 1 households, agricultural activities are not the primary source of income.
- 2. **Group 2** involves households whose income is largely dependent on agricultural activities, contributing from 50 to less than 95 percent of their total income. Although farming activities for those households are considered to be important income-generating activities, they still have at least one other source of income that could balance or compensate any losses or failures they are or will potentially encounter within the agricultural sector.
- 3. **Group 3** encompasses households who are almost wholly dependent on agricultural activities for their livelihood, including subsistence food production, and where income from agricultural activities represents 95 percent or more of their total income. Hence, working in agriculture is a vital component of their livelihood.

This categorisation into three groups helps to generate an understanding of the key (social, economic and cultural) differences between rural households. It also allows the author to elaborate on how different groups of farmers perceive climate change. The three groups could be considered a reflection of the model of agrarian transition or agricultural transformation explained broadly in Chapter 2 (Section 2.2). During this process, societies pass through a process from being an agricultural-based country to a developed or industrialised country (World Bank 2015; Barrett, Christian & Shiferaw 2017). A vital characteristic of the agricultural transformation of societies is that the financial and labour resources increasingly flow out of the agricultural sector and into other non-farm sectors (Timmer 1988).

At the micro-level analysis presented in this study, the three groups of households could be seen to reflect the first three stages of a labour transition in relation to the agricultural transformation presented by the World Bank (2015): from chiefly agricultural-based economies which represent Group 3 in this case; to a pre-transition or transition group (i.e. Group 2); to a group more typical of an urbanising or late-transition economy (Group 1). Group 1 (n=75) represents 21.4 percent of the interviewed households; Group 2 (n=151) represents the majority of households at 43.1 percent; Group 3 (n=124) includes 35.4 percent of households, and these people depend almost solely on agriculture as the primary source of their income (Table 5.11).

		Frequency	Percentage
Group	1 <50 %	75	21.4
	2 50-95 %	151	43.1
	3 ≥95 %	124	35.4
	Total	350	100.0

Table 5.11: Distribution of households by income from agricultural activities

Source: Field survey 2018

Most respondents (generally the heads of the households), or 71.7 percent, within Group 1 work in highly skilled and semi-skilled non-agricultural sectors, whereas almost all respondents in Group 3 are working as farmers and waged labourers. About 66.0 percent of respondents within households in Group 2 are farmers and waged-labourers, while around a quarter work in highly skilled non-agricultural occupations (Table 5.12). Results from Pearson Chi-square test revealed no significant difference (p=0.693) in the distribution of household heads or respondents across the different occupation groups

living in the two studied administrations (Table 5.13). This lack of distinction between the two places further supports a focus on the differences in household types based on the contribution of the agricultural sector to their livelihoods.

Table 5.12: The distribution of agricultural dependence groups by the respondent	s'
primary occupations in Damietta	

			The primary occupation of respondent							
				Unskilled	Semi-skilled	Highly-	-			
			Farmers	non-	non-	skilled non-				
			and waged	agricultural	agricultural	agricultural				
			labourers	occupations	occupations	occupations	Total			
Income	1 <50 %	Count	5	8	23	35	71			
from		%	7.0	11.3	32.4	49.3	100.0			
agriculture		within								
by Group	2 50–95 %	Count	96	5	9	36	146			
		%	65.8	3.4	6.2	24.7	100.0			
		within								
	3 ≥95 %	Count	114	1	0	0	115			
		%	99.1	0.9	0.0	0.0	100.0			
		within								
Total		Count	215	14	32	71	332			
		%	64.8	4.2	9.6	21.4	100.0			
		within								

Source: Field survey 2018

Table 5.13: The distribution of households living in the two studied areas by the respondents' primary occupations in Damietta

				The primary occu	pation of respond	lent	_
				Unskilled	Semi-skilled	Highly-	_
			Farmers	non-	non-	skilled non-	
			and waged	agricultural	agricultural	agricultural	
			labourers	occupations	occupations	occupations	Total
Local	Faraskur	Count	115	7	19	43	184
Adminis		%	62.5	3.8	10.3	23.4	100.0
tration		within					
	Kafr	Count	100	7	13	28	148
	Saad	%	67.6	4.7	8.8	18.9	100.0
		within					
Total		Count	215	14	32	71	332
		%	64.8	4.2	9.6	21.4	100.0
		within					

Source: Field survey 2018

5.5.1 Agricultural dependency and perceptions of climate variability and extremes

Agriculture is an activity that is particularly sensitive to climate change (Parry & Carter 1989; Smit & Skinner 2002). Hence, it is of great importance to understand the differences present across the three household groups in perceiving climate change, and the social and economic differences that would contribute to shaping their perceptions. To do so, the Kruskal–Wallis test was used to analyse the strength of differences among

the three groups in perceiving climate variability and extremes (Table 5.14). Households in Group 3, comprising those who are wholly dependent on the agricultural sector for their livelihood, have a higher perception of the adverse impacts of climate variability and extremes than the other two groups (K–W p=<0.001). Group 3 also perceived increased incidences of climate variability and extreme weather than Group 1 (K–W p=0.023). This heightened perception of climate change could be a result of their high dependence on and work within the climate-dependent agricultural sector, where any minor climate stress could be considered a significant potential threat to their livelihood conditions.

Table 5.14: Agricultural dependence groups' mean rankings of householdperceptions of climate change

	< 50%		50-95 %		≥95 %		Kruskal– Wallis (K–
	Group 1 (mean)	Group 1 (mean rank)	Group 2 (mean)	Group 2 (mean rank)	Group 3 (mean)	Group 3 (mean rank)	W) significance level (p)
Overall perception of adverse impacts of climate variability	5.33	144.42 ^a	5.32	141.46 ^a	6.14	184.59 ^b	<0.001
Frequency of experiencing climate variability and extremes in a year	4.00	125.76 ^a	4.51	133.20 ^{a,b}	4.57	156.92 ^b	0.023

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of ^{a,b,c}. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

To help to discuss these results in more detail, Table (5.15) shows the differences between groups in their experiences with and perceptions of the adverse effects of climate variability and extremes on agriculture and other livelihood conditions. Group 3 has higher mean ranks of perceived climatic impacts on crop productivity (K–W p=<0.001), crop quality (K–W p=< 0.001), livestock productivity (K–W p=<0.001), health conditions (K–W p=0.001) and individuals' or labourers' productivity (K–W p=0.003), in comparison with the other two groups. Furthermore, households in Group 3 also have a higher perception of climate variability and extremes on soil fertility and quality (K–W p=0.001), income (K–W p=0.004) and the amount of time spent on agricultural operations (K–W p=0.001) than households in Group 2. Again, the perceived exposure of agricultural systems to climate change could be a result of their lack of options in diversifying their income and their high engagement in agricultural activities. Accordingly, it appears that respondents from households in Group 3 generally have

higher perception levels of climate change adverse impacts and perceive that their livelihoods are more vulnerable to its impacts.

Perception of the	< 50%		50-95 %		≥95 %		Kruskal– Wallis (K–
climate variability and	Group	Group 1	Group	Group 2	Group 3	Group 3	W)
extremes on:	1 (mean)	(mean rank)	2 (mean)	(mean rank)	(mean)	(mean rank)	significance level (p)
Crop productivity	6.87	145.37 ^a	6.70	133.95 ^a	8.18	192.37 ^b	<0.001
Crop quality	6.54	149.81 ^a	6.44	135.70 ^a	7.81	187.74 ^b	<0.001
Livestock productivity	5.83	137.80 ^a	6.20	137.17 ^a	7.71	193.36 ^b	<0.001
Soil fertility or land quality	5.04	155.49 ^{a,b}	4.33	136.90 ^a	5.98	182.98 ^b	<0.001
Income	6.70	148.12 ^{a,b}	6.77	143.66 ^a	7.44	179.91 ^b	0.004
Health conditions	5.67	140.34 ^a	5.97	145.70 ^a	6.89	182.33 ^b	0.001
Productivity of individuals or labourers	6.46	139.97 ^a	6.89	147.43 ^a	7.50	180.64 ^b	0.003
Amount of time spent on agricultural activities	6.66	151.16 ^{a,b}	6.59	140.11 ^a	7.44	182.02 ^b	0.001

 Table 5.15: Agricultural dependence groups' mean rankings of household

 perceptions of climate variability and extremes adverse impacts

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of a,b,c. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

5.5.2 Household group perceptions of employment in the agricultural sector

This section addresses the differences between the three household groups in relation to their perceptions of working in the agricultural sector. Results from the Kruskal–Wallis test (Table 5.16) show that households in agriculturally-dependent Group 3 have a higher mean rank for the likelihood of experiencing income or crop losses as a result of climate variability and extremes (K–W p=>0.001), and the possibility of being poisoned by the application of pesticides (K–W p=>0.001) when compared with the other two groups. Respondents in Group 1, in general, work primarily in non-agricultural occupations and therefore, many did not respond to these questions.

		< 50%		50-95 %		≥95 %		Kruskal– Wallis (K–	
		Group 1	Group 1	Group 2	Group 2	Group 3	Group 3	W)	
		(mean)	(mean rank)	(mean)	(mean rank)	(mean)	(mean rank)	significance level (p)	
The likelihood of	Exposure to losses in income or work opportunity due to climate variability	6.13	169.33ª	5.49	148.87ª	7.42	211.66 ^b	>0.001	
	Getting poisoned because of pesticides application	4.19	137.69ª	4.74	153.29ª	7.15	225.42 ^b	>0.001	
Agreement with the future of agriculture in Egypt is uncertain		5.89	184.93ª	4.56	146.82 ^b	6.59	204.72ª	>0.001	

 Table 5.16: Agricultural dependence groups' mean rankings of household

 perceptions of working in agriculture

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of ^{a,b,c}. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

Results also revealed that there is a significant difference among the three groups in their agreement with the statement about the uncertainty of the future of agriculture in Egypt, with Group 2 having a lower mean than the other two groups (K–W p=>0.001). This result suggests that households in Group 2 which are still strongly involved in the sector, but not wholly dependent upon it, are more optimistic about the future of Egyptian agriculture.

5.5.3 Agricultural dependency and perceptions of future risks in the agricultural sector

It is perhaps unsurprising that the overall perception of future agriculture-related risks is higher amongst households highly dependent on farming activities for their livelihood. Households in Group 3 have a higher mean rank of their perception of future risks (K–W p=<0.001) than the other two groups (Table 5.17). They also have significant doubts about their future economic conditions and the sustainability of the natural resources they have access to. In particular, households in Group 3 perceived higher future risks linked to climate variability and extremes (K–W p=<0.001) than Group 2. In addition, the mean ranks of the perceived future potential decline in resources (K–W p=0.004) and environmental degradation (K–W p=0.004) were higher among households in Group 3 than those in Groups 1 or 2.

The perceived risks of the reduction of the future demand for their produce (K–W p=<0.001), wage reduction or lower purchasing power (K–W p=<0.001), unemployment (K–W p=<0.001), and lack of government support (K–W p=0.005) were found to be higher among households in Group 3, than Groups 1 or 2. Agricultural dependent households also seem to have more concerned about future financial stress than households in Group 2 (K–W p=0.002). This result could be due to their less-diversified income and high dependency on working in a sector that is characterised by low wages, unstable income, poor health and safety regulations and declining natural resource conditions (Powell et al. 2006; Hurst et al. 2007). Finally, households in Group 3 are more likely to perceive future risks of losing their farmlands (K–W p=0.002) and threats linked to shortages in the Nile river water (K–W p=0.003) than Group 2. In conclusion, households in Group 3 have higher levels of concern regarding their future economic and livelihood conditions.

Perception of future	< :	50%	50-9	95%	≥95%		Kruskal–
risks	Group	Group 1	Group	Group 2	Group 3	Group 3	Wallis (K–
	1	(mean	2	(mean	(mean)	(mean	vv) significance
	(mean)	rank)	(mean)	rank)		rank)	level (p)
Climate variability and		ab		9		h	
extremes	5.87	152.63 ^{a,b}	5.64	146.10 ^a	7.00	185.95	< 0.001
Environmental							
degradation (soil							
erosion, desertification,						h	
salinisation)	5.21	141.86"	5.72	152.78 "	6.73	184.99	0.004
Wage reduction or				a		a a a s=b	
lower purchasing power	5.44	137.33°	5.88	149.56"	7.54	202.675	< 0.001
Reduction in demand						h	
for your produce	3.74	130.49 ^a	4.90	153.81"	6.20	190.31	< 0.001
Lack of government						h	
support	6.36	149.48"	6.58	152.44"	7.43	186.78	0.005
Nile river water		a b				h	
shortages	6.64	158.11 ^{a,b}	6.37	150.09"	7.60	188.88	0.003
Decrease of resources							
(electricity, fuel, land)	5.91	146.40 ^a	6.06	149.50 ^a	7.09	184.36 ^b	0.004
Financial stress and						_	
related impacts	7.59	158.04 ^{a,b}	7.60	147.36 ^a	8.43	188.30 ^b	0.002
Loss of the agricultural		_ 1					
lands	5.69	154.43 ^{a,b}	5.48	143.73 ^a	6.70	184.28 ^b	0.002
Unemployment	5.17	144.89 ^a	5.57	147.89 ^a	6.93	194.47 ^b	< 0.001
Overall perception of		_		_			
future risks	5.80	147.48 ^a	5.96	152.77 ^a	7.08	202.93 ^b	< 0.001

 Table 5.17: Agricultural dependence groups' mean rankings of household

 perceptions of future risks within the agricultural sector

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of ^{a,b,c}. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

5.5.4 The socio-economic conditions of household groups

This section focuses on the differences between the three groups with differing dependence on agriculture in relation to a range of social and economic variables. As shown in Table 5.18, results from Kruskal–Wallis test revealed that there is a significant difference between the three groups in the total land area (owned, rented, shared), with Group 2 (agricultural sector contributes from 50 percent to less than 95 percent household income) having a higher mean rank in comparison with Groups 1 and 3 (K–W p=>0.001). This result also gives an indication of the importance of agricultural and non-agricultural activities to this group. In the case of Group 2, the non-agricultural income may be assisting in providing the financial capital needed to manage their agricultural businesses, which still contribute to a large part of their income, and could be working as a safeguard from any potential losses in agriculture, and vice versa.

	< 50%		50-9	50-95%		≥95%	
	Group 1	Group 1	Group 2	Group 2	Group 3	Group 3	Wallis (K–
	(mean)	(mean	(mean)	(mean	(mean)	(mean	W)
		rank)		rank)		rank)	significance
							level (p)
Total land area	1.63	121.31 ^a	3.35	207.95 ^b	2.36	168.76 ^c	>0.001
Age of the respondent	46.46	140.56 ^a	53.76	201.87 ^b	49.09	164.52 ^a	>0.001
Years of education of				h			
the respondent	11.08	233.51 ^a	7.46	178.94 ⁰	5.06	136.22 ^c	>0.001
Number cows or							
buffalos owned by the				1			
HH	2.11	129.63 ^a	4.18	201.69 ^D	3.04	169.90 ^c	>0.001
Total number of		1-		1			
employed members	1.6	180.79 ^b	1.8	167.16 ^b	1.34	145.93 ^a	>0.001
Total number of							
employed HH members				1			
in agriculture	0.13	78.82 ^a	0.74	171.62 ^D	1.27	238.70 [°]	>0.001
Total monthly		_		1.		_	
expenditure	6123.66	201.69 ^a	5006.16	153.73 ^D	5803.1290	186.17 ^a	0.001

 Table 5.18: Agricultural dependence groups' mean rankings of demographic and

 economic conditions

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of a,b,c. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

The mean rank of the years of education of the respondents is higher in Group 1 than in Groups 2 and 3 (K–W p=>0.001). Higher educational attainment could directly help explain the reason for their access to non-agricultural activities for a living, with more positions in business or professional, highly–skilled jobs. Most of the households in this group have higher degrees, and they are more likely to work in professional occupations

that fit with their degrees in other sectors, even if they maintain a small interest in agriculture. Even if they have a degree in agricultural sciences, they are likely to prefer to work as a governmental official or establish a private business. Respondents in Group 3 have the lowest mean rank of years of formal education and hence are the least educated. Moreover, Group 3 was found to have a smaller number of employed members, either in the agricultural or non-agricultural sectors, in comparison to the other two Groups (K–W p=>0.001). The number of household members employed in the agricultural sector was found to be higher among households in Group 3 in comparison with Group 1 or 2 (K–W p=>0.001).

The mean rank of the respondent age was also highest in Group 2 in comparison with the other two groups (K–W p=>0.001). This relative old age of respondents, who were generally the head of the household, could indicate that household heads in Group 2 have more accumulated experience in farming activities. Also, there is a significant difference between the three groups in asset acquisition, represented in the number of cows or buffaloes owned by each household category, where Group 2 has a higher mean rank than the other two groups (K–W p=>0.001). Households in Group 2 possess an average of 4 cows or buffaloes, which means that they have also acted to diversify their income sources within the agricultural sector to generate income from both crop and animal production, suggesting again that households in Group 2 are a group of relatively wealthy landowners.

This argument seems to be supported by another significant difference that was found in the expenditure/income of households among the three groups. Households in Group 2 have lower monthly expenditure in comparison to the other two groups (K–W p=0.001). This result might suggest that households in this group could be effectively managing their agricultural business, and saving money, especially that they have other sources for non-farm income, bigger farmland areas, and more assets (i.e. cows and buffaloes). This result is further clarified by analysing each expenditure item for each group of households (Table 5.19). Households in Group 1 were found to be spending more money on food (K–W p=0.002) and clothes (K–W p=0.001) in comparison with the other two groups. Less reliance on farming activities could be a reason for their relative high expenditure on food as the farmland they possess might not be providing sufficient. Households in Group 1 also have a higher mean rank of expenditure on education (K–W p=0.018) and gas (K–W p=>0.001) than households in Group 2. Households in Group 1 are mainly engaged in

highly-skilled non-agricultural professions, which might be a reason for their high spending on education. They are encouraging the younger members of their households to continue their education which could guarantee them a better future in non-agricultural occupations. On the contrary, households in Group 3 were found to be spending more money on health (K–W p=>0.001) and house rent (K–W p=0.009) than households in Group 2. All households in Group 2 (except for one household) own their houses, whereas about 8 percent of households in Group 3 are renting their homes, which clarifies the relatively high figure for expenditure on house rents. Moreover, households in Group 3 are more exposed to health trade-offs linked to farming activities, such as pesticide application and frequent exposure to polluted water, which could justify the more spending on health. Households in Group 3 also have the highest mean rank of expenditure on the water than those households in both other groups, perhaps due to the remotely-located houses in Group 3 households raising the costs of water delivery.

	< 5	< 50%		50-95%		≥95%	
	Group 1	Group 1	Group	Group 2	Group	Group 3	Wallis (K–
	(mean)	(mean	2	(mean	3	(mean	W)
		rank)	(mean)	rank)	(mean)	rank)	Significance
							level (p)
Expenditure on food	2742.6	210.61 ^a	2260.9	168 . 48 ^b	2171.0	162.82 ^b	0.002
Expenditure on clothes	610.9	212.99 ^a	374.9	162.25 ^b	396.0	168.96 ^b	0.001
Expenditure on health	484.7	177.92 ^{a,b}	364.7	146.61 ^a	660.0	209.22 ^b	>0.001
Expenditure on education		0		h		a b	
	681.1	198.51 ^a	502.1	160.22^{0}	715.4	180.19 ^{a,b}	0.018
Expenditure on gas		9		h		ab	
	114.5	180.07 ^a	107.0	143.68	137.7	211.48 ^{a,b}	>0.001
Expenditure on house rent							
	43.3	177.45 ^{a,b}	0.26	169.11 ^a	56.9	182.10 ^b	0.009
Expenditure on water	8/11	172 74 ^a	76.0	149 99 ^a	135.2	208 23 ^b	>0.001

 Table 5.19: Agricultural dependence groups' mean rankings of household

 expenditure items

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of ^{a,b,c}. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

Regarding social participation of the households (as shown in Table 5.20), those in Group 2 are likely to be more actively participating in social activities that require more formal networks represented in the discussions on public affairs in the village (K–W p=0.002), organising public activities in the area (K–W p=>0.001) and attending local community events (K–W p=0.001). However, they are less likely to give advice to other villagers in comparison with Group 3 households (K–W p=0.043). Group 3 households are more likely to participate in solving conflicts, if any, between other villagers than the other two

groups (K–W p=>0.001). Moreover, households in Group 2 were significantly less likely to have asked for help in the year before the survey (K–W p=>0.001), and have relatively poorer relationships with their relatives (K–W p=>0.001), neighbours and other villagers (K–W p=>0.001). These results could indicate that households in Group 2 are more dependent on formal social networks than the informal ones, given their high participation and involvement in social activities. In general, households of Group 3 have the least mean rank in their overall social participation (K–W p=0.045) in comparison to the other two groups. Although their relationship with informal social networks seems to be more developed, households in Group 3 are less involved in informal social activities.

Table 5.20: Agricultural dependence groups'	mean rankings of social participation
and relationships	

	< 5	50%	50-	50-95%		5%	Kruskal–
	Group 1	Group 1	Group 2	Group 2	Group 3	Group	Wallis (K–
	(mean)	(mean	(mean)	(mean	(mean)	3	W)
		rank)		rank)		(mean	Significance
						rank)	level (p)
The number of times you							
asked for help in the past				h			
year	43.26	124.60 ^a	9.28	78.88 ⁰	16.95	133.54 ^a	.000
Rate the relationship with		h				h	
your relatives	9.09	191.33 ⁰	8.62	152.15 ^a	9.23	194.36 ⁰	>0.001
Rate the relationship with							
neighbours and village		h				ь	
fellows	8.72	188.84 ⁰	8.11	148.76 ^a	9.10	199.99 ⁰	>0.001
Participation in discussing		ab				h	
local public affairs	4.64	166.74 ^{<i>a</i>,0}	5.81	196.68 ^a	4.36	155.01	0.002
Participation in organizing		h				h	
public activities	3.72	170.23	4.87	204.9 ^a	2.83	142.87	>0.001
Participation in giving		ab				h	
advice to other villagers	6.71	169.50 ^{a,0}	6.80	163.80 ^a	7.32	193.37 ⁰	0.043
Participation in solving						h	
conflicts between others	6.79	167.06 ^a	6.64	154.60 ^a	7.84	206.06 ⁰	>0.001
Participation in attending		a h				h	
local community events	3.68	173.00 ^{a,b}	4.17	197.13 ^a	2.89	150.67 ⁰	0.001
Overall social participation	5.11	166.49 ^a	5.7	190.83 ^a	5.04	162.28 ^b	0.045

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of ^{a,b,c}. Different letters indicate a significant difference between groups. Groups with same letter are not significantly different.

There are significant differences between respondents within household groups who were members of syndicates (Pearson Chi-squared p=>0.001), or farming organisations and unions (Pearson Chi-squared p=>0.001). There are large minorities of 45.3 percent of respondents in Group 1 and 37.7 percent in Group 2 who are members of syndicates, while almost no respondents in Group 3 are members of syndicates (Table 5.21).

Table 5.21: Distribution of agricultural dependence groups by respondent'smembership in syndicates

			Are you a n		
			syn	_	
			Yes	No	Total
Income from	1 <50 %	Count	34	41	75
agriculture by Group		% within	45.3	54.7	100.0
	2 50–95 %	Count	57	94	151
		% within	37.7	62.3	100.0
	3 ≥95 %	Count	5	119	124
		% within	4.0	96.0	100.0
Total		Count	96	254	350
		% within	27.4	72.6	100.0

Source: Field survey 2018

Close to 17.0 percent of respondents in Group 2 are members of one or more farming organisations or unions, while most respondents within the other two groups are not (Table 5.22).

 Table 5.22: Distribution of agricultural dependence groups by respondent's

 membership in farmers' organisations or unions

			Membership in a organisations		
			Yes	No	Total
Income from agriculture by Group	1 <50 %	Count	1	74	75
		% within	1.3	98.7	100.0
	2 50-95 %	Count	26	125	151
		% within	17.2	82.8	100.0
	3 ≥95 %	Count	2	122	124
		% within	1.6	98.4	100.0
Total	Count		29	321	350
		% within	8.3	91.7	100.0

Source: Field survey 2018

These results suggest that there is a significant percentage of households within Group 2 who have connections to several formal bodies and organisations and hence, could have a disproportionate influence on local decision-making on agricultural issues, including such decisions that influence their adaptive capacity. This result could also be directly related to having relatively large farmlands, as individuals who have relatively large land areas would hold or acquire more power through their relative wealth, or *vice versa*, and hence are more likely to be actively participating in local community politics and have strong connections with formal organisations.

Regarding housing and living conditions for the households in the study area, Group 3 has higher mean ranks than the other two groups regarding the distance between their houses and the nearest primary school (K–W p=>0.001), medical unit (K–W p=>0.001), market (K–W p=>0.001) or main road (K–W p=>0.001) (Table 5.23). Individuals in Group 3 are more likely to take longer to reach their farms or workplaces (K–W p=0.016). This result suggests that these households are more likely to be sited in more remote areas, which may infer that they are more likely to be marginalised from core social and natural resources.

	< 50%		50-	<95 %	≥95 %		Kruskal–
	Group 1	Group 1	Group 2	Group 2	Group 3	Group 3	Wallis
	(mean)	(mean	(mean)	(mean	(mean)	(mean	(K–W)
		rank)		rank)		rank)	significance
							level (p)
Satisfaction with local							
health services	2.75	153.71 ^a	4.10	201.87 ^b	2.81	156.56 ^a	>0.001
Number of rooms in the							
house	3.43	197.18 ^{a,b}	3.57	191.29 ^a	3.12	154.05 ^b	0.004
Distance to nearest							
primary school							
(kilometres)	1.03	172.71 ^{a,b}	0.94	148.12 ^a	1.38	204.98 ^b	>0.001
Distance to nearest							
medical unit							
(kilometres)	1.42	153.39 ^a	1.33	150.79 ^a	2.38	212.00 ^b	>0.001
Distance to nearest							
market (kilometres)	1.79	150.37 ^a	1.76	145.75 ^a	3.13	207.21 ^b	>0.001
Distance to main road							
(kilometres)	0.66	164.99 ^a	0.80	151.42 ^a	1.51	202.62 ^b	>0.001
Satisfaction with the							
house	7.96	186.00 ^{a,b}	7.98	186.88 ^a	7.35	155.29 ^b	0.019
Satisfaction with local							
refuse removal	3.20	162.68 ^a	4.31	196.89 ^b	2.94	157.21ª	0.002
Rate product selling in						154.67 ^{a,}	
past 5 years	5.39	133.17 ^a	6.71	173.81 ^b	6.05	Ь	0.008
Degree of job							
satisfaction	7.51	181.44 ^a	7.23	172.78 ^{a,b}	6.55	146.39 ^b	0.022
Time taken to reach							
work (in minutes)	22.98	170.98 ^{a,b}	18.81	146.23 ^a	26.44	178.14 ^b	0.016

 Table 5.23: Agricultural dependence groups' mean rankings of housing and other

 living conditions in the study area

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of ^{a,b,c}. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

In addition, there is a significant difference between groups in the degree of satisfaction with their houses (K–W p=0.019) and the number of rooms in them (K–W p=0.004), where members of Group 3 are less satisfied with their houses, which have fewer rooms in comparison with Group 2. Households in Group 2 are also more likely to be satisfied with health services provided in their village (K–W p=>0.001), and with local refuse removal (K–W p=0.002) than households in the other two groups. Finally, the

respondents of households in Group 3 are less satisfied with their occupations compared to those in Group 1, who work primarily in non-agricultural occupations (K–W p=0.022).

As shown in Table 5.24, households in Group 3 were found to be highly dependent on family members for most of the agricultural operations, including land preparation (K–W p=0.003), planting (K–W p=0.003), irrigation (K–W p=>0.001), crop operations (K–W p=>0.001) and harvesting (K–W p=0.001) in comparison to those in Group 1.

	< 50%		50-95 %		≥95 %		Kruskal–
Percentage of dependency	Group 1	Group 1	Group 2	Group 2	Group 3	Group 3	Wallis
on	(mean)	(mean	(mean)	(mean	(mean)	(mean	(K–W)
		rank)		rank)		rank)	significan
							ce level
							(p)
Family members in land		_		- 1-		1-	
preparation	26.59	131.33 ^a	34.39	162.19 ^{a,b}	42.17	180.18 ^D	0.003
Family members in planting	27.46	132.08 ^a	35.41	161.44 ^{a,b}	44.20	180.74 ^b	0.003
Family members in						1	
irrigation	49.57	124.72 ^a	67.82	153.66 ^a	84.76	196.32^D	>0.001
Family members in crop		_		_		1.	
operation	41.67	135.29 ^a	44.2	144.19 ^a	71.93	202.57 ^D	>0.001
Family members in		_		_		1-	
harvesting	23.77	137.08 ^a	26.40	155.29 ^a	36.93	186.01 ^D	0.001
Paid agricultural labour in		_		_		1-	
irrigation	45.29	194.01 ^a	29.42	168.89 ^a	14.29	130.08 ^D	>0.001
Paid agricultural labour in		_		_		1.	
crop operation	51.45	178.31 ^a	53.23	181.60 ^a	26.70	122.68 ^D	>0.001
Paid agricultural labour in		_		- 1-		1.	
harvesting	69.06	176.27 ^a	70.27	168.80 ^{a,b}	59.95	141.76 ^D	0.023

Table 5.24: Agricultural dependence groups' mean rankings of household reliance on unpaid/paid labourers for agricultural operations

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of a,b,c. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

Although households in Group 1 have a smaller mean area of farmland, their dependency on paid agriculture labourers is higher than those in Group 3, particularly in crop operations (i.e. pesticides or application of fertilisers, etc.) (K–W p=>0.001), irrigation (K–W p=>0.001) and harvesting (K–W p=0.023). This high reliance on paid labourers might suggest that households in Group 1 are more reluctant to participate even in basic agricultural activities like irrigation and crop operations. Their reluctance to participate in farming activities could be a result of working primarily in non-agricultural sectors and hence, have much less time available to spend farming that already contributes to less than 50 percent of their living.

There are also some significant differences between the three household groups concerning their satisfaction with several aspects of the agricultural situation in the study areas (Table 5.25). Group 3 has the highest mean ranks in the satisfaction with the availability of agricultural input (fertilizers, pesticides etc.) (K–W p=>0.001) and quality (K–W p=0.002); in comparison to Group 2, Group 3 householders also have the highest mean rank regarding their satisfaction with the quality of the agricultural drainage system (K–W p=0.003) and with the general agricultural situation in Egypt (K–W p=>0.001), compared to the other two groups. However, Group 3 is less satisfied with the prices of the different agricultural inputs (K–W p=0.036) in comparison to Group 2.

< 50%		50%	50-95 %		≥95 %		Kruskal– Wallis (K–
Satisfaction with	Group 1	Group 1	Group 2	Group 2	Group 3	Group 3	W)
	(mean)	(mean rank)	(mean)	(mean rank)	(mean)	(mean rank)	significance level (p)
A ari aultural inputa							
availability	7.00	167.49 ^{a,b}	5.84	135.49 ^a	7.39	192.45 ^b	>0.001
Agricultural inputs prices	2.72	159.88 ^{a,b}	2.75	173.84 ^a	2.00	143.75 ^b	0.036
Agricultural inputs quality	5.87	159.14 ^{a,b}	5.39	144.47 ^a	6.53	185.37 ^b	0.002
Agricultural drainage system quality	6.16	149.98 ^a	6.03	148.12 ^a	6.98	186.27 ^b	0.003
The agricultural situation in general	2.33	126.80 ^a	3.07	157.20 ^a	4.24	188.80 ^b	>0.001

Table 5.25: Agricultural dependence groups' mean rankings of aspects related to the satisfaction with the agricultural situation in the study area

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of ^{a,b,c}. Different letters indicate a significant difference between groups. Groups with same letter are not significantly different.

These findings suggest that although respondents in Group 3 are not satisfied with their occupation as farmers and waged labourers, and have relatively poor housing conditions and lower social participation and connections, their satisfaction with the agricultural situation is still relatively high, possibly because it is all that they know. In other words, this could result from their relatively low educational attainment and their high dependency on agriculture for a living. Hence, their opportunities for exposure to different external occupations are limited, having little chance of comparing their lifestyles with those of others.

5.7 Summary

Rural households in the study area were found to experience great climate variability. The most observed patterns of climate variability are related to a higher temperature and more frequent heatwaves, in addition to experiences with changes in rainfall patterns and cold spells. Such increasing variability in climatic conditions was also found to be having negative impacts upon the agricultural productivity of crops and livestock in Damietta, which in turn had adverse influences on the income and other living conditions of the sampled households. Most households perceive of increasing risks linked to climate variability and extremes, and environmental degradation. Perceptions of future risks were also high when looking at financial stress, unemployment, lack of government support, urban development pressure, water deficiency and the deterioration of the local resources.

It was found that there were no significant differences in the perceptions of climate variability and other future risks across the two studied local administrations. Stemming from this result and guided by the insights of the phases of agrarian transition, three groups of households were proposed based upon the contribution of farming or agricultural activities to their overall income. A focus on the classification of households according to their dependency on agriculture to generate their livelihoods could help to reflect the differences in the needs and levels of vulnerability within agrarian communities. Significant differences were identified between the three groups of households in relation to their perceptions of climate change and other future risks related to Egyptian agriculture. It was found that households in Group 3, whose livelihoods are vitally dependant on agriculture, were more likely to experience and perceive climate variability and have more concerns about the future of agriculture in Egypt than those in Groups 1 and 2, who were less dependent on agriculture. Household heads in Group 3 were found to have lower social participation and educational attainment, and their households live in smaller and remotely-located houses – again, issues that are likely to limit their adaptive capacities to climate change.

Chapter Six: Households' adaptation to climate variability and extremes in Damietta

6.1 Introduction

Egypt has been experiencing increased climate variability and extreme weather events over the recent past, and those climate changes are expected to get worse in the future (Saber 2009; Met Office et al. 2011; Seyam 2011; Hereher 2016; Nashwan, Shahid & Abd Rahim 2018). Such conditions are adversely affecting agriculture and are threatening the livelihood of households who are dependent on the agricultural sector for subsistence or sales. Adaptation is hence crucial to overcome the adverse effects of climate change and to protect the livelihood of rural dwellers, and reduce the vulnerabilities of rural communities.

This chapter describes how the rural households studied adapt to climate variability and extreme weather events. It addresses both their agro-based livelihood strategies, together with other strategies involving alterations in the livelihood options of rural households. It also discusses in more detail the specific coping strategies used by households in overcoming the risks paired with sudden shocks generated by the extreme weather events that occurred in the two years that preceded the survey. Following the general discussion about adaptation, the influence of households' socio-economic characteristics upon their level of adaptation and choices for adaptation is then addressed. Since mobility can be considered one means of adaptation to environmental change (Bardsley & Hugo 2010), the chapter also focuses on the current migration patterns of the households studied, and whether migration as an adaptation choice for the different groups of households could be considered as being effective. Moreover, analytical emphasis is placed on household cultures of migration, which is vital in understanding their potential decisions to move, particularly in cases of current and future variability in climate. Households' conceptions of, and engagement in, migration is of particular importance to the study area, as Damietta is highly vulnerable to potential rising sea levels. Hence, an understanding of how people use migration could help to guide policy-makers when developing policies suitable for both the current situation and the emerging risk cultures of the affected inhabitants.

6.2 Household adaptation to long-term climate variability and short-term, environmental shocks in Damietta

As explained earlier in Chapter 5, most households perceive of and experience increasing climate variability and extreme weather events within their area. Such climate change has also adversely affected agricultural production, natural resources, incomes and other livelihood conditions. Those households that were perceiving the adverse effects of climate change were asked whether they had adopted strategies to reduce losses or adapt to the changing climatic patterns within the five years that preceded the survey. Adaptation strategies were grouped into two general categories: the first of which included on-farm measures that involved any direct changes in the farming operations, while the second group involved strategies related to any changes in the livelihood conditions of the households, such as those involving decisions about migrating.

On-farm adaptation measures relate mainly to those households who both perceived changes in climate variability and who had access to farmland, and hence, could make farm-related decisions. Subsequently, landless households who perceived increases to climate variability (n=21), and those households who hadn't perceived any change in climate conditions (n=35), including landless (n=2) and land-holders (n=33)) have been excluded from the current analysis. Overall, results revealed that only a few households (n=10, 3.4 percent) within the total households who perceived increased climate variability and who were able to make farm-related decisions (n=294), did not introduce any on-farm adaptation methods. Whereas the majority, or 96.6 percent, had adopted at least one on-farm adaptation strategy to overcome risks associated with the perceived climate change.

For forms of on-farm adaptation measures (Table 6.1), most households, (76.2 percent) have increased irrigation and more than half of them (56.1 percent) mentioned increasing the use of pesticides and fertilisers. The problem of irrigating with mixed water (sewerage, agricultural wastewater and Nile water) and the excessive use of chemicals may exacerbate the problem of soil salinity in the study area in the near future. Rural households have already been complaining about the poor quality of water in the study area especially in Faraskur.
To illustrate that point, a farmer (Interview 2, Kafr Saad) said:

"Desertification of agricultural lands in the area will happen sooner or later as long as we continue irrigating from sewage water".

Approximately 39.0 percent of households have made some changes in planting dates and nearly 37.0 percent of the cohort had stopped cultivating certain crops. Some farmers (n=7) commented further on this specific adaptation strategy, mentioning that beans, potatoes, tomatoes and cucumbers were particularly sensitive to slight changes in climatic conditions and extreme weather events, so they had stopped cultivating those crops. Some of the respondents (n=15) also mentioned stopping the cultivation of cotton as the warmer weather has encouraged pest infestations.

Table 6.1: On-farm adaptation strategies adopted by the households to reduce losses or adapt to climate variability and extremes from 2012 to 2017

	Percentage		
On-farm adaptation (n= 294)	Yes	No	
Increasing irrigation	76.2	23.8	
Improving irrigation systems	34.4	65.6	
Diversifying or increasing water sources for irrigation	13.6	86.4	
Planting different crop varieties	23.8	76.2	
Stop planting certain crops	37.4	62.6	
Increase use of chemicals fertilizers/ insecticides, pesticides or fungicides	56.1	43.9	
Changing planting dates	39.1	60.9	
Change livestock types or sizes	24.8	75.2	
Mulching for soil conservation or moisture protection	17.7	82.3	

Source: Field survey 2018

In addition, improving irrigation systems and changing the type and/or size of livestock are another two strategies adopted by 34.4 percent and 24.8 percent of households, respectively. Of the studied households, 23.8 percent had used different crop varieties that were more drought or heat resistant, or varieties with shorter growing seasons. Some farmers (17.7 percent) had adopted soil conservation or moisture protection measures such as mulching, while other farmers, representing 13.6 percent, had diversified or increased the sources of irrigation water.

Regarding other adaptation strategies, all households who perceived climate change have been included in this analysis. Such strategies involve mobility decisions that took place to overcome economic losses associated with increasing climate variability and extremes. Consequently, members of landless households, who were merely engaged as paid agricultural labourers, were also included in this analysis as they might have made some decisions to move, to avoid or reduce environmental risks and the associated income losses within the sector.

In general, few households have adopted other adaptation strategies such as migration or other occupational shifts as a result of climate change (Table 6.2). More specifically, only 9.9 percent of households mentioned that a formally educated family member had to join the agriculture sector, whereas 4.8 percent of the cohort indicated a shift from agricultural to non-agricultural occupations. Regarding migration as an adaptation strategy, only 2.2 percent of households stated that one or more family members migrated to nearby towns or market centres for work outside the agricultural sector, and 1.3 percent of the cohort had sent a family member to the city for a different occupation. Migration of a family member to another country or rural area to work in an agricultural-based occupation were two potential adaptation strategies adopted by only 2.9 percent and 1.0 percent of households, respectively.

Table	6.2: Liv	elihood	adaptation	strategies	adopted	by the	households	to r	educe
losses	or adapt	t to incr	easing clima	te variabil	ity and ex	xtremes	from 2012 t	o 20	17

	Perce	ntage
Other livelihood adaptations (n=315)	Yes	No
Formally educated family member joined agriculture	9.9	90.1
Shifted from agriculture to non-agriculture occupation	4.8	95.2
Migration of family members to nearby town/market centre for different occupation	2.2	97.8
Migration of family members to the city for different occupation	1.3	98.7
Migration of family members to another rural area for agro-based occupation	1.0	99.0
Migration of family member to another country	2.9	97.1

Source: Field survey 2018

Within the household group sampled, migration was not a widely implemented adaptation strategy. This result could be due to the wide diversification of sources of income for most households, and/or the high dependence on family members for helping in the

farmlands. Another aspect that could influence the low migration rate is the households' migration culture that is discussed later in this chapter (Section 6.4).

In order to understand the level of adaptation in households, the number of strategies adopted was counted in each, and households were then grouped into four categories based upon their level of adaptation as follows:

- 1. No adaptation although perceiving climate variability;
- 2. Low adaptation, i.e. 1 to 2 adaptation measures have been adopted;
- 3. Medium adaptation, i.e. 3 to 4 adaptation measures; and
- 4. High adaptation, i.e. equal to, or greater than, five strategies adopted by each household.

As shown in Figure 6.1, almost one-third of the households, or 27.7 percent, are considered to have low climate change adaptation, whereas 24.4 percent had high adaptation, and 39.0 percent had medium adaptation to changing climatic variability and extremes. Results also revealed that some households (n=20, or 8.9 percent) had not exploited any strategies for reducing the climatic risks. It was found that all of those, 20 households had no accessibility to land and were only engaged in the agricultural sector as waged agricultural labourers. Consequently, they are incapable of implementing major decisions regarding the adoption of any on-farm and agricultural-related adaptation strategies. Besides, they had not implemented any occupational or geographical mobility decisions to overcome the risks of climate change during the period 2012 to 2017.

Figure 6.1: Distribution of households by their adaptation level to climate change (in percentages)



Source: Field survey 2018

As previously mentioned in Chapter 4, most households had been exposed to one or more economic and environmental shocks in the two years 2016–2017 preceding the fieldwork

(Section 4.3.3). Households have also been exposed to loss of crops or income due to climatic variability or extremes. Roughly half the households (52.8 percent) had been exposed to crop or income losses as a result of extreme weather events during 2016 and 2017 in the study area.

In relation to households who were exposed to crop or income losses due to climate variability and extremes, and how they coped, 34.8 percent declared that they had to borrow money from relatives and friends and 27.1 percent sold their livestock (Figure 6.2). About 18.4 percent of the households took no action, and 10.6 percent of the cohort took an advance payment from the tradesman as a means of spreading their risks. Taking out a loan or working in a secondary occupation besides agriculture were two coping strategies that were followed by 4.3 percent and 3.3 percent of households, respectively. A few households (0.5 percent) mentioned other coping strategies such as the migration of a family member, selling assets from the house, and a child or woman working.

Figure 6.2: Household strategies for coping with income losses as a result of climate variability and extremes that took place in 2016 – 2017 (multiple-response question)



Source: Field survey 2018

Again it can be concluded that almost all of the households investigated did not adopt migration as an adaptation strategy to short-term sudden extreme weather events. This result was also evident in the in-depth interviews when respondents were asked whether the observed increases in climate variability and extremes would be likely to drive rural dwellers to migrate or move to different non-agricultural sectors. For example, a farmer (Interview 5, Faraskur) responded with the following:

"Nobody leaves agriculture. Where the farmer would go; there are no other secondary occupations available today. When the farmer feels very hopeless, he will lease his farmland".

Another farmer (Interview 1, Faraskur) mentioned the following:

"No, no, no, migration has no relation with that [i.e. climate change]. There are some cases of migration where people seek opportunities to "eat bread" [i.e. generate income]. Someone would see that farming is not worthy anymore and he has kids, schools, education, marriage, daughters, so he sees his livelihood in other places".

While another (Interview 2, Kafr Saad) also mentioned the following when he was asked about whether or not climate variability is causing migration in his local area:

"At this stage, no, but if the situation remains the same, I will leave the agricultural sector myself, and I think all farmers would do the same. The costs of farming are high, and the production doesn't cover the costs".

Similarly, Interviewee 11, from Faraskur mentioned:

"Of course no, and it will never happen because the farmer has only a sole source of income which is coming from farming; what other work would be done than that? The farmer can't find anything other than farming to do, he can't find any other occupations, and he is unable to work in any other sector".

Likewise, a farmer from Kafr Saad (Interview 10) reported:

"We can withstand any environmental problems, even if the water dried. The government controls the water flow in the canals anyway, and we could spend a week without water for irrigation, so we can adapt to any environmental issue. I have no occupation other than agriculture. I don't know how to work in any other sector. I would never think about quitting the sector or leaving my farmland".

6.2.1 Climate change adaptation according to the dependence of households on agriculture

The Kruskal–Wallis test was used to identify whether there were statistically significant differences between the three independent groups of households according to their level of dependence on agriculture in relation to their level of adaptation. As introduced in chapter 5, the three household groups were categorised according to the contribution of agricultural activities to the total income of the households:

- **Group 1** households who depend upon agriculture for less than 50 percent of their income;
- **Group 2** households' for which agricultural activities make up 50 to 95 percent of their income; and
- Group 3 households for which agricultural activities make up more than 95 percent of their income (i.e. who depend primarily on agricultural activities for a living).

Results showed that households in Group 3, who depend solely on agriculture for their livelihood, were adapting less to increased climate variability and extremes (K-W p=0.001). However, they were more likely to perceive and recognise the adverse effects of climate change than the other two groups (K-W p=<0.001) (see Section 5.5.1 in Chapter 5). Households in Group 2 were found to be more adapting to increasing climate variability and extremes, as shown in Table 6.3. This relatively high adaptation could be as a result of the presence of other non-agricultural sources of income that could be allocated for investments and improvements in the agricultural sector, which still contributes to a significant portion of their income. The respondents within Group 2 (generally the household heads) have a higher mean age; hence they could be more experienced in managing their agricultural businesses (see Section 5.5.4 in Chapter 5). Moreover, those household heads were also found to have higher social participation and were likely to be members of syndicates and/or one or more farming organisations (see Section 5.5.4 in Chapter 5). Households within Group 2 also seem to be living in more centrally-located houses in towns and, therefore, have better access to services and sources of information. Such accessibility to markets and those forms of social and natural resources could be improving their adaptive capacity to climate variability.

Table 6.3: Agricultural dependence groups' mean rankings of climate change adaptation levels

	< 50%		50-95%		>95%		Kruskal– Wallis (K–W)
	Group 1 (mean)	Group 1 (mean rank)	Group 2 (mean)	Group 2 (mean rank)	Group 3 (mean)	Group 3 (mean rank)	significance level (p)
Number of adaptation measures adopted by the HH	3.18	155.09 ^{a,b}	3.68	179.41 ^a	2.74	135.95 ^b	0.001

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of a,b,c. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

The Chi-square statistical test was used to examine further the specific distributions of the three groups of households by their choices of climate change adaptation strategy. Households in Group 2 were found to have adopted more advanced adaptation strategies such as planting different crop varieties (Pearson Chi-squared p=0.004), changing planting dates (Pearson Chi-squared p=0.004), and using strategies to conserve soil and protect moisture (Pearson Chi-squared p=>0.001), particularly when compared with Group 3 that was the least to adopt those three strategies (Table 6.4, 6.5 and 6.6).

Table 6.4: The distribution of agricultural dependence groups by their choice of planting different crop varieties from 2012 to 2017

			Yes	No	Total
Income from agriculture	<50 %	Count	13	55	68
		% within	19.1%	80.9%	100.0%
	50-95 %	Count	42	85	127
		% within	33.1%	66.9%	100.0%
	≥95 %	Count	15	84	99
		% within	15.2%	84.8%	100.0%
Total		Count	70	224	294
		% within	23.8%	76.2%	100.0%

Source: Field survey 2018

Table 6.5: The distribution of agricultural dependence groups by their choice of changing planting dates to reduce losses from 2012 to 2017

			Yes	No	Total
Income from agriculture	<50 %	Count	28	40	68
		% within	41.2%	58.8%	100.0%
	50-95 %	Count	61	66	127
		% within	48.0%	52.0%	100.0%
	≥95 %	Count	26	73	99
		% within	26.3%	73.7%	100.0%
Total		Count	115	179	294
		% within	39.1%	60.9%	100.0%

Source: Field survey 2018

Table 6.6: The distribution of agricultural dependence groups by their choice ofmulching to reduce losses from 2012 to 2017

			Yes	No	Total
Income from agriculture	e <50 %	Count	6	62	68
		% within	8.8%	91.2%	100.0%
	50-95 %	Count	38	89	127
		% within	29.9%	70.1%	100.0%
	≥95 %	Count	8	91	99
		% within	8.1%	91.9%	100.0%
Total		Count	52	242	294
		% within	17.7%	82.3%	100.0%

Source: Field survey 2018

Households in Group 3 were found to be more likely to have adopted strategies relying upon increasing the amounts of inputs from pesticides and fertilisers than other groups (Pearson Chi-squared p=>0.001) (Table 6.7). Again this result shows that households in Group 2 seemed to have better access to modern technology and to people who could provide them with the required advice to overcome business-related losses caused by climate change – consequently implying the better adaptive capacity of this group in comparison with households in Group 3.

Table 6.7: The distribution of agricultural dependence groups by their choice ofincreasing the use of chemicals/fertilizers/insecticides to reduce losses from 2012 to2017

			Yes	No	Total
Income from agriculture	<50 %	Count	31	37	68
		% within	45.6%	54.4%	100.0%
	50-95 %	Count	56	71	127
		% within	44.1%	55.9%	100.0%
	≥95 %	Count	78	21	99
		% within	78.8%	21.2%	100.0%
Total		Count	165	129	294
		% within	56.1%	43.9%	100.0%

Source: Field survey 2018

Roughly 41.0 percent of households within Group 2 stated that they had improved their observation of meteorological information as a result of climate variability and extremes, a significantly greater percentage than households in Group 3 (Pearson Chi-squared p=>0.001) (Table 6.8). The low educational attainment of the respondents in Group 3 could also limit their capacity to access or make use of meteorological information, or perhaps they are less likely to understand that timely meteorological information will become more critical with a changing climate.

Table 6.8: The distribution of agricultural dependence groups by their choice of improving their observation of meteorological information from 2012 to 2017

			Yes	No	Total
Income from agriculture	<50 %	Count	14	56	70
		% within	20.0%	80.0%	100.0%
	50-95 %	Count	53	76	129
		% within	41.1%	58.9%	100.0%
	≥95 %	Count	18	98	116
		% within	15.5%	84.5%	100.0%
Total		Count	85	230	315
		% within	27.0%	73.0%	100.0%

Source: Field survey 2018

6.3 Current migration pattern in the study area

Migration is considered as a means for adapting to climate change (Massey 1990; Adger & Adams 2013) and people migrate as an effective response (Bardsley & Hugo 2010;

Black et al. 2011a, 2011b; Foresight 2011; Adger & Adams 2013; Baldwin & Gemenne 2013; Gemenne & Blocher 2017). However, as discussed earlier in this chapter, mobility with its two forms (geographical and occupational) is not considered an adaptation strategy to climate variability and extreme weather events for most of the studied rural households in Damietta. This section addresses the current migration pattern of the households interviewed and highlights some of the limiting or enabling factors that influence their decisions. Results show that there were only 45 household members who lived outside their residential area for at least six months during the period from 2012 to 2017 (Table 6.9). About 42.0 percent were returned migrants at the time the survey was conducted, and 40.0 percent were considered permanent migrants (i.e. those who had been living outside the area for more than five years). About 9.0 percent of those who migrated were considered seasonal migrants, 6.7 percent of the cohort were new migrants, and 2.2 percent were temporary migrants.

		Frequency	Percentage
Nature of migration	New migrant	3	6.7
	Return migrant	19	42.2
	Seasonal migrant	4	8.9
	Temporary migrant	1	2.2
	Permanent migrant	18	40.0
Total		45	100.0

Table 6.9: Nature of migration in Damietta in 2018

Source: Field survey 2018

As Table 6.10 shows, the place of destination for 68.9 percent of those who migrated was outside Egypt, particularly neighbouring Saudi Arabia and Libya. About 20.0 percent of the cohort migrated to a town or city outside Damietta, followed by 6.7 percent and 4.4 percent migrating to a town within Damietta and a rural village outside the governorate, respectively.

Table 6.10: Place of destination for household members migrating from Damiettain 2018

		Frequency	Percentage
Within Damietta	A town within the governorate	3	6.7
Outside Damietta	Town/city	9	20.0
	Rural village	2	4.4
Overseas	Unidentified	5	11.1
	Saudi Arabia	13	28.9
	Nigeria	1	2.2
	UAE	1	2.2
	Jordan	1	2.2
	Libya	6	13.3
	Kuwait	3	6.7
	Iran	1	2.2
Total		45	100.0
a 5 111a			

Source: Field Survey 2018

Regarding drivers of migration, about 46.4 percent of responses stated that employment was the main reason for migration, followed by accumulating savings, which is also income-related, and represent 38.1 percent of responses (Table 6.11). Other reasons for migration involved education (7.1 percent), the rise in the price of agricultural lands (4.8 percent), and better infrastructure in the place of destination (3.6 percent).

 Table 6.11: Reasons for migration from Damietta 2018 (multiple-response question)

		Frequency	Percentage
	Work	39	46.4
	Education	6	7.1
	Accumulate savings	32	38.1
	Better infrastructure	3	3.6
	The rise in the prices of agricultural lands	4	4.8
Total		84	100.0

Source: Field Survey 2018

Only 15 households (or 4.2 percent) were receiving remittances from migrants at the time of the survey. The average amount of remittances received by these households was about EGP51,800 (roughly equivalent to AUD4,500) per year. The majority (34.8 percent) indicated that they received the remittances monthly; some other households received it bi-monthly (23.1 percent), quarterly (23.1 percent), and half-yearly (15.4 percent). Remittances were widely used for accumulating savings, as mentioned by 18.8 percent

of respondents, and about 17.2 percent of the cohort mentioned using it for living costs (Table 6.12). Households also mentioned spending the remittance money on marriage expenditures, health and education with 14.1 percent, 12.5 percent, and 7.8 percent of responses, respectively. Equal percentages of responses (6.3 percent) gave other uses for remittances, including buying lands or houses, investing in non-agricultural businesses, and repaying debts. Finally, only 3.1 percent of responses mentioned using the money for agricultural intensification or improvement. This result suggests that households who receive remittances prefer to save the money or spend it on day-to-day expenditure rather than investing it in their agricultural businesses or buying farmland. Rural households might fear spending money on the agricultural sector due to the increased uncertainties, and market fluctuations that are linked with the sector.

 Table 6.12: Uses of remittance money as stated by the studied households in

 Damietta 2018 (multiple-response question)

		Frequency	Percentage
	Living costs	11	17.2
	Savings	12	18.8
	Agricultural intensification/ improvement	2	3.1
	Education	5	7.8
	Health related expenditure	8	12.5
	Business in non-agricultural activity	4	6.3
	Buying land/house	4	6.3
	Buying assets	5	7.8
	Marriage expenditure	9	14.1
	Repayment of debts	4	6.3
Total		64	100.0

Source: Field Survey 2018

The reasons why most households do not migrate was also asked (Table 6.13). Of the household respondents questioned, 34.6 percent talked about stable family and living conditions within the area, followed by 21.6 percent, stating that they had no opportunity to leave their local area. Some other responses involved not wanting to leave their homes or area (7.9 percent), the inability to travel as a result of being old (7.5 percent), and being financially incapable as migration was seen as a costly action (7.61 percent). Some other respondents (6.7%) mentioned holding on to agriculture and being unable to leave their land. About 5.1 percent of responses stated that they already had a variety of income

sources within their local areas, and 3.1 percent of the cohort indicated that there were only poor income opportunities abroad and hence it was not worth migrating. A few respondents mentioned not wanting to leave their parents (2.4 percent), having a chronic disease (1.2 percent), and lack the capability of working in non-agricultural sectors (1.2 percent). Finally, very few responses mentioned limited capacities of family members to migrate and no social networks living abroad, with 0.8 percent of responses each.

		Frequency	Percentage
	A variety of income sources within the area	13	5.1
	Stable family and living conditions	88	34.6
	I am the only son and can't leave my parents	6	2.4
	Bad financial conditions and migration is costly	18	7.1
	Old age unable to travel/can't travel at this age	19	7.5
	Have chronic disease	3	1.2
	Don't want to leave home/place	20	7.9
	No good income abroad/ not worth migrating	8	3.1
	No opportunity for migration appeared for me	55	21.6
	Holding on agriculture/land	17	6.7
	Incapable of working in sectors other than agriculture	3	1.2
	No networks to help in finding an opportunity outside	2	0.8
	Limited capacities of family members	2	0.8
Total		254	100

Table 6.13: Reasons for not migrating from Damietta 2018

Source: Field Survey 2018

There was a general attitude that migration is not a practical action. One interviewee (Interview 7, Faraskur) speaking about migration patterns in his residential community stated the following:

"No, people don't migrate, and now countries have complicated work permits and travel regulations. To migrate in recent days, you need a lot of money and a sponsor or patron. Moreover, money abroad is not good anymore".

While another farmer (Interview 9, Kafr Saad) had the following opinion:

"Migration rates these days are lower than the past, most people have now returned, nothing is rewarding abroad".

6.4 Culture of migration in rural Damietta

Cultural values and traditions are among other essential factors that influence the adaptation decisions within households (Adger et al. 2013), particularly migration-related decisions (Cohen & Serkici 2011). The culture of migration itself is shaped by the structure of migration within the sending communities, or in other words, the history and continuity of migration within a particular place (Cohen 2004). Understanding the culture of migration of the rural households in the study area could explain their potential migration behaviour in times of climate-related stressors.

Since migration is generally a household decision rather than an individual decision according to the New Economics of Labour Migration theory (Stark & Bloom 1985; Massey et al. 1993; de Haas 2010b), it was important in this study to identify the degree to which household members influenced the migration decisions of each other. Figure 6.3 shows that most respondents stated that their spouses, parents, siblings, children, closest friends or neighbours had no influence on their decision about migration at all. In other words, they neither encouraged the respondents to stay in the local area nor to move away during the five years (2012–2017) that preceded the survey.

Figure 6.3: The influence of important household community members on encouraging the respondents to stay or move away from the local area between 2012-2017 in Damietta (n= 248)



Source: Field Survey 2018

In more detail, an almost equal percentage of respondents, approximately 14.7, mentioned that their spouses and children had encouraged them to stay. Whereas, only 11.2 and 6.3 percent of the cohort said that their spouses and children encouraged them to move away

from their local community, respectively. Parents and siblings were also found to encourage respondents to stay in the local area as stated by 13.7 percent and 14.9 percent of the respondents, respectively. About 12.6 percent of the respondents stated that their closest friends were most likely to encourage them to stay in their local community, and 12.9 percent of the cohort stated that they were also encouraged not to leave the area by a community leader.

Similarly, results suggest that household members were unlikely to encourage the respondent to leave the agricultural sector, as 70.3 percent of the respondents stated (Figure 6.14). Only roughly one in four respondents reported that their household members are likely to encourage them to leave the sector, and 6.6 percent of the cohort were neutral. Importantly for the future of local agriculture and investment in the sector, about 47.4 percent of respondents stated that they were unlikely to encourage their children or younger generations to cease working in the agricultural sector. Rather almost half or about 45.0 percent of the cohort were more likely to encourage younger generations to leave the sector.

Table 6.14:	Respondents'	opinion	about	occupational	mobility	and	migration	in
Damietta (n	=350)							

			Mean		
		Very unlikely and unlikely	Neutral	Likely and very likely	
d of	Household members encouraging you to leave the agricultural sector	70.3	6.6	23.1	2.6
Likelihoo	Encouraging your household members and other community members from younger generations to quit working in the agricultural sector	47.4	7.7	44.9	4.6
		Strongly disagree and disagree	Neutral	Agree and strongly agree	Mean
	If I found a better work opportunity in other sectors, I would definitely leave the agricultural sector	63.1	4.9	32.0	3.5
lent with	Migration of you or any other family member is the best solution to improve your economic situation	58.0	8.3	33.7	3.7
Agreem	If I am financially capable, another family member or I would definitely migrate	75.4	4.9	19.7	2.3
1	I would leave this area if I found a guaranteed and good opportunity to move	68.3	3.4	28.3	3.0

Source: Field Survey 2018

Most respondents (63.1 percent) stated that they were unlikely to leave the agricultural sector even if they found a better work opportunity outside the sector, whereas 32.0 percent of the cohort indicated the likelihood of leaving the sector upon finding a better work opportunity in other non-agricultural sectors. Most respondents (58.0 percent) did not think that migration of one of their family members was the best solution for improving the economic situation of the household. About 33.7 percent of the cohort agreed that migration was the best way to improve the economic conditions of the household. Nearly eight in ten respondents stated that they were unwilling to migrate, even if they were financially capable of migrating. On the contrary, 19.7 percent of respondents thought that they would emigrate if they were financially capable of it. Nearly 68.3 percent of respondents were unlikely to leave their local area even if they found a guaranteed employment opportunity. In comparison, 28.3 percent of the cohort mentioned their willingness to leave the area if they saw an opportunity for migration.

Respondents' scores in each of the previously mentioned statements were compiled to calculate their overall migration culture. Most of them, representing 66.0 percent of households, were reluctant to migrate, while 14.9 percent of the cohort were more willing to migrate and leave the agricultural sector (Table 6.15). A further 19.1 percent of respondents fell in the second category, and had a neutral opinion about migration.

Table 6.15: The distribution of the respondents by their migration culture Damietta(n=350)

	Frequency	Percentage
Very reluctant and reluctant to migrate	231	66.0
Neutral	67	19.1
Willing and very willing to migrate	52	14.9
Total	350	100.0
Courses Field Survey 2018		

Source: Field Survey 2018

Some of the respondent's opinions about migration were reflected in the in-depth interviews. For example, one farmer (Interview 4, Faraskur) mentioned the following, speaking about the migration of farmers:

"The farmer is always in debt, and he has many problems accumulated on his shoulders, today the productivity of a feddan is much less than he spends on. So, where would he go? He just can't leave." While another farmer (Interview 6, Faraskur) illustrating the importance of the farmland as a concept to farmers mentioned:

"The farmer cannot give away his land; he must strive as much as he can. How can I leave farming? It is the source of food for my children and me. I can't do any other thing, and I do have nothing except the land and two cows. If I left it, from where shall I feed myself and my family? I am an old man, and I cannot work on the land with all my full capacity, so all my sons help me, all of them are farmers, they can't leave me or the land".

Another interviewee (interview 7) from Faraskur stated:

"Look, I have worked in many non-agricultural occupations; carpenter, smith, wall painter, etc., but I eventually returned to agriculture. I just take a turn in other occupations and go back to agriculture. At least I will be able to eat from my farmland, even having a cow or a buffalo, I can guarantee to have some milk every day. My per diem pay from other occupations barely gets a box of cigarettes and a bottle of oil, and at the end of the day, I return home without a single pound".

He added:

"I want to leave here, but I can't find any opportunity to migrate. We are a big family, and we only own 0.5 feddans. I wish to move elsewhere and start a new life for myself – a newly reclaimed area like the New Valley, for example, buy a house with land and live there instead of suffering here. I wish to go to any of the government's newly reclaimed lands in the desert, where the government resettle people and help them until they establish their lives".

An interviewee from Faraskur (Interview 11) responded the following when he was asked about any plans of leaving the agricultural sector:

"What shall I work? If drought or flood happened in the area, where shall I go, and what would I work? Even if the agricultural situation gets worse, which is already bad, what shall I do? I have no option other than agriculture ... even if I want to migrate, today travelling requires a lot of money". On the contrary, some other interviewees reported their keenness to migrate; for instance, one farmer (Interview 1, Faraskur) stated the following when he was asked how he would act if the agricultural situation in Egypt worsens:

"I will quit the agriculture sector; this is for sure. Currently, I trade in livestock besides agriculture, and I cultivate my land to feed the livestock. I even wish to leave the area here, not specifically Damietta, I mean Egypt. I want to leave Egypt and go to Kuwait, for example. They say that living there is good and it has nice people".

Similarly, another farmer (Interview 2, Kafr Saad) stated:

"If the situation gets worse and I found a good opportunity to leave, I would certainly leave for any other place".

Results from the Pearson Chi-squared test showed that there were no spatial differences between the two studied local administrations in relation to their overall culture of migration (Pearson Chi-squared p=0.008). However, upon running the Pearson Chi-squared test on each statement, results showed that there was a significant difference between the two local administrations concerning which households in Faraskur would be more willing to migrate if they were financially capable of doing so (Pearson Chi-squared p=0.003) in comparison with those living in Kafr Saad (Table 6.16). It is not clear why there is such a difference, but it could be a result of the particularly poor environmental conditions and water pollution prevailing in the Faraskur area that limits local agricultural activities (see Section 4.4, Chapter 4).

Table 6.16: The distribution of respondents residing in the two local administrations by their agreement with: "If I am financially capable, another family member or I would definitely migrate".

		Strongly disagree and disagree	Neutral	Agree and strongly agree	Total
Faraskur	Count	122	11	47	180
	% within	67.8%	6.1%	26.1%	100.0%
Kafr Saad	Count	142	6	22	170
	% within	83.5%	3.5%	12.9%	100.0%
Total	Count	264	17	69	350
	% within	75.4%	4.9%	19.7%	100.0%

Source: Field Survey 2018

6.4.1 Household culture of migration in relation to their dependence on agriculture

As previously mentioned, the degree of perception of increasing climate variability by households was found to be different based upon their dependency upon agricultural activities for generating their livelihoods. Results also revealed that there were some significant social and economic differences present among the three groups depending on their level of agricultural dependency. Unsurprisingly, results also showed the presence of some significant differences between the three groups of households in their migration culture, which will be addressed in this section. Again the Kruskal–Wallis test was used to analyse any significant differences present between the three groups of households in their migration their views and opinions about migration (Table 6.17).

Group 3, households who are fundamentally dependent upon farming activities for their livelihoods, was found to have the lowest mean rank in their total mobility/migration culture (K–W p=<0.001) in comparison with Groups 1 and 2. More specifically, households in Group 3 have a lower mean rank in the likelihood that their members would encourage the respondent (generally the head of the household) to leave the agricultural sector (K–W p=<0.001) in comparison with the other two groups. These households were also less likely to encourage their household members and other community members from younger generations to cease working in agriculture (K–W p=<0.001) than the other two groups.

On the contrary, households who depend on agricultural activities for less than 50 percent of their total income, namely Group 1, were more likely to choose to migrate if they had the financial capacity to do so (K–W p=0.001), and to leave the agricultural sector upon finding better work opportunities in other sectors (K–W p=<0.001) compared with the other two groups. The mean rank for any of the family members viewing migration as the best solution for improving the economic situation of the household was also higher in Group 1 (K–W p=0.001) than Group 3. Moreover, Group 1 has a higher mean rank relating to the likelihood of leaving the studied area upon finding a guaranteed way or good opportunity for migration (K–W p=0.001) compared with Groups 2 and 3.

Table 6.17: Agricultural dependence groups' mean rankings of household cultures

of migration in the study area

Migration/mobility	< 50%		50-95%		≥95 %		Kruskal–
culture	Group 1 (mean)	Group 1 (mean rank)	Group 2 (mean)	Group 2 (mean rank)	Group 3 (mean)	Group 3 (mean rank)	Wallis (K– W) significance level (p)
Overall migration culture or willingness for mobility	4.96	234.84 ^b	2.94	165.71 ^b	2.65	151.53 ^a	<0.001
The likelihood that any member of the HH encourages you to leave the agricultural sector	4.95	224.91 ^b	2.00	164.10 ^b	1.99	159.50 ^a	<0.001
The likelihood of encouraging HH members and other community members to quit working in the agricultural sector	5.69	205.35 ^b	4.87	183.33 ^b	3.56	147.92 ^a	<0.001
Migration of you or any other family member is the best solution to improve the economic situation	4.97	206.37 ^a	3.72	178.29 ^{a,b}	2.80	153.43 ^b	0.001
If I am financially capable, another family member or I would definitely migrate	3.80	210.18 ^a	1.91	160.25 ^b	1.98	173.09b	0.001
I would leave this area if I found a guaranteed way or good opportunity for migration	4.65	213.06 ^a	2.43	162.23 ^b	2.54	168.95 ^b	0.001
If I found a better work opportunity in other sectors, I would definitely quit working in the agricultural sector.	5.71	224.11 ^a	2.70	157.77 ^b	3.02	167.69 ^b	<0.001

Pairwise statistical difference between groups at 0.05 level is indicated with the use of a,b,c. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

Consequently, households in Group 3 were more likely to have stronger ties to their local place, farmlands and agricultural occupations, whereas households in Group 1 were more willing to migrate and leave the agricultural sector. The educational attainment and the age of the head of the household or the respondent could also be critical factors in shaping their opinions, capacities and opportunities for migration.

6.5 Migration plans and intentions of the studied households in Damietta

Respondents were requested to state whether or not they and their household members had any plans or intentions to migrate during the two years following the survey 2019–2020 (Table 6.18). Most of the respondents, representing 78.9 percent, stated that they

were unlikely to leave the agricultural sector permanently to join another sector in the next two years that followed the survey. About 16.0 percent of the cohort had intentions to leave the sector, and about 4.8 percent were neutral. Also, approximately eight in ten respondents stated that they were unlikely to leave the area either permanently or temporarily in the next two years that followed fieldwork. Only 13.1 percent of respondents mentioned that they were likely to leave the local area permanently, and 15.7 percent of the cohort stated that they would likely leave their residential areas temporarily in 2019–2020. Most of the respondents, or 75.4 percent, were also reluctant to move in the near future, whereas about 21.2 percent of the cohort expressed their willingness to move in the future.

Table 6.18: Household respondents' future migration intentions in Damietta

			Mean		
		Very unlikely and unlikely	Neutral	Likely and very likely	
	Leaving the agricultural sector and joining another sector in the next two years	78.9	4.8	16.3	3.2
hood of	Permanently leaving the area in the next two years	83.1	3.7	13.2	2.8
Likeli	Temporarily leaving the area in the next two years	80.6	3.7	15.7	2.9
		Very reluctant and reluctant to	Neutral	Willing and very willing to	
	Are you and your family members willing to move in the future	75.4	3.4	21.2	3.5

Source: Field Survey 2018

Furthermore, almost all respondents, or 91.4 percent, stated that neither they nor any of their household members had plans to move away from the area in 2019.

6.5.1 Household migration plans in relation to their dependence on agriculture

Although most of the respondents and their household members are less likely to migrate or occupationally move to other non-agricultural sectors in the near future, results from the Kruskal–Wallis test showed that there were some significant differences between household groups in their future intentions. As shown in Table 6.19, households in Group 3 have lower mean rank than Groups 1 and 2 in their willingness to move in the future (K–W p=0.003). Households in Group 3 were also found to be less likely to permanently

leave the area in 2018–2019 (K–W p=0.009) than Group 1, or to leave the area even temporarily (K–W p=0.003) in comparison with both groups.

Migration intentions	< 50%		50-95 %		≥95 %		Kruskal–
	Group 1 (mean)	Group 1 (mean rank)	Group 2 (mean)	Group 2 (mean rank)	Group 3 (mean)	Group 3 (mean rank)	Wallis (K– W) significance level (p)
You and your family members willing to move in the future	3.39	192.77 ^b	2.52	185.59 ^b	1.51	152.77 ^a	0.003
Leaving the agricultural sector permanently and joining another sector in the next two years	3.37	202.52 ^a	1.62	171.05 ^b	1.37	164.58 ^b	0.009
Permanently leaving the area in the next two years	2.39	194.56 ^a	1.63	180.89 ^{a,b}	1.05	157.41 ^b	0.009
Temporarily leaving the area in the next two years	2.47	195.57 ^b	1.92	183.09 ^b	1.10	154.11 ^a	0.003

 Table 6.19: Agricultural dependence groups' mean rankings of household migration intentions in the study area

Pairwise statistical difference between groups at 0.05 significance level is indicated with the use of a,b,c. Different letters indicate a significant difference between groups. Groups with the same letter are not significantly different.

Finally, Group 1 has a higher mean rank for their likelihood of permanently leaving the agricultural sector and joining another sector in the two years following the fieldwork (K–W p=0.009) than Groups 2 and 3. Expectedly, again this result shows that households in Group 3 are particularly less likely to have any potential migration decisions in the near future.

To conclude, results suggest that the household respondents in Damietta were very reluctant to move, and as will be discussed in chapter 7, may increasingly represent a stuck or trapped population as the climate changes (i.e. Black et al. 2011b) because they will be unlikely to exploit positive options to migrate if needed in the near future. This result is of key importance, as Damietta is considered to be one of the most vulnerable areas to future rising sea-levels and other impacts of climate change in Egypt, and their unwillingness to move either geographically or occupationally could result in failure of adaptation efforts within the area.

6.6 Summary

This chapter addressed the adaptation strategies implemented by the studied households to reduce losses in the agricultural sector as a result of increasing climate variability and extremes. Some on-farm adaptation methods have been widely implemented by the respondents, including increasing the use of chemicals, increasing irrigation water, changing planting dates and ceasing to plant those crops that are sensitive to climate variability and extremes.

To cope with sudden shocks of climate variability and extreme weather events, respondents were borrowing money from relatives, selling livestock, and taking advance payment from the traders in the short-term. Migration as a short or long-term adaptation strategy is not being adopted by most households, as became evident when the migration culture of households in the study area was investigated – most were highly reluctant to migrate. Most respondents were unwilling to leave the agricultural sector or to move away from their local areas. There were a few households from whom a household member had migrated and who were receiving remittances by the time the survey was conducted. Most of the money received as remittances was allocated for expenses other than investment in the agricultural sector. This result could also provide information about rural households have chosen migration as a means of adaptation to environmental change, they might be disinclined to invest remittance money in improving their farm business.

Chapter Seven: Livelihood challenges and dimensions of the vulnerability of agrarian communities in Egypt

Multiple challenges face the agricultural sector in Egypt and hinder its growth and development. Rural areas in Egypt have witnessed decades of neglect, and there has been a lack of national drive for real, sustainable forms of rural development (Bush 2000, 2004; El Nour 2015; Bush 2016; de Lellis 2019). This chapter discusses some of the critical stressors, climatic and non-climatic, which small-scale farmers in northern Egypt face. It then deliberates on the factors influencing farmers' perceptions of, and adaptation to, climate change, highlighting some of the potential burdens for effective transformation in Egypt, the Middle East and the other rural communities across the globe with similar circumstances. Finally, the chapter discusses some of the differences found among households within the same agrarian community in their level of vulnerability, risk perceptions and adaptive capacities by integrating the agrarian transition model.

7.1 Livelihood challenges for rural households in Damietta

As explained in Chapters 1 and 2, climate change is a global problem, but its effects are mostly defined at the local level. Hence, it is vital to recognise the current challenges presented to local communities to better understand how climate change generates obstacles and shapes rural communities' vulnerabilities in Egypt, as well as recognising that the extreme situation in Egypt provides many lessons for other developing countries with similar situations. Understanding the local context also allows for a reflection on the factors that can influence the adaptive capacity of rural regions, and different social groups within such regions. This section will look at the critical socio-economic, environmental and agricultural problems already reported by the rural households targeted and discuss how these emerging issues are increasingly linked with a range of pressing challenges found at the national and regional scales.

7.1.1 Socio-economic challenges

Nearly half of the respondents in Damietta, who were mainly the heads of the households, have low educational levels, and one-third of the total group has received no formal education (Table 4.6). The sample findings are similar to broader rural illiteracy rates amongst adults in Egypt, reaching 32.2 percent in 2017 as indicated in the national census

data (CAPMAS 2020b). Such low levels of educational attainment increase the social vulnerability of many small-scale farmers in Egypt (Drzewiecki et al. 2020; Poudel et al. 2020) and could potentially limit their capacity to adapt to climate change both *in situ* and *ex situ* (Ekemhonye et al. 2020). In particular, low educational levels of individuals limit their opportunities for income diversification and social mobility (Barnett & Adger 2007). This argument seems to be supported by the findings of this research, which indicate that a large proportion of respondents work primarily as farmers or waged labourers, of whom nearly half have no secondary occupations and are highly dependent upon agriculture for their livelihoods.

On the other hand, results show that another 40 percent of respondents have completed secondary schooling or higher. This result implies that there is a wide divergence in the educational attainment among respondents in Damietta, which may entail a variation in the vulnerabilities of small-scale farmers to climate change. Similarly, the results presented in Chapter 4 (Section 4.2.2) show higher levels of educational attainment among other adult household members (aged 16 or above), both males and females. This result could suggest that although most household heads have low educational attainment, they tend to encourage younger household members to complete at least secondary education. At the household level, education may be taken as means of improving individuals' employability and income (Barnett & Adger 2007) and thus improve the overall economic conditions of the household. This argument is supported by this research, which indicates that most of the other working household members, excluding the respondents themselves, are primarily employed in non-agricultural occupations (Figure 4.2) and most households have more than one source of income (Figure 4.6). This result also suggests that if younger family members inherit farmland in later years, they would probably be operating agricultural activities as a secondary job or occupation -asituation that could undermine local agricultural production and, over time, threaten the national food security considering the reduced effort invested in it.

It was also noted that younger members of rural households in Damietta, especially males, were becoming more reluctant to work in the agricultural sector or even to assist in farming activities (Section 4.2.2). Similarly, Aboulnaga et al. (2017) state that small-scale family agriculture in Egypt, like many countries in the developing world, is characterised by an increased dependency upon older people, rather than younger individuals who have been shown to be more disinclined to participate in farming activities. The reasons leading

to low levels of youth participation in agricultural activities were not explored in the current study. However, it is likely to be associated with their engagement with education and a growth in alternative opportunities, limiting their free time available to participate in agricultural activities actively (Kabbani 2019). They may also merely be unwilling to participate in the difficult, routine activities of farming which provide relatively small financial reward (Ojebiyi et al. 2015). Earlier youth studies have identified one of the main reasons that turn young people away from agriculture relate to the problems of inheritance and customs in developing countries increasingly limiting the ability of young rural people to access land while still young (White 2012; IFAD 2014; FAO, CTA & IFAD 2014). The growing dependency on ageing farmers in agriculture in many countries around the world is seen as a problem due to the perceived loss of potential in creating profitable, innovative and sustainable farms (FAO, CTA & IFAD 2014; Zagata & Sutherland 2015), which are attributes that will be particularly needed in the context of global climate change.

Although it might appear that households are diversifying their incomes, the degree to which most households depend upon the agricultural sector for a living is still high, particularly amongst the relatively poor. Income from agriculture contributes to 50 percent or more of the total income for the majority of sampled households (Table 5.11). Such a high dependency on an increasingly climate-sensitive activity and a potentially declining primary resource base make most of the studied farmers vulnerable to climate change (Paavola & Adger 2002; Agrawal & Perrin 2009; Betts 2010; Dasgupta et al. 2014). Despite the high livelihood dependency on agriculture, more than half of the households own, share or rent less than 3 feddans of agricultural land. According to Abouelnaga et al. (2017), farms of less than 3 feddans do not allow households to develop livelihoods above the international poverty line in Egypt (USD 1.25 per day per capita). Poor economic conditions increase the vulnerability of many rural households in Egypt and many other developing countries, and hinder their ability to mitigate the risks of climate change (Mertz et al. 2009a; Waly et al. 2020) resulting in trends of increasing poverty (Bohle et al. 1994; Adger et al. 2003; Thomalla et al. 2006; Parry et al. 2007; Mirza 2011; IPCC 2014; Warner, Hoffmaister & Milan 2015; Islam & Winkel 2017; Otto et al. 2017).

Moreover, the social participation of most households in organizing and attending community activities and in discussing public affairs is generally low (Table 4.10). In addition, nearly half of the studied farmers find themselves distanced from government officials, with one-quarter of the total group rated their relationship with government officials as very poor and poor – a result that could have significant implications on the acceptance and implementation of policies regarding adaptation, if any, in the studied areas. Poor relationships with government officials are unlikely to facilitate and promote sustainable adaptation to climate change, as Pelling et al. (2008) and Fatti and Patel (2013) suggested. On the contrary, another half of the group rated their relationship with government cadres as good and very good. Again, this wide variation implies that the degree of vulnerability of rural households is not equal and requires further explanation as presented later in this Chapter (Section 7.4).

Interestingly, the degree of social participation of respondents is associated with their primary occupations (Section 4.2.3.1). Respondents working primarily as farmers or waged labourers and unskilled labourers in non-agricultural occupations have relatively low social participation levels in community events. The majority of respondents even lack the institutional arrangements that could provide them with some influence over the decision-making processes in Egypt. For instance, only a few of the studied respondents are members of one or more formal associations such as syndicates, farmers unions and organisations, government agencies, NGOs, and political parties. Such low social capital could have a huge impact on adaptive capacities, as well as a considerable effect on their vulnerability and the recognition of their social, economic, political values, needs and circumstances by decision-makers in Egypt. In terms of collaboration and social relationships and networks, social capital is a critical factor to boost accessibility to adaptation measures (Adger 2003; Warner, Hoffmaister & Milan 2015).

More than half of the households in Northern Egypt are dissatisfied with the local health services (Table 4.15) and nearly 60 percent of the group rated the refuse removal services in the studied areas as bad (Figure 4.9). Almost half of their houses are not connected to a public or a private sewerage system (Figure 4.8). Similar records were obtained from Damietta's census data which shows that only 55.1 percent of households in the two local administrations are connected to a public sewerage system. The lack of access to basic public facilities is another dimension that contributes to the social vulnerability of rural communities in Egypt to climate change, especially as those poor amenities diminish the quality of the local environment (Waly, Ayad & Saadallah 2020).

7.1.2 Environmental challenges

Nearly half of the studied farmers are not satisfied with water quality in the local areas and rated their satisfaction "6 and below" on a 10-point scale (Figure 4.13). Water pollution is a predominant environmental problem observed in Damietta and it has been a persistent issue since 2003 (EEAA 2003). Deficiency in sewerage systems and poor solid waste management has forced dwellers to get rid of wastes in exposed water canals, and a shortage in the number of water treatment units has contributed to the pollution of drinking, irrigation and drainage water in the area, as it does in Egypt more broadly (EEAA 2003; Aboulnaga 2017).

Salinisation of agricultural lands is another major environmental issue reported by the studied households during the interviews. These findings are consistent with several studies addressing Egypt's ecological problems specifically and the broader MENA region in general (Koohfkan 2001; Lofgren & Richards 2003; GLOWA 2009; Koocheki 2010; Sowers et al. 2011). However, Damietta and the Nile Delta are of specific concern for Egypt due to their location near the Mediterranean Sea, where high soil salinity levels are prevalent (Mohamed 2017). Sea-level rise has the potential to further threaten the northern coast of Egypt through inundation and rising soil salinity of the fertile agricultural areas in the Nile Delta region, including Damietta (Kreimer, Arnold & Carlin 2003; Dasgupta et al. 2009; Werner & Simmons 2009; El-Raey 2010; Hunt & Watkiss 2011; Brecht et al. 2012; EEAA 2016). The persistent water pollution could also be progressively adding to local salinisation, land degradation and deteriorating soil quality (Nunzio 2013; Abdelaal & Thilmany 2019), and more than half of the studied households perceive further future land degradation in their local areas (Table 5.8).

Farmers from Northern Egypt are facing water shortages, particularly at the beginning of the summer season cultivations. Shortages in Nile river water is perceived as an important future risk by roughly two-thirds of the studied households (Table 5.8). Due to a lack of sufficient freshwater for irrigation, Mohamed (2017) argues that farmers are already forced to reuse agricultural waste water. The availability of Nile water is crucial for agricultural production as farmlands in Damietta, as like many parts of Egypt, the region depends solely upon it for irrigation (Information and Decision Support Unit of Damietta 2017). However, several factors are expected to exacerbate water shortages in Egypt further in the coming years. First, population growth is increasing the demand for water

for domestic and industrial use; consequently, water is already less likely be allocated to the agricultural sector and that trend is likely to continue in the future, as has already happened in Yemen, Jordan, and Libya (Sowers et al. 2011). Second, water pollution trends in Egypt in general, as well as the studied area, are expected to continue to reduce the quantity of clean water available for agricultural use (Abdelaal & Thilmany 2019). Third, there are escalating geopolitical challenges over Nile water. Egypt is losing control over Nile water which has been a pillar of security for the country, particularly due to the Ethiopian Grand Renaissance Dam (GDR) development (Nunzio 2013; Power 2014; ECC n.d). Fourth, the poor management of the water sector in Egypt in terms of inefficient irrigation systems, urban expansion mismanagement, and poor water delivery infrastructure are all diminishing the available resources (Soussa 2010; Karajeh et al. 2011; Swain 2011; El Bedawy 2014; Power 2014; Dakkak 2020). Added to all of those issues, climate change is projected to have adverse effects on the volumes of Nile river water flowing into Egypt (Conway 2005; Sowers et al. 2011). Consequently, the agricultural sector could face progressively increasing water shortages and reductions in quality. Such changes would adversely affect agricultural production by reducing the cultivated areas and crop productivity (Abdelaal & Thilmany 2019).

The combination of factors exacerbates risks associated with water availability, a fundamental input for agricultural production, consequently threatening many rural dwellers' livelihoods, particularly of those depending mainly on agricultural activities for living. On the national level, land degradation, water shortages and risks of climate change will undermine goals of achieving food security in Egypt, particularly when associated with rising levels of poverty and malnutrition and the declining levels of food production (Abdelaal & Thilmany 2019).

7.1.3 Agricultural-related challenges

Most respondent farmers stated that they are less than satisfied with the situation of the agricultural sector in general than they were in the past (Figure 4.13). They reported their discontent with the rising costs of agricultural inputs, which have dramatically increased in Egypt in recent years. For example, the price of 50kg of urea (a type of nitrogen fertiliser) increased by 17.5 percent in a year, reaching EGP 285 due to the rise in fuel prices (Al-Fiqi 2018). The majority of households were highly exposed to rising agricultural input costs during 2016 and 2017 (Table 4.25). Similarly, almost two-thirds

of the studied farmers who leased their agricultural lands were exposed to a sudden increase in the land rent prices in 2016-2017 (Table 4.25). The high increase in prices of renting agricultural land has been witnessed in Egypt since the 1990s, with the introduction of structural adjustment policies (Bush 2000; Kassim et al. 2018). However, the costs of leasing agricultural lands have almost doubled in several villages in the Nile Delta region since 2015 (Al-Fiqi 2018). The high leasing prices of agricultural lands might limit the capacity of ambitious, small landholder farmers to exploit a potential adaptation pathway involving accessing extra land to create more economically viable and profitable market-oriented farms, as suggested by Collier & Dercon (2014). The increased costs of agricultural inputs are becoming more difficult to tolerate for many Egyptian farmers, with no profit guaranteed under the context of climate change. This conclusion was further reinforced by the high percentage of households who were exposed to crop and income losses due to climate variability and extreme weather events in 2016-2017 (Table 4.25). The problem is not only restricted to the rising prices of agricultural inputs and resources, but also their availability in the future. For instance, more than half of the studied farmers perceive future threats related to losing their agricultural lands and a decline in their capacity to access resources such as fuel, land and electricity that are crucial for sustaining their farm businesses (Table 5.8).

Egyptian agricultural policies were also reported by farmers as being disappointing (Section 4.3.2). More than three-quarters of them perceive that the current agricultural policies are unlikely to reinforce or improve the agricultural situation in Egypt (Table 5.7). A policy of particular complaint relates to the decreasing area for cultivated rice, which the government imposed in 2018 to rationalise water usage, meaning that Egypt has stopped being the largest rice producer in the Middle East (Nigatu & Motamed 2015) and now searches the international market to import rice (Reuters 2018). As a result of this policy, many farmers in the Nile Delta region were forced to plant crops such as vegetables that required less irrigation but less profit, with non-compliance resulting in fines or, in extreme cases, imprisonment (Kassim et al. 2018).

Although reducing the area used for cultivating crops requiring high water usage is a recommended adaptation measure to reduce the vulnerability of the Egyptian agricultural sector to climate change (McCarl et al. 2015), banning rice cultivation might impose further environmental, cultural and socio-economic pressures on the farming communities. Rice is considered a subsistence crop for rural dwellers, a major cash crop

with high economic returns, in addition to its ability to reduce soil salinity (Tarek 1998). In fact, experts advise that the increased level of soil salinity prevalent in the northern part of the Delta necessitates the cultivation of rice as a reclamation crop to leach salts from the soils, and which is necessary since the absence of Nile flooding due to Aswan High Dam and the Grand Renaissance Dam in Ethiopia (Mohamed 2017; El Nour 2018). Such a recommendation runs counter to government plans to reduce the area under rice in the Nile Delta, thus putting the agricultural land at massive risk of deterioration.

Egypt is already one of the highest global importers of wheat in association with the Camp David Accords of 1978, and its decision to import rice, another strategic crop, could threaten local food security. The high dependency on food imports raises critical questions about global food markets' reliability to supply the country's needs, especially under climate change conditions (Waha et al. 2017; Woertz 2017; OECD-FAO 2018). In addition, the reliance upon food imports is likely to continue and is predicted to increase over time due to population growth (Woertz 2017). Such high reliance on food imports makes the region vulnerable to fluctuations in global markets and food production in other parts of the world (Waha et al. 2017; Woertz 2017; OECD-FAO 2018). Simultaneously, ongoing geopolitical conflicts and any new forms of potential economic or environmental instability generate considerable uncertainties regarding both food demand in, and supply to Egypt and many countries in the region (OECD-FAO 2018).

The majority of farmers choose traders as the main marketing channel for their produce (Table 4.22), a traditional marketing system in rural Egypt linked mainly to small and medium landholdings (Aboulnaga et al. 2017). However, a large proportion of them (43.0 percent) were generally dissatisfied with this marketing arrangement (Table 4.23). The traders are highly influential over product prices, and they deal directly with the farmers, who have weak bargaining powers, at the peak of production and when prices are lowest (Aboulnaga et al 2017). Hence, farmers generally get low prices for their produce, while traders earn high profits. Economic liberalisation in the 1980s resulted in the withdrawal of government marketing of major crops, and many small-scale producers have been unable to compete in the national or international markets (Bush & Martiniello 2017), leaving small-scale farmers under the control of middlemen and traders. Consequently, improving the marketing system in Egypt and increasing farmers' access to better market arrangements is necessary to improve local adaptive capacities and to reduce risks of climate and market-related failures (Ado et al. 2020; Marie et al. 2020).

Although most of the farmers perceive that the future of agriculture is uncertain and are generally dissatisfied with the agricultural situation in Egypt (Section 4.3.2), a quite high percentage (reaching 41.1 percent) of them still do believe that working in the farming sector, until now, has been more stable and secure than other sectors (Table 5.7). Interestingly, the results of this study found that nearly two-thirds of the studied households are not inclined to leave the agricultural sector (Table 6.14), and that more than three-quarters of the group do not have any plans to leave the sector in the near future (Table 6.18). These findings were quite surprising and suggest that those farmers might not have reached the threshold of stress, yet, that would force them to move away from the sector or their local areas (Bardsley & Hugo 2010). The incremental environmental pressures that are expected to grow in the future, however, are likely to impose more burdens on local socio-economic conditions, particularly for small landholders with limited resources, financial capital or *in situ* adaptive capacities (Morton 2007; IFAD 2010; Oppenheimer et al. 2014; Warner, Hoffmaister & Milan 2015).

With a growing population and limited economic diversification in Egypt, national labour markets are already facing challenges of absorbing an expanding urban workforce, intensifying unemployment (Akhtar et al. 2013; Nigatu & Motamed 2015; Bilgili & Marchand 2016). This scenario raises an important question of how small scale farming communities could respond effectively if their local agricultural situation continues to worsen under climate change. With the high rates of urban and rural unemployment in the region, finding alternative non-farming sources of income could become increasingly problematic for many rural households (Dixon, Gulliver & Gibbon 2001). The threat of future unemployment was already evident to more than half of the studied households as presented in Table 5.8.

7.2 Perceptions of climate change risks

Respondent smallholder farmers have been made aware of climate change through their lived personal experiences (Section 5.2). This result is in line with findings from other studies on natural resource users' perceptions of climate change in Egypt, such as those by Adoho and Wodon (2014b), Omar et al. (2015), Froehlich and Al-Saidi (2018), and Hafez (2020). The most frequently observed forms of climate change experienced by the studied households are higher temperatures and more frequent heatwaves (Figure 5.1). These particular perceptions of patterns of variability coincide with the findings of other

studies also looking at perceptions of climate change in Egypt (Adoho & Wodon 2014b; Omar et al. 2015; Froehlich & Al-Saidi 2018; Hafez 2020), and with the actual climate data recorded and published in a number of studies on Egypt and MENA region (Saber 2009; Kuglitsch et al. 2010; Met office et al. 2011; Seyam 2011; Seneviratne et al. 2012; Hereher 2016; Waha et al. 2017; Nashwan et al. 2018). Other patterns of extreme weather events perceived by the studied households are more frequent incidences of frost and more erratic rain (Figure 5.1). Again, these results are consistent with the findings of Nashwan (2016) and Met. Office et al. (2011) who document a trend towards more temperature extremes (heat and cold) and rainfall extremes in Egypt over the last decade. Those climate change trends are expected to continue in the MENA region, and Egypt in particular, over time (Lelieveld et al. 2016; Waha et al. 2017), and more than half of the studied households perceive the future threats of increasing climate variability and extremes as high (Table 5.8).

Egypt has been suffering from the negative effects of climate change on health, agriculture, and livestock (EEAA 2010, 2016), and these were apparent to roughly 70 percent of households studied, who reported that climate change had adversely affected crop productivity, crop quality, livestock productivity, and subsequently their incomes (Figure 5.2). Health conditions were also perceived to be adversely affected by climate change by nearly 60 percent of the studied group as presented in Figure 5.2. These results are in line with other studies in Egypt that found that climate change has resulted in a decline in crop productivity and livestock productivity (Adoho & Wodon 2014b; Omar et al. 2015), income losses (Adoho & Wodon 2014b) and increased health problems or exhaustion (Froehlich & Al-Saidi 2018). The majority of households also perceived the productivity of farmers or labourers as being impacted negatively by climate change, which can be associated with the increased feeling of exhaustion due to climatic extremes. For illustration, Dunne, Stouffer and John (2013) project that heat stress would reduce labour capacity to work and increase risks in many parts of the world where agriculture relies still on manual labour. Additionally, most of the studied households observed that climate change had impacted heavily on the amount of time spent on agricultural operations. Likewise, Ebow et al. (2019) suggest that climate change can particularly affect the amount of effort spent on agricultural activities, given the extra work needed to overcome the impacts of climate change.

Despite the low educational attainment of most respondents, they have a high level of perception of climate change's adverse effects, which contradicts some other findings for Egyptian farmers (Omar et al. 2015), and other farmers among the developing world (Maddison 2007; Gbetibouo 2009; Nnko et al. 2021). While there are several socioeconomic, demographic and biophysical factors that shape individuals' perceptions of climate change, this study found that the degree of dependency on agricultural activities for a living is a significant contributor to increasing perceptions of current and future risks associated with climate change, as explained further in Section 7.4. Agriculture is an activity sensitive to minor variations in climate; consequently, those who depend more on the sector for livelihood are more exposed to climate change risks.

7.3 Adaptation to climate change

Despite the high level of awareness of climate change amongst households, their adaptation levels are low (Figure 6.1) – a result that concurs with the notion that there is no assurance that climate change perceptions result in effective adaptation responses (Weber 2010). Even their attempts to cope with sudden short term climate risks rely on seeking help from other members of the family and community, selling assets and/or taking no action (Figure 6.2)

7.3.1 In situ adaptation methods

Broadly speaking, the results of this study found that increasing irrigation use is the most adopted on-farm adaptation method (Table 6.1). Although this strategy might appear to be a feasible and rational solution to climate change for many farmers, it could exacerbate the Egyptian water crisis. Moreover, the escalating water shortages and hydrologic drought caused by climate change (Hameed et al. 2020) would limit many agrarian communities' future efforts to adapt by freely utilising and increasing irrigation under extreme weather conditions.

Another on-farm adaptation method widely adopted by the studied households is the increased use of pesticides and fertilisers (Table 6.1) – a result that is similar to what Adaho & Wodon (2014c) found in several MENA countries including Egypt. The application of inorganic fertilisers and pesticides serves as a marker of agricultural intensification in the early stages of agrarian transformation (Barrett, Christian & Shiferaw 2017), and has been widely used for several decades to improve product yield

and quality (Delcour, Spanoghe & Uyttendaele 2014). However, the increasing usage of chemicals in agriculture, which is projected to further increase under climate change conditions (Koohfkan 2001; Delcour, Spanoghe & Uyttendaele 2014; Koocheki 2010), could intensify the removal of soil nutrients, especially in fragile soil (Valdivia, Antle & Stoorvogel 2017). As previously highlighted, the northern Nile Delta is already facing deteriorating land quality due to the rising water tables, intrusion of seawater, inadequate drainage systems, high water pollution levels, and shortages of freshwater that are driving farmers to use more saline drainage water on their farms (Mohamed 2017). Consequently, the excessive application of chemicals could further reduce the quality of irrigation water in the studied area, which will ultimately enhance land degradation.

Although increasing the application of fertilisers and pesticides correlates with an increased harvest value, it also increases health costs and time wasted due to farmers being unable to work from sickness (Sheahan, Barrett & Goldvale 2017). One of the challenges already reported by nearly half of the respondents is the likelihood of being poisoned during the application of pesticides (Table 5.5). Many low-income farmers do not know or understand the health and safety procedures that they should follow to protect themselves when applying chemicals, and in being unaware or less informed about the health effects of these chemicals, there are significant trade-offs between food production and farmers' health (Sheahan, Barrett & Goldvale 2017). There is no accurate data about the amounts and types of pesticide used by rural households in Egypt or even the numbers of poisoning or contamination cases (Mansour 2008).

With the expected increased application of pesticides due to climate change and the lack of government regulations governing their use, the health of farmers and even many consumers could be adversely affected. Consequently, regulating the use of pesticides in Egypt is a very important part of building a sustainable agricultural system.

7.3.2 Ex situ adaptation methods

Migration-related adaptation strategies were found to be unpopular with most of the rural households studied (Table 6.2). Likewise, earlier studies addressing environmentally induced migration in Egypt have found that climate change is unlikely to encourage decisions to migrate (Afifi 2011; Adoho & Wodon 2014a). Again, farmers from Northern Egypt might not have reached the threshold of stress that would push them away from their local communities, but there are signs that farming is going to become more difficult,

which raises the question about whether they are more likely to need to move in the future? Afifi (2011) also found that people in Egypt are not willing to leave their homes if they are not facing the immediate impacts of a sudden natural disaster, such as earthquakes or floods. Within the stable rural communities, the incremental effects of climate change are likely to keep undermining many small scale farmers' livelihoods, and especially those highly dependent on the sector for livelihood, which eventually will lead to critical situations with an absence of possible sustainable adaptation pathways.

The state of relative immobility could also be directly related to the households' migration culture shaped by the experiences and history of migration within their households and the broader communities they live in, as presented in Section 6.3. Most households have not experienced migration, and only a few households had a member who was a migrant at the time of the survey (Table 6.9). Similarly, community-level data gathered from indepth interviews also showed limited migration experience within the local community. Even the pattern of mobility found in the studied areas is mostly restricted to movements within Damietta's regional boundaries (Table 4.8). In an ongoing cycle, the lack of experience with migration could have influenced their migration culture (Cohen 2004), which in turn has made household members more reluctant to move away from their communities. For example, most households in Northern Egypt are unlikely to leave their local communities even if they have found a guaranteed and better opportunity elsewhere (Table 6.14). Consequently, *ex situ* adaptation strategies might not be the best possible pathway for effective agricultural transformation in Egypt under the climate change context.

Importantly, the statistics and field observations from this research reveal that small landholder farmers in Egypt are not single blocks of households with a homogenous base and interests. Small-scale farmers in Egypt are considered vulnerable to climate change; however, the degree of such vulnerability varies significantly. The next section will explain the dimensions of vulnerability found among the studied households by framing the study results within the agrarian transition model.

7.4 Climate change vulnerability across the agrarian transition process

The vulnerabilities of different groups of farmers vary across the different phases of agricultural transformation within rural societies. Three groups of households were categorised based upon their livelihood dependency on agricultural activities, and used in
the analysis to typify farming households with a range of different attributes, challenges and opportunities. The three household groups reflect examples of the first three phases of societal agricultural transformation, namely agricultural-based, pre-transition and transition households, and classified according to the contribution of agriculture economic activity. Group 3 includes households who depend almost wholly or wholly on agriculture to generate their livelihoods, with more than 95 percent of their income sourced from agricultural activities. For households in Group 2, income from agriculture contributes 50 to less than 95 percent of their total income. Finally, Group 1 includes households whose incomes mainly depend on non-agricultural activities, where agricultural activities only contribute less than 50 percent of their income.

Small-scale agriculture is among the most vulnerable sectors to the impacts of climate change due to its inherent sensitivity to climate (Parry & Carter 1989; Smit & Skinner 2002). Hence, individuals and households who depend on natural resources to generate their livelihoods are believed to be more vulnerable to climate stressors (Betts 2010; Black et al. 2011b). Consequently, more attention here is given to the category of households in Group 3 and the dimensions of their vulnerability to climate change. The data reveals important attributes of this most vulnerable group of households in relation to climate change risk, which suggests important lessons for adaptation in rural areas, and Egypt more broadly.

7.4.1 Group Three: Agricultural dependence and climate change

The first stage in the agrarian transition is defined as societies that are wholly dependent on agriculture, and such a population are exemplified by households in Group 3, who heavily depend on agricultural activities for their livelihoods. The agricultural sector is not only contributing to more than 95 percent of the total income of households in this Group, but results also reveal that the agricultural sector absorbs the majority of the local labour force (Table 5.18). These are the main two characteristics of agricultural-based societies, as identified by the World Bank (2015). This Group also reflects the Egyptian class structure referred to as "petty commodity producers" (Hopkins 1987), who may even have control over or ownership of a small land area, but have limited access to other means of production. The household heads of the households in Group 3 are the least educated, with an average of 5 years of formal education. The income of households in Group 3 is highly dependent on agricultural activities, yet, most respondents were either landless or held only small areas of farmland, with an average below three *feddans* (equivalent to 1.26 ha) (Table 5.18). As previously mentioned, previous work by Aboulnaga et al. (2017) has already indicated that agricultural dependent households with less than three *feddans* are the most vulnerable category of farmers in Egypt, as their income generally falls below the international poverty line (US\$ 1.25 per day per capita). In Damietta, households in Group 3 were found to have fewer assets (particularly livestock) and live in small houses located away from community services, with inadequate local refuse removal (Table 5.23). They also have low social participation rates, particularly within formal social settings, and they mostly do not hold membership in any farmers' organisations (Section 5.5.4).

The household heads of Group 3 have high levels of awareness of the negative impacts of climate change, despite their low educational attainment (Tables 5.11 and 5.18). The results from northern Egypt suggest that risk perceptions are more linked to risk exposure, which in this case is partly reflected by the individual or household dependency on the climate-sensitive activity of agriculture for a living (Koubi, Stoll & Spilker 2016). In line with Adoho and Wodon (2014b) study, poor or less-wealthy households, prevalent in Group 3, are more likely to suffer from crop and income losses due to climate change (Tables 5.15 and 5.16). Thus, irrespective of their education level, respondents who rely immediately on the weather to generate their livelihoods also appear to have a greater recognition of climate change risks.

Despite their high levels of perception of current and future climate change-related risks, and in contrast to earlier findings (Omar 2015), households in Group 3 are adapting less to climate change than other groups (Table 6.3). Many factors could limit the adaptive capacity of individuals and households, including education (Abid et al. 2019; Ekemhonye et al. 2020), income diversification (Thulstrup 2015; Ekemhonye et al. 2020; Khan et al. 2021), mobility (Adger et al. 2009; Bohensky et al. 2010; Sobczak-Szelc & Fekih 2020; Benveniste, Oppenheimer & Fleurbaey 2020; Maharjan et al. 2020), social networks and capital (Smit & Wandel 2006; Tinch et al. 2015; Abid et al. 2017), assets and wealth (Warner, Hoffmaister and Milan 2015), and land size and tenure (Defiesta & Rapera 2014; Abid et al. 2019; Khan et al. 2021). Consequently, the agriculturally-dependent households in Group 3 appear to have less capacity than other groups to adapt

either *in situ* or *ex situ* to climate change due to their low social capital and assets, limited income diversification opportunities, and low educational attainment of their household heads.

Poor socio-economic factors limit their capacities to adapt and may force them to make unfavourable decisions that could often lead to a situation of maladaptation that does not reduce vulnerability but increase it instead (UNFCCC 2007). This situation was already apparent in the adaptation strategies adopted by most households and specifically those in Group 3 (Section 6.2.1). They have, for example, increased the use of chemicals in the form of pesticides and fertilisers to overcome problems imposed by climate variability and extremes. Excessive use of pesticides was found to increase health costs and time wasted by the farmers due to being ill (Sheahan, Barrett & Goldvale 2017). For illustration, the results of this study indicate that household members in Group 3 are more likely to get poisoned while applying pesticides (Table 5.16), and their expenditure on health is relatively high (Table 5.19).

7.4.1.1 A potentially trapped population? Place attachment and migration culture as barriers to *ex situ* adaptation

The vast majority of households, particularly in Group 3, have strong ties to their communities, farmlands and agricultural occupations. Even though many are suffering from poor working conditions, are dissatisfied with their primary professions, and have high levels of perception of current and future economic and environmental risks, they still state that they are unwilling to leave the sector or migrate (Section 6.5.1). Interestingly, the satisfaction of households in Group 3 with the agricultural situation is still relatively high - possibly because it is all that they know. In other words, people develop knowledge and skills in relation to a particular place, system or occupation. Accordingly, staying in a place, with a system or in an occupation makes economic sense even though the quality of their livelihoods may be declining (Schewel 2019). Such households might have developed a "bounded rationality" and chosen alternatives that are "good enough" in the short-term instead of the best option among all those available over the long-term (Simon 1982). For example, farming households may only know how to farm, so shifting their work to other non-agricultural occupations would be difficult or even unimaginable when they need to compete in a difficult labour market in Egypt. The low educational attainment of household heads in Group 3 and lack of income diversification might have limited their opportunities for previous exposure to other offfarm occupations. So for many Egyptian farmers who have spent their whole lives working in agriculture, they perceive of little alternative to move in the labour market.

Group 3 respondents even appear to wish to discourage their children from working in non-agricultural sectors (Table 6.17), which was also reflected in their high dependency on household members in the different farm operations (Table 5.24). Encouraging younger generations to stay in the sector could result from the need for workers to help on the farm and, hence, act to reduce business costs, especially as they may experience extra work with a declining local environment. This notion is reinforced by previous findings in the literature suggesting that impoverished and marginalised households can neither finance migration of their members, nor spare labour, and consequently have the potential to become trapped in deteriorating livelihood and spatial situations (Standing 1981; Black et al. 2011b; Foresight 2011; Black & Collyer 2014; Nawrotzki & DeWaard 2018). They may also wish to keep family close to look after them in their old age. Culturally, the family in Egypt is considered the most important institution for the care of the elderly, and children are aware of their responsibility towards their elderly parents (Fadel-Girgis 1983). Similarly in many other countries, family play a central role in elder care. For example, in China, there is a strong intergenerational solidarity and family members have the primary responsibility for taking care of their elderly parents by law and constitution (Li & Tracy 1999; Zhang, Gu & Luo 2014).

Migration for households in Group 3 is not a "clear strategy" that they follow despite the absence of sufficient local income diversification alternatives, as also suggested by Stark & Levhari (1982). These findings from Damietta could imply that some agricultural communities in the developing world might choose to be "voluntary immobile" even with the continuing deterioration in their living conditions because of climate change. Similarly, evidence from several parts of the developing world has shown that many people choose to stay, citing place attachment and cultural motivations as reasons for their immobility (Mortreux & Barnett 2009; Adams 2016; Farbotko 2018; Zickgraf 2019; Singh et al. 2020; Farbotko et al. 2020; Nunn et al. 2020).

In contrast, some countries such as Nepal, Haiti and the Philippines have large populations who are traditionally more mobile than others, readily seeking livelihood opportunities in other places either internally or externally in their countries (Quisumbing & McNiven 2005; Kelly 2011; Joseph & Neiburg 2020). These countries are likely to either develop or maintain a strong migration culture as human mobility forms an increasingly important foundation to many aspects of economic, cultural and political power in their rural areas (Kelly 2011). In Damietta, high levels of attachment to place or specific cultural attributes or institutions could make it more likely that many individuals will stay in an increasingly risky location, threatened by increasing climate change impacts, including sea-level rise. These findings may also imply that *ex situ* adaptation strategies might not be a useful pathway for many agrarian communities and may only be exploited once their livelihoods are undermined, or risks become too evident and lives are put at immediate risk (Bardsley & Hugo 2010). In such a context and given their socio-economic and cultural constraints, enhancing the capacities for more impoverished, less educated smallholder farmers to adapt *in situ* becomes even more critical for generating local resilience to climate change impacts.

Spatial immobility magnifies the vulnerability of those individuals who are unable or unwilling to move (Warner, Hoffmaister and Milan 2015). Therefore, even if they need to move under extreme conditions in the future, such vulnerable Group of households will likely be trapped or forced to move in a manner in which they will be distressed and fail to find new livelihood opportunities (Bardsley & Hugo 2010; Black & Collyer 2014). For trapped populations who are unable or unwilling to relocate, repeated climate change stressors can continue to undermine their already fragile economic livelihoods and erode their asset base, making them extremely vulnerable to climate change (Black et al. 2011b; Foresight 2011; Gray & Mueller 2012; Warner & Afifi, 2014; Bronon 2015). It can be concluded that people who are poor and highly dependent on natural resources for a living such as households in Group 3, are more exposed to climate stressors and yet may also be the least able to move effectively to facilitate adaptation through the development of alternative livelihoods in other locations (Betts 2010; Black et al. 2011b).

Another dimension that contributes to the potentially "trapped" rural populations in developing countries is that some farming households might not effectively transition out of agriculture, especially under climate change situations. As mentioned earlier in Chapter 2, the early stages of transition involve agricultural production intensification, increased market-oriented production (Johnston & Mellor 1961; Timmer 1988) and the de-agrarianisation and diversification of livelihoods within rural areas (Kelly 2011;

Castella 2012; Barrett et al. 2017a, 2017b). The availability of off-farm jobs and the capacities to exploit skills and labour to move into new occupations are crucial for any individual to effectively transition out of agriculture (Christiaensen, De Weerdt & Todo 2013; Dorosh & Thurlow 2014; Christiaensen & Kanbur 2016). Therefore, it is both physical and labour-market mobility that will facilitate effective adaptation for many within transitioning societies. In other words, if societies do not have the means to intensify their agricultural production and diversify income sources for their population, they are likely to be limited in their abilities to improve farm and labour productivity and overall living conditions (Johnston & Mellor 1961; Timmer 1988; Barrett, Reardon & Webb 2001). More specifically, the proven incapability of the labour markets in Egypt and other countries in the MENA region to provide sufficient job opportunities in more productive sectors, and thereby absorb a growing population seeking opportunities away from small-scale agriculture, could further undermine the capacity of many agrarian communities to find alternative employment pathways. Hence, feedback loops could develop where a lack of industrialisation results in a stumbling agrarian transition process, limiting further opportunities for development and wealth creation.

7.4.1.2 Structural issues that are intertwined with being "trapped" in place

The implementation of neoliberal policy reforms since the 1980s in Egypt has exacerbated the fragmentation of the labour class by limiting state-led industrial investment, and many poorer, uneducated migrants to towns and cities have relied upon informal, low-productivity activities (Morsy & Levy 2020). The failure to shift labour from agriculture towards high value-added sectors not only disadvantages the more impoverished farmers typified by Group 3, as was anticipated by the agrarian transition framework suggested by Byres (1977), but is disadvantaging the country economically and politically. In other words, while capitalist influences over agriculture have been deepening in Egypt, they have taken a form that does not maximise opportunities for real economic development for the society at large and particularly entrenches disadvantage for a large, poor rural sector - and climate change will add substantially to that disadvantage.

The educational attainment of young members of low-income rural households is substantially dependent on parents' backgrounds and community characteristics (Salehi-Isfahani, Hassine & Assaad 2014; Assaad & Krafft 2020). Given the low educational attainment of household heads and lack of other household resources in Group 3, the younger members might not receive the quality of education to meet the labour market's demand for new skills. Additionally, the lack of social connections deprives young members of low-income families to secure jobs, especially highly skilled occupations (World Bank 2014; Sika 2016, 2017, 2020; Bremer 2018). Again, even if agricultural-based households have the education and skills needed for other productive non-agricultural occupations, the failure of national labour markets to create jobs and relocate labour to other productive sectors would lock millions of rural populations in a "low-value trap" (Morsy & Levy 2020), especially with the growing demographic and environmental challenges.

These struggling "trapped" households could be a potential latent group that could continue to drive radical changes in Egypt if agriculture gets worse in the future under climate change conditions. Climate change has already been seen to be destabilising many parts of the MENA region. Prolonged droughts that took place in 2006–2007 in Syria and accelerated water scarcity in Yemen due to climate change have already been identified as significant drivers of conflict in these countries over the last decade (Gleick 2014; Kelley et al. 2015; Suter 2017). In this case, the severe drought caused agricultural failures that affected millions of farmers and herders, and their communities, leading to a massive increase in the rates of migration to urban areas in search of better income provisions (Gleick 2014; Kelley et al. 2015; Suter 2017). The internal mismanaged displacement and the high unemployment rates in the major cities put more pressure on the socio-economic tensions already existing in the area, and helped to trigger the mass protests of 2011 and subsequent conflict (De Châtel 2014; Gleick 2014; Kelley et al. 2015; Suter 2017; Selby et al. 2017). In fact, climate change was an indirect driver of the Egyptian uprisings in 2011 as a result of drought and heatwaves affecting the world's largest producers of wheat, such as Russia, Ukraine, Australia and China, causing the rapid increases in global food prices (Null & Prebble 2013; Sternberg 2013; Ayeb & Bush 2019). Pressures on natural resources in Egypt are already generating some societal instability. For instance, a number of rural villages in Egypt have already witnessed several protests, both violent and non-violent, driven by frustrated farmers in response to water shortages, pollution and water-intensive land reclamation projects (IRIN 2010; Swain 2011; OOSKNews Correspondent 2012a, 2012b, 2015; Pacific Institute 2020). Consequently, Egypt is not immune from another wave of societal instability driven by environmental change, rising poverty rates, unemployment, growing population and high dependency on food imports.

The urban sector in Egypt has already proven its incapability to absorb and provide livelihoods for farmers who were dispossessed from their lands in the 1990s, leading to successive farmer protests often involving violent uprisings (Bush 2000; El Nour 2015). Future pressures arising from economic, social, and environmental factors could erode millions of rural people's livelihoods. Consequently, a critical question of how the government could manage and support those people throughout this stumbling agrarian transition process is crucial for national societal stability. Governments need to consider the absence of potential pathways or mechanisms to support marginal local communities and households to sustain their livelihoods. Consequently, the state should consider pathways beyond simple rural economic assistance or otherwise, many rural dwellers will become stuck or move in a manner that has the potential to trigger societal instability, similar to what was experienced in Syria. Investing in rural education and reforming the rural micro-economy is crucial to developing marginal communities' living conditions and society at large. Likewise, developing primary manufacturing industries within rural communities could help create jobs and absorb de-skilled farm labour and new entrants (Tregenna 2008; Martins 2019) and reduce poverty (Loayza & Raddatz 2010). This pathway proved successful in some parts of the world, like East Asia and Turkey (Rodrik 2010; Gürbüz 2011; Dabla-Norris et al. 2013; Martins 2019). Improving the rural economy is much needed as climate change impacts would undermine many agrarian societies' living conditions and would act as a multiplier effect to the challenges already evident in every dimension of livelihoods, particularly among poor people in the developing world.

7.4.2 Group Two: Transitioning rural households

Agricultural activities contribute 50 percent to 95 percent of Group 2 household income. Although these households have had the capacity to diversify their income sources, income from agriculture remains significant to their livelihoods. Households in Group 2 have access to relatively large farmlands (Table 5.18). Group 2 has also diversified their farm income through both crop and animal production, with many households possessing relatively large numbers of livestock (Table 5.18). They are also a group of relatively wealthy landowners living in centrally located big houses (Table 5.23). Most seem to manage their agricultural business effectively, and save money to intensify farming or support alterative livelihood activities. This argument seems justified by the relatively low amount of money spent by Group 2 households on the different household expenditure items (Table 5.19). In particular, given their low expenditure on food, their farms appear able to provide them with sufficient food for household consumption (Table 5.19). Their centrally located houses might have also reduced the money spent on commuting to and from schools and reducing their education expenses (Table 5.19).

Households in Group 2 are able to diversify their livelihoods within the rural locales (Kelly 2011; Castella 2012; Barrett et al. 2017a, 2017b) and are forming linkages between farm and non-farm sectors (Johnston & Mullor 1961; Barrett, Reardon & Webb 2001), which are two key components that drive the agrarian transition. An important question is whether the partial transition out of agriculture within the pre-transition society of Egypt is sufficient to reduce poverty, improve their living conditions in the long term, and move them out of agriculture effectively, especially under climate change. The nonagricultural income may be assisting in providing financial capital to manage their agricultural businesses and could be working as a safeguard from any potential losses in agriculture. For those reasons, households in Group 2 have better adaptive capacities to climate change. They have relatively high adaptation levels, despite their lower levels of perception of climate change risks than households in Group 3. Another dimension that contributes to adaptive capacity is the adoption of sustainable and effective adaptation strategies (IPCC 2007). Beside increasing irrigation and chemicals, the adoption of more advanced in situ agricultural adaptation strategies was found to be adopted by households in Group 2. More specifically, of those small number of farmers who planted different crop varieties and used some soil conservation and moisture protection strategies, more than half were from households in Group 2 (Section 6.2.1).

Several factors could be seen to boost the adaptive capacities of households in Group 2. Firstly, livelihood diversification through the farm and non-farm incomes (IFAD 2008; Deressa et al. 2009; Thulstrup 2015; Barrett, Reardon & Webb 2001; Barrett, Bezuneh & Aboud 2001; Ekemhonye et al. 2020). Secondly, the household heads' older age could be another factor that influenced their adaptation choices and reinforced their adaptive capacity. Older farmers have considerable experience in farming and are more likely to understand the necessity of adaptation to their livelihoods and know-how to apply

appropriate adaptive measures (Nhemachena & Hassan 2007; Hassan & Nhemachena 2008; Deressa et al. 2009; Dang et al. 2019). For example, a higher age in household heads increased the probability of adopting measures such as planting trees and irrigation among Ethiopian farmers (Deressa et al. 2009).

Finally, strong social capital is another factor that can boost household adaptive capacity in Group 2, and households in Group 2 have relatively strong social capital. Households in Group 2 have high levels of social participation and involvement in community events (Table 5.20), and some of the household heads are members of formal bodies and organisations (Tables 5.21 and 5.22). Strong social participation could be directly related to possessing large areas of farmland. Individuals who own large agricultural land areas naturally hold or acquire more power, or vice versa, and become more socially and economically dominant. In other words, possessing large land areas could be why those households actively participate in community events and have strong connections with formal organisations as they may represent a more dynamic elite within the rural area. Consequently, they are likely to have a disproportionate influence on local decisionmaking, including such decisions that influence their adaptive capacity. In line with earlier findings, social capital and networks are important factors that reduce vulnerability and enhance adaptive capacity (Smit & Wandel 2006; Tinch et al. 2015; Warner, Hoffmaister & Milan 2015). Strong social networks and frequent exposure to government officials and agricultural organisation personnel could also boost chances for marketing of crop produce (Hassan & Nhemachena 2008; Gbetibouo 2009; Fosu-Mensah, Vlek & MacCarthy 2012; Balew, Agwata & Anyango 2014). This argument was further justified with Group 2's high levels of satisfaction with their produce, rating it to an average of 7 on a 10-point scale (Table 5.23). This result might also imply that households in Group 2 have better accessibility to marketing channels, which has also been proven to improve adaptive capacity (Johnston & Hesseln 2012; Fernández-Giménez et al. 2015; Below, Schmid & Sieber 2015).

Although they note strong participation, their social relationships with other villagers appear weak (Table 5.20). They might have the knowledge needed to adapt to climate variability and to operate farmlands but are perhaps less inclined to share with other farmers. This finding might have important implications for policymakers if they want to involve farmers of Group 2 who are adapting to climate change, through leadership, to

transfer their knowledge and experiences with other smallholder and less wealthy farmers who are at high risk of climate change but less able to adapt.

7.4.3 Group One: Households moving to independence from agriculture

Households' members in Group 1 are more likely to be educated and satisfied with their highly skilled non-agricultural occupations (Tables 5.18 and 5.23). This Group of households typically represents what Springborg (1990) refers to as "agrarian bourgeoise", who have more education qualifications than the traditional capitalist farmers and earn primarily from non-farm activities as a result.

The high educational attainment of respondents within this Group appears to have improved their capabilities to seize other employment alternatives outside the agricultural sector, diversify their income sources, and *vice versa* (Barnett & Adger 2007). Agriculture is a secondary source of income for them, and households seem more reluctant to exert effort or invest money in their agricultural lands. These households have a low level of perceptions of climate change risks (Section 5.5.1). This low level of perception could be a result of their little dependency on agriculture for a living and their high dependence on waged labourers to operate their farms (Table 5.24). Hence, the livelihoods of households in Group 1 are less exposed to the changes in climate patterns and the resulting potential losses (Table 5.14).

This result concurs well with the importance of exposure and socio-economic status as key factors in determining the degree of vulnerability to climate change (IPCC 2014). The results also support previous findings that the higher the household dependency on natural resources for a living, the more exposed they are to climate change risks (Betts 2010; Black, Kniveton & Schmidt 2011). For example, the study of Adoho and Wodon (2014b) in Egypt and other MENA countries found that households whose heads have higher levels of education and better jobs are least likely to suffer from the impacts of climate change in comparison with those household heads who are employed in agriculture. For the households in Group 1, the social and economic risks of climate variability are relatively unimportant. They can voluntarily leave or reduce their investments into the agricultural sector if the agricultural situation in Egypt worsens. Households in Group 1 appear dissatisfied with their product sales and the national agricultural situation (Tables 5.23 and 5.25). That may also be why they are less tied to their agricultural lands and were more willing to migrate whenever they find a better

opportunity elsewhere (Table 6.17) or permanently leave the agricultural sector (Tables 6.17 and 6.19).

Even if it seems that households in Group 1 have successfully transitioned out of the agricultural sector, the overall national progression through an agrarian transition appears to be less than effective - or incomplete. During the transition from a low-income agrarian society to a high-income developed society, the agriculture sector should become relatively small, yet remain productive (Barrett, Christian & Shiferaw 2017). However, Egypt has witnessed a real decline in the agricultural share of both GDP and employment over time (World Bank 2020). Additionally, the level of dissatisfaction with the agricultural situation and the willingness of the households in the transition stage to leave the sector might indicate that the sector is perhaps unproductive and doesn't drive real economic growth for those households and society at large.

7.5 Summary

This chapter discussed the key findings presented in Chapters 4, 5 and 6, with respect to the current and future environmental, institutional, socio-economic and political challenges affecting rural communities in Egypt. Results were also explained within the theoretical frameworks of agrarian transition, adaptation and (im)mobility, emphasising differences in rural households' vulnerabilities and adaptive capacities within the same agrarian system. The next chapter will further build upon these discussions and conclude the study findings in relation to the broader structural challenges prevalent in Egypt and the wider MENA region.

Chapter Eight: Conclusion

8.1 Introduction

This thesis examined the degree to which rural households in Egypt perceive and are adapting to climate change risks, and the factors that shape their perceptions of climate change and adaptive responses. The study classified three groups of households, based upon the contribution of agricultural activities to their overall income – guided by the insights of the agrarian transition phases that countries pass through as they develop. This classification reflected some of the differences in the needs and levels of vulnerability of rural households in each group. The analysis applied in the research is based on primary data collected through a questionnaire survey of 350 households at four villages in Damietta governorate in Egypt and eleven in-depth interviews with local farmers. The study also utilised secondary data such as census and government reports to support the analysis and discussion of results. The study covered a special geographical location, Damietta on the Egyptian Mediterranean coast, which is considered to be one of the most vulnerable areas to a potential rise in sea level, while the area already suffers from high levels of water pollution and soil salinity of agricultural lands.

The Middle East and North Africa region (MENA) has distinctive features that are unlike other parts of the world. The region embraces one of the fastest-growing populations in the world. In four decades, population numbers have quadrupled, rising from about 138 million in 1970 to 456 million in 2019 (World Bank 2019b). About 50 percent of the population is younger than 25 years of age, making the region one of the most youthful parts of the globe (Tür 2018; Forouheshfar, El Mekkaoui & d'Albis 2020).

In addition to population growth, multiple environmental challenges face the region. The scarcity of water resources and the projected decline in precipitation due to climate change have significant consequences for water and food security in the region. The damaged environmental systems, the scarcity of natural resources (land and water), and the ongoing degradation of agricultural lands limit the region's capacity for domestic agricultural production. Subsequent changes in agricultural production due to climate change will have huge consequences on rural livelihoods, poverty, national economies and food security (Breisinger et al. 2012; Waha et al. 2017; Sofuoğlu & Ay 2020) and exemplify a significant challenge for sustainable development in the region, which are

likely to have ongoing and increasingly global consequences (Sowers, Vengosh & Weinthal 2011).

The Egyptian small-scale agricultural sector is already challenged by demographic, environmental, institutional and regional issues. Small landholders and the landless are among the most vulnerable to climate change (Morton 2007; IFAD 2010; EEAA 2010; Oppenheimer et al. 2014; Warner, Hoffmaister & Milan 2015; Aboulnaga et al. 2017; Abid et al. 2019). Alongside the extensive water pollution and scarcity, population growth, land degradation and fragmentation, the threat of the Grand Renaissance dam on Egypt's share of Nile water, climate change is jeopardising the agricultural sector and the livelihoods of many rural dwellers. Many Egyptian agricultural labourers are predicted to lose their jobs in the near future due to climate change (Gouda 2020; Ahmed et al. 2021). Consequently, Egypt could face rising economic inequalities, food insecurity and unemployment, which, in turn, could spark anti-state protests as previously experienced (Climate Diplomacy n.d-b; Selwaness & Roushdy 2019; Forouheshfar, El Mekkaoui & d'Albis 2020), or even lead to political instability in the Nile basin region (Nunzio 2013; Pemunta et al. 2021), sparked by resource-related conflicts (Selby & Hoffmann 2014; Schleussner et al. 2016; Von Uexkull et al. 2016; Detges 2016; Feitelson & Tubi 2017; Ide et al. 2021). Egypt is a key player in the Middle East, and its political and economic stability is critical for ensuring broader regional stability.

Several climate-modelling studies indicate that the region is experiencing significant climatic change compared to other parts of the world. The changes are in terms of increased summer temperatures and heatwaves (Lange 2019; Varela, Rodríguez-Díaz & deCastro 2020; Ozturk, Saygili-Araci & Kurnaz 2021), as well as decreased precipitation accompanied by long drought periods (Ozturk et al. 2018; Lange 2019). Such changes in climatic patterns have significant effects on agriculture and the livelihood of farmers in the MENA region, as they are also having on many other parts of the developing world. The research here produced important findings in relation to the broader structural and environmental changes in Egypt and the MENA region more broadly.

8.2 Major findings

The key findings of the study are presented here as answers to each of the six key research questions outlined in Chapter 1. Chapter 5, the first results chapter from the work in Damietta, presented data and discussion relating to the first three questions:

- Questions 1: What patterns of climate change have been experienced by rural households?
- Question 2: How do rural households perceive climate change risks and other risks associated with working in the agricultural sector? And,
- Question 3: Do rural households perceive additional future risks associated with climate change?

In Chapter 5, a detailed description of rural householders' perceptions of climate change was presented and discussed in relation to the associated current and future risks to the agricultural sector. This study found that 90 percent of rural households have experienced increasing climatic variability and reported various implications of climate change (Section 5.2). Increased frequency of heatwaves, higher temperatures, more cold spells and decreased rainfall are the most important forms of climate change experienced by households (Figure 5.1). Most households perceive of important adverse effects caused by climate change, particularly on crop productivity and quality, livestock productivity, income, health conditions, labourer productivity and time spent on agricultural activities (Figure 5.2).

The findings also indicate that most of the studied households perceive of further risks of climatic variability and extreme weather events, environmental degradation, shortage of resources (mainly land and Nile water), financial and business-related stresses (Table 5.8). The role of the government in supporting local farmers seems to be negligible, and many farmers reveal discontent with government policies and the general situation for agriculture in Egypt (Figure 4.13). They also believe that the future of agriculture in Egypt is uncertain (Table 5.7).

Chapter 6 provided a comprehensive analysis of households *in situ* and *ex situ* adaptation decisions. That work responds directly to the fourth research question posed in the introductory chapter of this thesis:

• Question 4: What kinds of adaptation strategies have been followed by farmers and their households to reduce the risks of impact of climate change?

The results clearly reveal that local farmers and their households are generally not adapting successfully to climate change, although their perception of the adverse impacts of climate change is high. Most of the studied farmers' adaptation strategies were *in situ* incremental adaptations – concurring with the views of Dowd et al. (2014) that the

global agricultural sector mostly implements incremental adaptation rather than the necessary transformational adaptation strategies to generate long-term resilience. This study also suggests that most of the *in situ* strategies being applied in rural Damietta are likely to be ineffective for developing a sustainable agricultural sector in the longer term, given the current environmental risks prevailing in Egypt, and could eventually lead to maladaptation (Barnett & O'Neill 2010 Howden, Crimp & Nelson 2010). In fact, most of the *in situ* adaptive methods followed by the households were more related to increasing the use of agricultural inputs such as water and chemicals to deal with immediate short-term emergent problems.

The *ex situ* adaptation methods of occupational and geographical mobility were found to be adopted by few rural households in Damietta (Table 6.2). In fact, and importantly in relation to further discussion below, farmers and their households appear highly reluctant to move (Section 6.4). Their responses suggest that they are strongly tied to the agricultural sector and their local communities, and they are unwilling to move in the future, in spite of their growing discontent with the agricultural situation and their livelihood conditions.

The revelation of the low level of adaptation among local households in Damietta brings me to the last two questions of the research:

- Question 5: What are the socio-economic, demographic and cultural characteristics that influence the capacity of farmers to adapt to the risks associated with climate change?
- Question 6: Does the degree of household dependency on agriculture for their livelihood influence their level of perception of and adaptation to climate change?

These questions were addressed through a synthesis of the three results chapters. Results presented in Chapter 4 suggested that households depend on more than one income source; however, the contribution of agricultural activities to their total income remains high. Therefore, households within this study were assembled into three groups based upon the contribution of agricultural activities to their overall income and reflecting the different stages of agrarian transition. The agricultural sector contributes 95 percent and more to the incomes of households in Group 3, while Group 2 depends on agricultural activities for a living with percentages ranging from 50 to below 95. The final group of

households (Group 1) main depend on non-agricultural livelihood activities, with income from agriculture contributing to below 50 percent of their total income.

The findings revealed that demographic, economic, social and physical resources accessed by the households mainly determined their vulnerability and adaptive capacity to the prevailing climate change in the study area. The case study highlights some of the socio-economic inequalities found among small-scale farmers within the local community, and mirrors the views of Clay & King (2019) that within any group of small landholder farmers, vulnerabilities and capacities to adapt to climate change are unevenly distributed.

It is vital to understand the importance of the adaptive capacities of these groups. Some households are fundamentally dependent on agriculture (Group 3) and have few alternatives for income diversification, and are more exposed to environmental losses and less capable of adapting to climate change. Besides these elements of vulnerability, the low educational attainment of the heads of Group 3 households and the lack of social capital of the majority of these agricultural dependent households are also contributing to their low adaptive capacity. Together Group 3 could be considered a particularly vulnerable group of poor rural households with limited access to agricultural land and poor social capital and live in remotely located small houses with inadequate public services. The evidence from this study suggests that the prevalent culture of immobility (both occupationally and spatially) was also remarkably strong among these agriculturally-dependent households, suggesting that they are not looking to use *ex situ* opportunities to adapt either.

On the other hand, there are a group of small-scale transitioning farming households (i.e. Group 2) who are revealing some capacities to adapt to climate change (Section 6.2.1), and are less exposed to income losses due to climate change because they are not solely dependent on agriculture (Table 5.16). This group of households were found to have strong social capital and were diversifying their income sources between farm and non-farm income. Finally, the social and economic impacts of climate change appear largely unimportant for households who depend primarily on non-agricultural activities for a living (Group 1). The heads of these increasingly agriculturally-independent households have high educational attainment and predominantly work in highly skilled and semi-skilled non-agricultural occupations (Table 5.12). They are satisfied with their non-

agricultural occupation (Table 5.23) and are more willing to migrate or leave the agricultural sector if they find the opportunity to do so (Table 6.17). Hence, the findings show that agrarian populations in Egypt are not homogenous with respect to vulnerability to climate change or the capacity to adapt to its negative impacts.

8.3 Implications for theory

Much of the focus of migration studies conducted in MENA and Egypt has been about migrants and the role of environmental change in influencing their migration decisions (e.g. Warner et al. 2008; Afifi 2010; Adoho & Wodon 2014a). Perhaps research has not focused sufficiently on the category of non-movers and the causes of their immobility, when in many cases those groups could be seen to be the most vulnerable and are likely to move only when forced and often in a manner that is unplanned and distressed (Lubkemann 2008; Gleick 2014; Kelley et al. 2015; Abel et al. 2019; Nagabhatla et al. 2021). The importance of this research lies in the recognition that some portion of the Egyptian rural populations is largely immobile: a) spatially, either because they are incapable of moving, or simply that they prefer not to move as shown in Chapter 6, and b) economically, because they are not exploiting alternative livelihood opportunities as highlighted in Chapter 5.

This study attempted to explain some of the reasons behind the immobility of people in Damietta communities despite their exposure to climate-related stressors and environmental pressures. Several reasons might explain this issue within the Egyptian context. Firstly, the present stresses from climatic events might have not yet reached the threshold that threatens the basic survival needs of rural people (Bardsley & Hugo 2010). Secondly, Egypt is different from other more mobile poor rural populations from other countries, where migration is an integral part of the culture of many people (Lee 1985; Quisumbing & McNiven 2005; Kelly 2011; Knerr 2017). Migration experiences, history, and continuity generally create a culture of migration of societies (Cohen 2004; Cohen & Sirkici 2011) and over its history, Egypt has revealed comparatively low levels of migration (World Bank 2014; Forouheshfar, El Mekkaoui & d'Albis 2020). Even the rate of internal migration is low, as indicated by the stagnant rate of urbanisation in Egypt since the 1970s associated with limited rural-urban migration flows (World Bank 2014). Thirdly, the limited resources, capital, education and aspirations of poor, small landholders and landless agricultural labourers, restrict their opportunities for mobility

even if they were willing to leave the agricultural sector or their local areas (Ramos 2019). The low educational attainment of many small landholder farmers in Egypt could be limiting their current and future adaptive capacities, both in relation to anticipation of *in situ* adaptation responses, and also in relation to developing prospects for occupational or geographical mobility.

Expanding on the views of Black et al. (2011b), the state of limited mobility increases the probability that the Group 3 households would be potentially "trapped" in less-rewarding occupations and fragile local environments as the climate changes, with few options for local transformations of their agricultural systems or options for moving. In Egypt and many countries of the MENA region, the main reason for the potential trapped populations likely goes beyond the lack of households' assets and willingness to move. However, the incapability, and perhaps the unwillingness, of governments to provide competitive employment and educational pathways to those stuck in their agrarian communities also exacerbates this situation.

The Egyptian labour market has been unable to provide a sufficient number of job opportunities to absorb new entrants and thereby reallocating those leaving the agricultural labour force to other productive sectors (Sika 2016, 2019; Dimova, Elder & Stephan 2016; Morsy & Levy 2020; Dimova & Stephan 2020), which is an essential step for an efficient agrarian transition. Towards the final stages of the agrarian transition, Timmer (1988) argues that the emergent industrial sectors in urban areas across many African countries have not been able to keep pace with the liberation of labour from farming and to sustainably absorb rural migrants, which leads to ongoing rural–urban tensions. The outcome of this societal pressure is a rise in unemployment, forcing agricultural labour to remain in the sector, often in association with rising rural landlessness and poverty (Timmer 1988). The arguments of that seminal work are still playing out across Africa, but are likely to be accentuated by climate change.

With the absence of possible social and economic pathways, rural households will keep suffering from climate change's negative impacts. The current and future threats of climate change are already evident to many farmers in Damietta (Tables 5.5 & 5.8) and some of the quotes are very powerful. One farmer mentioned, "*My losses are severe*. *I am afraid that we would lose more in the upcoming years*". At the same time, farmers in Damietta are aware of the limited livelihood opportunities they have; as one reported, "*If*

drought or flood happened in the area, where shall I go and what would I work? Even if the agricultural situation gets worse, which is already bad, what shall I do? I have no option other than agriculture ... even if I want to migrate, today travelling requires a lot of money". Another farmer mentioned, "I have no occupation other than agriculture. I don't know how to work in any other sector," and another one added, "Where would the farmer go? There are no other secondary occupations available today".

Consequently, rural dwellers might keep holding on for as long as possible even when they fail to sustain sufficient livelihood conditions and become stuck in a deteriorating environment with limited opportunities (Black et al. 2013; Black & Collyer 2014; Adams 2016; Ayeb-Karlsson, Smith & Kniveton 2018; Nawrotzki & DeWaard 2018; Niva, Taka & Varis 2019). The ongoing poverty, the high dependency on small-scale agriculture for many rural dwellers, lack of opportunities for income diversification or remittances from elsewhere will generate problematic situations for Egypt, especially where people move into urban cities only after losing their rural livelihoods. This situation of uncontrolled rural-urban migration is already exacerbating underemployment and unemployment in urban areas such as Cairo (Chaudhuri 2000; Herrmann & Svarin 2009; Zenou 2011; Suter 2017; Selby et al. 2017), and has at times, created groups of angry citizens who are not able to participate fully in societal development, are entrenched in their situations of relative poverty and have the potential to spark political unrest (World Bank 2014; Heyne & Gebel 2016; Al-Shammari & Willoughby 2019; Selwaness & Roushdy 2019; Forouheshfar, El Mekkaoui & d'Albis 2020). Hence, it could be anticipated that unless an effective adaptation policy is implemented, Egypt could reach situations similar to those experienced in Syria, where the escalating impacts of climate change upon agriculture leave no choice for stuck farmers but to migrate to urban areas ineffectively. Already, the uncontrolled movements from rural to urban areas in Syria increased societal pressures - and the implications of the disrupted political security of the country are still playing out (De Châtel 2014; Gleick 2014; Kelley et al. 2015; Suter 2017; Selby et al. 2017).

In fact, social movements in relation to changing environmental conditions have become an increasing characteristic of rural areas across the world. India for example, has lately encountered farmers' movements in response to droughts in semi-arid regions (Chaudhuri 2021). Likewise, drought has sparked violence in Syria (Gleick 2014; Selby et al. 2017), Yemen (Glass 2010; Sowers & Weinthal 2010; Suter 2017), Sudan (Selby & Hoffmann 2014; De Juan 2015) and Kenya (Opiyo et al. 2012; Adano et al. 2012; Scheffran, Ide & Schilling 2014). Riots in Egypt have also been seen to be driven primarily by climate change-induced food crises (Null & Prebble 2013; Sternberg 2013; Maystadt, Trinh Tan & Breisinger 2014; Bush & Martiniello 2017; Ayeb & Bush 2019) and shortages in resources, particularly water (IRIN 2010; Swain 2011; OOSKAnews Corrspeondent 2012a; 2012b; Hamama & Charbel 2015; Local Press Report 2015; Pacific Institute 2020). Those movements are only likely to be empowered by a combination of exposure to climate change and limited adaptation capacities.

Beyond environmental-induced social movements and protests, farmers were at the heart of rural movements that engaged in struggles against liberalisation in Egypt (Bush 2011; El Naggar 2012; Keshk 2012; El Nour 2015; Bush & Martiniello 2017; de Lellis 2019); India (Murari 2015), Burkina Faso (Engles 2021), and Latin America (Vergara-Camus 2014; Tilzey 2020). Over history, small landholder farmers and peasants were often at the centre of political and economic struggles and have already created networks of resistance against the neoliberal processes of "depeasantisation" and "de-agrarianisation" (Bush 2016; Bush & Martiniello 2017). Edelman & Borras (2016, p.3 cited by Engles 2021) believe that the "incompleteness of the transition to capitalism in agriculture" has been driving peasant movements in recent decades. The institutions of neo-liberal governance have failed to represent the needs of an agrarian poor (Engles 2021), and in many cases, added to the economic burdens of small-scale producers who are unable to compete with large scale producers in international markets (Bush & Martiniello 2017). Climate change is also likely to exacerbate these broader political-economic risks.

The message must be heard by decision-makers that unless pathways are developed to provide rural groups with a way forward out of stuck situations, unrest may follow. With the growing water crisis, land depletion and losses in rural livelihoods due to climate change, as well as the ever-increasing unemployment, population growth, rural poverty and marginalisation of rural areas, Egypt could potentially face an agricultural crisis and a growing threat of political instability. Future risk of societal instability should be "anticipated" and incorporated into the present functioning of the current systems as a way of orienting or modulating the influence of the projected future conditions.

8.4 Implications for policy and practice

One of the major contributions of this research is the integration of the agrarian transition framework with climate adaptation theory to present the main findings in a form that acknowledge structural differences in the adaptive capacities of rural households within the single agrarian system. The three groups of households outlined in this study are a good reflection of the range of agrarian transition paths being followed within Egypt, and in fact, in the MENA region or developing countries more broadly. This classification reflects some of the differences in the needs and levels of vulnerabilities within and between groups of rural households, and consequently, could guide further research and also policies that target the situations of rural households, as they experience a particular phase of the transformation in their rural economy. To produce tangible solutions, intervention regarding adaptation should consider the differences found among households in relation to their perceptions of climate change, their intentions to respond to the change; and their capacities to adapt to the change in situ or ex situ. In fact, this study advocated that the choices of policies must consider the biophysical, cultural, economic and social heterogeneity of the underlying agrarian systems when aiming for sustainable adaptation policy.

Assisting trapped marginal communities with different sustainable pathways for adapting *in situ* should be key priority for policy makers to alleviate the adverse effects of climate change on agrarian households and stimulate the sustainable utilisation of resources in the rural locales. MENA's growing population, its dependence on food imports, its limited water resources, its deteriorated soil conditions, and its vulnerability to climate change all reinforce the need for integral agricultural policy that are guaranteed to raise farmers' living conditions and to secure food for its growing population. To achieve sustainable adaptation, a mix of mechanisms that combine policy, technology, education and awareness-raising, and economic or financial pathways is required (Stinger et al. 2020).

As part of any adaptation response, it is vital that the region's governments identify opportunities to reform the labour market to efficiently allocate its growing labour force and provide opportunities to those agricultural labourers who might potentially lose their jobs due to climate change. Investing in reforming the rural micro-economy is crucial to developing marginal communities' living conditions and society at large. That may involve the development of primary-manufacturing industries and attracting both domestic and foreign investment within and across rural communities to create jobs and absorb new entrants from low-skilled farm labour to reduce poverty (Tregenna 2008; Loayza & Raddatz 2010; Martins 2019). This pathway proved successful in supporting positive agrarian transitions in parts of the world like East Asia and Turkey (Rodrik 2010; Gürbüz 2011; Dabla-Norris et al. 2013; Roncolato & Kucera 2014; Martins 2019). In fact, the study of Ingham, Read and Elkomy (2020) found that the impacts of foreign investments were positive on the manufacturing sector in Egypt over the period from 1990 and 2007, which also have beneficial growth spill-overs in other sectors such as agriculture, finance and retail, tourism and construction. The further development of this approach may be needed to reach the large marginalised populations of rural Egypt.

The absence of economic and social opportunities within rural areas raises a critical question of how much the resources of urban centres in Egypt, and many other developing countries, can absorb more migration flows from rural areas in an efficient and sustainable way. For instance, 60 percent of Cairo's urban population are living in informal areas, mostly illegal subdivisions of former agricultural lands (Sims 2012), having poor services and infrastructure (Shehayeb 2009; Abdelhalim & Abou Samra 2010). By understanding the challenges presented to communities such as those in Damietta, policy-makers can anticipate future risks arising from economic, social, and environmental pressures that could potentially erode millions of rural people's livelihoods and lead to unplanned rural-urban migration.

Another important pillar for sustainable inclusive adaptation is that people in rural communities must feel that they are able to voice their concerns and get that represented within a political sphere. It is crucial that farmers work in conjunction with researchers, policy-makers, and government and non-government bodies in order to adapt in a sustainable way and establish resilient agrarian systems (Pretty et al. 2018; Bardsley & Knierim 2020). Policy-makers need to provide economic and social alternatives that can improve the adaptive capacities of agricultural communities against climate change and at the same time suit their socio-economic needs and cultural contexts. Key to that reform would be the evolution of educational systems that run in parallel with the creation of job opportunities that absorb the emerging labour force in Egypt. Investing in education and schools in marginal areas is fundamental so that rural people are provided with the tools and skills that would enable them to diversify their income sources or move away from

rural areas in a sustainable manner. Education is also fundamental to improve the adaptive capacities of agrarian communities to climate change (Abid et al. 2019; Ekemhonye et al. 2020) and allow people to effectively seek employment in emerging industries. In other words, rural adaptation extends well beyond the immediate natural resource and chemical inputs at the farm level – it raises important questions about ongoing structural change in the country.

8.5 Future research

The research findings reflect the critical situation of small landholder farmers in Damietta, one of the Northern governorates of the Egyptian Nile Delta. However, it is important to understand whether these findings reflect a societal challenge specific to Damietta or is it something that is going to be affecting other parts or all of rural Egypt, or other parts of the MENA region, in similar ways. Much of the climate change research has focussed on the impacts of sudden climate extremes (Budhathoki et al. 2020), such as cyclones, floods, droughts etc. However, attention should also be given to those people with limited opportunities who are gradually losing their livelihoods over time, through slow-onset long-term climate events. Based upon the findings of this study, some rural farmers are mitigating the risks of climate change but the majority are not. This situation is going to exacerbate inequalities and may drive more complicated challenges across Egyptian society than if people were just needing to respond to a natural disaster. In fact, the size of societal disruption from the gradual changes in resource availability due to climate change that is already witnessed in many developing countries around the world is starting to challenge the way many systems and societies are operating.

This study concluded that there is a large group of small landholder farmers that are not willing to move although their livelihoods are declining due to climate change. They appear to wish to keep holding on to their current agricultural activities until some point until they might not be able to stay immobile. So it is crucial to understand at what stage will people be desperate to move (Bardsley and Hugo 2010)? To where? And what is next? Modelling migration destinations and flows and the combination of factors that could predict migration decisions is crucial to plan for effective future migration patterns and avoid migration crises similar to what happened in Syria. The question of why and when people protest against governmental institutions also requires further consideration (Ide et al. 2021). Aspects like the broader impacts of neoliberal policies, and the pre-

existing conditions of unemployment, dispossession of natural resources, poverty and socio-economic marginalization are likely to spark environmental-related conflicts and anti-state opposition (Bush & Martiniello 2017; Ide, Fröhlich & Donges 2020; Ide et al. 2021). Consequently, addressing the links between climate change, environmental degradation, resource constraints, inequality, and liberalisation policies is crucial to understand the depth of vulnerability of rural people in Egypt and many developing countries and find solutions to alleviate potential societal instability.

This study found that younger members of rural households are becoming reluctant to participate in the agricultural sector. There is ample room for further research to determine the particular reasons for the unwillingness of younger members to participate or work in agriculture and how this might interact with capacities to provide other economic alternatives for those members. In fact, it might be effective inter-generational mobility that provides the key mechanism for *ex situ* adaptation. At the same time, however, young people's reluctance to invest in agriculture could affect future adaptation efforts, especially with the growing dependency of agricultural systems around the world on old farmers.

8.6 Summary

It is clear that there are several climate and non-climate concerns that are important for decision-making in the agricultural sector. The growing political insecurity, economic inequalities, poverty, unemployment, and conflicts have historically been key features in many countries of the MENA region. Climate change would serve as a "threat multiplier", exacerbating current environmental, economic, political and social challenges. Small-scale marginalized agricultural-based societies will suffer the most from climate change. Effective and inclusive adaptation is crucial to sustain the livelihoods of agriculturally-dependent groups and at the same time avoid potential food crisis and societal instabilities and conflicts. Agrarian communities need to be supported to build resilient and sustainable systems that consider demographic challenges, resources scarcity and opportunities for mobility in relation to future climate change.

Appendix 1: Survey on climate change perceptions and adaptation in Damietta 2018

Questionnaire number

Date	
Local Unit	
Village	
Name and code of the interviewer	

Part 1: Household information, Employment situation and perception of working in agriculture

A1 Household members

Ala	Are you the HH head?	1. Ye	s	2. N	0									
A1b	Other members living in the HH (1 refers to the respondent, from 2 to 13 people are sorted in a descending order from the oldest to	1	2	3	4	5	6	7	8	9	10	11	12	13
	the youngest age).													
	What is the relationship of the other members to the HH													
	head?													
	1. Head of the HH 2. Spouse 3. Son/daughter 4. Step													
	son/daughter 5. Parent 6. Parent in Law /. Son/daughter in													
	Other													
A1c	Gender 1. Male 2. Female													
A1d	Age													
Ale	Marital Status: 1. Married 2. Divorced 3. Widowed 4. Single													
	5. Separated 6. Contracted													
A1f	Years of schooling/education													
Alg	Number of students pursuing education													

A2 Employment situation

A1b	Other members living in the HH (Referring to people same as A1b)	1	2	3	4	5	6	7	8	9	10	11	12	13
A2a	Has any member of the household (aged 16 or older) currently taken up paid work? 1. Yes 2. No													
A2b	In case of No: Why: Unemployed/have no work and looking for work Not wishing to work Housewife Young Child (not yet attending school) Retired person Full time student Have a disability that doesn't allow him/her to work Other (specify) 													
A2c	 Please indicate the primary occupation of your family members (Main occupation that provides the highest wage/income) Farmer Agricultural wage Labourer Craftsman Government employee Technical or Professional Services Non-agricultural Labourer (construction, manufacturing, transportation, and service industries) Small-medium enterprise owner Tradesman Other unskilled work/servant Retired 													
A2d	When did you start this job Year													
A2e	Place of main occupation: 1. In Village 2. Other village within the governorate 3. Town 4. Other governorate													

	5. Overseas									
A2f	a. Does this job require the member to travel									
	b. How often? 1. Daily 2. Several days/ weeks per month 2. Once weekly									
A2g	How many times did you change your main occupation since 2012? (<i>if no change write 0</i>)									
A2h	For family members who took up paid work over the period from 2012-2017: How									
	satisfied or dissatisfied you are with your job?									
	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10(very satisfied)									
A2i	Please indicate the secondary occupation of your family members if any?									
	1. Farmer									
	2. Agricultural waged Labourer									
	3. Craftsman									
	4. Government employee									
	5. Technical or Professional Services									
	6. Non-agricultural Labourer (construction, manufacturing, transportation, and									
	service industries)									
	7. Small-medium enterprise owner									
	8. Tradesman									
	9. Other unskilled work/servant									
	10. None									
A2j	a. Does this job require the member to travel									
	b. How often? 1. Daily 2. Several days/ weeks per month 2. Once weekly									
A2k	Does any member regularly participate in any unpaid agricultural activities?									
	1. Yes 2. No									
A21	a. Do any of the children participate in any child labour? 1. Yes 2. No									
	b. What is the nature of work									
	1. Paid agricultural labour									
	2. Help in farm without cash returns									
	3. Paid labour in other sectors									
A2m	In case of working (primary or secondary occupation) as a paid agricultural labourer:								T	
	a. How many hours per day do you normally work?		1							
1		1	1	1		1	1 1			

1. Regular/Permanent 2. Seasonal/casual/temporary							
c. How do you work? 1 Through a contractor 2 Agreement with the landlord 3 Other specify							

A3 Perception of working in the agricultural sector

A3a	How likely are you exposed to the following as being employed in the agricultural sector?	
	a. Getting poisoned as a result of pesticides application	0(very unlikely) 1 2 3 4 5(neutral) 6 7 8 9 10 (very likely)
	b. Accidents during moving to - from work	0(very unlikely) 1 2 3 4 5(neutral) 6 7 8 9 10 (very likely)
	c. Not getting paid for your work	0(very unlikely) 1 2 3 4 5(neutral) 6 7 8 9 10 (very likely)
	d. Paid less than what was agreed upon	0(very unlikely) 1 2 3 4 5(neutral) 6 7 8 9 10 (very likely)
	e. Not finding work easily	0(very unlikely) 1 2 3 4 5(neutral) 6 7 8 9 10 (very likely)
	f. Income losses as a result of climate variability	0(very unlikely) 1 2 3 4 5(neutral) 6 7 8 9 10 (very likely)
A3b	Do you agree or disagree with the following statements?	
	a. People working in the agriculture sector are well paid in comparison with other sectors	0(strongly disagree) 1 2 3 4 5(neutral) 6 7 8 9 10 (strongly agree)
	b. Working in agriculture is more stable and secured than other sectors	0(strongly disagree) 1 2 3 4 5(neutral) 6 7 8 9 10 (strongly agree)
	c. Current agricultural policies are reinforcing the Egyptian agricultural situation	0(strongly disagree) 1 2 3 4 5(neutral) 6 7 8 9 10 (strongly agree)
	d. The future of agriculture in Egypt is uncertain	0(strongly disagree) 1 2 3 4 5(neutral) 6 7 8 9 10 (strongly agree)

Part 2: Previous movements, migration norms and remittance

B1 Previous movements

A 11	Other members living in the HH (Same order as A1b)	1	2	3	4	5	6	7	8	9	10	11	12	13
Alb														
B1a	Has this person ever lived outside "this area" for any period of six months or													
	longer within the period from 2012-2017?													
	1. Yes 2. No													
B1b	Place of destination													
	Within the governorate: 1. Town 2. Rural village													
	Outside the governorate: 3. Town/city 4. Rural village													
	5 . Abroad/overseas where?													
B1c	When did you move? (the latest movement) (mm/yyyy – mm-yyyy) (if not													
	returned yyyy)													
B1d	Nature of migration:													
	1. New migrant													
	2. Return migrant													
	3. Seasonal migrant													
	4. Temporary migrant													
	5. Permanent migrant													
B1e	Reason for movement (mention the three main reasons)													
	1. Work 2. Marriage/joining family 3. Education 4.													
	Patient/Accompanying patient 5. Poor quality of land or depleted													
1	soils 6. Accumulate savings 7. Political reasons/ violence 8.								1					
	Escape flood/drought 9. Better infrastructure 10. Family													
	problems, divorce, separation 11. Other (specify)													

.....

B1f In case of no migration: What are the reasons of no migration?

B2a	During the past 5 years, have anyone of your h	ousehold/ family member	rs and other pe	eople who ma	y be important to you e	ver encouraged you to move
	away from this area or to stay in this area or ex	erted no influence?				
		No Such Person	Encourag	ed you to	No influence	Encouraged you to move
		N/A	sta	ay		away
	a. Spouse	0	1	l	2	3
	b. Parents	0	1	l	2	3
	c. Siblings	0	1	l	2	3
	d. Children	0	1	l	2	3
	e. Closest friend/neighbour	0	1	l	2	3
	f. Community leader	0	1	l	2	3
	g. Other (Specify)		1	l	2	3
B2b	How likely would your household members en	courage you to leave the	agricultural	0(very unli	kely) 1 2 3 4 5(neutral)	6 7 8 9 10 (very likely)
	sector?					
B2c	How likely would you encourage your HH me	mbers and other commun	ity members	0(very unli	kely) 1 2 3 4 5(neutral)	678910 (very likely)
	from younger generations to quit working in th	e agricultural sector?				
B2d	Migration of you or any other family member	is the best solution to imp	prove your	0(strongly	disagree) 1 2 3 4 5(neutr	ral) 6 7 8 9 10 (strongly agree)
	economic situation					
B2e	If I am financially capable, another family mer	nber or I would definitely	/ migrate	0(strongly	disagree) 1 2 3 4 5(neutr	ral) 6 7 8 9 10 (strongly agree)
DOF	I mould have this area if I found a more thank	and accid any automites to a		0(atrease alay	diag ang a) 1 2 2 4 5 (4 and	(1)
B21	I would leave this area if I found a guaranteed	and good opportunity to i	move	0(strongly	uisagree) 1 2 5 4 5(neut	ral) 6 7 8 9 10 (strongly agree)
B2g	If I found a better work opportunity in other se	ectors. I would definitely	leave the	0(strongly	disagree) 1 2 3 4 5(neut	ral) 678910 (strongly agree)
8	agricultural sector	,		- (~ 6- J	<i>G i i j i j i j i j i j i j i j j j j j j j j j j</i>	······································

B2 Family migration culture/norms

B3 Remittance

B3a	Do the HH receive remittance from the migrant? If		1. Yes 🗖	2. No 🗖				
	no Part 3							
	If yes: How much per year?							
B3b	How often do you receive it	1. M	onthly	2. Bi-monthly	3.	Quarterly 🗖	4. Half	f yearly 🗖
B3c	Which purpose(s) does the HH use the remittances?							
	a. Living costs (food)		d. Education/ trai	ining		g. Buying land/hou	ise	
	b. Savings		e. Health related	expenditure		h. Buying assets		
	c. Agricultural intensification/ improvement		f. Business in nor	n-agricultural activity		i. marriage expendi	iture	

Part 3: Perception and adaptation to climate variability and extremes

C1 Perception of Effects of climate variability and extremes

C1a	Did you or your household experience remarkable climate extreme weather events by 2018?	1	1. Yes □	2. No 🗖						
C1b	How often have you experienced such climate variability at (number of times in a year)	nd extreme ev	ents?							
Clc	Ic Which types of climate variability and extremes have you experienced?									
	a. Hotter temperature		e. Ch	anges in duration of ra	ainy season					
	b. More erratic rain		f. Dec	f. Decrease in annual rainy days						
	c. Changes in timing of rainy season		g. Inc							
	d. More heat waves		h. Bo							
C1d	By 2018, did climate variability and extremes adversely im	pact:								
	a. productivity of crops			0(not at all) 1 2 3 4 5	5(neutral) 6 7 8 9 10 (most seve	ere impacts)				
	b. quality of crop production			0(not at all) 1 2 3 4 5	5(neutral) 6 7 8 9 10 (most seve	ere impacts)				
	c. livestock production			0(not at all) 1 2 3 4 5	5(neutral) 6 7 8 9 10 (most seve	ere impacts)				
	d. water quantity/shortage			0(not at all) 1 2 3 4 5	5(neutral) 6 7 8 9 10 (most seve	ere impacts)				

e. quality of irrigation water	0(not at all) 1 2 3 4 5(neutral) 6 7 8 9 10 (most severe impacts)
f. soil fertility/ land quality	0(not at all) 1 2 3 4 5(neutral) 6 7 8 9 10 (most severe impacts)
g. income	0(not at all) 1 2 3 4 5(neutral) 6 7 8 9 10 (most severe impacts)
h. health conditions	0(not at all) 1 2 3 4 5(neutral) 6 7 8 9 10 (most severe impacts)
i. housing	0(not at all) 1 2 3 4 5(neutral) 6 7 8 9 10 (most severe impacts)
j. transportation and communication	0(not at all) 1 2 3 4 5(neutral) 6 7 8 9 10 (most severe impacts)
k. Productivity of individuals/labourers	0(not at all) 1 2 3 4 5(neutral) 6 7 8 9 10 (most severe impacts)
1. Amount of time spent on agricultural operations	0(not at all) 1 2 3 4 5(neutral) 6 7 8 9 10 (most severe impacts)

C1e Do you have any comments you would like to make about climate variability and extremes?

.....

.....

C2 Adaptation to climate variability and extremes

C2a	Which of the following strategies have you/your HH adopted the	ha fallou	ring means to reduce losses or adapt to climate variability and extremes wi	thin
	the past 5 years?		ing means to reduce loses of adapt to chinate variability and extremes wi	uIIII
	Strategies adopted for continuation of Agro-based liveliho strategies (for those HHs owning/renting farmlands)	ood	Other strategies involved change in livelihood options	
	a. Increasing irrigation		k. Shifted from agriculture to non-agriculture occupation	
	b. Improving irrigation systems		1. Migration of family members to nearby town/market centre for different occupation	
	c. Increased water sources for irrigation		m. Migration of family members to city for different occupation	
	 d. Planting different crop varieties (more resistant to drought, heat/varieties with shorter growing seasons) 		n. Migration of family members to other rural area for agro-based occupation	
	e. Stop planting certain crops		o. Migration of family member to other country	
	f. Increase use of chemicals fertilizers/ insecticides, pesticides or fungicides		p. improving accessibility to weather information	
	g. Changing planting dates			
	h. Formally educated family member joined agriculture			
	i. Change livestock types/ sizes			
	j. Mulching/ soil conservation/ moisture protection			

C2d Do you wish to comment on the previously mentioned adaptation strategies?

Part 4: Livelihood and economic conditions

D1 Social capital

D1a1	Did anyone other than your family members provide you any kind of help in times of difficulties?	1. Yes 🗆	2. No 🗖			
D1a2	How many times did you ask for help in the past year?					
D1a3	Kind of help offered					
	1. Financial help (borrowing money or products/ finding job/ helping in agricultural land)					
	2. Psychological help(emotional support/ giving advice)					
D1b	1bHow do you rate your relationship with 0(very poor) 1 2 3 4 5(neutral) 6 7 8 9 10(very good)					
	a. Family members					
	d. Colleagues					
	e. Government cadres					
D1c	How often do your family participate in the following affairs 0(never) 1 2 3 4 5(neutral) 7 8 9 10(always)					
	a. Discussion of public affairs in your village					
	b. Organizing public activities in your village					
	c. Giving advice to other villagers					
	e. Attendance of local community events					
D1d	D1d Are you a member in any of the following:					
	b. Farmers or labour organizations/unions					
	c. Non-governmental organization					
	d. Governmental agencies					
	e. Political party					
D1e1	What information sources are accessible to you and your family members?					
	a. Internet					
	c. Mobile phones					

	d. Radio broadcast	
	e. Newspapers	
	f. Land-line phone	
D1e2	What is the main source of information for your HH?	

2 Economic conditions

D2a1	HH expenditure (per month	a. Food				
	in Egyptian Pounds)	b. Clothes				
		c. Health				
		d. Education				
		e. House maintenance				
		f. Cigarettes/ mobile recharge				
		g. Electricity				
		h. Gas				
		i. Transportation				
		j. House rent				
		k. Gifts/Donations				
		1. water				
		m. Entertaining activities				
		n. non-food groceries				
		o. other				
D2a2		Total monthly income				
D2a3		Total annual income				
D2b	In percentage, how does each income source contribute to the	Total				
-----	---	-------	------------	------------	---------------	----------------
	total annual income of your family?	100%				
			1. Animal	2.Plant	3. Paid	4. Regular day
			Production	production	seasonal work	to day
				•		labouring
	a. Agricultural production and related activities					
	b. Non-agricultural production and businesses (manufacturing,					
	construction, tourism and servicesetc)					
	c. Social security, subsidy and compensation from government					
	d. Remittances					

D2c Do you or your family members own any of the following? Please indicate the quantity

a. Business Shop		d. motorcycle		g. Chicken/ducks	
b. Land		e. Car		h. Goats/sheep	
c. apartment		f. bicycle		i. Cows/bulls/buffalo	

D3 Health conditions

	Members living in the HH (1 refers to the respondent)	1	2	3	4	5	6	7	8	9	10	11	12	13
D3a	In general, how would you rate the health condition of the family members? 0(very poor) 1 2 3 4 5(neutral) 6 7 8 9 10(very good)													
D3b	Does any member in the household have a chronic disease? 1. Yes 2. No													
D3c	Do you have any health insurance?	1. Ye	s 🗆	2. No										
D3d	How satisfied or dissatisfied are you with the health services provided in your village?	0(ver	y dissa	tisfied) 1 2 3	4 5(ne	utral)	5789	10 (to	tally sa	tisfied)		

D4 Housing and transportation

D4a	Property type: 1. Renting house 2. Owned		
D4b	Number of rooms in the house / How large is the house (m ²)		
D4c	Please tick if the property has the following facilities:	a. tap water	
		b. electricity	
		c. toilet	
		d. separated room for livestock and poultry	
		e. Sanitation	
D4d	Please indicate the average distance in kilometers between your	a. primary school	
	home and the following destinations	b. nearest public transportation	
		c. medical unit	
		d. market	
		e. Main road	
		f. Town	
D4e	How satisfied or dissatisfied are you with your house?	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10	(totally satisfied)
D4f	In your opinion, what is the quality of the following services in the an	rea you live in?	
	a. Water	0(very poor quality) 1 2 3 4 5(neutral) 6 7 8 9 10) (very good quality)
	b. Electricity	0(very poor quality) 1 2 3 4 5(neutral) 6 7 8 9 10) (very good quality)
	c. Sewerage system	0(very poor quality) 1 2 3 4 5(neutral) 6 7 8 9 10) (very good quality)
	d. Refuse removal	0(very poor quality) 1 2 3 4 5(neutral) 6 7 8 9 10) (very good quality)

Part 5: Land use and agricultural situation

E1 Land Use and patterns of production

E1a1	Do you or any member of your HH own or rent any agricultural farmland?	Own		
	If yes: write down the area (in feddans) of the farmland	Rent		
	If No: Skip to E1h1	Shared		
E1a2	How far is the farmland?			
	1. Near 2. Not very far 3. Faraway			
E1a3	The current use of this agricultural land			
	1. crops 2. Livestock 3. Fruits 4. Vegetables 5. Medicinal Plants 6. Fallow			
E1b	Which main marketing channel does your family adopt to sell your products?			
	1. Self-selling 2. Tradesman 3. Collective marketing 4. Government-			
	arranged purchase			
E1c	What kind of training on agriculture have you or your household receive in the			
	past 2 years?			
E1d	What kind of training on off-farm skills have you or your household receive in			
	the past 2 years?			
E1e	Who was responsible for the following agricultural operations in past 2 years?			
	1. HH head 2. Spouse 3. Sons/daughters 4. Relatives			
	a. Land Preparation			
	b. Planting			
	c. Irrigation			
	d. Crop operations (pesticides application, fertilization etc.)			
	e. Harvesting			
E1f	In percentage how much do you use/depend on paid agricultural labour in such	1. Family members	2. Paid Labour	3. Unpaid labour/relatives/
	operations?			friends
	a. Land preparation			
	b. Planting			
	c. Irrigation			
	d. Crop operations (pesticides application, fertilization etc.)			

	e. Harvesting	
E1g	How satisfied or dissatisfies are you with the following	
	a. Availability of agricultural inputs	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10 (totally satisfied)
	b. Prices of the different agricultural inputs	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10 (totally satisfied)
	c. Quality of the inputs	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10 (totally satisfied)
	d. Irrigation water flow and availability	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10 (totally satisfied)
	e. Irrigation water quality	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10 (totally satisfied)
	f. Soil quality	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10 (totally satisfied)
	g. Quality of the agricultural drainage system	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10 (totally satisfied)
	h. Agricultural policies imposed by the government	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10 (totally satisfied)
	i. The agricultural situation in general	0(very dissatisfied) 1 2 3 4 5(neutral) 6 7 8 9 10 (totally satisfied)

E1h1 In the previous two years 2016-2017, have you and your family members been exposed to the following shocks/problems and if yes what did you do to solve such problems?

	E1 h2 Adaptation action (please use "," to separate different choices)
	1. Lending money from 8. Selling livestock
	relatives/friends 9. Selling assets from your house
	2. Taking a loan 10. Agricultural production
	3. Donation from a religious intensification
	institution/people 11. Changed occupation
	4. Advance payment from a 12. Worked in a secondary occupation
	5. Decreasing number of meals 13. Withdrawing Children from
	6. Changing types of food school
	consumed 14. Child/women started working
	7. A family member had to 15. No action
	migrate 16. Other (specify)
a. Poor crop returns due to water shortage	
b. Crop/income losses due to extreme weather events]
c. Reduction in yields as a result of pests and diseases]
d. Reductions in crop yields due to irregularities in]
agricultural operations (irrigation, fertilization, pesticide	
application, etc.,)	
e. Crop prices dropped upon selling your production	1
f. Food shortage in the HH	
g. Labour shortage	
h. Increase in the land rent prices	
i. Any increase in the prices of the agricultural inputs]
j. Lack of cash liquidity when purchasing production	
inputs	

E2 Perception of future risks and related governmental actions

E2a	22a What do you perceive to be the greatest risk to your farm/business/agricultural job in the next five years? Please give a value for each risk					
			Unsure			
	a. Reduction in demand for your produce	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	b. Impacts of large agricultural businesses/corporations	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	c. Environmental degradation/soil erosion/desertification/salinization	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	d. Wage reduction/ lower purchasing power	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	e. Climate variability and extremes	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	f. Urban development pressure	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	g. Lack of government support	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	h. Nile river water shortage	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	i. Decrease of resources (electricity, fuel, land)	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	j. Financial stress and related impacts	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	k. Loss of agricultural land	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				
	1. Unemployment	0(no risk) 1 2 3 4 5(neutral) 6 7 8 9 10(high risk)				

E2b Do you have any comments you would like to make about future risks to agriculture?

E2c Are there any government policies you would like to see implemented to address these risks, for example water shortage, climate variability and extremes?

E2d	What kind of government support have you received in the last 5 years? If received please rate its impact on your business/returns				
	a. Marketing assistance		0(negative effect) 1 2 3 4 5(neutral) 6 7 8 9 10(positive effect)		
	b. Research and development assistance		0(negative effect) 1 2 3 4 5(neutral) 6 7 8 9 10(positive effect)		
	c. Subsidized input prices		0(negative effect) 1 2 3 4 5(neutral) 6 7 8 9 10(positive effect)		
	d. Exceptional circumstances payment		0(negative effect) 1 2 3 4 5(neutral) 6 7 8 9 10(positive effect)		

E2e	What further action would you like the government to be taking to support you as a farmer or hired agricultural labourer? Please rate how					
	important each item is.					
	a. More financial investment in farming cooperative	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				
	b. Reduce costs of inputs	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				
	c. Marketing support	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				
	d. Apply a minimum hourly rate/ increase wages	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				
	e. Provide exceptional circumstances payment in times of crisis	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				
	f. Ensure supply of agricultural inputs	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				
	g. More financial investment in agricultural research and training	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				
	h. Financial investment in climate variability and extremes adaptation programs	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				
	i. Integrating farmers and labourers in DMP and in setting policy	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				
	j. Providing loans with less restrictions (lower interest rate- accepting tenants)	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				
	k. Improving climate information and early warning systems	0(not important) 1 2 3 4 5(neutral) 6 7 8 9 10(very important)				

Part 6 F Movement intentions and plans

f1	How likely are you to leave the agricultural sector permanently and join another sector/occupation in the next 2 years?	0(very unlikely) 1 2 3 4 5(neutral) 6 7 8 9 10 (very likely)		
f2	Are you and family members willing to move in the future?	0(very reluctant to) 1 2 3 4 5(neutral) 6 7 8 9 10(very willing to)		
f3	How likely are you or some of your family members to leave this area permanently in the next 2 years?	0(very unlikely) 1 2 3 4 5(neutral) 6 7 8 9 10 (very likely)		
f4	How likely are you or some of your family members to leave this area temporarily in the next 2 years?	0(very unlikely) 1 2 3 4 5(neutral) 6 7 8 9 10 (very likely)		
f5	Do your family or some family members plan to move away from "this area" in the next 12 months?	1.Yes □ 2. No □		
f6	If yes: to where:			
	Within the governorate: 1. Town 2. Rural village			
	Outside the governorate: 3. Town/city 4. Rural village			
	5. Abroad/overseas			

REFERENCES

Abbott, ML & McKinney, J 2013, *Understanding and Applying Research Design*, John Wiley & Sons, Incorporated, Somerset, United States.

Abdel-Fadil, M 1975, Development, income distribution, and social change in rural Egypt, 1952-1970 : a study in the political economy of agrarian transition, *Cambridge University Press Cambridge*; New York

Abdel-Shafy, HI & Aly, RO 2002, 'Water issue in Egypt: Resources, pollution and protection endeavors', *Central European Journal of Occupational and Environmental Medicine*, vol. 8, no. 1, pp. 3-21.

Abdelaal, HSA & Thilmany, D 2019, 'Grains Production Prospects and Long Run Food Security in Egypt', *Sustainability*, vol. 11, no. 16, p. 4457.

Abdou, AI 2013, 'Agricultural Policies and the Egyptian Farmer, Modifications in Favor of Agricultural Development', *Middle-East Journal of Scientific Research*, vol. 15, no. 6, pp. 763-767.

Abdulrahman, SA 2018, 'Agreements that favour Egypt's rights to Nile waters are an anachronism', *The Conversation*. <u>https://theconversation.com/agreements-that-favour-egypts-rights-to-nile-waters-are-an-anachronism-103353</u>

Abel, GJ, Brottrager, M, Crespo Cuaresma, J & Muttarak, R 2019, 'Climate, conflict and forced migration', *Global Environmental Change*, vol. 54, 2019/01/01/, pp. 239-249.

Abid, M, Ngaruiya, G, Scheffran, J & Zulfiqar, F 2017, 'The Role of Social Networks in Agricultural Adaptation to Climate Change: Implications for Sustainable Agriculture in Pakistan', *Climate*, vol. 5, no. 4.

Abid, M, Scheffran, J, Schneider, UA & Elahi, E 2019, 'Farmer Perceptions of Climate Change, Observed Trends and Adaptation of Agriculture in Pakistan', *Environmental Management*, vol. 63, no. 1, pp. 110-123.

Abid, M, Schilling, J, Scheffran, J & Zulfiqar, F 2016, 'Climate change vulnerability, adaptation and risk perceptions at farm level in Punjab, Pakistan', *Sci Total Environ*, vol. 547, Mar 15, pp. 447-460.

Abou-Hadid, AF 2006, Assessment of impacts, adaptation and vulnerability to climate change in North Africa: food production and water resources. *Final report. Assessments of Impacts and Adaptations to Climate Change (AIACC), Project No. AF 90.*

Abou-Hadid, AF 2008, Climate change and the Egyptian agriculture conundrum, *Technical Report of the Association of International Civil Servants (AFICS-Egypt)*, Egypt.

Aboulnaga, A, Siddik, L, Megahed, W, Salah, E, Ahmed, S, Nageeb, R, Yassin, D & Abdelzaher, M 2017, Study on Small-Scale Family Farming in the Near East and North African Region. Focus Country: Egypt, *FAO*, *CIHEAM-IAMM and CIRAD*, Cairo.

Aboulnaga, Z 2017, 'Water pollution challenge and its impact on egypt's life an overview', *Mans. Univ. J.* <u>http://mena.exceed-swindon.org/wp-content/uploads/2018/01/Abou-Elnaga-Aqaba-Final-2017-01-09.pdf</u>

Abu-Lughod, L 2012, 'Living the "revolution" in an Egyptian village: Moral action in a national space', *American Ethnologist*, vol. 39, no. 1, 2012/02/01, pp. 21-25.

Adams, H 2016, 'Why populations persist: mobility, place attachment and climate change', *Population and Environment*, vol. 37, no. 4, 2016/06/01, pp. 429-448.

Adano, WR, Dietz, T, Witsenburg, K & Zaal, F 2012, 'Climate change, violent conflict and local institutions in Kenya's drylands', *Journal of Peace Research*, vol. 49, no. 1, 2012/01/01, pp. 65-80.

Adger, WN & Adams, H 2013, Migration as an adaptation strategy to environmental change. In: *World Social Science Report 2013. Changing Global Environments*. pp 261-264.

Adger, WN, Agrawala, S, Mirza, MMQ, Conde, C, O'Brien, K, Pulhin, J, Pulwarty, R, Smit, B & Takahashi, K 2007, 'Assessment of adaptation practices, options, constraints and capacity. Climate change 2007: impacts, adaptation and vulnerability'. In: *Contribution of working group II to the fourth assessment report of the intergovernmental panel on climate change*, 01/01, pp. 717-743.

Adger, WN, Huq, S, Brown, K, Conway, D & Hulme, M 2003, 'Adaptation to Climate Change in the Developing World', *Progress in Development Studies*, vol. 3, 07/01, p. 179.

Adger, WN 2003, 'Social Capital, Collective Action, and Adaptation to Climate Change', *Economic Geography*, vol. 79, no. 4, pp. 387-404.

Adger, WN, Arnell, NW, Black, R, Dercon, S, Geddes, A & Thomas, DSG 2015, 'Focus on environmental risks and migration: Causes and consequences', *Environmental Research Letters*, vol. 10 060201, no. 6.

Adger, WN, Arnell, NW & Tompkins, EL 2005, 'Adapting to climate change: perspectives across scales', *Global Environmental Change*, vol. 15, no. 2, pp. 75-76.

Adger, WN, Barnett, J, Brown, K, Marshall, N & O'brien, K 2013, 'Cultural dimensions of climate change impacts and adaptation', *Nature Climate Change*, vol. 3, no. 2, pp. 112-117.

Adger, WN, Dessai, S, Goulden, M, Hulme, M, Lorenzoni, I, Nelson, DR, Naess, LO, Wolf, J & Wreford, A 2009, 'Are there social limits to adaptation to climate change?', *Climatic Change*, vol. 93, no. 3, 2009/04/01, pp. 335-354.

Adger, WN, Pulhin, JM, Barnett, J, Dabelko, GD, Hovelsrud, GK, Levy, M, Spring, ÚO & Vogel, CH 2014, Human security, Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. In: *Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, Intergovernmental Panel on Climate Change (IPCC)*, Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA,.

Ado, AM, Savadogo, P, Pervez, AKMK & Mudimu, GT 2020, 'Farmers' perceptions and adaptation strategies to climate risks and their determinants: insights from a farming community of Aguie district in Niger', *GeoJournal*, vol. 85, no. 4, 2020/08/01, pp. 1075-1095.

Adoho, F & Wodon, Q 2014a, 'Do Changes in Weather Patterns and the Environment Lead to Migration?', in Q Wodon, A Liverani, G Joseph & N Bougnoux (eds). In: *Climate Change and Migration: Evidence from the Middle East and North Africa*, The World Bank, Washington, USA.

Adoho, F & Wodon, Q 2014c, 'How Do Households Cope with and Adapt to Climate Change?', in Q Wodon, A Liverani, G Joseph & N Bougnoux (eds), *Climate Change and Migration: Evidence from the Middle East and North Africa*, World Bank, Washington, DC.

Adoho, F & Wodon, Q 2014b, 'Perceptions of Climate Change, Weather Shocks, and Impacts on Households', in Q Wodon, A Liverani, G Joseph & N Bougnoux. (eds), *Climate Change and Migration: Evidence from the Middle East and North Africa*, World Bank, Washington, DC.

Adriansen, HK 2009, 'Land reclamation in Egypt: A study of life in the new lands', *Geoforum*, vol. 40, no. 4, pp. 664-674.

Afifi, T 2010, 'The Environmental Root Causes Tiggering Economic Migration: The Case of Egypt', in T Afifi & J Jager (eds), *Environment, Forced Migration and Social Vulnerability*, Springer, Berlin Germany, pp. 197-209.

Afifi, T, Milan, A, Etzold, B, Schraven, B, Rademacher-Schulz, C, Sakdapolrak, P, Reif, A, van der Geest, K & Warner, K 2016, 'Human mobility in response to rainfall variability: opportunities for migration as a successful adaptation strategy in eight case studies', *Migration and Development*, vol. 5, no. 2, 2016/05/03, pp. 254-274.

Afifi, T & Warner, K 2008, The impact of environmental degradation on migration flows across countries, *UNU-EHS Working Paper*, Bonn.

Agoubi, B 2021, 'A review: saltwater intrusion in North Africa's coastal areas—current state and future challenges', *Environmental Science and Pollution Research*, pp. 1-15.

Agrawal, A & Perrin, N 2009, 'Climate adaptation, local institutions and rural livelihoods', *Adapting to climate change: thresholds, values, governance*, pp. 350-367.

Ahmed, AU, Bouis, HE, Gutner, T & Löfgren, H 2001, The Egyptian Food Subsidy System: Structure, Performance, and Options for Reform, Research Report 119, International Food Policy Research Institute (IFPRI), Washington, DC:.

Ahmed, Tarek A. (1998): Worth of rice cultivation in the Nile Delta. Loughborough University. Conference contribution. <u>https://hdl.handle.net/2134/29059</u>

Ahmed, YN, Delin, H, Belford, C, Shaker, V & Abdelrahaman, NAM 2021, 'An estimate of the potential economic impacts of climate change on Egypt's agriculture: a multimarket model approach', *Climate and Development*, vol. 13, no. 3, pp. 228-241. Akhtar, SI, Bolle, MJ & Nelson, RM 2013, U.S. Trade and Investment in the Middle East and North Africa: Overview and Issues for Congress, Congressional Research Service, Washington, DC. 46p <u>https://fas.org/sgp/crs/misc/R42153.pdf</u>

Akter, S & Bennett, J 2011, "Household perceptions of climate change and preferences for mitigation action: the case of the Carbon Pollution Reduction Scheme in Australia," *Climatic Change*, Springer. vol. 109, no. 3-4, pp. 417-436.

Al-Fiqi, M 2018, 'Fertiliser price hikes in Egypt', *Ahram online*. <u>https://english.ahram.org.eg/NewsPrint/308889.aspx</u>

Al-Shammari, N & Willoughby, J 2019, 'Determinants of political instability across Arab Spring countries', *Mediterranean Politics*, vol. 24, no. 2, pp. 196-217.

Al Taweel, AM, Ugursal, VI & Boodlal, D 2015, 'Sustainable Management of Climate Change: The Case of the Middle East and North Africa Region', *Renewable Energy and Sustainable Development*, vol. 1, no. 1, pp. 146-159.

Alboghdady, M & El-Hendawy, ES 2016, 'Economic impacts of climate change and variability on agricultural production in the Middle East and North Africa region', *International Journal of Climate Change Strategies and Management*, vol. 8, no. 3, pp. 463-472.

Alderman, H 1986, The Effect of Food Price and Income Changes on the Acquisition of Food by Low- Income Households, International Food Policy Research Institute (IFPRI). Washington, DC. <u>https://www.ifpri.org/publication/effect-food-price-and-income-changes-acquisition-food-low-income-households</u>

Amare, M & Shiferaw, B 2017, 'Nonfarm employment, agricultural intensification, and productivity change: empirical findings from Uganda', *Agricultural Economics*, vol. 48, no. S1, pp. 59-72.

Amer, M 2015, 'Patterns of Labor Market Insertion in Egypt, 1998-2012', *The Egyptian labor market in an era of revolution*, pp. 70-89.

Anderson, K 1983, 'Growth of agricultural protection in East Asia', *Food Policy*, vol. 8, no. 4, pp. 327-336.

Anjum, G & Fraser, A 2021, 'Vulnerabilities associated with slow-onset events (SoEs) of climate change: multi-level analysis in the context of Pakistan', *Current Opinion in Environmental Sustainability*, vol. 50, pp. 54-63.

Antle, JM 2009, Agriculture and the food system: Adaptation to climate change. *RFF Report, Resources for the Future (RFF)*, Washington, DC, p. 28.

Araghi, F 2009, 'Accumulation by displacement: Global enclosures, food crisis, and the ecological contradictions of capitalism', *Review (Fernand Braudel Center)*, pp. 113-146.

Armitage, D 2005, 'Adaptive Capacity and Community-Based Natural Resource Management', *Environmental Management*, vol. 35, no. 6, 2005/06/01, pp. 703-715.

Armstrong, J 2013, 'Positivism and Constructivism', in *Improving International Capacity Development*, Springer, pp. 28-37.

Arnell, NW 1999, 'Climate change and global water resources', *Global Environmental Change*, vol. 9, 1999/10/01/, pp. S31-S49.

Asfaw, A, Simane, B, Bantider, A & Hassen, A 2019, 'Determinants in the adoption of climate change adaptation strategies: evidence from rainfed-dependent smallholder farmers in north-central Ethiopia (Woleka sub-basin)', *Environment, Development and Sustainability*, vol. 21, no. 5, pp. 2535-2565.

Assaad, R & Krafft, C 2014, Youth Transitions in Egypt: School, Work, and Family Formation in an Era of Changing Opportunities. *Doha: Silatech*

Assaad, R & Krafft, C 2015, 'The evolution of labor supply and unemployment in the Egyptian economy: 1988-2012', *The Egyptian labor market in an era of revolution*, pp. 1-26.

Assaad, R & Krafft, C 2020, 'Excluded generation: the growing challenges of labor market insertion for Egyptian youth', *Journal of Youth Studies*, vol. 24, no. 2, pp. 186-212.

Assaad, R, Krafft, C & Yassin, S 2020, 'Job creation or labor absorption? An analysis of private sector job growth in Egypt', *Middle East Development Journal*, vol. 12, no. 2, 2020/07/02, pp. 177-207.

Attaher, S, Medany, M & El-Gindy, A 2010, 'Feasibility of some adaptation measures of on- farm irrigation in Egypt under water scarcity conditions', *Options. Mediterraneennes* vol. A no. 95/307, p.12.

Attaher, SM, Medany, MA & Abou-Hadid, AF 2009, 'Possible adaptation measures of agriculture sector in the Nile Delta to climate change impacts', *Adv. Sci. Res.*, vol. 3, no. 1, pp. 123-126.

Awazi, NP, Tchamba, MN, Temgoua, LF & Avana, M-LT 2020, 'Appraisal of smallholder farmers' vulnerability to climatic variations and changes in the Western Highlands of Cameroon', *Scientific African*, vol. 10, 2020/11/01/, p. e00637.

Ayeb-Karlsson, S, Smith, CD & Kniveton, D 2018, 'A discursive review of the textual use of 'trapped' in environmental migration studies: The conceptual birth and troubled teenage years of trapped populations', *Ambio*, vol. 47, no. 5, pp. 557-573.

Ayeb, H & Bush, R 2019, Food Insecurity and Revolution in the Middle East and North Africa: Agrarian Questions in Egypt and Tunisia, Anthem Press.

Ayers, JM & Huq, S 2009, 'The value of linking mitigation and adaptation: a case study of Bangladesh', *Environmental Management*, vol. 43, no. 5, pp. 753-764.

Babbie, ER 2011, *The basics of social research*, 5th ed. edn, Wadsworth / Cengage Learning, Australia.

Babbie, ER 2013, *Social research counts*, International ed. edn, Wadsworth Cengage Learning, Belmont, Calif.

Babbie, ER 2016, *The practice of social research*, Fourteenth edition. edn, Cengage Learning, Boston, MA.

Bagagnan, A, Ouedraogo, I & Fonta, W 2019, 'Perceived Climate Variability and Farm Level Adaptation in the Central River Region of The Gambia', *Atmosphere*, vol. 10, no. 7.

Bakewell, O 2008, "Keeping them in their place': The ambivalent relationship between development and migration in Africa', *Third World Quarterly*, vol. 29, no. 7, pp. 1341-1358.

Baldwin, A & Gemenne, F 2013, *The paradoxes of climate change and migration*. In: International Social Science Council/United Nations Educational, Scientific and Cultural Organization, World Social Science Report 2013: Changing Global Environments, OECD/UNESCO, Paris. p. 265.

Balew, S, Agwata, J & Anyango, S 2014, 'Determinants of adoption choices of climate change adaptation strategies in crop production by small scale farmers in some regions of Central Ethiopia', *J Nat Sci Res*, vol. 4, 01/01, pp. 78-92.

Bani Salameh, MAR, Bani Salameh, MT & Al-Shra'h, MK 2012, 'The Camp David Accords: Lessons and Facts', *Association of Arab Universities Journal for Arts*, vol. 10, 03/24, pp. 2-39.

Barakat, AO 2004, 'Assessment of persistent toxic substances in the environment of Egypt', *Environment International*, vol. 30, no. 3, 2004/05/01/, pp. 309-322.

Bardsley, D 2015, 'Limits to adaptation or a second modernity? Responses to climate change risk in the context of failing socio-ecosystems', *Environment, Development and Sustainability*, vol. 17, no. 1, pp. 41-55.

Bardsley, DK & Hugo, GJ 2010, 'Migration and climate change: examining thresholds of change to guide effective adaptation decision-making', *Population and Environment*, vol. 32, no. 2, December 01, pp. 238-262.

Bardsley, DK & Knierim, A 2020, 'Hegel, Beck and the reconceptualization of ecological risk: The example of Australian agriculture', *Journal of Rural Studies*, vol.80, pp. 503-512.

Barnes, R 2020, *Grand Renaissance Dam (GERD) project timeline and what you need to know*, Construction Review Online (CR), <u>https://constructionreviewonline.com/project-timelines/grand-renaissance-dam-gerd-project-timeline-and-what-you-need-to-know/</u>

Barnett, J 2001, 'Adapting to climate change in Pacific Island countries: the problem of uncertainty', *World Development*, vol. 29, no. 6, pp. 977-993.

Barnett, J & Adger, WN 2007, 'Climate change, human security and violent conflict', *Political Geography*, vol. 26, no. 6, 2007/08/01/, pp. 639-655.

Barnett, J & O'Neill, S 2010, 'Maladaptation', Pergamon. pp. 211-213.

Barrett, C, Reardon, T & Webb, P 2001, 'Nonfarm income diversification and household livelihood strategies in rural Africa: concepts, dynamics, and policy implications', *Food Policy*, vol. 26, no. 4, pp. 315-331.

Barrett, CB, Bezuneh, M & Aboud, A 2001, 'Income diversification, poverty traps and policy shocks in Côte d'Ivoire and Kenya', *Food Policy*, vol. 26, no. 4, pp. 367-384.

Barrett, CB, Christiaensen, L, Sheahan, M & Shimeles, A 2017, 'On the StructuralTransformation of Rural Africa', Policy Research working paper; no. WPS 7938Washington, D.C.:WorldBankGroup.https://documents1.worldbank.org/curated/en/527271484057904631/pdf/WPS7938.pdf

Barrett, CB, Christian, P & Shiferaw, BA 2017, 'The structural transformation of African agriculture and rural spaces: introduction to a special section', *Agricultural Economics*, vol. 48, pp. 5-10.

Barsoum, G, Ramadan, M & Mostafa, M 2014, *Labour market transitions of young women and men in Egypt*, vol 16, Work4Youth publication series. International Labour Organization, Geneva p. 72. <u>https://www.ilo.org/wcmsp5/groups/public/---dgreports/---dcomm/documents/publication/wcms_247596.pdf</u>

Bartlett, S 2008, 'The Implications of Climate Change for Children in Lower-Income Countries', *Children, Youth and Environments*, vol. 18, no. 1, pp. 71-98.

Beck, U 1989, 'On The Way To The Industrial Risk-Society? Outline Of An Argument', *Thesis Eleven*, vol. 23, no. 1, 1989/05/01, pp. 86-103.

Beck, U 1992, 'From Industrial Society to Risk Society: Questions of Survival, Social Structure, and Ecological Enlightenment', *Theory, Culture & Society*, vol. 9, pp. 97-123.

Beck, U 1995, *Ecological enlightenment : essays on the politics of the risk society*, Humanities Press, Atlantic Highlands, N.J.

Beck, U 2002, 'The terrorist threat: World risk society revisited', *Theory, Culture & Society*, vol. 19, no. 4, pp. 39-55.

Beck, U 2008, 'World at risk: The new task of critical theory', *Development and society*, Vol. 37, No. 1, pp. 1-21

Beck, U 2010, 'Climate for Change, or How to Create a Green Modernity?', *Theory, Culture & Society*, vol. 27, no. 2-3, 2010/03/01, pp. 254-266.

Beck, U, Giddens, A & Lash, S 1994, *Reflexive modernization: Politics, tradition and aesthetics in the modern social order*, Stanford University Press.

Below, T, Artner, A, Siebert, R & Sieber, S 2010, 'Micro-level practices to adapt to climate change for African small-scale farmers', *A review of selected literature*, vol. 953, pp. 1-20.

Below, TB, Schmid, JC & Sieber, S 2015, 'Farmers' knowledge and perception of climatic risks and options for climate change adaptation: a case study from two Tanzanian villages', *Regional environmental change*, vol. 15, no. 7, 2015/10/01, pp. 1169-1180.

Benveniste, H, Oppenheimer, M & Fleurbaey, M 2020, 'Effect of border policy on exposure and vulnerability to climate change', *Proceedings of the National Academy of Sciences of the United States of America*, vol. 117, no. 43, pp. 26692-26702.

Betts, A 2010, 'Survival Migration: A New Protection Framework', *Global Governance*, vol. 16, no. 3, pp. 361-382.

Bezu, S, Barrett, CB & Holden, ST 2012, 'Does the Nonfarm Economy Offer Pathways for Upward Mobility? Evidence from a Panel Data Study in Ethiopia', *World Development*, vol. 40, no. 8, pp. 1634-1646.

Bilgili, O & Marchand, K 2016, *Migration, Development and Climate Change in North Africa*, Thematic input paper prepared for the Thematic Regional Meeting of the Swiss Agency for Development and Cooperation on "Climate Change, Disaster Risks, Migration and Economic Development, Swiss Agency for Development and Cooperation (SDC)/ Maastricht University, Agadir, May 2016. p. 15.

Black, R, Adger, WN, Arnell, NW, Dercon, S, Geddes, A & Thomas, D 2011a, 'The effect of environmental change on human migration', *Global Environmental Change*, vol. 21, no. 1, pp. S3-S11.

Black, R, Arnell, NW, Adger, WN, Thomas, D & Geddes, A 2013, 'Migration, immobility and displacement outcomes following extreme events', *Environmental Science and Policy*, vol. 27, pp. S32-S43.

Black, R, Bennett, SR, Thomas, SM & Beddington, JR 2011b, 'Climate change: Migration as adaptation', *Nature*, vol. 478, no. 7370, Oct 20, pp. 447-449.

Black, R & Collyer, M 2014, "Trapped " populations: Limits on mobility in times of crisis', in SF Martin, S Weerasinghe & A Taylor (eds), *Humantarian Crisis and Migration: Causes, consequences and responses*, Routledge, New York, pp. 287-305.

Black, R, Kniveton, D & Schmidt, K 2011, 'Migration and Climate Change: Toward an Integrated Assessment of Sensitivity', *Environment and Planning A*, vol. 43, 01/01, pp. 431-450.

Bohensky, E, Stone-Jovicich, S, Larson, S & Marshall, N 2010, 'Adaptive Capacity in Theory and Reality: Implications for Governance in the Great Barrier Reef Region', in pp. 23-41.

Bohle, HG, Downing, TE & Watts, MJ 1994, 'Climate change and social vulnerability: toward a sociology and geography of food insecurity', *Global Environmental Change*, vol. 4, no. 1, pp. 37-48.

Bolin, B 2007, 'Race, Class, Ethnicity, and Disaster Vulnerability', in H Rodríguez, EL Quarantelli & RR Dynes (eds), *Handbook of Disaster Research*, Springer New York, New York, NY, pp. 113-129.

Borras Jr, SM, Edelman, M & Kay, C 2009, *Transnational agrarian movements* confronting globalization, John Wiley & Sons.

Bouchaou, L, Michelot, JL, Vengosh, A, Hsissou, Y, Qurtobi, M, Gaye, CB, Bullen, TD & Zuppi, GM 2008, 'Application of multiple isotopic and geochemical tracers for investigation of recharge, salinization, and residence time of water in the Souss–Massa aquifer, southwest of Morocco', *Journal of Hydrology*, vol. 352, no. 3, 2008/05/15/, pp. 267-287.

Brecht, H, Dasgupta, S, Laplante, B, Murray, S & Wheeler, D 2012, 'Sea-Level Rise and Storm Surges: High Stakes for a Small Number of Developing Countries', *The Journal of Environment & Development*, vol. 21, no. 1, 2012/03/01, pp. 120-138.

Breisinger, C, Ecker, O, Perrihan, A-R & Yu, B 2012, *Beyond the Arab awakening: Policies and investments for poverty reduction and food security*, International Food Policy Research Institute (IFPRI), Washington, DC, p.68.

Bremer, J 2018, 'Youth Unemployment and Poverty in Egypt', *Poverty and Public Policy*, vol. 10, no. 3, pp. 295-316.

Bromley, S & Bush, R 1994, 'Adjustment in Egypt? The political economy of reform', *Review of African Political Economy*, vol. 21, no. 60, 1994/06/01, pp. 201-213.

Bronen, R 2015, 'Climate-induced community relocations: using integrated socialecological assessments to foster adaptation and resilience', *Ecology and Society*, vol. 20, no. 3.

Bruton, HJ 1983, 'Egypt's Development in the Seventies', *Economic Development and Cultural Change*, vol. 31, no. 4, pp. 679-704.

Bryan, E, Deressa, TT, Gbetibouo, GA & Ringler, C 2009, 'Adaptation to climate change in Ethiopia and South Africa: options and constraints', *Environmental science & policy*, vol. 12, no. 4, pp. 413-426.

Bryant, CR, Smit, B, Brklacich, M, Johnston, TR, Smithers, J, Chiotti, Q & Singh, B 2000, 'Adaptation in Canadian agriculture to climatic variability and change', in *Societal adaptation to climate variability and change*, Springer, pp. 181-201.

Bryceson, DF 2004, 'Agrarian vista or vortex: African rural livelihood policies', *Review* of African Political Economy, vol. 31, no. 102, pp. 617-629.

Bryman, A 1984, 'The Debate about Quantitative and Qualitative Research: A Question of Method or Epistemology?', *The British journal of sociology*, vol. 35, no. 1, pp. 75-92.

Bryman, A 2006, *Mixed methods*, Sage benchmarks in research methods, Sage, London [et autres].

Bryman, A 2016, *Social research methods*, Fifth edition edn, Oxford University Press, Oxford.

Burton, I, Kates, RW & White, GF 1993, The environment as hazard, Guilford press.

Bush, R 1994, 'Crisis in Egypt: Structural Adjustment, Food Security and the Politics of USAID', *Capital & Class*, vol. 18, no. 2, pp. 15-37.

Bush, R 2000, 'An agricultural strategy without farmers: Egypt's countryside in the New Millennium', *Review of African Political Economy*, vol. 27, no. 84, 2000/06/01, pp. 235-249.

Bush, R 2004, 'Civil society and the Uncivil State': Land Tenure Reform in Egypt and the Crisis of Rural Livelihoods, *Civil Society and Social Movements Programme Paper Number 9*, United Nations Research Institute for Social Development, Geneva, Switzerland, p. 36.

Bush, R 2007, 'Politics, power and poverty: twenty years of agricultural reform and market liberalisation in Egypt', *Third World Quarterly*, vol. 28, no. 8, 2007/12/01, pp. 1599-1615.

Bush, R 2011, 'Coalitions for Dispossession and Networks of Resistance? Land, Politics and Agrarian Reform in Egypt', *British Journal of Middle Eastern Studies: The Dynamics of Opposition Cooperation in the Arab World*, vol. 38, no. 3, pp. 391-405.

Bush, R 2016, 'Uprisings without Agrarian Questions', in A Kadri (ed.), *Development Challenges and Solutions after the Arab Spring*, Palgrave Macmillan UK, London, pp. 153-172.

Bush, R & Martiniello, G 2017, 'Food Riots and Protest: Agrarian Modernizations and Structural Crises', *World Development*, vol. 91, 2017/03/01/, pp. 193-207.

Byres, TJ 1977, 'Agrarian transition and the Agrarian question', *The Journal of Peasant Studies*, vol. 4, no. 3, 1977/04/01, pp. 258-274.

Byres, TJ 1986, 'The Agrarian Question, Forms of Capitalist Agrarian Transition and the State: An Essay with Reference to Asia', *Social Scientist*, vol. 14, no. 11/12, pp. 3-67.

Byres, TJ 2016, 'In Pursuit of Capitalist Agrarian Transition', *Journal of Agrarian Change*, vol. 16, no. 3, pp. 432-451.

CAPMAS 2018, *Egypt in Figures 2018*, Central Agency for Public Mobilization and Statistics (CAMPAS), Cairo, Egypt, March Issue, ref. no. 71-01112-2018.

CAPMAS 2019a, Important Indicators: Research on Income, Expenditure and Consumption 2017/2018, Central Agency for Public Mobilization and Statistics (CAMPAS), Cairo, Egypt.

CAPMAS 2019b, *Statistical Yearbook 2019*, Central Agency for Public Mobilization and Statistics (CAMPAS), Cairo, Egypt, issue no. 110, ref. no. 71-01111-2019.

CAPMAS 2020a, *Egypt in Figures*, Central Agency for Public Mobilization and Statistics (CAMPAS), Cairo, Egypt.

CAPMAS 2020b, *Egypt Statistical Abstract*, Central Agency for Public Mobilization and Statistics (CAMPAS), Cairo, Egypt, ref. no. 71-01113-2020.

Carling, Jr 2002, 'Migration in the age of involuntary immobility: Theoretical reflections and Cape Verdean experiences', *Journal of Ethnic and Migration Studies*, vol. 28, no. 1, 2002/01/01, pp. 5-42.

Cassing, J, Nassar, S, Siam, G & Moussa, H 2009, *Arab Rebublic of Egypt*, Distortions to agricultural incentives in Africa (English), World Bank Group. Washington, D.C. p. 71.

Castella, J-C 2012, 'Keynote 1: Agrarian transition and farming system dynamics in the uplands of South-East Asia', in *The 3rd International Conference*, p. 4.

Castells-Quintana, D, Lopez-Uribe, MdP & McDermott, TKJ 2018, 'Adaptation to climate change: A review through a development economics lens', *World Development*, vol. 104, 2018/04/01/, pp. 183-196.

Chaudhary, P & Bawa, KS 2011, 'Local perceptions of climate change validated by scientific evidence in the Himalayas', *Biology Letters*, vol. 7, no. 5, 2011/10/23, pp. 767-770.

Chaudhuri, NR 2021, 'Social movements and grassroots discourse of climate justice in the context of droughts in semi-arid regions: A case study in India', *Onati Socio-Legal Series*, vol. 11, no. 1, pp. 69-107.

Chaudhuri, S 2000, 'Rural–Urban Migration, the Informal Sector, Urban Unemployment, and Development Policies: A Theoretical Analysis', *Review of Development Economics*, vol. 4, no. 3, pp. 353-364.

Chenery, HB & Syrquin, M 1975, *Patterns of development, 1950-1970*, Oxford University Press for the World Bank, London.

Chenery, HB & Taylor, L 1968, 'Development Patterns: Among Countries and Over Time', *The Review of Economics and Statistics*, vol. 50, no. 4, pp. 391-416.

Christiaensen, L, De Weerdt, J & Todo, Y 2013, 'Urbanization and Poverty Reduction -The Role of Rural Diversification and Secondary Towns', *Agricultural Economics*, vol. 44, pp. 435-447.

Christiaensen, L & Kanbur, R 2016, 'Secondary towns and poverty reduction : refocusing the urbanization agenda', IZA Discussion Papers 10637, Institute of Labor Economics (IZA), p. 26.

Clay, N & King, B 2019, 'Smallholders' uneven capacities to adapt to climate change amid Africa's 'green revolution': Case study of Rwanda's crop intensification program', *World Development*, vol. 116, pp. 1-14.

Cliff, T 1946, '*The Agrarian Question in Egypt*', In: The Problem of the Middle East (Egypt, Palestine, Syria, Lebanon and Iraq), pp 75-85. <u>https://www.marxists.org/archive/cliff/works/1946/probme/chap09.html</u>

Cline, W 2007, *Global Warming and Agriculture: Impact Estimates by Country*, Centre for Global Development; Peterson Institute for International Economics.

Cochrane, L 2016, 'Land Grabbing', In: P.B. Thompson, D.M. Kaplan (eds.), *Encyclopedia of Food and Agricultural Ethics*, Springer Science+Business Media Dordrecht.

Cohen, JH 2004, The culture of migration in southern Mexico, University of Texas Press.

Cohen, JH & Sirkeci, I 2011, *Cultures of Migration: The Global Nature of Contemporary Mobility*, University of Texas Press, USA.

Collier, P & Dercon, S 2014, 'African agriculture in 50 years: smallholders in a rapidly changing world?', *World Development*, vol. 63, pp. 92-101.

Commander, S & Hadhoud, AA 1986, 'From Labour Surplus to Labour Scarcity? The Agricultural Labour Market in Egypt', *Development Policy Review*, vol. 4, no. 2, pp. 161-180.

Conisbee, M & Simms, A 2003, *Environmental refugees: The case for recognition*, New Economics Foundation London, pp. 17-18.

Conway, D 2005, 'From Headwater Tributaries to International River: Observing and Adapting to Climate Variability and Change in the Nile basin', *Global Environmental Change*, vol. 15, no. 2, 2005/07/01/, pp. 99-114.

Conway, D & Hulme, M 1996, 'The Impacts of Climate Variability and Future Climate Change in the Nile Basin on Water Resources in Egypt', *International Journal of Water Resources Development*, vol. 12, no. 3, 1996/09/01, pp. 277-296.

Creswell, JW 2003, *Research design : qualitative, quantitative, and mixed method approaches*, 2nd ed. edn, Sage Publications, Thousand Oaks, Calif.

Creswell, JW 2015, A Concise Introduction to Mixed Methods Research, SAGE, Los Angeles.

Creswell, JW & Clar, VLP 2017, *Designing and Conducting Mixed Methods Research*, SAGE Publications, Thousand Oaks.

Cunningham, E 2012, 'Could Egypt run out of water by 2025?', *Global Post*. <u>https://www.pri.org/stories/2012-04-06/could-egypt-run-out-water-2025</u>

Cutter, SL & Finch, C 2008, 'Temporal and spatial changes in social vulnerability to natural hazards', *Proceedings of the National Academy of Sciences*, vol. 105, no. 7, p. 2301-2306.

Czaika, M & Vothknecht, M 2014, 'Migration and aspirations-are migrants trapped on a hedonic treadmill?', *IZA Journal of migration*, vol. 3, no. 1, p. 1.

Dabla-Norris, ME, Thomas, MAH, Garcia-Verdu, MR & Chen, MY 2013, *Benchmarking structural transformation across the world*, International Monetary Fund.

Dakkak, A 2020, *Egypt's Water Crisis – Recipe for Disaster*, Echoing Sustainability in MENA (EcoMENA), <u>https://www.ecomena.org/egypt-water/</u>

Dane, FC 1990, Research methods, Thomson Brooks/Cole Publishing Co.

Dang, HL, Li, E, Nuberg, I & Bruwer, J 2019, 'Factors influencing the adaptation of farmers in response to climate change: a review', *Climate and Development*, vol. 11, no. 9, 2019/10/21, pp. 765-774.

Dasgupta, P, Morton, JF, Dodman, D, Karapinar, B, Meza, F, Rivera-Ferre, MG, Sarr, AT & Vincent, KE 2014, *Rural areas.*, In: Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, Intergovernmental Panel on Climate Change (IPCC), Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

Dasgupta, S, Laplante, B, Meisner, C, Wheeler, D & Yan, J 2007, *The Impact of Sea Level Rise on Developing Countries: A Comparative Analysis*, World Bank Policy Research Working Paper 4136, The World Bank, Washington, DC.

Dasgupta, S, Laplante, B, Meisner, C, Wheeler, D & Yan, J 2009, 'The impact of sea level rise on developing countries: a comparative analysis', *Climatic Change*, vol. 93, no. 3, 2009/04/01, pp. 379-388.

Dasgupta, S, Laplante, B, Murray, S & Wheeler, D 2011, 'Exposure of developing countries to sea-level rise and storm surges', *Climatic Change*, vol. 106, no. 4, 2011/06/01, pp. 567-579.

DaVanzo, J 1980, *Microeconomic approaches to studying migration decisions*, The Rand Corporation. N-1201-NICHD.

Davies, M & Hughes, N 2014, *Doing a successful research project: using qualitative or quantitative methods*, Second edition. edn, Palgrave Macmillan, Basingstoke, Hampshire.

De Châtel, F 2014, 'The Role of Drought and Climate Change in the Syrian Uprising: Untangling the Triggers of the Revolution', *Middle Eastern Studies*, vol. 50, no. 4, 2014/07/04, pp. 521-535.

De Haas, H 2007, 'Turning the tide? Why development will not stop migration', *Development and change*, vol. 38, no. 5, pp. 819-841.

de Haas, H 2010a, 'The Internal Dynamics of Migration Processes: A Theoretical Inquiry', *Journal of Ethnic and Migration Studies*, vol. 36, no. 10, pp. 1587-1617.

de Haas, H 2010b, 'Migration and Development: A Theoretical Perspective', *The International Migration Review*, vol. 44, no. 1, pp. 227-264.

De Haas, H 2014, *Migration theory: Quo vadis?*, International Migration Institute, University of Oxford.

De Juan, A 2015, 'Long-term environmental change and geographical patterns of violence in Darfur, 2003–2005', *Political Geography*, vol. 45, 2015/03/01/, pp. 22-33.

De Koninck, R 1992, *Malay peasants coping with the world : breaking the community circle?*, Social Issues in Southeast Asia, Institute of Southeast Asian Studies, Singapore.

de Lellis, F 2019, 'Peasants, dispossession and resistance in Egypt: an analysis of protest movements and organisations before and after the 2011 uprising', *Review of African Political Economy*, vol. 46, no. 162, 2019/10/02, pp. 582-598.

De Sherbinin, A, Castro, M, Gemenne, F, Cernea, M, Adamo, S, Fearnside, P, Krieger, G, Lahmani, S, Oliver-Smith, A & Pankhurst, A 2011, 'Preparing for resettlement associated with climate change', *Science*, vol. 334, no. 6055, pp. 456-457.

Defiesta, G & Rapera, C 2014, 'Measuring Adaptive Capacity of Farmers to Climate Change and Variability: Application of a Composite Index to an Agricultural Community in the Philippines', *Journal of Environmental Science and Management*, vol. 17, 01/01, pp. 48-62.

DeJuan, JP & J. Seater, J 1999, 'The permanent income hypothesis: Evidence from the consumer expenditure survey', *Journal of Monetary Economics*, vol. 43, no. 2, 1999/04/01/, pp. 351-376.

Delcour, I, Spanoghe, P & Uyttendaele, M 2014, 'Literature review: Impact of climate change on pesticide use', *Food Research International*, vol. 68, pp. 7-15.

Dercon, S & Gollin, D 2014, 'Agriculture in African development: theories and strategies', *Annu. Rev. Resour. Econ.*, vol. 6, no. 1, pp. 471-492.

Deressa, T, Hassan, R & Ringler, C 2011, 'Perception and adaptation to climate change by farmers in the Nile basin of Ethiopia', *The Journal of Agricultural Science*, vol. 149, 02/01, pp. 23-31.

Deressa, TT, Hassan, RM, Ringler, C, Alemu, T & Yesuf, M 2009, 'Determinants of farmers' choice of adaptation methods to climate change in the Nile Basin of Ethiopia', *Global Environmental Change*, vol. 19, no. 2, 2009/05/01/, pp. 248-255.

Desmarais, A-A 2007, 'Globalization and the power of peasants: La Vía Campesina', Halifax: Fernwood Publishing.

Detges, A 2016, 'Local conditions of drought-related violence in sub-Saharan Africa: The role of road and water infrastructures', *Journal of Peace Research*, vol. 53, no. 5, pp. 696-710.

Diggs, D 1991, 'Drought Experience and Perception of Climatic Change among Great Plains Farmers', *Great Plains Research*, vol. 1, no. 1, pp. 114-132.

Dimova, R, Elder, S & Stephan, K 2016, 'Labour market transitions of young women and men in the Middle East and North Africa', *Work4Youth Publication Series*, International Labour Organization, Geneva, vol. 44. p. 80. <u>https://youtheconomicopportunities.org/sites/default/files/uploads/resource/Labour% 20t</u> <u>ransition.pdf</u>

Dimova, R & Stephan, K 2020, 'Inequality of opportunity and (unequal) opportunities in the youth labour market: How is the Arab world different?', *International Labour Review*, vol. 159, no. 2, pp. 217-242.

Dixon, J, Gulliver, A & Gibbon, D 2001, *Farming Systems and Poverty: Improving Farmers' Livelihoods in a Changing World*, ed. M Hall, FAO, Rome, p. 49.

Dixon, MW 2020, 'The agrarian question revisited: smallholders and corporate food in Egypt', *Canadian Journal of Development Studies / Revue canadienne d'études du développement*, vol. 41, no. 2, 2020/04/02, pp. 279-295.

Donat, MG, Peterson, TC, Brunet, M, King, AD, Almazroui, M, Kolli, RK, Boucherf, D, Al-Mulla, AY, Nour, AY, Aly, AA, Nada, TAA, Semawi, MM, Al Dashti, HA, Salhab, TG, El Fadli, KI, Muftah, MK, Dah Eida, S, Badi, W, Driouech, F, El Rhaz, K, Abubaker, MJY, Ghulam, AS, Erayah, AS, Mansour, MB, Alabdouli, WO, Al Dhanhani, JS & Al Shekaili, MN 2014, 'Changes in extreme temperature and precipitation in the Arab region: long-term trends and variability related to ENSO and NAO', *International Journal of Climatology*, vol. 34, no. 3, 2014/03/15, pp. 581-592.

Dorosh, P & Thurlow, J 2014, 'Can Cities or Towns Drive African Development? Economywide Analysis for Ethiopia and Uganda', *World Development*, vol. 63, no. C, pp. 113-123.

Dowd, A-M, Marshall, N, Fleming, A, Jakku, E, Gaillard, E & Howden, M 2014, 'The role of networks in transforming Australian agriculture', *Nature Climate Change*, vol. 4, no. 7, 2014/07/01, pp. 558-563.

Droogers, P, Immerzeel, W, Terink, W, Hoogeveen, J, Bierkens, M, Van Beek, L & Debele, B 2012, 'Water resources trends in Middle East and North Africa towards 2050', *Hydrology and Earth System Sciences*, vol. 16, pp. 3101-3114.

Drzewiecki, DM, Wavering, HM, Milbrath, GR, Freeman, VL & Lin, JY 2020, 'The association between educational attainment and resilience to natural hazard-induced disasters in the West Indies: St. Kitts & Nevis', *International Journal of Disaster Risk Reduction*, vol. 47.

Dunne, JP, Stouffer, RJ & John, JG 2013, 'Reductions in labour capacity from heat stress under climate warming', *Nature Climate Change*, vol. 3, June 01, 2013, pp. 563-566.

EACH-FOR 2009, Environmental Change and Forced Migration Scenarios: D.3.4. Synthesis Report, p. 81.

Easterling, WE, Crosson, PR, Rosenberg, NJ, McKenney, MS, Katz, LA & Lemon, KM 1993, 'Paper 2. Agricultural impacts of and responses to climate change in the Missouri-Iowa-Nebraska-Kansas (MINK) region', *Climatic Change*, vol. 24, no. 1-2, pp. 23-61.

Ebrahim, N 2019, 'This is the water crisis that Egypt is facing', *World Economic Forum*. <u>https://www.weforum.org/agenda/2019/11/water-crisis-builds-in-egypt-as-dam-talks-falter-temperatures-rise/</u>

Climate Diplomacy n.d-a, *Conflict Factsheet: Food Price Shocks in Egypt*, adelphi and Federal Forgien Office of Germany, accessed March 2020, <u>https://climate-diplomacy.org/case-studies/food-price-shocks-egypt</u>

Climate Diplomacy n.d-b, *Conflict Factsheet: Security Implications of Growing Water Scarcity in Egypt*, adelphi and Fedral Forigen Office of Germany, accessed March 2020, <u>https://climate-diplomacy.org/case-studies/security-implications-growing-water-</u> <u>scarcity-egypt</u>

EEAA 2003, *Damietta Governorate Environmental Profile: SEAM programme*, Ministry of State for Environmental Affairs, Egyptian Environmal Affairs Agency, Cairo, Egypt, p. 190.

EEAA 2010, Second National Communication. Arab Republic of Egypt, Egyptian Environmental Affairs Agency, Cairo, Egypt p. 137.

EEAA 2016, *Egypt State of the Environment Report*, Egyptian Environmental Affairs Agency. Cairo, Egypt, p. 288.

Eid, H, El-Marsafawy, S & Ouda, S 2007, 'Assessing the Economic Impacts of Climate Change on Agriculture in Egypt: A Ricardian Approach', *The World Bank, Policy Research Working Paper Series*, 01/01, p. 39.

Ekemhonye, S, Nmadu, J, Coker, A & Ndanitsa, R 2020, 'Factors Influencing Farmers Adaptive Capacities to Climate Change Alone River Niger communities in Edo and Kogi States, Nigeria', *International Journal of Innovative Science and Research Technology*, vol. 5, no. 7, pp. 475-479.

El-Gammal, MI, Ali, RR & Eissa R, 2014, 'Land use assessment of barren areas in Damietta Governorate, Egypt using remote sensing', *Egyptian Journal of Basic and Applied Sciences*, vol. 1, no. 3–4, pp. 151-160.

El-Marsafawy, S 2008, *Vulnerability and Adaptation of Climate Change on the Agricultural Sector in Egypt*, Technical Report 2008, Association of International Civil Servants (AFICS-Egypt), Egypt.

El-Nashar, WY & Elyamany, AH 2018, 'Managing risks of the Grand Ethiopian Renaissance Dam on Egypt', *Ain Shams Engineering Journal*, vol. 9, no. 4, 2018/12/01/, pp. 2383-2388.

El-Raey, M 2010, *Impact of Sea Level Rise on the Arab Region*, University of Alexandria and Regional Center for Disaster Risk Reduction Arab Academy of Science, Technology and Maritime Transport, Egypt, p. 89.

El-Ramady, HR, El-Marsafawy, SM & Lewis, LN 2013, 'Sustainable Agriculture and Climate Changes in Egypt', in E Lichtfouse (ed.), *Sustainable Agriculture Reviews: Volume 12*, Springer Netherlands, Dordrecht, pp. 41-95.

El Bedawy, R 2014, 'Water Resources Management: Alarming Crisis for Egypt', *Journal of Management and Sustainability*, vol. 4, no. 3, pp. 108-124.

El Megharbel, N 2011, *Local Administration System In Egypt*, European Association For State Territorial Representatives (Eastr), Local Administration Reform Unit, Cairo, Egypt.

El Naggar, AS 2012, 'Egypt's economy in the past three decades', in H Shaarawy & A Shokr (eds), *Social protest movements in Egypt and the preparation to the January 25 Revolution*, Jazeret Alward, Egypt.

El Nour, S 2015, 'Small farmers and the revolution in Egypt: the forgotten actors', *Contemporary Arab Affairs*, vol. 8, no. 2, pp. 198-211.

El Nour, S 2018, 'The Dam, the rice and the state: Environmental racism towards Delta farmers.', *almanassa*. <u>https://almanassa.com/ar/story/8895</u>

El Nour, S 2019, 'Grabbing from below: a study of land reclamation in Egypt', *Review of African Political Economy*, vol. 46, no. 162, 2019/10/02, pp. 549-566.

Elasha, BO 2010, *Mapping of Climate Change Threats and Human Development Impacts in the Arab Region*, Arab Human Development Report: Research Paper Series, United Nations Development Programme (UNDP), p. 51.

Elliott, JR & Pais, J 2006, 'Race, class, and Hurricane Katrina: Social differences in human responses to disaster', *Social science research*, vol. 35, no. 2, pp. 295-321.

Elliott, JR & Pais, J 2010, 'When Nature Pushes Back: Environmental Impact and the Spatial Redistribution of Socially Vulnerable Populations*', *Social Science Quarterly*, vol. 91, no. 5, 2010/12/01, pp. 1187-1202.

Elshamy, M, Sayed, A & Badawy, B 2009, 'Impacts of Climate Change on the Nile Flows at Dongola Using Statistical Downscaled GCM Scenarios', *Nile Basin Water Engineering Scientific Magazine*, vol. 2, 01/01, pp. 1-14.

Elshinnawy, IA & Almaliki, AH 2021, 'Vulnerability Assessment for Sea Level Rise Impacts on Coastal Systems of Gamasa Ras El Bar Area, Nile Delta, Egypt', *Sustainability*, vol. 13, no. 7, p. 3624.

Engels, B 2021, 'Peasant Resistance in Burkina Faso's Cotton Sector', *International Review of Social History*, vol. 66, no. S29, pp. 93-112.

Eriksen, S, Aldunce, P, Bahinipati, CS, Martins, RDA, Molefe, JI, Nhemachena, C, O'brien, K, Olorunfemi, F, Park, J & Sygna, L 2011, 'When not every response to climate change is a good one: Identifying principles for sustainable adaptation', *Climate and Development*, vol. 3, no. 1, pp. 7-20.

Eriksen, SH, Nightingale, AJ & Eakin, H 2015, 'Reframing adaptation: The political nature of climate change adaptation', *Global Environmental Change*, vol. 35, pp. 523-533.

Evans, JP 2009, '21st century climate change in the Middle East', *Climatic Change*, vol. 92, no. 3, 2009/02/01, pp. 417-432.

Fadel-Girgis, M 1983, 'Family support for the elderly in Egypt', *The Gerontologist*, vol. 32, no. 6, pp. 589-59.

FAO 2017, 'AQUASTAT Country Statistics', *Online*, Food and Agriculture Organisation (FAO), Rome, Italy,

FAO 2020, 'Country Briefs: Egypt', Global Information and Early Warning System(GIEWS);FoodandAgricultureOrganization(FAO).

http://www.fao.org/giews/countrybrief/country.jsp?code=EGY#:~:text=The%20country %20remains%20the%20world's.of%20the%20last%20five%20years

FAO, CTA & IFAD 2014, *Youth and agriculture: key challenges and concrete solutions*, Food and Agriculture Organization (FAO), Italy.

FAOSTAT 2018, 'Wheat Import Quantity: Egypt', Food and Agriculture Organization (FAO), Italy.

Farbotko, C 2018, 'Voluntary immobility: Indigenous voices in the Pacific', *Forced Migration Review*, no. 57, pp. 81-83.

Farbotko, C, Dun, O, Thornton, F, McNamara, KE & McMichael, C 2020, 'Relocation planning must address voluntary immobility', *Nature Climate Change*, vol. 10, no. 8, pp. 702-704.

Fatti, CE & Patel, Z 2013, 'Perceptions and responses to urban flood risk: Implications for climate governance in the South', *Applied Geography*, vol. 36, pp. 13-22.

Fautras, M & Iocco, G 2019, 'Land, politics and dynamics of agrarian change and resistance in North Africa', *Review of African Political Economy*, vol. 46, no. 162, 2019/10/02, pp. 535-548.

Feenstra, JF, Burton, I, Smith, JB & Tol, RSJ 1998, *Handbook on methods for climate change impact assessment and adaptation strategies*, United Nations Environment Programme (UNEP).

Fei, X 1939, 'Peasant life in China', London: Kegan Paul.

Feilzer, YM 2010, 'Doing Mixed Methods Research Pragmatically: Implications for the Rediscovery of Pragmatism as a Research Paradigm', *Journal of Mixed Methods Research*, vol. 4, no. 1, 2010/01/01, pp. 6-16.

Feitelson, E & Tubi, A 2017, 'A main driver or an intermediate variable? Climate change, water and security in the Middle East', *Global Environmental Change*, vol. 44, pp. 39-48.

Ferede, W & Abebe, S 2014, 'The Efficacy of Water Treaties in the Eastern Nile Basin', *Africa Spectrum*, vol. 49, no. 1, 2014/04/01, pp. 55-67.

Fernández-Giménez, ME, Batkhishig, B, Batbuyan, B & Ulambayar, T 2015, 'Lessons from the Dzud: Community-Based Rangeland Management Increases the Adaptive Capacity of Mongolian Herders to Winter Disasters', *World Development*, vol. 68, 2015/04/01/, pp. 48-65.

FIAN 2000, *The Right to Adequate Food in Egypt*, FIAN Parallel Report, The International Human Rights Organisation for the Right to Feed Oneself (FIAN), Heidelberg.

Field, CB, Barros, VR, Mach, KJ, Mastrandrea, MD, Aalst, Mv, Adger, WN, Arent, DJ, Barnett, J, Betts, R, Bilir, TE, Birkmann, J, Carmin, J, Chadee, DD, Challinor, AJ, Chatterjee, M, Cramer, W, Davidson, DJ, Y.O. Estrada, Gattuso, J-P, Hijioka, Y, Hoegh-Guldberg, O, Huang, HQ, Insarov, GE, Jones, RN, Kovats, RS, Romero-Lankao, P, Larsen, JN, Losada, IJ, Marengo, JA, McLean, RF, Mearns, LO, Mechler, R, Morton, JF, Niang, I, Oki, T, Olwoch, JM, Opondo, M, Poloczanska, ES, Pörtner, H-O, Redsteer, MH, Reisinger, A, Revi, A, Schmidt, DN, Shaw, MR, Solecki, W, Stone, DA, Stone, JMR, Strzepek, KM, Suarez, AG, Tschakert, P, Valentini, R, Vicuña, S, Villamizar, A, Vincent, KE, Warren, R, White, LL, Wilbanks, TJ, Wong, PP & Yohe, GW 2014, *Technical summary*, Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, Intergovernmental Panel on Climate Change (IPCC), Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

Findlay, AM 2011, 'Migrant destinations in an era of environmental change', *Global Environmental Change*, vol. 21, 2011/12/01/, pp. S50-S58.

Foresight 2011, *Migration and Global Environmental Change: Future Challenges and Opportunities*, Final Project Report, The Government Office of Science, London.

Forouheshfar, Y, El Mekkaoui, N & d'Albis, H 2020, 'Demographics in MENA Countries: A Major Driver for Economic Growth', *Economist (Netherlands)*, vol. 168, no. 2, pp. 183-213.

Foster, JB & Magdoff, F 1998, 'Liebig, Marx, and the depletion of soil fertility: Relevance for today's agriculture', *Monthly Review*, vol. 50, no. 3, p. 32.

Fosu-Mensah, BY, Vlek, PLG & MacCarthy, DS 2012, 'Farmers' perception and adaptation to climate change: a case study of Sekyedumase district in Ghana', *Environment, Development and Sustainability*, vol. 14, no. 4, 2012/08/01, pp. 495-505.

Fothergill, A, Maestas, EGM & Darlington, JD 1999, 'Race, Ethnicity and Disasters in the United States: A Review of the Literature', *Disasters*, vol. 23, no. 2, 1999/06/01, pp. 156-173.

Friedman, M 1957, Theory of the consumption function, Princeton university press.

Frihy, OE & El-Sayed, MK 2013, 'Vulnerability risk assessment and adaptation to climate change induced sea level rise along the Mediterranean coast of Egypt', *Mitigation and Adaptation Strategies for Global Change*, vol. 18, no. 8, 2013/12/01, pp. 1215-1237.

Froehlich, P & Al-Saidi, M 2018, 'Local community perception of climate change adaptation in Egypt', *IOP Conference Series, Earth and Environmental Science*, vol. 191, Institute of Physics Publishing.

Füssel, H-M & Klein, R 2006, 'Climate Change Vulnerability Assessments: An Evolution of Conceptual Thinking', *Climatic Change*, vol. 75, 01/03, pp. 301-329.

Gaballah, M, Khalaf, K, Beck, A & Lopez, J 2005, 'Water pollution in relation to agricultural activity impact in Egypt', *Journal of Applied Sciences Research*, vol. 1, no. 1, pp. 9-17.

Gallopín, GC 2006, 'Linkages between vulnerability, resilience, and adaptive capacity', *Global Environmental Change*, vol. 16, no. 3, pp. 293-303.

Gbetibouo, GA 2009, 'Understanding farmers' perceptions and adaptations to climate change and variability: The case of the Limpopo Basin, South Africa', vol. 15, no. 8, International Food Policy Research Institute (IFPRI), p. 52.

Gemenne, F 2013, 'Migration doesn't have to be a failure to adapt: An escape from environmental determinism', in J Palutikof, SL Boulter, AJ Ash, MS Smith, M Parry, M Waschka & D Guitart (eds), *Climate Adaptation Futures*, John Wiley & Sons, pp. 235-241.

Gemenne, F 2015, 'Migration as a Climate Adaptation Strategy', in *State of the World 2015: Confronting Hidden Threats to Sustainability*, Island Press/Center for Resource Economics, Washington, DC, pp. 117-126.

Gemenne, F & Blocher, J 2017, 'How can migration serve adaptation to climate change? Challenges to fleshing out a policy ideal', *Geographical Journal*, vol. 183, no. 4, pp. 336-347.

Gervasio, G & Manduchi, P 2020, 'Introduction: reading the revolutionary process in North Africa with Gramsci', *The Journal of North African Studies*, pp. 1-6.

Ghoneim, AF 2012, *The Political Economy of Food Price Policy in Egypt*, WIDER Working Paper 2012/096, UNU-WIDER., Helsinki.

Glass, N 2010, 'The water crisis in Yemen: causes, consequences and solutions', *Global Majority E-Journal*, vol. 1, no. 1, pp. 17-30.

Gleick, PH 2014, 'Water, Drought, Climate Change, and Conflict in Syria', *Weather, Climate, and Society*, vol. 6, no. 3, pp. 331-340.

GLOWA 2009, An integrated approach to sustainable management of water resources under global change, *GLOWA Jordan River, phase II final report*, p. 358.

Goldberg, E 2009, 'Peasants in Revolt — Egypt 1919', *International Journal of Middle East Studies*, vol. 24, no. 2, pp. 261-280.

Golmohammadi, F 2020, Land fragmentation and small holdings major barriers for sustainable agricultural development in Iran: Situations and problems, *EasyChair*, no. 2687, p. 17.

Gouda, DM 2020, 'Climate Change, Agriculture and Rural Communities' Vulnerability in the Nile Delta', in E-S Ewis Omran & AM Negm (eds), *Climate Change Impacts on Agriculture and Food Security in Egypt: Land and Water Resources—Smart Farming— Livestock, Fishery, and Aquaculture*, Springer International Publishing, Cham, pp. 525-576.

Gray, C & Mueller, V 2012, 'Drought and population mobility in rural Ethiopia', *World Development*, vol. 40, no. 1, pp. 134-145.

Grossman-Cohen, B 2011, 'Teetering on the edge', *Politics of Poverty*, OXFAM, USA. <u>https://politicsofpoverty.oxfamamerica.org/teetering-on-the-edge/</u>

Grothmann, T & Patt, A 2005, 'Adaptive capacity and human cognition: The process of individual adaptation to climate change', *Global Environmental Change*, vol. 15, no. 3, 2005/10/01/, pp. 199-213.

Gürbüz, AA 2011, 'Comparing trajectories of structural change', *Cambridge Journal of Economics*, vol. 35, no. 6, pp. 1061-1085.

Hafez, MR 2020, 'Impacts of Climate Change on Agriculture, Livelihoods, and Women in Nile Delta, Egypt', in WL Filho (ed.), *Handbook of Climate Change Resilience*, Springer Nature, Switzerland, pp. 765-784.

Haggblade, S, Hazell, PB & Reardon, T 2007, *Transforming the rural nonfarm economy: Opportunities and threats in the developing world*, Intl Food Policy Res Inst, IFPRI Issue Brief 58, p. 4.

Hallegatte, S 2012, 'A framework to investigate the economic growth impact of sea level rise', *Environmental Research Letters*, vol. 7, no. 1, pp. 1-7.

Hamama, M & Charbel, J 2015, 'Giza residents protest ongoing water cuts, doubt promises of resolution ', *Mada Masr*. <u>https://www.madamasr.com/en/2015/09/21/feature/politics/giza-</u> residents-protest-ongoing-water-cuts-doubt-promises-of-resolution/

Hamdan, G 1989, *The Normalization: The Zionist Penetration Strategy*, Dar El-Aman, Beirut, Lebanon.

Hameed, M, Ahmadalipour, A & Moradkhani, H 2020, 'Drought and food security in the middle east: An analytical framework', *Agricultural and Forest Meteorology*, vol. 281, pp. 1-12.

Hart, GP, Turton, A & White, BNF 1989, *Agrarian transformations : local processes and the state in Southeast Asia*, eds GP Hart, A Turton & BNF White, University of California Press, Berkeley.
Hassan, R & Nhemachena, C 2008, 'Determinants of African farmers' strategies for adapting to climate change: Multinomial choice analysis', *African Journal of Agricultural and Resource Economics*, vol. 02, no. 1, pp. 83-104.

Henderson, C 2017, 'Gulf Land Acquisitions in Egypt and Sudan: Food Security or the Agro-commodity Supply Chains?', *Cairo Papers in Social Science*, vol. 34, no. 4, pp. 127-140.

Hereher, M 2016, 'Time series trends of land surface temperatures in Egypt: a signal for global warming', *Environmental Earth Sciences*, vol. 75, no. 17, pp. 1-11.

Herrmann, M & Svarin, D 2009, 'Environmental pressures and rural-urban migration: The case of Bangladesh', MPRA Paper No. 12879, p. 28.

Hesse-Biber, S 2015, 'Mixed Methods Research: The "Thing-ness" Problem', *Qualitative Health Research*, vol. 25, no. 6, 2015/06/01, pp. 775-788.

Heyne, S & Gebel, M 2016, 'Education effects on the school-to-work transition in Egypt: A cohort comparison of labor market entrants 1970–2012', *Research in Social Stratification and Mobility*, vol. 46, 2016/12/01/, pp. 37-49.

Hibberts, M, Johnson, RB & Hudson, K 2012, 'Common survey sampling techniques', in L Gideon (ed.), *Handbook of Survey Methodology for the Social Sciences*, Springer, USA, pp. 53-74.

Hoerling, M, Eischeid, J, Perlwitz, J, Quan, X, Zhang, T & Pegion, P 2011, 'On the Increased Frequency of Mediterranean Drought', *Journal of Climate*, vol. 25, no. 6, 2012/03/01, pp. 2146-2161.

Hopkins, NS 1987, Agrarian Transformation in Egypt, Westview Press.

Houghton, J 2009, *Global Warming: The Complete Briefing*, 4 edn, Cambridge University Press, Cambridge.

Howden, S, Crimp, S & Nelson, R 2010, 'Australian agriculture in a climate of change', *Managing Climate Change*, 01/01, pp. 101-111.

Hugo, G 1996, 'Environmental concerns and international migration', *International Migration Review*, vol. 30, no. 1, pp. 105-131.

Hugo, G 2006, 'Globalization and changes in Australian international migration', *Journal of Population Research*, vol. 23, no. 2, 2006/09/01, pp. 107-134.

Hugo, G, Bardsley, D, Tan, Y, Sharma, V, Williams, M & Bedford, R, 2009 'Climate change and migration in the Asia-Pacific region', Asian Development Bank, Manila.

Hugo, G 2011, 'Future demographic change and its interactions with migration and climate change', *Global Environmental Change*, vol. 21, 2011/12/01/, pp. S21-S33.

Human Rights Watch 2020, *Deadly Consequences: Obstruction of Aid in Yemen During Covid-19*, Human Rights Watch, USA. <u>https://www.hrw.org/report/2020/09/14/deadly-consequences/obstruction-aid-yemen-during-covid-19</u>

Hunt, A & Watkiss, P 2011, 'Climate change impacts and adaptation in cities: a review of the literature', *Climatic Change*, vol. 104, no. 1, 2011/01/01, pp. 13-49.

Hunter, LM, Luna, JK & Norton, RM 2015, 'Environmental Dimensions of Migration', *Annual Review of Sociology*, vol. 41, no. 1, 2015/08/14, pp. 377-397.

Ianchovichina, E, Mottaghi, L & Devarajan, S 2015, *Inequality, uprisings, and conflict in the Arab World (English)*. *Middle East and North Africa (MENA) Economic Monitor.*, World Bank Group, Washington, D.C.

Ibrahim, FN & Ibrahim, B 2003, *Egypt : an economic geography*, I.B. Tauris, London ; New York.

Ide, T, Lopez, MR, Fröhlich, C & Scheffran, J 2021, 'Pathways to water conflict during drought in the MENA region', *Journal of Peace Research*, vol. 58, no. 3, pp. 568-582.

IFAD 2008, *IFAD and climate change*, Consultation on the Eighth Replenishment of IFAD's Resources — Fourth Session Rome, 21-22 October, International Fund for Agricultural Development (IFAD).

IFAD 2010, *Climate Change: Strategy*, International Fund for Agricultural Development (IFAD), Rome, Italy.

IFAD 2014, *Youth and Land Tenure: Land Tenure Toolkit*, Lessons learned, Internation Fund for Agricultural Development, Rome, Italy.

IFAD 2019, *Investing in rural people in Egypt*, International Fund for Agricultural Development (IFAD). Cairo, Egypt.

Information and Decision-making Support Unit of Damietta 2010, *Monthly Information Bulletin of Damietta: December 2010*, Egypt.

Information and Decision-making Support Unit of Damietta 2017, *Monthly Information Bulletin of Damietta: December 2017*, Egypt.

IOM 2019, *Migration in Nepal: A Country Profile 2019*, International Organization for Migration, Thirbam Sadak, Nepal, p. 188.

IOM 2020a, 'International Emigrants (Total): Egypt', International Organisation for Migration,

https://migrationdataportal.org/data?amp%3Bfocus=profile&i=stock_abs_origin&t=202 0&cm49=818

IOM 2020b, 'International Emigrants (Total): Jordan', International Organisation for Migration,

https://migrationdataportal.org/data?amp%3Bfocus=profile&i=stock_abs_origin&t=202 0&cm49=400

IOM 2020c, 'International Emigrants (Total): Lebanon', nternational Organisation for Migration.,

https://migrationdataportal.org/data?amp%3Bfocus=profile&i=stock_abs_origin&t=202 0&cm49=422

IPCC 2007a, *Climate Change 2007: Impacts, Adaptation and Vulnerability*, Contribution of Working Group II to the Fourth Assessment Report of the Intergovernomental Panel on Climate Change., Intergovernmental Panel on Climate Change (IPCC), CU Press, Cambridge, UK.

IPCC 2007b, Climate Change 2007: Synthesis Report. Contribution of Working Groups I, II and III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, Intergovernmental Panel for Climate Change (IPCC), Geneva, Switzerland.

IPCC 2014, *Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects*, Working Group II Contribution to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, Intergovernmental Panel on Climate Change, CU Press, Cambridge, United Kingdom and New York, NY, USA.

IRIN 2010, 'Growing protests over water shortages', *The New Humanitarian*, <u>https://www.thenewhumanitarian.org/news/2010/07/27/growing-protests-over-water-shortages</u>

Isham, J 2002, 'The Effect of Social Capital on Fertiliser Adoption: Evidence from Rural Tanzania', *Journal of African Economies*, vol. 11, no. 1, pp. 39-60.

Ishaya, S & Abaje, IB 2008, 'Indigenous People's Perception on Climate Change and Adaptation Strategies in Jema'a Local Government Area of Kaduna State, Nigeria', *Journal of Geography and Regional Planning.*, vol. 1, 10/23, pp. 138-143.

Islam, N & Winkel, J 2017, 'Climate change and social inequality', UN Department of Economic and Social Affairs (DESA) Working Papers, No. 152, UN, New York.

Jäger, J, Frühmann, J, Grünberger, S & Vag, A 2009, *Environmental Change and Forced Migration Scenarios EACH-FOR Project Synthesis Report*, United Nations University, Spain.

Johnson, RB & Onwuegbuzie, AJ 2004, 'Mixed Methods Research: A Research Paradigm Whose Time Has Come', *Educational Researcher*, vol. 33, no. 7, 2004/10/01, pp. 14-26.

Johnston, BF & Mellor, JW 1961, 'The role of agriculture in economic development', *The American Economic Review*, vol. 51, no. 4, pp. 566-593.

Johnston, M & Hesseln, H 2012, 'Climate change adaptive capacity of the Canadian forest sector', *Forest Policy and Economics*, vol. 24, 11/01, pp. 29–34.

Jones, L, Ludi, E & Levine, S 2010, 'Towards a characterisation of adaptive capacity: a framework for analysing adaptive capacity at the local level', Background Paper, *Overseas Development Institute*, p.8.

Joseph, H & Neiburg, F 2020, '(I)mmobility and the Pandemic in Haitian Landscapes', *Horizontes Antropologicos*, vol. 26, no. 58, pp. 1-17.

Hassanein, MK, Elsayed, M & Khalil, AA 2012, 'Impacts of Sowing Date, Cultivar, Irrigation Regimes and Location on Bread Wheat Production in Egypt under Climate Change Conditions', *Nature and Science*, vol. 10, no. 12, pp. 141-150.

Kabbani, N 2019, *Investing in rural youth in the Near East, North Africa, Europe and Central Asia*, IFAD Research Series 50, International Fund for Agricultural Development (IFAD), Italy.

Karajeh, F, Oweis, T, Swelam, A, El-Gindy, A, El-Quosy, D, Khalifa, H, El-Kholy, M & Abd El-Hafez, S 2011, *Water and Agriculture in Egypt*, Food and Agriculture Organisation (FAO), Beirut, Lebanon.

Kassem, HS, Bello, ARS, Alotaibi, BM, Aldosri, FO & Straquadine, GS 2019, 'Climate Change Adaptation in the Delta Nile Region of Egypt: Implications for Agricultural Extension', *Sustainability*, vol. 11, no. 3, pp. 1-22.

Kassim, Y, Mahmoud, M, Kurdi, S & Breisinger, C 2018, *An Agricultural Policy Review* of Egypt: First Steps Towards a New Strategy, Middle East and North Africa Regional Program Working Paper 11, International Food Policy Research Institute (IFPRI), Egypt.

Kates, RW, Travis, WR & Wilbanks, TJ 2012, 'Transformational adaptation when incremental adaptations to climate change are insufficient', *Proceedings of the National Academy of Sciences*, vol. 109, no. 19, pp. 7156–7161.

Kelley, CP, Mohtadi, S, Cane, MA, Seager, R & Kushnir, Y 2015, 'Climate change in the Fertile Crescent and implications of the recent Syrian drought', *Proceedings of the National Academy of Sciences*, vol. 112, no. 11, pp. 3241-3246.

Kelly, P & Adger, W 2000, 'Theory and Practice in Assessing Vulnerability to Climate Change and Facilitating Adaptation', *Climatic Change*, vol. 47, no. 4, pp. 325-352.

Kelly, PF 2011, 'Migration, Agrarian Transition, and Rural Change in Southeast Asia', *Critical Asian Studies*, vol. 43, no. 4, 2011/12/01, pp. 479-506.

Kerkvliet, B 1990, '*Everyday Politics in the Philippines : Class and Status Relations in a Central Luzon Village*', University of California Press, Berkeley, CA.

Keshk, H 2012, 'The possibilities of the protests and the peasant movement in social and political change 1977-2008', in H Shaarawy & A Shokr (eds), *Social protest movements in Egypt and the preparation to the January 25 Revolution.*, Jazeret Alward., Egypt.

Keyder, Ç 1983, 'Paths of rural transformation in Turkey', *The Journal of Peasant Studies*, vol. 11, no. 1, 1983/10/01, pp. 34-49.

Khan, NA, Qiao, J, Abid, M & Gao, Q 2021, 'Understanding farm-level cognition of and autonomous adaptation to climate variability and associated factors: Evidence from the rice-growing zone of Pakistan', *Land Use Policy*, vol. 105, pp. 1-12.

Kim, U & Kaluarachchi, JJ 2009, 'Climate Change Impacts on Water Resources in the Upper Blue Nile River Basin, Ethiopia1', *JAWRA Journal of the American Water Resources Association*, vol. 45, no. 6, pp. 1361-1378.

Klein, R 1998, 'Towards better understanding, assessment and funding of climate adaptation', *Change*, vol. 44, pp. 15-19.

Klein, RJ 2003, 'Adaptation to climate variability and change: what is optimal and appropriate', In: *Climate Change in the Mediterranean: Socio-Economic Perspectives of Impacts, Vulnerability and Adaptation*, vol. 32, pp. 32-52.

Knerr, B 2017, 'International labour migration and remittances in Nepal: An overview', In: *International Labor Migration and Livelihood Security in Nepal: Considering the Household Level*, vol. 22, p. 63.

Kniveton, D, Schmidt-Verkerk, K, Smith, C & Black, R 2008, *Climate Change and Migration: Improving Methodologies to Estimate Flows*, IOM International Organization for Migration Paper No.33, International Organization for Migration (IOM), Geneva, Switzerland.

Koerth, J, Vafeidis, AT, Hinkel, J & Sterr, H 2013, 'What motivates coastal households to adapt pro-actively to sea-level rise and increasing flood risk?', *Regional environmental change*, vol. 13, no. 4, pp. 897-909.

Kohler, T, Giger, M, Hurni, H, Ott, C, Wiesmann, U, von Dach, SW & Maselli, D 2010, 'Mountains and climate change: a global concern', *Mountain Research and Development*, vol. 30, no. 1, pp. 53-55.

Koocheki, A 2010, 'The Middle East: Adapting Food Production to Local Biophysical Realities', in SR Gliessman & M Rosemeyer (eds), *The Conversion to Sustainable Agriculture: Principles, Processes and Practices*, CRC Press, USA, pp. 313-326.

Koohfkan, P 2001, 'Food, security, and sustainable development in desert communities. In Sustainable development of desert communities: A regional symposium, 169–181. Technical Paper 2, ' in United Nation Development Programme in Iran, Tehran.

Kothari, U 2003, 'Staying put and staying poor?', *Journal of International Development*, vol. 15, no. 5, 2003/07/01, pp. 645-657.

Koubi, V, Stoll, S & Spilker, G 2016, 'Perceptions of environmental change and migration decisions', *Climatic Change*, vol. 138, no. 3, October 01, pp. 439-451.

Kouzana, L, Mammou, AB & Felfoul, MS 2009, 'Seawater intrusion and associated processes: Case of the Korba aquifer (Cap-Bon, Tunisia)', *Comptes Rendus Geoscience*, vol. 341, no. 1, 2009/01/01/, pp. 21-35.

Kreimer, A, Arnold, M & Carlin, A 2003, *Building safer cities - the future of disaster risk*, Disaster Risk Management Seriers no. 3, World Bank., Washington, DC.

Kuglitsch, FG, Toreti, A, Xoplaki, E, Della-Marta, PM, Zerefos, CS, Türkeş, M & Luterbacher, J 2010, 'Heat wave changes in the eastern Mediterranean since 1960', *Geophysical Research Letters*, vol. 37, no. 4, pp. 1-5.

Kuznets, SS 1966, *Modern economic growth: rate, structure, and spread*, Studies in comparative economics, Yale University Press, New Haven.

Lange, M 2019, 'Impacts of Climate Change on the Eastern Mediterranean and the Middle East and North Africa Region and the Water–Energy Nexus', *Atmosphere*, vol. 10, no. 8, pp. 1-22.

Lee, SH 1985, Why people intend to move: individual and community-level factors of outmigration in the Philippines, Westview Press, Boulder. Leiserowitz, A 2006, 'Climate Change Risk Perception and Policy Preferences: The Role of Affect, Imagery, and Values', *Climatic Change*, vol. 77, no. 1, 2006/07/01, pp. 45-72.

Lelieveld, J, Hadjinicolaou, P, Kostopoulou, E, Chenoweth, J, El Maayar, M, Giannakopoulos, C, Hannides, C, Lange, MA, Tanarhte, M, Tyrlis, E & Xoplaki, E 2012, 'Climate change and impacts in the Eastern Mediterranean and the Middle East', *Clim Change*, vol. 114, no. 3-4, pp. 667-687.

Lelieveld, J, Proestos, Y, Hadjinicolaou, P, Tanarhte, M, Tyrlis, E & Zittis, G 2016, 'Strongly increasing heat extremes in the Middle East and North Africa (MENA) in the 21st century', *Climatic Change*, vol. 137, no. 1-2, pp. 245-260.

Lemos, MC & Agrawal, A 2006, 'Environmental Governance', Annual Review of Environment and Resources, vol. 31, no. 1, pp. 297-325.

Lewis, LN 2008, *Egypt's future depends on agriculture and wisdom*, Peleman Industries/UniBook (web-publishing),

http://www.landpedia.org/landdoc/Analytical materials/Lewis Egypt's Future Depend s_on_Agriculture.pdf

Li, H & Tracy, MB 1999, 'Family support, financial needs, and health care needs of rural elderly in China: a field study', *J Cross Cult Gerontol*, vol. 14, no. 4, Dec, pp. 357-371.

Li, S, Juhász-Horváth, L, Harrison, PA, Pintér, L & Rounsevell, MDA 2017, 'Relating farmer's perceptions of climate change risk to adaptation behaviour in Hungary', *Journal of Environmental Management*, vol. 185, 2017/01/01/, pp. 21-30.

Loayza, NV, Olaberría, E, Rigolini, J & Christiaensen, L 2012, 'Natural Disasters and Growth: Going Beyond the Averages', *World Development*, vol. 40, no. 7, 2012/07/01/, pp. 1317-1336.

Loayza, NV & Raddatz, C 2010, 'The composition of growth matters for poverty alleviation', *Journal of Development Economics*, vol. 93, no. 1, 2010/09/01/, pp. 137-151.

Local Press Report 2015, *Egyptians Block Ring Road in Giza Over Water Shortages*, OOSKAnews, Cairo, Egypt, <u>https://www.ooskanews.com/story/2015/09/egyptians-block-ring-road-giza-over-water-shortages_167915</u>

Lofgren, H & Richards, A 2003, *Food, agriculture, and economic policy in the Middle East and North Africa*, In: Food, Agriculture, and Economic Policy in the Middle East and North Africa, JAI.

Long, J & Ferrie, J 2006, 'Labour Mobility: Oxford Encyclopedia of Economic History'.

Lubkemann, SC 2008, 'Involuntary Immobility: On a Theoretical Invisibility in Forced Migration Studies', *Journal of Refugee Studies*, vol. 21, no. 4, pp. 454-475.

Lücke, M 2011, 'Climate migration: A strategy to adapt to climate change or a failure to adapt?', in *Climate Change: Current Issues*, Kiel: Kiel Institute for the World Economy, pp. 8-9.

Lumumba, PLO 2007, 'The Interpretation of the 1929 Treaty and its Legal Relevance and Implications for the Stability of the Region', *African Sociological Review / Revue Africaine de Sociologie*, vol. 11, no. 1, pp. 10-24.

Maddison, D 2007, *The Perception of and Adaptation to Climate Change in Africa*, Policy Research Working Paper 4308, The World Bank.

Maharjan, A, de Campos, RS, Singh, C, Das, S, Srinivas, A, Bhuiyan, MRA, Ishaq, S, Umar, MA, Dilshad, T, Shrestha, K, Bhadwal, S, Ghosh, T, Suckall, N & Vincent, K 2020, 'Migration and Household Adaptation in Climate-Sensitive Hotspots in South Asia', *Current Climate Change Reports*, vol. 6, no. 1, pp. 1-16.

Malkki, L 1992, 'National geographic: The rooting of peoples and the territorialization of national identity among scholars and refugees', *Cultural anthropology*, vol. 7, no. 1, pp. 24-44.

MALR 2009, *Agricultural Sustainable Development Strategy 2030*, Ministry of Agriculture and Land Reclamation, Cairo, Egypt.

Mansour, S 2008, 'Environmental Impact of Pesticides in Egypt', *Reviews of environmental contamination and toxicology*, vol. 196, 02/01, pp. 1-51.

Mansour, SA 2004, 'Pesticide exposure—Egyptian scene', *Toxicology*, vol. 198, no. 1-3, pp. 91-115.

Marenya, PP & Barrett, CB 2007, 'Household-level determinants of adoption of improved natural resources management practices among smallholder farmers in western Kenya', *Food Policy*, vol. 32, pp. 515-536.

Margold, S 1957, 'Agrarian Land Reform in Egypt', *The American Journal of Economics and Sociology*, vol. 17, no. 1, pp. 9-19.

Marie, M, Yirga, F, Haile, M & Tquabo, F 2020, 'Farmers' choices and factors affecting adoption of climate change adaptation strategies: evidence from northwestern Ethiopia', *Heliyon*, vol. 6, no. 4.

Marii, S 1954, 'The agrarian reform in Egypt.', *International Labour Review*, vol. 69, no. 2, pp. 140-150.

Marshall, NA 2010, 'Understanding social resilience to climate variability in primary enterprises and industries', *Global Environmental Change*, vol. 20, no. 1, 2010/02/01/, pp. 36-43.

Martin, SF, Weerasinghe, S & Taylor, A 2014, 'Setting the Scene', in SF Martin, S Weerasinghe & A Taylor (eds), *Humanitarian crises and migration : causes, consequences and responses*, Routledge, New York, pp. 3-27.

Martins, PMG 2019, 'Structural change: Pace, patterns and determinants', *Review of Development Economics*, vol. 23, no. 1, 2019/02/01, pp. 1-32.

Massey, DS 1990, 'Social Structure, Household Strategies, and the Cumulative Causation of Migration', *Population Index*, vol. 56, no. 1, pp. 3-26.

Massey, DS, Arango, J, Hugo, G, Kouaouci, A, Pellegrino, A & Taylor, JE 1993, 'Theories of International Migration: A Review and Appraisal', *Population and Development Review*, vol. 19, no. 3, pp. 431-466.

Massey, DS, Axinn, WG & Ghimire, DJ 2010, 'Environmental Change and Out-Migration: Evidence from Nepal', *Population and Environment*, vol. 32, no. 2, pp. 109-136.

Massey, DS & España, FG 1987, 'The Social Process of International Migration', *Science*, vol. 237, no. 4816, pp. 733-738.

Maystadt, J-F, Trinh Tan, J-F & Breisinger, C 2014, 'Does food security matter for transition in Arab countries?', *Food Policy*, vol. 46, 2014/06/01/, pp. 106-115.

Mbow, C, Rosenzweig, C, Barioni, LG, Benton, TG, Herrero, M, Krishnapillai, M, Liwenga, E, Pradhan, P, Rivera-Ferre, MG, Sapkota, T & Xu, FNTY 2019, *Food Security*, In: Climate Change and Land: an IPCC special report on climate change, desertification, land degradation, sustainable land management, food security, and greenhouse gas fluxes in terrestrial ecosystems, Intergovernmental Panel for Climte Change (IPCC).

McCarl, BA, Musumba, M, Smith, JB, Kirshen, P, Jones, R, El-Ganzori, A, Ali, MA, Kotb, M, El-Shinnawy, I, El-Agizy, M, Bayoumi, M & Hynninen, R 2015, 'Climate change vulnerability and adaptation strategies in Egypt's agricultural sector', *Mitigation and Adaptation Strategies for Global Change*, vol. 20, no. 7, 2015/10/01, pp. 1097-1109.

McGuire, LC, Ford, ES & Okoro, CA 2007, 'Natural disasters and older US adults with disabilities: implications for evacuation', *Disasters*, vol. 31, no. 1, Mar, pp. 49-56.

McHugh, ML 2013, 'The chi-square test of independence', *Biochemia medica*, vol. 23, no. 2, pp. 143-149.

McKendrick, JH 2009, 'Mixed and Multiple Methods', in R Kitchin & N Thrift (eds), *International Encyclopedia of Human Geography*, Elsevier, pp. 128–133.

McLeman, R & Smit, B 2006, 'Migration as an Adaptation to Climate Change', *Climatic Change*, vol. 76, no. 1, 2006/05/01, pp. 31-53.

Mekonnen, Z & Kassa, H 2019, 'Living with Climate Change: Assessment of the Adaptive Capacities of Smallholders in Central Rift Valley, Ethiopia', *American Journal of Climate Change*, vol. 8, pp. 205-227.

Mendelsohn, R 2008, 'The impact of climate change on agriculture in developing countries', *Journal of Natural Resources Policy Research*, vol. 1, no. 1, pp. 5-19.

Mertz, O, Halsnæs, K, Olesen, J & Rasmussen, K 2009a, 'Adaptation to Climate Change in Developing Countries', *Environmental Management*, vol. 43, 02/01, pp. 743-752.

Mertz, O, Mbow, C, Reenberg, A & Diouf, A 2009b, 'Farmers' Perceptions of Climate Change and Agricultural Adaptation Strategies in Rural Sahel', *Environmental Management*, vol. 43, no. 5, 2009/05/01, pp. 804-816.

Met Office, The University of Nottingham, Walker Institute, Centre for Ecology and Hydrology, University of Leeds & Tyndall Centre 2011, *Climate: Observations, Projections and Impacts, Egypt*, Department of Energy and Climate Change, UK.

Milly, PC, Dunne, KA & Vecchia, AV 2005, 'Global pattern of trends in streamflow and water availability in a changing climate', *Nature*, vol. 438, no. 7066, Nov 17, pp. 347-350.

Mirza, MMQ 2011, 'Climate change, flooding in South Asia and implications', *Regional environmental change*, vol. 11, no. 1, pp. 95-107.

Mohamed, NN 2017, 'Land Degradation in the Nile Delta', in AM Negm (ed.), *The Nile Delta*, Springer International Publishing, Cham, pp. 235-264.

Mohtar, R, Assi, A & Daher, B 2017, 'Current Water for Food Situational Analysis in the Arab Region and Expected Changes Due to Dynamic Externalities', in *The Water, Energy, and Food Security Nexus in the Arab Region. Water Security in a New World.*, pp. 193-208.

Moore, JW 2010, 'Cheap food & bad money: Food, frontiers, and financialization in the rise and demise of neoliberalism', *Review (Fernand Braudel Center)*, pp. 225-261.

Moore, JW 2015, 'Cheap food and bad climate: From surplus value to negative value in the capitalist world-ecology', *Critical Historical Studies*, vol. 2, no. 1, pp. 1-43.

Morse, J 2003, Principles of Mixed Methods and Multimethod Research Design, Sage, USA.

Morsy, H & Levy, A 2020, 'Growing without changing: A tale of Egypt's weak productivity growth', *African Development Review*, vol. 32, no. 3, pp. 261-287.

Morsy, M, El-Sayed, T & Ouda, SAH 2016, 'Potential Evapotranspiration Under Present and Future Climate', in SAH Ouda & AE-H Zohry (eds), *Management of Climate Induced Drought and Water Scarcity in Egypt: Unconventional Solutions*, Springer International Publishing, Cham, pp. 5-25. Morton, JF 2007, 'The impact of climate change on smallholder and subsistence agriculture', *Proceedings of the National Academy of Sciences*, vol. 104, no. 50, pp. 19680-19685.

Mortreux, C & Barnett, J 2009, 'Climate change, migration and adaptation in Funafuti, Tuvalu', *Global Environmental Change*, vol. 19, no. 1, pp. 105-112.

Murari, K 2015, 'Farmers' Movements in Independent India', *Indian Journal of Public Administration*, vol. 61, no. 3, 2015/07/01, pp. 457-479.

Nagabhatla, N, Cassidy-Neumiller, M, Francine, NN & Maatta, N 2021, 'Water, conflicts and migration and the role of regional diplomacy: Lake Chad, Congo Basin, and the Mbororo pastoralist', *Environmental Science and Policy*, vol. 122, pp. 35-48.

Nashwan, MS, Shahid, S & Abd Rahim, N 2018, 'Unidirectional trends in annual and seasonal climate and extremes in Egypt', *Theoretical and Applied Climatology*, vol. 136, no. 1-2, pp. 457-473.

Nawar, MHA & Abdel-Hakim, T 2013, 'Current Status and Prospective of Farmers' Unions and Syndicates in Egypt', Abis, S. (ed.), In: Mobilisations rurales en méditerranée, L'Harmattan, Bibliothèque de l'IReMMO pp. 81-93.

Nawrotzki, RJ & DeWaard, J 2018, 'Putting trapped populations into place: Climate change and inter-district migration flows in Zambia', *Regional environmental change*, vol. 18, no. 2, pp. 533-546.

Negm, AM, Saavedra, O & El-Adawy, A 2017, *Nile Delta Biography: Challenges and Opportunities*, The Nile Delta, ed. AM Negm, Cham: Springer International Publishing, pp. 3-18.

Neuman, WL 2012, 'Designing the Face-to-Face Survey', in L Gideon (ed.), *Handbook of Survey Methodology for the Social Sciences*, Springer New York, New York, NY, pp. 227-248.

Neuman, WL 2014, *Social research methods : qualitative and quantitative approaches*, Seventh edition, Pearson New International Edition. edn, Pearson.

Neumayer, E, Plümper, T & Barthel, F 2014, 'The political economy of natural disaster damage', *Global Environmental Change*, vol. 24, no. 1, pp. 8-19.

Nigatu, G & Motamed, M 2015, *Middle East and North Africa Region: An Important Driver of World Agricultural Trade*, A Report from the Economic Research Service, AES-88, United States Department of Agriculture (USDA), USA.

Nightingale, AJ 2017, 'Power and politics in climate change adaptation efforts: Struggles over authority and recognition in the context of political instability', *Geoforum*, vol. 84, 2017/08/01/, pp. 11-20.

Niles, MT & Mueller, ND 2016, 'Farmer perceptions of climate change: Associations with observed temperature and precipitation trends, irrigation, and climate beliefs', *Global Environmental Change*, vol. 39, 2016/07/01/, pp. 133-142.

Nnko, H, Gwakisa, P, Ngonyoka, A & Estes, A 2021, 'Climate change and variability perceptions and adaptations of pastoralists' communities in the Maasai Steppe, Tanzania', *Journal of Arid Environments*, vol. 185, pp. 1-7.

Null, S & Prebble, M 2013, 'Spring Thaw: What Role Did Climate Change and Natural Resource Scarcity Play in the Arab Spring?', New Security Beat (online), <u>https://www.newsecuritybeat.org/2013/05/spring-thaw-role-climate-change-natural-resource-scarcity-play-arab-spring/</u>

Nunn, PD, McLean, R, Dean, A, Fong, T, Iese, V, Katonivualiku, M, Klöck, C, Korovulavula, I, Kumar, R & Tabe, T 2020, '*Adaptation to climate change: Contemporary challenges and perspectives*', In: Climate Change and Impacts in the South Pacific. Springer Nature, Switzerland, pp. 499-524.

Nunzio, JD 2013, *Conflict on the Nile: The future of transboundary water disputes over the world's longest river*, Strategic Analysis Paper, Future Directions International, Australia, pp. 1-9.

Nyanga, PH, Johnsen, FH & Aune, JB 2011, 'Smallholder Farmers' Perceptions of Climate Change and Conservation Agriculture: Evidence from Zambia', *Journal of Sustainable Development*, vol. 4, no. 4, pp. 73-85.

O'Brien, K, Eriksen, S, Nygaard, LP & Schjolden, A 2007, 'Why different interpretations of vulnerability matter in climate change discourses', *Climate policy*, vol. 7, no. 1, pp. 73-88.

O'Brien, K, Eriksen, S, Schjolden, A, Nygaard, L, O'Brien, K & Alfsen, K 2004, 'What's in a word? Conflicting interpretations of vulnerability in climate change research', CICERO Working Paper 04, p. 1.9

O'Connor, RE, Bard, RJ & Fisher, A 1999, 'Risk perceptions, general environmental beliefs, and willingness to address climate change', *Risk Analysis*, vol. 19, no. 3, pp. 461-471.

OECD-FAO 2018, *The Middle East and North Africa: Prospects and challenges*, OECD-FAO Agricultural Outlook 2018-2027, OECD Publishing, Paris/Food and Agriculture Organization of the United Nations, Rome.

Ojebiyi, W, Ashimolowo, W, Odediran, O, Soetan, O, Aromiwura, O & Adeoye, O 2015, 'Willingness to Venture into Agriculture-related Enterprises after Graduation among Final Year Agriculture Students of Federal University of Agriculture, Abeokuta', vol. 11, 01/01, pp. 103-114.

Omar, A 2015, 'Study of Farmers' Awareness and Perception of Climate Change and its impacts in Rural Damietta Governorate', Rural Sociology, Master of Science thesis, Cairo University, Egypt.

Omar, A, Nawar, M, El-Bendary, A & Abou-Bakr, H 2015, 'Farmers' Perception of Climate Change in Rural Damietta Governorate ', *Bulletin of Faculty of Agriculture, Cairo University*, 02/01, pp. 1-9.

Omar, MEDM, Moussa, AMA & Hinkelmann, R 2021, 'Impacts of climate change on water quantity, water salinity, food security, and socioeconomy in Egypt', *Water Science and Engineering*, vol. 14, no. 1, pp. 17-27.

OOSKAnews Correspondent 2008, *Protests in 3 Egyptian Governorates Over Lack of Water*, OOSKAnews, Cairo, Egypt. <u>https://www.ooskanews.com/middle-east-africa/protests-3-egyptian-governorates-over-lack-water</u>

OOSKAnews Correspondent 2012a, *Egypt Water Shortage Protests Are Widespread and Violent*, OOSKAnews, Cairo, Egypt. <u>https://www.ooskanews.com/story/2012/07/egypt-water-shortage-protests-are-widespread-and-violent_151910</u>

OOSKAnews Correspondent 2012b, Egyptian Villagers Protest Water Contamination in Menoufiya; Government Announces \$1.3 Billion USD for Water, Sanitation, Cairo, Egypt

https://www.ooskanews.com/daily-water-briefing/egyptian-villagers-protest-watercontamination-menoufiya-government-announces-1

Opiyo, F, Wasonga, O, Schilling, J & Mureithi, S 2012, 'Resource-based conflicts in drought-prone Northwestern Kenya: The drivers and mitigation mechanisms', *Wudpecker Journal of Agricultural Research*, vol. 1, no. 11, pp. 442 – 453.

Oppenheimer, M, Campos, M, Warren, R, Birkmann, J, Luber, G, O'Neill, B & Takahashi, K 2014, 'Emergent risks and key vulnerabilities Climate Change 2014: Impacts, Adaptation, and Vulnerability. In: Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change ed CB Field et al', Cambridge: Cambridge University Press.

Osberghaus, D, Finkel, E & Pohl, M 2010, 'Individual Adaptation to Climate Change: The Role of Information and Perceived Risk', ZEW Discussion Papers 10-061, ZEW -Leibniz Centre for European Economic Research.

Otto, I, Reckien, D, Reyer, C, Marcus, R, Le Masson, V, Jones, L, Norton, A & Serdeczny, O 2017, 'Social vulnerability to climate change: a review of concepts and evidence', *Regional environmental change*, vol. 17, pp. 1651–1662.

Oxford Analytica 2015, *Water stress will drive government cooperatoin*, <u>https://dailybrief.oxan.com/Analysis/GA201615/Water-stress-will-drive-government-cooperation</u>

Ozturk, T, Ceber, ZP, Türkeş, M & Kurnaz, ML 2015, 'Projections of climate change in the Mediterranean Basin by using downscaled global climate model outputs', *International Journal of Climatology*, vol. 35, no. 14, 2015/11/30, pp. 4276-4292.

Ozturk, T, Saygili-Araci, FS & Kurnaz, ML 2021, 'Projected changes in extreme temperature and precipitation indices over CORDEX-MENA domain', *Atmosphere*, vol. 12, no. 5, pp. 1-24.

Ozturk, T, Turp, MT, Türkeş, M & Kurnaz, ML 2018, 'Future projections of temperature and precipitation climatology for CORDEX-MENA domain using RegCM4.4', *Atmospheric Research*, vol. 206, pp. 87-107.

Paavola, J & Adger, WN 2002, 'Justice and adaptation to climate change', *Tyndall Centre* for Climate Change Research Working Paper 23, p. 24.

Paciello, MC & Pioppi, D 2020, 'Working class youth transitions as a litmus test for change: labour crisis and social conflict in Arab Mediterranean countries', *Mediterranean Politics*, vol. 26, no. 3, pp. 308-329.

Park, S, Marshall, N, Jakku, E, Dowd, A, Howden, S, Mendham, E & Fleming, A 2012, 'Informing adaptation responses to climate change through theories of transformation', *Global Environmental Change*, vol. 22, no. 1, pp. 115-126.

Parry, M, Parry, ML, Canziani, O, Palutikof, J, Van der Linden, P & Hanson, C 2007, *Climate change 2007-impacts, adaptation and vulnerability: Working group II contribution to the fourth assessment report of the IPCC*, vol. 4, Cambridge University Press.

Parry, ML & Carter, T 1989, 'An assessment of the effects of climatic change on agriculture', *Climatic Change*, vol. 15, no. 1-2, pp. 95-116.

Patt, A & Schröter, D 2008, 'Perceptions of climate risk in Mozambique: Implications for the success of adaptation strategies', *Global Environmental Change*, vol. 18, no. 3, pp. 458-467.

Pawar, MS 2004, *Data collecting methods and experiences: A guide for social researchers*, Sterling Publishers Pvt. Ltd.

Peek, L 2008, 'Children and disasters: Understanding vulnerability, developing capacities, and promoting resilience—An introduction', *Children Youth and Environments*, vol. 18, no. 1, pp. 1-29.

Peek, L & Stough, LM 2010, 'Children with disabilities in the context of disaster: A social vulnerability perspective', *Child development*, vol. 81, no. 4, pp. 1260-1270.

Pelling, M, High, C, Dearing, J & Smith, D 2008, 'Shadow Spaces for Social Learning: A Relational Understanding of Adaptive Capacity to Climate Change within Organisations', *Environment and Planning A: Economy and Space*, vol. 40, no. 4, 2008/04/01, pp. 867-884.

Pemunta, NV, Ngo, NV, Fani Djomo, CR, Mutola, S, Seember, JA, Mbong, GA & Forkim, EA 2021, 'The Grand Ethiopian Renaissance Dam, Egyptian National Security, and human and food security in the Nile River Basin', *Cogent Social Sciences*, vol. 7, no. 1, 2021/01/01, pp. 1-18.

Laczko, F & Piguet, E 2014, Regional Prespectives on Migration, the Environment and Climate Change, in: *People on the Move in a Changing Climate: The Regional Impact of Environmental Change on Migration*, eds E Piguet & F Laczko vol. 2, Springer, Netherland, pp. 1-20.

Piguet, E, Pecoud, A & de Guchteneire, P 2011, 'Migration and Climate Change: An Overview', *Refugee Survey Quarterly*, vol. 30, no. 3, pp. 1-23.

Pincus, J 1996, 'Class Power and Agrarian Change: Land and Labour in Rural West Java', Palgrave macmillan.

Piore, MJ 1979, *Birds of Passage : Migrant Labor and Industrial Societies*, Cambridge University Press, Cambridge.

Pohoryles, S 2009, *Agricultural Cooperation: A Prototype of Post-conflict Resolution*, The Legacy of Camp David: 1979-2009. A Special Edition of Viewpoints., The Middle East Institute, Washington, DC.

Poudel, S, Funakawa, S, Shinjo, H & Mishra, B 2020, 'Understanding households' livelihood vulnerability to climate change in the Lamjung district of Nepal', *Environment, Development and Sustainability*, vol. 22, no. 8, pp. 8159-8182.

Power, L 2014, *Death on the Nile: Egypt's Burgeoning Food and Water Security Crisis*, Strategic Analysis Paper, Future Directions International, Australia, p.8.

Pradeep, K & Shane, R 2003, *Climate Change and Agriculture: A Review of Impacts and Adaptations*, Climate Change Series: Paper No. 91, The International Bank for Reconstruction and Development/The World Bank, Washington, DC.

Preston, V 2009, 'Questionnaire Survey', in R Kitchin & N Thrift (eds), *International Encyclopedia of Human Geography*, Elsevier, pp. 46-52.

Pretty, J, Benton, TG, Bharucha, ZP, Dicks, LV, Flora, CB, Godfray, HCJ, Goulson, D, Hartley, S, Lampkin, N, Morris, C, Pierzynski, G, Prasad, PVV, Reganold, J, Rockström,

J, Smith, P, Thorne, P & Wratten, S 2018, 'Global assessment of agricultural system redesign for sustainable intensification', *Nature Sustainability*, vol. 1, no. 8, pp. 441-446.

Pyhälä, A, Fernández-Llamazares, Á, Lehvävirta, H, Byg, A, Ruiz-Mallén, I, Salpeteur, M & Thornton, TF 2016, 'Global environmental change: local perceptions, understandings, and explanations', *Ecology and society : a journal of integrative science for resilience and sustainability*, vol. 21, no. 3, p. 25.

Quisumbing, AR & McNiven, S 2005, *Migration and the rural-urban continuum: Evidence from the Rural Philippines*, International Food Policy Research Institute (IFPRI), Washington, DC.

Raha, S & Gayen, SK 2021, 'Drought-Induced Human Mobility in Purulia District of West Bengal', In: *Habitat, Ecology and Ekistics*, Springer, pp. 263-277.

Ramadan, E, Negm, A, Smanny, M & Helmy, A 2013, *Environmental Impacts Of Great Ethiopian Renaissance Dam On The Egyptian Water Resources Management And Security*. Conference paper: The 23rd. International Conference On: Environmental Protection is a Must 11 – 13 May 2013, Alexandria.

Ramos, R 2019, 'Migration aspirations among youth in the Middle East and North Africa region', *Journal of Geographical Systems*, vol. 21, no. 4, pp. 487-507.

Ray-Bennett, NS 2009, 'The influence of caste, class and gender in surviving multiple disasters: A case study from Orissa, India', *Environmental Hazards*, vol. 8, no. 1, 2009/01/01, pp. 5-22.

Renaud, F, Bogardi, JJ, Dun, O & Warner, K 2007, *Control, Adapt or Flee: How to Face Environmental Migration?*, InterSecTions, Publication Series of UNU-EHS No. 5/2007, UNU-EHS, Bonn.

Reuters 2018, 'Egypt to begin importing rice after slashing its own cultivation', Cairo, Egypt, <u>https://www.reuters.com/article/ozabs-uk-egypt-rice-idAFKCN1J11RV-OZABS</u>

Reuveny, R 2007, 'Climate change-induced migration and violent conflict', *Political Geography*, vol. 26, no. 6, 2007/08/01/, pp. 656-673.

Ribot, J 2011, 'Vulnerability Before Adaptation: Toward Transformative Climate Action', *Global Environmental Change*, vol. 21, no.4 10/01, pp. 1160–1162.

Ribot, J 2013, 'Vulnerability does not just fall from the sky: Toward multi-scale pro-poor climate policy', in MR Redclift & M Grasso (eds), *Handbook on Climate Change and Human Security*, Edward Elgar Publishing Limited, UK, pp. 164-172.

Rickards, L & Howden, SM 2012, 'Transformational adaptation: agriculture and climate change', *Crop and Pasture Science*, vol. 63, no. 3, pp. 240-250.

Rigg, J 2001, *More than the soil : rural change in Southeast Asia*, Rural change in Southeast Asia, Prentice Hall, Harlow.

Rigg, J 2006, 'Land, farming, livelihoods, and poverty: Rethinking the links in the Rural South', *World Development*, vol. 34, no. 1, pp. 180-202.

Rigg, J 2007, 'Moving lives: migration and livelihoods in the Lao PDR', *Population, Space and Place*, vol. 13, no. 3, pp. 163-178.

Rodrik, D 2010, 'Structural transformation and economic development', Economic Policy Research Foundation of Turkey. *TEPAV: Ankara*. p.28.

Rodrik, D 2016, 'Premature deindustrialization', *Journal of economic growth*, vol. 21, no. 1, pp. 1-33.

Rodrik, D 2017, 'Premature deindustrialisation in the developing world', *Frontiers of Economics in China*, vol. 12, no. 1, pp. 1-7.

Roncolato, L & Kucera, D 2014, 'Structural drivers of productivity and employment growth: a decomposition analysis for 81 countries', *Cambridge Journal of Economics*, vol. 38, no. 2, pp. 399-424.

Rothermich, K, Johnson, EK, Griffith, RM & Beingolea, MM 2021, 'The influence of personality traits on attitudes towards climate change – An exploratory study', *Personality and Individual Differences*, vol. 168, 2021/01/01/, p. 110304.

Roumasset, JA, Boussard, J-M & Singh, I 1979, *Risk, uncertainty, and agricultural development*, eds JA Roumasset, J-M Boussard & I Singh, Agricultural Development Council, New York, N.Y.

Saad, R 2002, 'Egyptian politics and the tenancy law.', in R Bush (ed.), *Counter revolution in Egypt's countryside.*, Zed Books, London.

Saber, M 2009, 'Environment in Jeopardy: Consequences of Climate Change in Egypt', *Journal of Ecology and the Natural Environment*, vol. 1, no.5, November 2009, pp. 191-195.

Sahu, PK 2013, 'Collection of Data', in *Research Methodology: A Guide for Researchers In Agricultural Science, Social Science and Other Related Fields*, Springer India, India, pp. 63-73.

Salam, K 2020, 'The Blue Nile Is Dammed: Geopolitics, water security, and health will keep the dispute over the Grand Ethiopian Renaissance Dam alive.', *Foreign Policy (FP)*, <u>https://foreignpolicy.com/2020/07/24/the-blue-nile-is-dammed/</u>

Salehi-Isfahani, D, Hassine, NB & Assaad, R 2014, 'Equality of opportunity in educational achievement in the Middle East and North Africa', *The Journal of Economic Inequality*, vol. 12, no. 4, pp. 489-515.

Sampei, Y & Aoyagi-Usui, M 2009, 'Mass-media coverage, its influence on public awareness of climate-change issues, and implications for Japan's national campaign to reduce greenhouse gas emissions', *Global Environmental Change*, vol. 19, no. 2, 2009/05/01/, pp. 203-212.

Sanchez, E, Gallardo, C, Gaertner, M, Arribas, A & Castro, M 2004, 'Future climate extreme events in the Mediterranean simulated by a regional climate model: A first approach', *Global and Planetary Change*, vol. 44, no. 1-4, pp. 163-180.

Scheffran, J, Ide, T & Schilling, J 2014, 'Violent climate or climate of violence? Concepts and relations with focus on Kenya and Sudan', *The International Journal of Human Rights*, vol. 18, no. 3, 2014/04/03, pp. 369-390.

Schewel, K 2019, 'Understanding Immobility: Moving Beyond the Mobility Bias in Migration Studies', *International Migration Review*, vol. 54, no. 2, 2020/06/01, pp. 328-355.

Schleussner, CF, Donges, JF, Donner, RV & Schellnhuber, HJ 2016, 'Armed-conflict risks enhanced by climate-related disasters in ethnically fractionalized countries',

Proceedings of the National Academy of Sciences of the United States of America, vol. 113, no. 33, pp. 9216-9221.

Schuh, GE 1976, 'The New Macroeconomics of Agriculture', *American Journal of Agricultural Economics*, vol. 58, no. 5, pp. 802-811.

Scott, JC 1976, *The moral economy of the peasant rebellion and subsistence in Southeast Asia*, Yale University Press, New Haven.

Scott, JC 1985, Weapons of the weak : everyday forms of peasant resistance, Yale University Press, New Haven.

Selby, J, Dahi, OS, Fröhlich, C & Hulme, M 2017, 'Climate change and the Syrian civil war revisited', *Political Geography*, vol. 60, pp. 232-244.

Selby, J & Hoffmann, C 2014, 'Beyond scarcity: Rethinking water, climate change and conflict in the Sudans', *Global Environmental Change*, vol. 29, pp. 360-370.

Selwaness, I & Roushdy, R 2019, 'Young people school-to-work transition in the aftermath of the Arab Spring: Early evidence from Egypt', *International Journal of Manpower*, vol. 40, no. 3, pp. 398-432.

Semenza, JC, Hall, DE, Wilson, DJ, Bontempo, BD, Sailor, DJ & George, LA 2008, 'Public Perception of Climate Change', *American Journal of Preventive Medicine*, vol. 35, no. 5, pp. 479-487.

Sen, AK 1983, *Poverty and famines: an essay on entitlement and deprivation.*, Oxford University Press, Oxford, UK.

Seneviratne, S, Nicholls, N, Easterling, D, Goodess, C, Kanae, S, Kossin, J, Luo, Y, Marengo, J, McInnes, K, Rahimi, M, Reichstein, M, Sorteberg, A, Vera, C & Zhang, X 2012, *Changes in climate extremes and their impacts on the natural physical environment*, Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change (IPCC). IPCC, Cambridge University Press, Cambridge, UK, and New York, NY, USA.

Seyam, MA 2011, 'Extreme Weather Events Impacted by Climate Variability over Egypt', Ph.D. thesis, Cairo University, Cairo.

Shaarawy, H 2012, 'The evolution of the sociopolitical scene in Egypt prior to the January 2011 unprising.', in H Shaarawy & A Shokr (eds), *Social protest movements in Egypt and the preparation to the January 25 Revolution.*, Jazeret Alward, Egypt.

Shalaby, A 2012, 'Assessment of urban sprawl impact on the agricultural land in the Nile delta of Egypt using remote sensing and digital soil map', *Int. J. Environ. Sci.*, vol. 1, 01/01, pp. 253-262.

Sharp, JM 2020, *Egypt: Background and U.S. Relations: CRS Report*, Congressional Research Service (CRS), United States.

Sheahan, M, Barrett, CB & Goldvale, C 2017, 'Human health and pesticide use in Sub-Saharan Africa', *Agricultural Economics*, vol. 48, no. S1, pp. 27-41.

Shinbrot, XA, Jones, KW, Rivera-Castañeda, A, López-Báez, W & Ojima, DS 2019, 'Smallholder Farmer Adoption of Climate-Related Adaptation Strategies: The Importance of Vulnerability Context, Livelihood Assets, and Climate Perceptions', *Environmental Management*, vol. 63, no. 5, May 01, pp. 583-595.

SIDA 2019, *Multidimensional Poverty Analysis for the Middle East and North Africa* 2019, The Swedish International Development Cooperation Agency (SIDA).

Sika, N 2016, 'The disguise of youth inclusion in Egypt', *Power for Youth Project, Working paper No. 4*, p.25.

Sika, N 2019, 'Neoliberalism, marginalization and the uncertainties of being young: the case of Egypt', *Mediterranean Politics*, vol. 24, no. 5, pp. 545-567.

Sika, N 2020, 'Youth socio-economic and political grievances: Bringing the 'Political' back into understanding contestation in the MENA', *Mediterranean Politics*, vol. 26, no. 3, pp. 393-406.

Simon, HA 1982, Models of bounded rationality, MIT Press, Cambridge, Mass.

Singh, P, Charan, D, Kaur, M, Railoa, K & Chand, R 2020, 'Place Attachment and Cultural Barriers to Climate Change Induced Relocation: Lessons from Vunisavisavi Village, Vanua Levu, Fiji', In: *Climate Change Management*, pp. 27-43.

SIS 2009, 'Damietta Governorate', State Information Service, Cairo, Egypt, [online], <u>https://www.sis.gov.eg/Story/171?lang=ar</u>

Slovic, P 1987, 'Perception of risk', Science, vol. 236, no. 4799, pp. 280-285.

Smit, B & Pilifosova, O 2003, *Adaptation to Climate Change in the Context of Sustainable Development and Equity*, Climate Change 2001: Impacts, Adaptation, and Vulnerability, Intergovernmental Panel on Climate Change (IPCC), Cambridge, UK.

Smit, B & Skinner, M 2002, 'Adaptation options in agriculture to climate change: a typology', *Mitigation and Adaptation Strategies for Global Change*, vol. 7, 03/01, pp. 85-114.

Smit, B & Wandel, J 2006, 'Adaptation, adaptive capacity and vulnerability', *Global Environmental Change*, vol. 16, no. 3, 2006/08/01/, pp. 282-292.

Smith, JB, McCarl, BA, Kirshen, P, Jones, R, Deck, L, Abdrabo, MA, Borhan, M, El-Ganzori, A, El-Shamy, M, Hassan, M, El-Shinnawy, I, Abrabou, M, Hassanein, MK, El-Agizy, M, Bayoumi, M & Hynninen, R 2014, 'Egypt's Economic Vulnerability to Climate Change', *Climate Research*, vol. 62, pp. 59-70.

Smithers, J & Smit, B 1997, 'Human adaptation to climatic variability and change', *Global Environmental Change*, vol. 7, no. 2, 1997/07/01/, pp. 129-146.

Smithers, J & Smit, B 2009, 'Adaptation to climatic variability and change', In: *Adaptation to Climate Change*. Earthscan, London, pp. 15-33

Sobczak-Szelc, K & Fekih, N 2020, 'Migration as one of several adaptation strategies for environmental limitations in Tunisia: evidence from El Faouar', *Comparative Migration Studies*, vol. 8, no. 1, pp. 1-20.

Sofuoğlu, E & Ay, A 2020, 'The relationship between climate change and political instability: the case of MENA countries (1985:01–2016:12)', *Environmental Science and Pollution Research*, vol. 27, no. 12, 2020/04/01, pp. 14033-14043.

Sousa, PM, Trigo, RM, Aizpurua, P, Nieto, R, Gimeno, L & Garcia-Herrera, R 2011, 'Trends and extremes of drought indices throughout the 20th century in the Mediterranean', *Nat. Hazards Earth Syst. Sci.*, vol. 11, no. 1, pp. 33-51.

Soussa, H 2010, 'Effects of Drip Irrigation Water Amount on Crop Yield, Productivity and Efficiency of Water Use in Desert Regions in Egypt', *Nile Basin Water Sci Eng J*, vol. 3, no. 2, pp. 96-109.

Sowers, J, Vengosh, A & Weinthal, E 2011, 'Climate change, water resources, and the politics of adaptation in the Middle East and North Africa', *Climatic Change*, vol. 104, no. 3, pp. 599-627.

Sowers, J & Weinthal, E 2010, *Climate Change Adaptation in the Middle East and North Africa: Challenges and Opportunities*, Working Paper No. 2, Dubai School of govenment; Belfer Center for Science and International Affairs, Dubai, United Arab Emirates.

Spence, A, Poortinga, W, Butler, C & Pidgeon, NF 2011, 'Perceptions of climate change and willingness to save energy related to flood experience', *Nature Climate Change*, vol. 1, no. 1, 2011/04/01, pp. 46-49.

Springborg, R 1978, 'Professional Syndicates in Egyptian Politics, 1952-1970', *International Journal of Middle East Studies*, vol. 9, no. 3, pp. 275-295.

Springborg, R 1990, 'Agrarian Bourgeoisie, Semiproletarians, and the Egyptian State: Lessons for Liberalization', *International Journal of Middle East Studies*, vol. 22, no. 4, pp. 447-472.

Standing, G 1981, 'Migration and modes of exploitation: Social origins of immobility and mobility', *The Journal of Peasant Studies*, vol. 8, no. 2, 1981/01/01, pp. 173-211.

Stark, O & Bloom, DE 1985, 'The New Economics of Labour Migration', *American Economic Review*, vol. 75, no. 2, May 1985, pp. pp. 173-178.

Stark, O & Levhari, D 1982, 'On Migration and Risk in LDCs', *Economic Development and Cultural Change*, vol. 31, no. 1, pp. 191-196.

Stephens, E, Nicholson, C, Brown, D, Parsons, D, Barrett, C, Lehmann, J, Mbugua, D, Ngoze, S, Pell, A & Riha, S 2012, 'Modeling the impact of natural resource-based poverty

traps on food security in Kenya: The Crops, Livestock and Soils in Smallholder Economic Systems (CLASSES) model', *Food Security*, vol. 4, no. 3, pp. 423-439.

Stern, N 2006, Stern review: the economics of climate change, HM Treasury, London.

Stern, N 2007, *The Economics of Climate Change: The Stern Review*, Cambridge University Press, Cambridge.

Sternberg, T 2013, Chinese Drought, Wheat and the Egyptian Uprising: How a Localized Hazard Became Globalized, In: *The Arab Spring and Climate Change: A Climate and Security Correlations Series*, The Center for Climate and Security.

Suppan, P, Kunstmann, H, Heckl, A & Rimmer, A 2008, 'Impact of Climate Change on Water Availability in', in F Zereini & H Hötzl (eds), *Climatic Changes and Water Resources in the Middle East and North Africa*, Springer Berlin Heidelberg, Berlin, Heidelberg, pp. 47-58.

Suter, M 2017, *Running out of water: Conflict and water scarcity in Yemen and Syria*, Atlantic Council, <u>https://www.atlanticcouncil.org/blogs/menasource/running-out-of-water-conflict-and-water-scarcity-in-yemen-and-syria/</u>

Sutherland, K, Smit, B, Wulf, V & Nakalevu, T 2005, 'Vulnerability in Samoa', *Tiempo*, vol. 54, pp. 11-15.

Swain, A 2011, 'Challenges for water sharing in the Nile basin: changing geo-politics and changing climate', *Hydrological Sciences Journal*, vol. 56, no. 4, 2011/06/01, pp. 687-702.

Sweeney, S, Mukora, R, Candfield, S, Guinness, L, Grant, AD & Vassall, A 2018, 'Measuring income for catastrophic cost estimates: limitations and policy implications of current approaches', *Social Science & Medicine*, vol. 215, pp. 7-15.

Tacoli, C 2009, 'Crisis or adaptation? Migration and climate change in a context of high mobility', *Environment & Urbanization*, vol. 21, no. 2, pp. 513-525.

Tashakkori, A & Teddlie, C 2003, *Handbook of mixed methods in social & behavioral research*, Mixed methods in social & behavioral research, SAGE Publications, Thousand Oaks, Calif.

Teddlie, C & Tashakkori, A 2009, Foundations of mixed methods research : integrating quantitative and qualitative techniques in the social and behavioral sciences, Sage Publications, Thousand Oaks.

Tellioglu, I & Konandreas, P 2017, *Agricultural Policies, Trade and Sustainable Development in Egypt*, International Centre for Trade and Sustainable Development (ICTSD) and Food and Agriculture Organization (FAO), Geneva and Rome.

Terink, W, Immerzeel, WW & Droogers, P 2013, 'Climate change projections of precipitation and reference evapotranspiration for the Middle East and Northern Africa until 2050', *International Journal of Climatology*, vol. 33, no. 14, pp. 3055-3072.

Tesfaye, A 2013, 'The Politics of the Imposed and Negotiation of the Emerging Nile Basin Regime', *International Journal of Ethiopian Studies*, vol. 7, no. 1 & 2, pp. 57-76.

The Economist 2015, *Sisi signs accord on Ethiopian dam*, <u>http://country.eiu.com/article.aspx?articleid=1443019528&Country=Ethiopia&topic=P</u> <u>olitics&subtopic_2#</u>

Thomalla, F, Downing, T, Spanger-Siegfried, E, Han, G & Rockström, J 2006, 'Reducing hazard vulnerability: towards a common approach between disaster risk reduction and climate adaptation', *Disasters*, vol. 30, no. 1, 2006/03/01, pp. 39-48.

Thomas, BY, William, MF & Tobias, W 2018, 'Can Social Capital influence Smallholder Farmers' Climate-Change Adaptation Decisions? Evidence from Three Semi-Arid Communities in Burkina Faso, West Africa', *Social Sciences*, vol. 7, no. 3, p. 33.

Thornton, F, McNamara, KE, Farbotko, C, Dun, O, Ransan-Cooper, H, Chevalier, E & Lkhagvasuren, P 2019, 'Human mobility and environmental change: a survey of perceptions and policy direction', *Population and Environment*, vol. 40, no. 3, 2019/03/01, pp. 239-256.

Thornton, PK, van de Steeg, J, Notenbaert, A & Herrero, M 2009, 'The impacts of climate change on livestock and livestock systems in developing countries: A review of what we know and what we need to know', *Agricultural Systems*, vol. 101, no. 3, 2009/07/01/, pp. 113-127.

Thulstrup, AW 2015, 'Livelihood Resilience and Adaptive Capacity: Tracing Changes in Household Access to Capital in Central Vietnam', *World Development*, vol. 74, 2015/10/01/, pp. 352-362.

Tilzey, M 2020, 'Capitalism, imperialism, nationalism: agrarian dynamics and resistance as radical food sovereignty', *Canadian Journal of Development Studies*, vol. 41, no. 3, pp. 381-398.

Timmer, C 1988, 'The agricultural transformation', in H Chenery & TN Srinivasan (eds), *Handbook of Development Economics*, vol. 1, Elsevier, pp. 275-331.

Timmer, CP 2009, 'A world without agriculture? The historical paradox of agricultural development', *American Enterprise Institute for Public Policy Research*, no. 1, p. 96.

Tinch, R, Jäger, J, Omann, I, Harrison, PA, Wesely, J & Dunford, R 2015, 'Applying a capitals framework to measuring coping and adaptive capacity in integrated assessment models', *Climatic Change*, vol. 128, no. 3, 2015/02/01, pp. 323-337.

Travers, M 2006, 'Postmodernism and qualitative research', *Qualitative Research*, vol. 6, no. 2, 2006/05/01, pp. 267-273.

Tregenna, F 2008, 'The contributions of manufacturing and services to employment creation and growth in South Africa', *South African Journal of Economics*, vol. 76, 02/01, pp. 175-175.

Tür, Ö 2018, 'Challenges of demographic pressures and resource scarcity on the political economy in the Levant & MENA region', *Uluslararasi Iliskiler*, vol. 15, no. 60, pp. 75-87.

UNDP 2011, Drought Impact Assessment, Recovery and Mitigation Framework and Regional Project Design in Kurdistan Region (KR), United Nations Development Programme (UNDP) - IRAQ, Iraq.

UNDP 2014, Human Development Report 2014. Sustaining Human Progress: Reducing Vulnerabilities and Building Resilience, The United Nations Development Programme (UNDP), New York, USA.

UNDP 2016, Arab Human Development Report 2016: Youth and the Prospects for Human Development in a Changing Reality, Regional Bureau for Arab States (RBAS), United Nations Development Programme (UNDP), New York, USA.

UNFCCC 2007, *Climate Change: Impacts, Vulnerabilities and Adaptation in Developing Countries*, United Nations Framework Convention on Climate Change (UNFCCC), Germany.

UNICEF 2018, Country Background: Egypt, UNICEF, https://www.unicef.org/egypt/country-background

UNICEF 2019, *MENA GENERATION 2030: Investing in children and youth today to secure a prosperous region tomorrow*, United Nations Children's Fund (UNICEF) Division of Data, Research and Policy.

UNOCHA 2019, Yemen Humanitarian needs overview, United Nations for the Coordination of Humanitarian Affairs, NY.

Valdivia, RO, Antle, JM & Stoorvogel, JJ 2017, 'Designing and evaluating sustainable development pathways for semi-subsistence crop–livestock systems: lessons from Kenya', *Agricultural Economics*, vol. 48, no. S1, pp. 11-26.

Van der Geest, K, Vrieling, A & Dietz, T 2010, 'Migration and environment in Ghana: a cross-district analysis of human mobility and vegetation dynamics', *Environment and Urbanization*, vol. 22, no. 1, pp. 107-123.

van Neuss, L 2019, 'The Drivers Of Structural Change', *Journal of Economic Surveys*, vol. 33, pp. 309-349.

Varela, R, Rodríguez-Díaz, L & deCastro, M 2020, 'Persistent heat waves projected for Middle East and North Africa by the end of the 21st century', *PLoS ONE*, vol. 15, no. 11.

Vedwan, N & Rhoades, RE 2001, 'Climate Change in the Western Himalayas of India: A Study of Local Perception and Response', *Climate Research - CLIMATE RES*, vol. 19, no.2, pp. 109-117.

Vengosh, A 2003, 'Salinization and Saline Environments', in HD Holland & KK Turekian (eds), *Treatise on Geochemistry*, Pergamon, Oxford, pp. 1-35.

Vergara-Camus, L 2014, Land and Freedom: The MST, the Zapatistas and Peasant Alternatives to Neoliberalism, ZED books.

Verme, P, Milanovic, B, Al-Shawarby, S, Tawila, SE, Gadallah, M & A.El-Majeed, EA 2014, *Inside Inequality in the Arab Republic of Egypt Facts and Perceptions across People, Time, and Space*, The World Bank, Washington, DC.

Verner, D 2012, Adaptation to a changing climate in the Arab countries a case for adaptation governance and leadership in building climate resilience, MENA development report, ed. D Verner, World Bank, Washington, D.C.

Voll, SP 1980, 'Egyptian Land Reclamation since the Revolution', *Middle East Journal*, vol. 34, no. 2, pp. 127-148.

Von Uexkull, N, Croicu, M, Fjelde, H & Buhaug, H 2016, 'Civil conflict sensitivity to growing-season drought', *Proceedings of the National Academy of Sciences of the United States of America*, vol. 113, no. 44, pp. 12391-12396.

Waha, K, Krummenauer, L, Adams, S, Aich, V, Baarsch, F, Coumou, D, Fader, M, Hoff, H, Jobbins, G, Marcus, R, Mengel, M, Otto, I, Perrette, M, Rocha, M, Robinson, A & Schleussner, C-F 2017, 'Climate change impacts in the Middle East and Northern Africa (MENA) region and their implications for vulnerable population groups', *Regional environmental change*, vol. 17, no. 6, pp. 1623-1638.

Walker, RJ 2016, 'Population Growth and its Implications for Global Security', *American Journal of Economics and Sociology*, vol. 75, no. 4, pp. 980-1004.

Walter, M 2006, *Social research methods : an Australian perspective*, Oxford University Press, South Melbourne, Vic.

Walter, M 2013, *Social research methods*, Third edition. edn, Oxford University Press, South Melbourne, Vic.

Waly, NM, Ayad, HM & Saadallah, DM 2020, 'Assessment of spatiotemporal patterns of social vulnerability: A tool to resilient urban development Alexandria, Egypt', *Ain Shams Engineering Journal*, vol 12, no. 1, pp. 1059-1072.

Warner, K 2010, 'Global environmental change and migration: Governance challenges', *Global Environmental Change*, vol. 20, no. 3, pp. 402-413.

Warner, K & Afifi, T 2014, 'Where the rain falls: Evidence from 8 countries on how vulnerable households use migration to manage the risk of rainfall variability and food insecurity', *Climate and Development*, vol. 6, no. 1, pp. 1-17.

Warner, K, Afifi, T, Dun, O, Stal, M & Schmidl, S 2008, *Human Security, Climate Change, and Environmentally Induced Migration*, In: Thanos Dokos (ed.) Climate Change: Addressing the Impact on Human Security, pp. 46 – 67.

Warner, K, Ehrhart, C, Sherbinin, Ad, Adamo, S & Chai-Onn, T 2009, *In Search of Shelter: Mapping the Effects of Climate Change on Human Migration and Displacement*, United Nations University, CARE International, CIESIN-Columbia University, UN Refugee Agency (UNHCR), and World Bank, Bonn, Germany, p. 31.

Warner, K, Hamza, M, Oliver-Smith, A, Renaud, F & Julca, A 2010, 'Climate change, environmental degradation and migration', *Natural Hazards*, vol. 55, no. 3, 2010/12/01, pp. 689-715.

Warner, K, Hoffmaister, J & Milan, A 2015, 'Human mobility and adaptation: Reducing susceptibility to climatic stressors and mainstreaming', in F Hillmann, M Pahl, B Rafflenbeul & H Sterly (eds), *Environmental Change, Adaptation and Migration: Bringing in the Region*, Palgrave Macmillan, London, pp. 41-56.

Wasimi, SA 2010, 'Climate change in the Middle East and North Africa (MENA) region and implications for water resources project planning and management', *International Journal of Climate Change Strategies and Management*, vol. 2, no. 3, pp. 297-320.

Weber, E 1997, 'Perception and expectation of climate change: Precondition for economic and technological adaptation', In Bazerman M., Messick D., Tenbrunsel A., and Wade-Benzoni K.(Eds.) *Psychological perspectives to environmental and ethical issues in management*, San Francisco, CA: Jossey-Bass, pp. 314–341.

Weber, EU 2010, 'What shapes perceptions of climate change?', *Wiley Interdisciplinary Reviews: Climate Change*, vol. 1, no. 3, pp. 332-342.

Weber, EU & Johnson, EJ 2009, 'Decisions under uncertainty: Psychological, economic, and neuroeconomic explanations of risk preference', in *Neuroeconomics*, Elsevier, pp. 127-144.

Weinbaum, MG 1982, *Food, development, and politics in the Middle East*, Westview special studies on the Middle East, Westview Press, Boulder, Colo.

Weinthal, E, Vengosh, A, Marei, A, Gutierrez, A & Kloppmann, W 2005, 'The Water Crisis in the Gaza Strip: Prospects for Resolution', *Groundwater*, vol. 43, no. 5, 2005/09/01, pp. 653-660.

Werner, AD & Simmons, CT 2009, 'Impact of Sea-Level Rise on Sea Water Intrusion in Coastal Aquifers', *Groundwater*, vol. 47, no. 2, 2009/03/01, pp. 197-204.

Wheeler, S, Zuo, A & Bjornlund, H 2013, 'Farmers' climate change beliefs and adaptation strategies for a water scarce future in Australia', *Global Environmental Change*, vol. 23, no. 2, 2013/04/01/, pp. 537-547.

Wheeler, SA, Zuo, A & Bjornlund, H 2014, 'Australian irrigators' recognition of the need for more environmental water flows and intentions to donate water allocations', *Journal of Environmental Planning and Management*, vol. 57, no. 1, 2014/01/01, pp. 104-122.

White, B 2012, 'Agriculture and the generation problem: rural youth, employment and the future of farming', *IDS Bulletin*, vol. 43, no. 6, pp. 9-19.

Whittington, D, Waterbury, J & Jeuland, M 2014, 'The Grand Renaissance Dam and prospects for cooperation on the Eastern Nile', *Water Policy*, vol. 16, no. 4, pp. 595-608.

Wilbanks, TJ 2003, 'Integrating climate change and sustainable development in a place-based context', *Climate policy*, vol. 3, no. sup1, 2003/11/01, pp. S147-S154.

Wisner, B 2006, Let Our Children Teach Us! A Review of the Role of Education and Knowledge in Disaster Risk Reduction, Books for Change: Bangalore, India.

Woertz, E 2017, 'Agriculture and Development in the Wake of the Arab Spring', *Combining Economic and Political Development : The Experience of MENA*, *International Development Policy*, vol. 7, no. 7, pp. 144-169.

Woo, SHL, Liu, JC, Yue, X, Mickley, LJ & Bell, ML 2020, 'Air pollution from wildfires and human health vulnerability in Alaskan communities under climate change', *Environmental Research Letters*, vol. 15, no. 9.

Wood, A 2019, 'Globalization and structural change around the world, 1985–2015', *World Bank Research Observer*, vol. 34, no. 1, pp. 65-94.

Wöppelmann, G, Le Cozannet, G, de Michele, M, Raucoules, D, Cazenave, A, Garcin, M, Hanson, S, Marcos, M & Santamaría-Gómez, A 2013, 'Is land subsidence increasing the exposure to sea level rise in Alexandria, Egypt?', *Geophysical Research Letters*, vol. 40, no. 12, 2013/06/28, pp. 2953-2957.

World Bank 2011, North African coastal cities: address natural disasters and climate change—summary of the regional study, The World Bank, Washington D.C.

World Bank 2014, *Arab Republic of Egypt – More jobs, better jobs: A priority for Egypt.*, Report No. 88447-EG., Washington, DC.

World Bank 2019a, 'The World Bank Data, Unemployment rates- MENA region', <u>https://data.worldbank.org/indicator/SL.UEM.TOTL.ZS?locations=ZQ-1W</u>

World Bank 2019b, 'The World Bank Data: Total Population- MENA region', <u>https://data.worldbank.org/indicator/SP.POP.TOTL?locations=ZQ</u>

World Bank 2020, 'The World Bank Data, Employment in Agriculture - Egypt', <u>https://data.worldbank.org/indicator/SL.AGR.EMPL.ZS?locations=EG</u>

World Bank Group 2015, *Enabling the Business of Agriculture*, Progress Report 2015, The World Bank Group, Washington DC.

WWAP 2012, *The United Nations World Water Development Report 4: Managing Water Report under Uncertainty and Risk*, World Water Assessment Programme (WWAP)/United Nations Educational, Scientific and Cultural Organization (UNESCO), Paris, France.

Yang, J, Owusu, V, Andriesse, E & Dziwornu Ablo, A 2019, 'In-Situ Adaptation and Coastal Vulnerabilities in Ghana and Tanzania', *The Journal of Environment & Development*, vol. 28, no. 3, 2019/09/01, pp. 282-308.

Zagata, L & Sutherland, LA 2015, 'Deconstructing the 'young farmer problem in Europe: Towards a research agenda', *Journal of Rural Studies*, vol. 38, pp. 39-51.

Zenou, Y 2011, 'Rural–urban Migration and Unemployment: Theory and Policy Implications', *Journal of Regional Science*, vol. 51, no. 1, pp. 65-82.

Zhang, Z, Gu, D & Luo, Y 2014, 'Coresidence with elderly parents in contemporary China: The role of filial piety, reciprocity, socioeconomic resources, and parental needs', *Journal of Cross-Cultural Gerontology*, vol. 29, no. 3, pp. 259-276.

Zickgraf, C 2019, 'Keeping People in Place: Political Factors of (Im) mobility and Climate Change', *Social Sciences*, vol. 8, no. 8, p. 228.

Zickgraf, C, Vigil Diaz Telenti, S, De Longueville, F, Ozer, P & Gemenne, F 2016, 'The impact of vulnerability and resilience to environmental changes on mobility patterns in West Africa', The World Bank, Washington, United States, p. 26.

Zohry, A 2002, 'Rural-to-Urban Labor Migration: A Study of Upper Egyptian Laborers in Cairo', Centre for the Comparative Study of Culture, Development and the Environment (CDE), Doctor of Philosophy thesis, University of Sussex, Brighton.

Zohry, A 2005, 'Interrelationships between internal and international migration in Egypt: a pilot study', Development Research Center on Migration, Globalization, and Poverty, University of Sussex, Brighton, p. 102.

Zohry, A 2007, *Migration and Development in Egypt*, Paper Prepared For Project on Migration as a Potential and Risk Funder by, Robert Bosch Foundation; Institute for Migration and Cultural Studies (IMIS); Osnabruck University.

Zohry, A 2014, 'Migration And Development In Egypt', in M Bommes, H Fassmann & W Sievers (eds), *Migration from the Middle East and North Africa to Europe: Past Developments, Current Status, and Future Potentials*, Amsterdam University Press, Amsterdam, pp. 75-98.

Zohry, AE-H & Ouda, SAH 2016, 'Crops Intensification to Face Climate Induced Water Scarcity in Nile Delta Region', in SAH Ouda & AE-H Zohry (eds), *Management of Climate Induced Drought and Water Scarcity in Egypt: Unconventional Solutions*, Springer International Publishing, Cham, pp. 47-62.